in Rule 12b-2 of the Exchange Act.

Large accelerated filer

Non-accelerated filer

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECT	ION 13 OR 15(d) OF THE SECURITIES	S EXCHANGE ACT OF 1934
	For the quarterly period ended June 30,	2024
	OR	
□ TRANSITION REPORT PURSUANT TO SECT	TON 13 OR 15(d) OF THE SECURITIES	S EXCHANGE ACT OF 1934
	Commission File No. 1-11083	
BOSTO	N SCIENTIFIC CORPO	RATION
(Ex	xact name of registrant as specified in its cha	arter)
<u>Delaware</u>		<u>04-2695240</u>
(State or other jurisdiction of incorporation or or	ganization)	(I.R.S. Employer Identification No.)
	tific Way, Marlborough, Massachusetts s of Principal Executive Offices)	<u>01752-1234</u> (Zip Code)
	508 683-4000	
(Re	gistrant's telephone number, including area	code)
Securities registered pursuant to Section 12(b) of the Act:	:	
Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, par value \$0.01 per share	BSX	New York Stock Exchange
0.625% Senior Notes due 2027	BSX27	New York Stock Exchange
		13 or 15(d) of the Securities Exchange Act of 1934 during ts), and (2) has been subject to such filing requirements for
Indicate by check mark whether the registrant has submit Regulation S-T ($\S232.405$ of this chapter) during the prec \square No \square	, , , , , , , , , , , , , , , , , , ,	e required to be submitted pursuant to Rule 405 of d that the registrant was required to submit such files). Yes
Indicate by check mark whether the registrant is a large at emerging growth company. See the definitions of "large a		ccelerated filer, a smaller reporting company, or an er reporting company," and "emerging growth company"

Accelerated filer

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or

Smaller reporting company Emerging growth company **√**

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes \square No \square The number of shares outstanding of Common Stock, \$0.01 par value per share, as of July 30, 2024 was 1,472,626,135.

revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

TABLE OF CONTENTS

		Page No.
PART I	FINANCIAL INFORMATION	<u>3</u>
<u>ITEM 1.</u>	Consolidated Financial Statements	<u>3</u>
	Consolidated Statements of Operations (Unaudited)	<u>3</u>
	Consolidated Statements of Comprehensive Income (Loss) (Unaudited)	<u>4</u>
	Consolidated Balance Sheets (Unaudited)	<u>5</u>
	Consolidated Statements of Stockholders' Equity (Unaudited)	<u>6</u>
	Consolidated Statements of Cash Flows (Unaudited)	7
	Notes to the Consolidated Financial Statements (Unaudited)	9
<u>ITEM 2.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>37</u>
<u>ITEM 3.</u>	Quantitative and Qualitative Disclosures About Market Risk	<u>56</u>
<u>ITEM 4.</u>	Controls and Procedures	<u>57</u>
PART II	OTHER INFORMATION	<u>58</u>
<u>ITEM 1.</u>	<u>Legal Proceedings</u>	<u>58</u>
<u>ITEM 1A.</u>	Risk Factors	<u>58</u>
<u>ITEM 5.</u>	Other Information	<u>58</u>
<u>ITEM 6.</u>	Exhibits	<u>58</u>
<u>SIGNATURE</u>		<u>60</u>

PART I FINANCIAL INFORMATION

ITEM 1. CONSOLIDATED FINANCIAL STATEMENTS

BOSTON SCIENTIFIC CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF OPERATIONS (UNAUDITED)

		Three Moi Jun			Six Months Ended June 30,				
(in millions, except per share data)		2024		2023		2024		2023	
Net sales	\$	4,120	\$	3,599	\$	7,977	¢	6,988	
Cost of products sold	Ф	1,270	Ф	1,058	Ф	2,479	Ф	2,098	
Gross profit		2,850		2,542		5,498		4,891	
Operating expenses:									
Selling, general and administrative expenses		1,446		1,354		2,810		2,570	
Research and development expenses		383		359		749		695	
Royalty expense		9		12		19		23	
Amortization expense		213		210		427		412	
Intangible asset impairment charges		276		57		276		57	
Contingent consideration net expense (benefit)		2		19		18		31	
Restructuring net charges (credits)		1		16		5		36	
		2,330		2,028		4,303		3,825	
Operating income (loss)		520		514		1,195		1,066	
Other income (expense):									
Interest expense		(77)		(70)		(146)		(135)	
Other, net		(23)		(18)		(21)		(61)	
Income (loss) before income taxes		420		426		1,028		870	
Income tax expense (benefit)		98		156		213		287	
Net income (loss)		322		270		815		584	
Preferred stock dividends				(9)		_		(23)	
Net income (loss) attributable to noncontrolling interests		(2)		_		(4)			
Net income (loss) attributable to Boston Scientific common stockholders	\$	324	\$	261	\$	819	\$	561	
Net income (loss) per common share — basic	\$	0.22	\$	0.18	\$	0.56	\$	0.39	
Net income (loss) per common share — diluted	\$	0.22	\$	0.18	\$	0.55	\$	0.39	
Weighted-average shares outstanding									
Basic		1,470.6		1,446.2		1,469.5		1,441.0	
Diluted		1,484.2		1,456.2		1,483.0		1,451.1	

BOSTON SCIENTIFIC CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (LOSS) (UNAUDITED)

	Three Months Ended June 30,				Six Months Ended June 30,				
(in millions)		2024		2023		2024		2023	
Net income (loss)	\$	322	\$	270	\$	815	\$	584	
Other comprehensive income (loss), net of tax:									
Foreign currency translation adjustment		45		15		102		(28)	
Net change in derivative financial instruments		(16)		15		6		(28)	
Net change in defined benefit pensions and other items		0		0		0		(5)	
Other comprehensive income (loss)		30		30		108		(61)	
Comprehensive income (loss)	\$	352	\$	300	\$	924	\$	523	
Net income (loss) attributable to noncontrolling interests		(2)		_		(4)			
Other comprehensive income (loss) attributable to noncontrolling interests		(1)		_		(6)		_	
Comprehensive income (loss) attributable to noncontrolling interests		(4)		_		(10)		_	
Comprehensive income attributable to Boston Scientific common stockholders	\$	356	\$	300	\$	933	\$	523	

BOSTON SCIENTIFIC CORPORATION AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (UNAUDITED)

ASSETS Current accounts receivable, net 2,000 2,222 1,200 2,200		As of			
Current des Carb S 2,913 S 8 Cash and cash equivalents 2,400 2,222 Inventories 2,608 2,48 Prepad income taxes 315 31 Other current assets 8,991 6,51 Property, plant and equipment, net 2,951 2,85 Goodwill 5,417 6,00 Defer intangible assets, net 5,417 6,00 Defer transpile assets, net 1,551 3,301 3,34 Obber from great assets 1,551 1,533 1,535 1,535 1,535 1,535 1,535 1,535 1,535 1,535 3,343 3,344	(in millions, except share and per share data)		June 30, 2024		December 31, 2023
Cash and cash equivalents \$ 2,913 \$ 8.86 Trade accounts receivable, net 2,608 2,222 Inventories 2,608 2,48 Prepaid income taxes 313 313 Other current assets 756 62 Cotal current assets 8,991 6,51 Crodicyll 14,397 14,38 Condowll 14,397 6,00 Other intengible assets, net 3,801 3,801 Other long-term assets 3,801 3,801 Other long-term assets 3,801 3,53 OTAL ASSETS 3,501 3,53 Current abilities 3,501 3,53 Current debt obligations \$ 1,550 \$ 35,30 Accounts payable 906 94 Accounts payable 906 94 Other current liabilities 5,576 4,93 Long-term liabilities 5,576 4,93 Long-term liabilities 1,80 1,80 Long-term liabilities 1,80 1,80 <t< th=""><th>ASSETS</th><th></th><th></th><th></th><th></th></t<>	ASSETS				
Trade accounts receivable, net	Current assets:				
Prepaid income taxes 2,608 2,48 Prepaid income taxes 315 315 Other current assets 756 62 Other current assets 8,991 6,51 Property, plant and equipment, net 2,951 2,85 Goodwill 14,397 14,38 Other long-term assets 3,801 3,43 Other long-term assets 3,801 3,43 Other long-term assets 3,801 3,53 OTAL ASSETS 3,71,08 3,51,31 OTAL ASSETS 3,71,08 3,51,31 OTAL ASSETS 3,51,31	Cash and cash equivalents	\$	2,913	\$	865
Prepaid income taxes 315 31 Other current assets 8,91 6,51 Property, plant and equipment, net 2,951 2,85 Goodwill 14,397 14,38 Other intangible assets, net 5,417 6,00 Deferred tax assets 3,801 3,84 Other long-term assets 1,515 1,535 TOTAL ASSETS 3,7108 3,53,138 Current liabilities: Current debt obligations 1,580 5 3,53,33 Accounts payable 906 94 Account accounts payable 9,96 94 Account accounts payable 9,91 8,75 Other current liabilities 5,576 4,93 Long-term debt 8,991 8,77 Deferred income taxes 13,20 1,96 Commitments and contingencies Stockholders' equity Preferred stock, \$0,01 par value - authorized 50,000,000 shares - issued as of June 30, 2024 and December 31, 2023 1,7 1 Common sto	Trade accounts receivable, net		2,400		2,228
Other current assets 756 62 Cotal current assets 8,991 6,51 Copperty, plant and equipment, net 2,951 2,858 Goodwill 14,397 14,388 Other intangible assets, net 5,417 600 Other red tax assets 3,801 3,84 Other long-term assets 1,551 1,53 TOTAL ASSETS 5 37,108 \$ 35,33 CLARRILITIES AND STOCKHOLDERS' EQUITY Current liabilities Current liabilities Stock page 18 5,33 5 35,33 Accounts payable 906 94 4 4 4 4 4 96 94 4 4 96 1,580 \$ 5,33 8 5 3,33 1,580 \$ 3,33 2 2,44 4 6 0,49 4 4 96 94 4 4 96 6 94 94 96 6 94 94 94 96 6	Inventories		2,608		2,484
Total current assets 8,991 6,51 Property, plant and equipment, net 2,951 2,855 Condwill 14,397 14,387 Other intangible assets, net 5,417 6,000 Other long-term assets 3,801 3,840 Other long-term assets 1,551 1,533 TOTAL ASSETS 3,7108 5,371,000 Current labilities	Prepaid income taxes		315		315
Property, plant and equipment, net 2,951 2,858 Goodwill 14,397 14,389 Other intangible assets, net 5,417 6,000 Deferred tax assets 3,801 3,84 Other long-term assets 1,551 1,551 TOTAL ASSETS \$ 37,108 \$ 35,13 IABILITIES AND STOCKHOLDERS' EQUITY Current debt obligations \$ 1,580 \$ 35,33 Accounts payable 906 94 Accounts payable 906 94 Account et liabilities 770 81 Long-term debt 5,576 493 Long-term liabilities 132 13 Other long-term liabilities 1,80 1,96 Commitments and contingencies Steckholders' equity - - Preferred stock, \$0,01 par value - authorized \$0,000,000 shares - 0 shares issued as of June 30, 2024 and December 31, 2023 - - Common stock, \$0,01 par value - authorized \$2,000,000,000 shares - issued 1,734,329,744 shares as of June 30, 2024 and 1,729,000,224 shares as of December 31, 2023 1,7 1<	Other current assets		756		621
Goodwill 14,397 14,380 Other intangible assets, net 5,417 6,00 Deferred tax assets 3,801 3,84 Other long-term assets 1,551 1,53 TOTALASSETS \$ 37,108 \$ 35,13 LIABILITIES AND STOCKHOLDERS' EQUITY Current debt obligations \$ 1,580 \$ 53 Accounts payable 906 94 Accounts payable 906 94 Accounts payable 770 81 Other current liabilities 5,576 4,93 Long-term debt 8,991 8,70 Deferred income taxes 132 13 Other long-term liabilities 1,80 1,96 Commitments and contingencies Stockholders' equity - - Preferred stock, \$0,01 par value - authorized 50,000,000 shares - issued as of June 30, 2024 and December 31, 2023 1,23 1 Common stock, \$0,01 par value - authorized 2,000,000,000 shares - issued 1,734,329,744 shares as of June 30, 2024 and December 31, 2023 1,23 2,3	Total current assets		8,991		6,514
Other intangible assets, net 5,417 6,00 Defered tax assets 3,801 3,34 Other long-term assets 1,551 1,535 ITOTAL ASSETS \$ 37,108 \$ 35,136 ILIABILITIES AND STOCKHOLDERS' EQUITY Current labibilities: Current debt obligations \$ 1,580 \$ 5,53 Accounts payable 906 94 Accound expenses 2,320 2,64 Other current liabilities 70 81 Ical current liabilities 5,576 4,93 Long-term debt 8,991 8,57 Deferred income taxes 132 13 Other long-term liabilities 1,800 1,96 Commitments and contingencies Stockholders' equity \$ - \$ - Preferred stock, \$0.01 par value - authorized 50,000,000 shares - 0 shares issued as of June 30, 2024 and December 31, 2023 17 1 Common stock, \$0.01 par value - authorized 2,000,000,000 shares - issued 1,734,329,744 shares as of June 30, 2024 and December 31, 2023 17 1 Treasury stoc	Property, plant and equipment, net		2,951		2,859
Deferred tax assets 3,801 3,84 Other long-term assets 1,531 1,531 ITALASSETS S 37,108 S 35,131 ITALASSETS S 3,131 ITALASSETS S 3,131 ITALASSETS S 3,131 ITALASSETS S 3,131 ITALASSETS S 35,131 ITALASSETS S 5,131 ITALASSETS S 5,131 ITALASSETS S 1,158 S 5,331 ITALASSETS S 1,158 S 1,331 ITALASSETS S 1,158 S 1,158 ITALASSETS S 1,158 S 1,	Goodwill		14,397		14,387
1,55	Other intangible assets, net		5,417		6,003
Current labilities	Deferred tax assets		3,801		3,841
Current liabilities:	Other long-term assets		1,551		1,531
Current liabilities: S 1,580 \$ 53 Current debt obligations \$ 1,580 \$ 53 Accounts payable 906 94 Accrude expenses 2,320 2,646 Other current liabilities 770 81- Total current liabilities 5,576 4,93 Long-term debt 8,991 8,57 Deferred income taxes 132 13 Other long-term liabilities 1,800 1,96 Commitments and contingencies 5 4,93 Stockholders' equity Preferred stock, \$0.01 par value - authorized 50,000,000 shares - 0 shares issued as of June 30, 2024 and December 31, 2023 2 - Common stock, \$0.01 par value - authorized 2,000,000,000 shares - issued 1,734,329,744 shares as of June 30, 2024 and 1,7729,000,224 shares as of December 31, 2023 17 1 Treasury stock, at cost - 25a,289,848 shares as of June 30, 2024 and December 31, 2023 2,251 2,251 2,251 Additional paid-in capital 20,803 20,644 20,803 20,644 3 Retained earnings 1,639 81 4 <td>TOTAL ASSETS</td> <td>\$</td> <td>37,108</td> <td>\$</td> <td>35,136</td>	TOTAL ASSETS	\$	37,108	\$	35,136
Current liabilities: S 1,580 \$ 53 Current debt obligations \$ 1,580 \$ 53 Accounts payable 906 94 Accrude expenses 2,320 2,646 Other current liabilities 770 81- Total current liabilities 5,576 4,93 Long-term debt 8,991 8,57 Deferred income taxes 132 13 Other long-term liabilities 1,800 1,96 Commitments and contingencies 5 4,93 Stockholders' equity Preferred stock, \$0.01 par value - authorized 50,000,000 shares - 0 shares issued as of June 30, 2024 and December 31, 2023 2 - Common stock, \$0.01 par value - authorized 2,000,000,000 shares - issued 1,734,329,744 shares as of June 30, 2024 and 1,7729,000,224 shares as of December 31, 2023 17 1 Treasury stock, at cost - 25a,289,848 shares as of June 30, 2024 and December 31, 2023 2,251 2,251 2,251 Additional paid-in capital 20,803 20,644 20,803 20,644 3 Retained earnings 1,639 81 4 <td>LIARH ITIES AND STOCKHOLDERS' FOULTV</td> <td></td> <td></td> <td></td> <td></td>	LIARH ITIES AND STOCKHOLDERS' FOULTV				
Current debt obligations \$ 1,580 \$ 53 Accounts payable 906 944 Accrued expenses 2,320 2,644 Cother current liabilities 770 811 Total current liabilities 5,576 4,93 Long-term debt 8,991 8,57 Deferred income taxes 132 13 Other long-term liabilities 1,800 1,966 Commitments and contingencies					
Accounts payable 906 944 Accounts payable 2,320 2,644 Accounts payable 770 814 Cother current liabilities 770 814 Cother current liabilities 5,576 4,933 Long-term debt 8,991 8,57 Deferred income taxes 132 133 Other long-term liabilities 1,800 1,966 Commitments and contingencies Stockholders' equity Preferred stock, \$0.01 par value - authorized 50,000,000 shares - 0 shares issued as of June 30, 2024 and December 31, 2023 Common stock, \$0.01 par value - authorized 2,000,000,000 shares - issued 1,734,329,744 shares as of June 30, 2024 and 1,729,000,224 shares as of December 31, 2023 17 Treasury stock, at cost - 263,289,848 shares as of June 30, 2024 and December 31, 2023 20,803 20,647 Retained earnings 1,639 814 Accumulated other comprehensive income (loss), net of tax 164 447 Cotal stockholders' equity 20,371 19,285 Noncontrolling interests 238 246 Total equity 20,609 19,534		\$	1 580	\$	531
Accrued expenses 2,320 2,644 Other current liabilities 770 81 Total current liabilities 5,576 4,93 Long-term debt 8,991 8,57 Deferred income taxes 132 13 Other long-term liabilities 1,800 1,96 Commitments and contingencies - - Stockholders' equity - - - Preferred stock, \$0.01 par value - authorized 50,000,000 shares - 0 shares issued as of June 30, 2024 and December 31, 2023 - - - Common stock, \$0.01 par value - authorized 2,000,000,000 shares - issued 1,734,329,744 shares as of June 30, 2024 shares as of December 31, 2023 17 1' 1' Treasury stock, at cost - 263,289,848 shares as of June 30, 2024 and December 31, 2023 (2,251) (2,251) Additional paid-in capital 20,803 20,64' Retained earnings 1,639 81' Accumulated other comprehensive income (loss), net of tax 164 44' Total stockholders' equity 20,371 19,28' Noncontrolling interests 238 24' To	<u> </u>	Ψ		Ψ	942
Other current liabilities 770 81- Total current liabilities 5,576 4,93 Long-term debt 8,991 8,57 Deferred income taxes 132 13- Other long-term liabilities 1,800 1,966 Commitments and contingencies - - Stockholders' equity - - Preferred stock, \$0.01 par value - authorized 50,000,000 shares - 0 shares issued as of June 30, 2024 and December 31, 2023 - - Common stock, \$0.01 par value - authorized 2,000,000,000 shares - issued 1,734,329,744 shares as of June 30, 2024 and 1,729,000,224 shares as of December 31, 2023 17 1' Treasury stock, at cost - 263,289,848 shares as of June 30, 2024 and December 31, 2023 (2,251) (2,251) Additional paid-in capital 20,803 20,64 Accumulated other comprehensive income (loss), net of tax 164 44 Total stockholders' equity 20,371 19,28 Noncontrolling interests 238 24 Total equity 20,609 19,534					2,646
Total current liabilities					814
Stockholders' equity				_	4,933
Deferred income taxes 132 13-					· · · · · · · · · · · · · · · · · · ·
Other long-term liabilities 1,800 1,960 Commitments and contingencies Stockholders' equity Preferred stock, \$0.01 par value - authorized 50,000,000 shares - 0 shares issued as of June 30, 2024 and December 31, 2023 — — — — — Common stock, \$0.01 par value - authorized 2,000,000,000 shares - issued 1,734,329,744 shares as of June 30, 2024 and 1,729,000,224 shares as of December 31, 2023 17 1° 1° 1° 1° 1° 1° 1° 1° 1° 1° 1° 1° 2° 2 2 2 2 5 2 2 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 5 2 3 2 4 <td>· ·</td> <td></td> <td></td> <td></td> <td>134</td>	· ·				134
Stockholders' equity Preferred stock, \$0.01 par value - authorized 50,000,000 shares - 0 shares issued as of June 30, 2024 and December 31, 2023 — — — — — — — — — — — — — — — — — —	Other long-term liabilities				1,967
Stockholders' equity Preferred stock, \$0.01 par value - authorized 50,000,000 shares - 0 shares issued as of June 30, 2024 and December 31, 2023 — — — — — — — — — — — — — — — — — —					
Preferred stock, \$0.01 par value - authorized 50,000,000 shares - 0 shares issued as of June 30, 2024 and December 31, 2023 — 1 1 1 1 1 1 1 1 1 2 2 2 1 3 2 2 8 1 3 2 3 2 3	Commitments and contingencies				
December 31, 2023	Stockholders' equity				
2024 and 1,729,000,224 shares as of December 31, 2023 17 17 Treasury stock, at cost - 263,289,848 shares as of June 30, 2024 and December 31, 2023 (2,251) (2,251) Additional paid-in capital 20,803 20,64* Retained earnings 1,639 819 Accumulated other comprehensive income (loss), net of tax 164 44 Total stockholders' equity 20,371 19,28* Noncontrolling interests 238 24 Total equity 20,609 19,530			_		_
Additional paid-in capital 20,803 20,644 Retained earnings 1,639 819 Accumulated other comprehensive income (loss), net of tax 164 44 Total stockholders' equity 20,371 19,28 Noncontrolling interests 238 24 Total equity 20,609 19,530	Common stock, $\$0.01$ par value - authorized $2,000,000,000$ shares - issued $1,734,329,744$ shares as of June 302024 and $1,729,000,224$ shares as of December $31,2023$,	17		17
Retained earnings 1,639 819 Accumulated other comprehensive income (loss), net of tax 164 44 Total stockholders' equity 20,371 19,280 Noncontrolling interests 238 24 Total equity 20,609 19,530	Treasury stock, at cost - 263,289,848 shares as of June 30, 2024 and December 31, 2023		(2,251)		(2,251)
Retained earnings 1,639 819 Accumulated other comprehensive income (loss), net of tax 164 44 Total stockholders' equity 20,371 19,28 Noncontrolling interests 238 24 Total equity 20,609 19,530	Additional paid-in capital				20,647
Accumulated other comprehensive income (loss), net of tax 164 49 Total stockholders' equity 20,371 19,28 Noncontrolling interests 238 24 Total equity 20,609 19,530	Retained earnings		1,639		819
Noncontrolling interests 238 24 Total equity 20,609 19,530	Accumulated other comprehensive income (loss), net of tax		164		49
Noncontrolling interests 238 24 Total equity 20,609 19,530	Total stockholders' equity		20,371		19,282
Total equity 20,609 19,530	Noncontrolling interests				248
	-		20,609		19,530
		\$	37,108	\$	35,136

BOSTON SCIENTIFIC CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (UNAUDITED)

		Three Months Ended June 30,				Six Months Ended June 30,			
(in millions, except share data)		2024		2023		2024		2023	
Preferred stock shares issued									
Beginning		_		10,062,500		_		10,062,500	
Conversion of mandatory convertible preferred stock to common stock		_		(10,062,500)		_		(10,062,500)	
Ending		_							
Common stock shares issued									
Beginning		1,733,293,885		1,700,828,873		1,729,000,224		1,696,633,993	
Impact of stock-based compensation plans		1,035,859		1,144,366		5,329,520		5,339,246	
Conversion of mandatory convertible preferred stock to common stock				23,982,902				23,982,902	
Ending		1,734,329,744		1,725,956,141		1,734,329,744		1,725,956,141	
Preferred stock									
Beginning	\$	_	\$	0	\$	_	\$	0	
Conversion of mandatory convertible preferred stock to common stock		<u> </u>		(0)		_		(0)	
Ending	\$		\$		\$		\$		
Common stock				,		,			
Beginning	\$	17	\$	17	\$	17	\$	17	
Impact of stock-based compensation plans		0		0		0		0	
Conversion of mandatory convertible preferred stock to common stock				0				0	
Ending	\$	17	\$	17	\$	17	\$	17	
Treasury stock									
Beginning	\$	(2,251)	\$	(2,251)	\$	(2,251)	\$	(2,251)	
Repurchase of common stock								_	
Ending	\$	(2,251)	\$	(2,251)	\$	(2,251)	\$	(2,251)	
Additional paid-in capital									
Beginning	\$	20,713	\$	20,356	\$	20,647	\$	20,289	
Impact of stock-based compensation plans		90		85		156		153	
Ending	\$	20,803	\$	20,441	\$	20,803	\$	20,441	
Retained earnings/(Accumulated deficit)	6	1 214	e e	(450)	ď.	010	e	(750)	
Beginning National (last)	\$	1,314	\$	(450)	\$	819	\$	(750)	
Net income (loss) Net (income) loss attributable to noncontrolling interests		322 2		270		815 4		584	
Preferred stock dividends		2		(9)		4		(23)	
	<u>\$</u>	1,639	\$	(189)	\$	1,639	\$	(189)	
Ending Accumulated other comprehensive income (loss), net of tax	3	1,037	J.	(103)	Þ	1,039	.	(103)	
Beginning	\$	132	\$	178	\$	49	\$	269	
Changes in other comprehensive income (loss)	J	31	Φ	30	φ	114	φ	(61)	
Ending	\$	164	\$	208	\$	164	\$	208	
Total stockholders' equity	<u>s</u>	20,371	\$	18,226	\$	20,371	\$	18,226	
Total stockholders equity	3	20,371	J	18,220	J	20,371	J	10,220	
Noncontrolling interests									
Beginning	\$	242	\$	259	\$	248	\$	_	
Net income (loss) attributable to noncontrolling interests		(2)		_		(4)		_	
Changes in other comprehensive income (loss)		(1)		_		(6)		250	
Changes to noncontrolling ownership interest	•	220	•	250	•	220	•	259	
Ending	\$	238	\$	259	\$	238	\$	259	
Total equity	\$	20,609	\$	18,485	\$	20,609	\$	18,485	

BOSTON SCIENTIFIC CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)

	Six Months E	nded June 30,
(in millions)	2024	2023
Net income (loss)	\$ 815	\$ 584
Adjustments to reconcile net income (loss) to cash provided by (used for) operating activities		
Depreciation and amortization	615	583
Deferred and prepaid income taxes	21	(45)
Stock-based compensation expense	129	115
Goodwill and other intangible asset impairment charges	276	57
Net loss (gain) on investments and notes receivable	46	35
Contingent consideration net expense (benefit)	18	31
Inventory step-up amortization	_	3
Other, net	8	23
Increase (decrease) in operating assets and liabilities, excluding purchase accounting:		
Trade accounts receivable	(228)	(161)
Inventories	(219)	(399)
Other assets	(126)	(55)
Accounts payable, accrued expenses and other liabilities	(379)	78
Cash provided by (used for) operating activities	977	848
Investing activities:		
Purchases of property, plant and equipment and internal use software	(334)	(254)
Proceeds from sale of property, plant and equipment	1	3
Payments for acquisitions of businesses, net of cash acquired	(95)	(1,018)
Payments for investments and acquisitions of certain technologies, net of investment proceeds	(139)	(73)
Proceeds from royalty rights	11	16
Proceeds from settlements of hedge contracts	<u> </u>	2
Cash provided by (used for) investing activities	(556)	(1,324)
Financing activities:		
Payment of contingent consideration previously established in purchase accounting	(34)	(39)
Payments for royalty rights	(15)	(34)
Payments for finance leases	(25)	(e i)
Payments on short-term borrowings	(504)	_
Net increase (decrease) in commercial paper	_	37
Proceeds from long-term borrowings, net of debt issuance costs	2,145	_
Cash dividends paid on preferred stock		(28)
Cash used to net share settle employee equity awards	(80)	(52)
Proceeds from issuances of common stock pursuant to employee stock compensation and purchase plans	106	90
Cash provided by (used for) financing activities	1,593	(26)
	(8)	(2)
Effect of foreign exchange rates on cash	(8)	(3)
Net increase (decrease) in cash, cash equivalents, restricted cash and restricted cash equivalents	2,006	(506)
Cash, cash equivalents, restricted cash and restricted cash equivalents at beginning of period	1,055	1,126
Cash, cash equivalents, restricted cash and restricted cash equivalents at end of period	\$ 3,062	\$ 620

BOSTON SCIENTIFIC CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (SUPPLEMENTAL INFORMATION)

	Six I	Months Ende	d June 30,
(in millions)	2	024	2023
Supplemental Information			
Stock-based compensation expense	\$	129 \$	115
Non-cash impact of transferred royalty rights		(11)	(16)

		As of J	une 30,	
Reconciliation to amounts within the unaudited consolidated balance sheets:	·	2024		2023
Cash and cash equivalents	\$	2,913	\$	426
Restricted cash and restricted cash equivalents included in Other current assets		75		135
Restricted cash equivalents included in Other long-term assets		74		60
Cash, cash equivalents, restricted cash and restricted cash equivalents at end of period	\$	3,062	\$	620

NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

NOTE A – BASIS OF PRESENTATION

The accompanying unaudited consolidated financial statements of Boston Scientific Corporation have been prepared in accordance with accounting principles generally accepted in the United States (GAAP) and with the instructions to Form 10-Q and Article 10 of Regulation S-X, and they do not include all of the information and footnotes required by GAAP for complete financial statements. When used in this report, the terms, "we," "us," "our," and "the Company" mean Boston Scientific Corporation and its divisions and subsidiaries. In the opinion of management, all adjustments (consisting only of normal recurring adjustments) considered necessary for fair presentation have been included. Operating results for the three and six months ended June 30, 2024 are not necessarily indicative of the results that may be expected for the year ending December 31, 2024. Accordingly, our unaudited consolidated financial statements and footnotes thereto should be read in conjunction with our audited consolidated financial statements and footnotes thereto included in Item 8 of our most recent Annual Report on Form 10-K.

The accompanying unaudited consolidated financial statements include the accounts of the Company's wholly owned- subsidiaries and entities for which the Company has a controlling financial interest. All intercompany balances and transactions have been eliminated in consolidation.

Amounts reported in millions within this Quarterly Report on Form 10-Q are computed based on the amounts in thousands. As a result, the sum of the components may not equal the total amount reported in millions due to rounding. Certain columns and rows within tables may not add due to the use of rounded numbers. Percentages presented are calculated from the underlying unrounded amounts.

Subsequent Events

We evaluate events occurring after the date of our accompanying unaudited consolidated balance sheet for potential recognition or disclosure within our financial statements. Those items requiring recognition in the financial statements have been recorded and disclosed accordingly.

Those items requiring disclosure (non-recognized subsequent events) in the financial statements have been disclosed accordingly. Refer to $Note\ H-Commitments\ and\ Contingencies$ for further details.

NOTE B – ACQUISITIONS AND STRATEGIC INVESTMENTS

Our accompanying unaudited consolidated financial statements include the operating results for acquired entities from the respective dates of acquisition. We have not presented supplemental pro forma financial information for completed acquisitions or divestitures given their results are not material to our accompanying unaudited consolidated financial statements. Further, transaction costs were immaterial to our accompanying unaudited consolidated financial statements and were expensed as incurred.

On June 18, 2024, we announced our entry into a definitive agreement to acquire 100 percent of Silk Road Medical, Inc. (Silk Road Medical), a publicly traded medical device company that has developed an innovative platform of products to prevent stroke in patients with carotid artery disease through a minimally invasive procedure called transcarotid artery revascularization (TCAR). The purchase price is \$27.50 in cash per share, or approximately \$1.260 billion for 100% of the fully diluted equity. The transaction is expected to be completed in the second half of 2024, subject to customary closing conditions. The Silk Road Medical business will be integrated into our Peripheral Interventions division.

On January 8, 2024, we announced our entry into a definitive agreement to acquire 100 percent of Axonics, Inc. (Axonics), a publicly traded medical technology company primarily focused on the development and commercialization of devices to treat urinary and bowel dysfunction. The purchase price is \$71.00 in cash per share, or approximately \$3.670 billion for 100% of the fully diluted equity. On April 3, 2024, we and Axonics each received a request for additional information (Second Request) from the United States Federal Trade Commission (FTC) in connection with the FTC's review of the transaction. The issuance of the Second Request extends the waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (HSR Act), until 30 days after both we and Axonics have substantially complied with the Second Request, unless the waiting period is extended voluntarily by the parties or terminated earlier by the FTC. We and Axonics are responding to the Second Request and continue to work cooperatively with the FTC in its review. The transaction is expected to be completed in the second half of 2024, subject to the expiration or termination of the waiting period under the HSR Act and the satisfaction (or waiver) of other customary closing conditions. The Axonics business will be integrated into our Urology division.

2023 Acquisitions

On April 4, 2023, we completed our acquisition of 100 percent of the outstanding equity of Apollo Endosurgery, Inc. (Apollo), a public company which offers a portfolio of devices used during endoluminal procedures to close gastrointestinal defects, manage gastrointestinal complications and aid in weight loss for patients suffering from obesity. The transaction consisted of an upfront cash payment of \$636 million, net of cash acquired. The Apollo business is being integrated into our Endoscopy division.

On February 20, 2023, we completed the acquisition of a majority stake investment in Acotec Scientific Holdings Limited (Acotec), a publicly traded Chinese manufacturer of drug-coated balloons and other products used in the treatment of vascular and other diseases. We consolidated this majority stake investment in Acotec based on the conclusion we control the entity, and recorded a noncontrolling interest for the portion we do not own. We acquired approximately 65 percent of the outstanding shares of Acotec, for an upfront cash payment of HK\$20.00 per share, or \$519 million at foreign currency exchange rates at closing. The Acotec portfolio complements our existing Peripheral Interventions portfolio.

Purchase Price Allocation

We accounted for these transactions as business combinations in accordance with Financial Accounting Standards Board (FASB) Accounting Standards Codification (ASC) Topic 805, Business Combinations (FASB ASC Topic 805). The final purchase prices were comprised of the amounts presented below:

(in millions)	A	.cotec ⁽¹⁾	Apollo
Payment for acquisition, net of cash acquired ⁽²⁾	\$	381 \$	636
	\$	381 \$	636

⁽¹⁾ Excludes approximately \$140 million of cash on hand at the closing of the transaction

We recorded the assets acquired, liabilities assumed and specific to Acotec, the noncontrolling interest, at their respective fair values as of the closing date of the transaction. The final purchase price allocations were comprised of the components presented below, with the excess of the purchase price over the fair value of net assets acquired recorded to goodwill:

(in millions)	Acotec	Apollo
Goodwill	\$ 337 \$	378
Amortizable intangible assets	334	248
Other assets acquired	93	50
Liabilities assumed	(48)	(33)
Net deferred tax liabilities	(76)	(5)
Fair value of noncontrolling interest	(259)	_
	\$ 381 \$	636

The fair value of Acotec's noncontrolling interest was based on the publicly traded market value of the remaining 35 percent of the outstanding shares we did not acquire as of the transaction date and is presented in *Stockholders' equity* within our accompanying unaudited consolidated balance sheets. Goodwill was primarily established for Acotec due to opportunities for collaboration in research and development, manufacturing and commercial strategies, and for Apollo, due to synergies expected to be gained from leveraging our existing operations, as well as revenue and cash flow projections associated with future technologies, none of which is deductible for tax purposes.

⁽²⁾ Related to Acotec, represents our majority stake investment

We allocated a portion of the purchase price to the specific intangible asset categories as follows:

		Amount A		Weighted Average Amortization Period (in years)	Risk-Adjusted Discount Rates used in Purchase Price Allocation
Acotec:					
Amortizable intangible assets:					
Technology-related	5	S	308	11	14%
Customer relationships			15	11	14%
Other intangible assets	_		11	13	14%
	5	3	334		
Apollo:					
Amortizable intangible assets:					
Technology-related	9	S	222	11	12%
Customer relationships			26	11	12%
	5	3	248		

Contingent Consideration

Changes in the fair value of our contingent consideration liability during the first six months of 2024 associated with prior period acquisitions were as follows:

(in millions)

(III IIIIIIOII)	
Balance as of December 31, 2023	\$ 404
Amount recorded related to current year acquisitions	29
Contingent consideration net expense (benefit)	18
Contingent consideration payments and other adjustments	(158)
Balance as of June 30, 2024	\$ 293

The payments made during the first six months of 2024 primarily related to our acquisition of Farapulse, Inc. (Farapulse) following the achievement of revenue-based earnouts as a result of over-performance. The maximum amount we could be required to pay for certain contingent consideration is not determinable as it is uncapped and based on a percent of certain sales. As of June 30, 2024, the fair value of such uncapped contingent consideration is estimated at \$163 million. As of June 30, 2024, the maximum amount that we could be required to pay under our other contingent consideration arrangements (undiscounted) is approximately \$320 million. Refer to *Note B – Acquisitions and Strategic Investments* to our audited financial statements contained in Item 8. Financial Statements and Supplementary Data of our most recent Annual Report on Form 10-K for additional information.

The recurring Level 3 fair value measurements of our contingent consideration liability that we expect to be required to settle include the following significant unobservable inputs:

Contingent Consideration Liability	Fair Value as of June 30, 2024 Valuation Technique Unobservable Input]	Range	Weighted Average ⁽¹⁾	
			Discount Rate	6%	- 15%	7%
Revenue-based Payments and Milestones	\$293 million	Discounted Cash Flow	Probability of Payment	90%	- 100%	99%
		Cush i low	Projected Year of Payment	2024	- 2029	2027

⁽¹⁾ Unobservable inputs were weighted by the relative fair value of the contingent consideration liability. For projected year of payment, the amount represents the median of the inputs and is not a weighted average.

Projected contingent payment amounts related to our revenue-based payments and milestones are discounted back to the current period, primarily using a discounted cash flow model. Significant increases or decreases in projected revenues, probabilities of payment, discount rates or the time until payment is made would have resulted in a significantly lower or higher fair value measurement as of June 30, 2024.

Strategic Investments

The aggregate carrying amount of our strategic investments was comprised of the following:

		As of						
(in millions)	June 3	30, 2024	December 31, 2023					
Equity method investments	\$	207 \$	219					
Measurement alternative investments ^(1, 2)		243	194					
	\$	450 \$	413					

⁽¹⁾ Measurement alternative investments are privately-held equity securities without readily determinable fair values that are measured at cost less impairment, if any, adjusted to fair value for any observable price changes in orderly transactions for the identical or a similar investment of the same issuer, recognized in *Other, net* within our accompanying unaudited consolidated statements of operations.

These investments are classified as *Other long-term assets* within our accompanying unaudited consolidated balance sheets, in accordance with GAAP and our accounting policies.

As of June 30, 2024, the cost of our aggregated equity method investments exceeded our share of the underlying equity in net assets by \$233 million, which represents amortizable intangible assets, in-process research and development (IPR&D), goodwill and deferred tax liabilities.

NOTE C - GOODWILL AND OTHER INTANGIBLE ASSETS

The gross carrying amount of goodwill and other intangible assets and the related accumulated amortization for intangible assets subject to amortization and accumulated goodwill impairment charges are as follows:

	 As of Jui	1e 3	As of December 31, 2023				
(in millions)	Gross Carrying Amount		Accumulated Amortization/ Write-offs		Gross Carrying Amount		Accumulated Amortization/ Write-offs
Technology-related	\$ 12,846	\$	(8,372)	\$	13,207	\$	(8,101)
Patents	476		(380)		480		(387)
Other intangible assets	2,188		(1,555)		2,130		(1,500)
Amortizable intangible assets	\$ 15,510	\$	(10,307)	\$	15,817	\$	(9,988)
Goodwill	\$ 24,297	\$	(9,900)	\$	24,287	\$	(9,900)
IPR&D	\$ 94			\$	54		
Technology-related	120				120		
Indefinite-lived intangible assets	\$ 214			\$	174		

⁽²⁾ Includes publicly-held securities and convertible notes measured at fair value with changes in fair value recognized in *Other*; net within our accompanying unaudited consolidated statements of operations.

The following represents a roll-forward of our goodwill balance by reportable segment:

(in millions)		MedSurg	Card	iovascular	Total	
As of December 31, 2023	\$	5,347	\$	9,041	\$	14,387
Goodwill acquired		24		46		71
Impact of foreign currency fluctuations and purchase price and other adjustments		(30)		(30)		(61)
As of June 30, 2024	\$	5,341	\$	9,057	\$	14,397

Goodwill and Other Intangible Asset Impairments

We did not record any goodwill impairment charges in the first six months of 2024 or 2023. We test our goodwill balances in the second quarter of each year as of April 1 for impairment, or more frequently if impairment indicators are present or changes in circumstances suggest an impairment may exist.

We assess goodwill for impairment at the reporting unit level, which is defined as an operating segment or one level below an operating segment, referred to as a component. We identified the following reporting units for purposes of our annual goodwill impairment test: Interventional Cardiology, Rhythm Management, Peripheral Interventions, Endoscopy, Urology and Neuromodulation. Based on the criteria prescribed in FASB ASC Topic 350, *Intangibles - Goodwill and Other* (FASB ASC Topic 350), we aggregated the Interventional Cardiology Therapies and Watchman components of our Cardiology operating segment into a single Interventional Cardiology reporting unit and aggregated the Cardiac Rhythm Management and Electrophysiology components of our Cardiology operating segment into a single Rhythm Management reporting unit.

In the second quarter of 2024, we performed our annual goodwill impairment test utilizing both the qualitative and quantitative approach described in FASB ASC Topic 350. The qualitative approach was used for testing certain reporting units where fair value has historically exceeded carrying value by greater than 100 percent, and all other reporting units were tested using the quantitative approach. For the reporting units tested using the qualitative approach, after assessing the totality of events, it was determined that it was not more likely than not that the fair value of the reporting units was less than their carrying value, and it was not deemed necessary to proceed to the quantitative test. For the reporting units tested using the quantitative approach, we determined that the fair value of the reporting units exceeded the carrying value and concluded that goodwill was not impaired or at risk of impairment.

We recorded *Intangible asset impairment charges* of \$276 million in the second quarter and first six months of 2024 and \$57 million in the second quarter and first six months of 2023. The impairment charges recorded in 2024 were associated with amortizable intangible assets established in connection with our acquisitions of Cryterion Medical, Inc. (Cryterion) and Devoro Medical, Inc. (Devoro), which were integrated into our Electrophysiology and Peripheral Interventions business units, respectively. Intangible assets acquired from Cryterion were impaired due to strong commercial adoption of our FarapulseTM Pulsed Field Ablation System and the resulting lower revenue projections and cannibalization of our cryoablation business in major markets like the U.S. Intangible assets acquired from Devoro were impaired following management's decision to cancel the related program in the second quarter of 2024. We calculated the fair value of our Cryterion and Devoro intangible assets as the present value of estimated future cash flows we expect to generate from the assets based on estimates and assumptions about future revenue contributions, cost structures and the remaining useful lives of the assets.

We review intangible assets subject to amortization quarterly to determine if any adverse conditions exist or a change in circumstances has occurred that would indicate impairment or a change in the remaining useful life. Conditions that may indicate impairment include, but are not limited to, a significant adverse change in legal factors or business climate that could affect the value of an asset, a product recall or an adverse action or assessment by a regulator. If we determine it is more likely than not that the asset is impaired based on our qualitative assessment of impairment indicators, we test the intangible asset for recoverability. If the carrying value of the intangible asset or asset group exceeds the undiscounted cash flows expected to result from the use and eventual disposition of the intangible asset or asset group, we will write the carrying value down to fair value in the period impairment is identified. We test our indefinite-lived intangible assets at least annually during the third quarter for impairment and reassess their classification as indefinite-lived assets. In addition, we review our indefinite-lived intangible assets for classification and impairment more frequently if impairment indicators exist.

Refer to Note A - Significant Accounting Policies to our audited financial statements contained in Item 8. Financial Statements

and Supplementary Data of our most recent Annual Report on Form 10-K for further discussion of our annual goodwill and intangible asset impairment testing.

NOTE D - HEDGING ACTIVITIES AND FAIR VALUE MEASUREMENTS

Derivative Instruments and Hedging Activities

We address market risk from changes in foreign currency exchange rates and interest rates through risk management programs which include the use of derivative and nonderivative financial instruments. We manage concentration of counterparty credit risk by limiting acceptable counterparties to major financial institutions with investment grade credit ratings, limiting the amount of credit exposure to individual counterparties and by actively monitoring counterparty credit ratings and the amount of individual credit exposure. We also employ master netting arrangements that limit the risk of counterparty non-payment on a particular settlement date to the net gain that would have otherwise been received from the counterparty. Although not completely eliminated, we do not consider the risk of counterparty default to be significant as a result of these protections. Further, none of our derivative instruments are subject to collateral or other security arrangements, nor do they contain provisions that are dependent on our credit ratings from any credit rating agency.

<u>Currency Hedging Instruments</u>

Risk Management Strategy

Our risk from changes in currency exchange rates consists primarily of monetary assets and liabilities; forecasted intercompany and third-party transactions; and net investments in certain subsidiaries. We manage currency exchange rate risk at a consolidated level to reduce the cost of hedging by taking advantage of offsetting transactions. We employ derivative and nonderivative instruments, primarily forward currency contracts, to reduce the risk to our earnings and cash flows associated with changes in currency exchange rates.

The success of our currency risk management program depends, in part, on forecasted transactions denominated primarily in euro, Chinese renminbi, Japanese yen, British pound sterling, Australian dollar and Swiss franc. We may experience unanticipated currency exchange gains or losses to the extent the actual activity is different than forecasted. In addition, changes in currency exchange rates related to any unhedged transactions may impact our earnings and cash flows.

Hedge Designations and Relationships

Certain of our currency derivative instruments are designated as cash flow hedges under FASB ASC Topic 815, Derivatives and Hedging (FASB ASC Topic 815), and are intended to protect the U.S. dollar value of forecasted transactions. The gain or loss on a derivative instrument designated as a cash flow hedge is recorded in the Net change in derivative financial instruments component of Other comprehensive income (loss), net of tax (OCI) within our unaudited consolidated statements of comprehensive income (loss) until the underlying third-party transaction occurs. When the underlying third-party transaction occurs, we recognize the gain or loss in earnings within Cost of products sold within our unaudited consolidated statements of operations. In the event the hedging relationship is no longer effective, or if the occurrence of the hedged forecast transaction becomes no longer probable, we reclassify the gains or losses within Accumulated other comprehensive income (loss), net of tax (AOCI) to earnings at that time. The cash flows related to the derivative instruments designated as cash flow hedges are reported as operating activities within our unaudited consolidated statements of cash flows.

We also designate certain forward currency contracts as net investment hedges to hedge a portion of our net investments in certain of our entities with functional currencies denominated in the Chinese renminbi and Japanese yen. For these derivative instruments, we elected to use the spot method to assess hedge effectiveness. We also elected to exclude the spot-forward difference, referred to as the excluded component, from the assessment of hedge effectiveness and are amortizing this amount separately, as calculated at the date of designation, on a straight-line basis over the term of the currency forward contracts. As such, we defer recognition of foreign currency gains and losses within the *Foreign currency translation adjustment* (CTA) component of OCI, and we reclassify amortization of the excluded component from AOCI to current period earnings within *Interest expense* within our unaudited consolidated statements of operations.

We designate certain euro-denominated debt as net investment hedges to hedge a portion of our net investments in certain of our entities with functional currencies denominated in the euro. As of June 30, 2024 and December 31, 2023, we designated as a net investment hedge our €900 million in aggregate principal amount of 0.625% senior notes issued in November 2019 and due in 2027 (2027 Notes). For these nonderivative instruments, we defer recognition of the foreign currency remeasurement gains and losses within the CTA component of OCI. We reclassify these gains and losses to current period earnings within

Other, net within our accompanying unaudited consolidated statements of operations only when the hedged item affects earnings, which would occur upon disposal or substantial liquidation of the underlying foreign subsidiary.

We also use forward currency contracts that are not part of designated hedging relationships as a part of our strategy to manage our exposure to currency exchange rate risk related to monetary assets and liabilities and related forecast transactions. These non-designated currency forward contracts have an original time to maturity consistent with the hedged currency transaction exposures, generally less than one year, and are marked-to-market with changes in fair value recorded to earnings within *Other*; net within our accompanying unaudited consolidated statements of operations.

Interest Rate Hedging Instruments

Risk Management Strategy

Our interest rate risk relates primarily to U.S. dollar and euro-denominated borrowings partially offset by U.S. dollar cash investments. We use interest rate derivative instruments to mitigate the risk to our earnings and cash flows associated with exposure to changes in interest rates. Under these agreements, we and the counterparty, at specified intervals, exchange the difference between fixed and floating interest amounts calculated by reference to an agreed-upon notional principal amount. We designate these derivative instruments either as fair value or cash flow hedges in accordance with FASB ASC Topic 815.

Hedge Designations and Relationships

We had no interest rate derivative instruments designated as cash flow hedges outstanding as of June 30, 2024 or December 31, 2023. In the event that we designate outstanding interest rate derivative instruments as cash flow hedges, we record the changes in the fair value of the derivatives within OCI until the underlying hedged transaction occurs.

We had no interest rate derivative instruments designated as fair value hedges outstanding as of June 30, 2024 or December 31, 2023. In the event that we designate outstanding interest rate derivative instruments as fair value hedges, we record the changes in the fair values of interest-rate derivatives designated as fair value hedges and of the underlying hedged debt instruments in *Interest expense*, which generally offset.

The following table presents the contractual amounts of our hedging instruments outstanding:

		As of							
(in millions)	FASB ASC Topic 815 Designation		June 30, 2024		December 31, 2023				
Forward currency contracts	Cash flow hedge	\$	1,969	\$	2,284				
Forward currency contracts	Net investment hedge		645		333				
Foreign currency-denominated debt(1)	Net investment hedge		997		997				
Forward currency contracts	Non-designated		3,284		3,282				
Total Notional Outstanding		\$	6,895	\$	6,896				

⁽¹⁾ Foreign currency-denominated debt is the €900 million debt principal associated with our 2027 Notes designated as a net investment hedge.

The remaining time to maturity as of June 30, 2024 is within 36 months for all forward currency contracts designated as cash flow hedges and generally less than one year for all non-designated forward currency contracts. The forward currency contracts designated as net investment hedges generally mature between one and two years. The euro-denominated debt principal designated as a net investment hedge has a contractual maturity of December 1, 2027.

The following presents the effect of our derivative and nonderivative instruments designated as cash flow and net investment hedges under FASB ASC Topic 815 within our accompanying unaudited consolidated statements of operations. Refer to *Note M* – *Changes in Other Comprehensive Income* for the total amounts relating to derivative and nonderivative instruments presented within our accompanying unaudited consolidated statements of comprehensive income (loss).

Effect of Hodging	Dolotionshins on	A commulated Other	· Comprehensive Income	_
Effect of Heaging	- Keiationsnibs on	Accumulated Other	· Comprenensive income	e

	Amou	nt Rec	ognized in OC	on Hedges	Unaudited Consolidate Operation	ments of	Amount Reclassified from AOCI into Earnings				
(in millions)	Pre-Tax (Loss		Tax Benefit (Expense)	Gain (Loss) Net of Tax		Location of Amount Reclassified and Total Amount of Line Item			-Tax 1) Loss	Tax (Benefit) Expense	(Gain) Loss Net of Tax
				Three Mont	hs Ended June 30, 2024						
Forward currency contracts											
Cash flow hedges	\$	31	\mathfrak{S} (7)	\$ 24	Cost of products sold	\$	1,270	\$	(52) \$	12	\$ (41)
Net investment hedges ⁽²⁾		20	(5)	16	Interest expense		77		(4)	1	(3)
Foreign currency-denominated	d debt										
Net investment hedges ⁽³⁾		9	(2)	7	Other, net		23		_	_	_
Interest rate derivative contra	cts										
Cash flow hedges		_	_	_	Interest expense		77		0	(0)	0

Effect of Hodging Deletionships on A	Accumulated Other Comprehensive Income
RITECT OF HEAGING REISTIONSHINS ON A	Accilmiliated Other Comprehensive Income

	An	ount Rec	ognized in OC	I on Hedges	Unaudited Consolidat Operation		ments of		Amount Reclassified from AOCI into Earnings				
(in millions)		Cax Gain Loss)	Tax Benefit (Expense)	Gain (Loss) Net of Tax	Location of Amount I Total Amount of		Pre-Tax (Gain) Loss		Fax (Benefit) Expense	(Gain) Loss Net of Tax			
				Three Mon	ths Ended June 30, 2023								
Forward currency contracts													
Cash flow hedges	\$	74 \$	(17)	\$ 58	Cost of products sold	\$	1,058	\$	(55) \$	12	\$ (43)		
Net investment hedges ⁽²⁾		22	(5)	17	Interest expense		70		(2)	1	(2)		
Foreign currency-denominate	d debt												
Net investment hedges ⁽³⁾		1	_	1	Other, net		18		_	_	_		
Interest rate derivative contra	cts												
Cash flow hedges		_	_	_	Interest Expense		70		1	(0)	1		

Effect of Hedging Relationships on Accumulated Other C	Comprehensive Income
--	----------------------

					I								
	Amount	Rec	ognized in OCI	on Hedges	Unaudited Consolidated Statements of Operations ⁽¹⁾				Amount Reclassified from AOCI into Earnings				
(in millions)	Pre-Tax G (Loss)	ain	Tax Benefit (Expense)	Gain (Loss) Net of Tax	Location of Amount Ro Total Amount of I			(Pre-Tax Gain) Loss	Tax (Benefit) Expense	(Gain) Loss Net of Tax		
				Six Months	s Ended June 30, 2024								
Forward currency contracts													
Cash flow hedges	\$ 1	08 \$	(24) \$	84	Cost of products sold	\$	2,479	\$	(101) \$	23	\$ (79)		
Net investment hedges ⁽²⁾		48	(11)	37	Interest expense		146		(8)	2	(6)		
Foreign currency-denominate	d debt												
Net investment hedges ⁽³⁾		32	(7)	25	Other, net		21		_	_	_		
Interest rate derivative contra	icts												
Cash flow hedges		_	_	_	Interest expense		146		1	(0)	1		

Effect of Hedging Relationships on Accumulated Other Comprehensive Income

	Amou	unt Rec	ognized in OCI	on Hedges	Unaudited Consolidated Statements of Operations ⁽¹⁾			Amount Reclassified from AOCI into Earnings			
(in millions)	Pre-Tax (Los		Tax Benefit (Expense)						Tax (Benefit) Expense	(Gain) Loss Net of Tax	
				Six Months	s Ended June 30, 2023						
Forward currency contracts											
Cash flow hedges	\$	87 3	\$ (20) 5	§ 67	Cost of products sold	\$	2,098	\$	(125) \$	28	\$ (97)
Net investment hedges ⁽²⁾		28	(6)	22	Interest expense		135		(5)	1	(4)
Foreign currency-denominated	d debt										
Net investment hedges ⁽³⁾		(18)	4	(14)	Other, net		61		_	_	_
Interest rate derivative contra	cts										
Cash flow hedges		_	_	_	Interest expense		135		1	(0)	1

⁽¹⁾ In all periods presented in the table above, the pre-tax (gain) loss amounts reclassified from AOCI to earnings represent the effect of the hedging relationships on earnings.

As of June 30, 2024, pre-tax net gains or losses for our derivative instruments designated, or previously designated, as cash flow and net investment hedges under FASB ASC Topic 815 that may be reclassified from AOCI to earnings within the next twelve months are presented below (in millions):

Designated Hedging Instrument	FASB ASC Topic 815 Designation	Location on Unaudited Consolidated Statements of Operations	ount of Pre-Tax Gain (Loss) nat may be Reclassified to Earnings
Forward currency contracts	Cash flow hedge	Cost of products sold	\$ 163
Forward currency contracts	Net investment hedge	Interest expense	11
Interest rate derivative contracts	Cash flow hedge	Interest expense	(1)

Net gains and losses on currency hedge contracts not designated as hedging instruments offset by net gains and losses from currency transaction exposures are presented below:

	Location on Unaudited Consolidated Statements of Three Months Ended June 30,			Si	x Months E	nde	nded June 30,	
(in millions)	Operations		2024	2023		2024		2023
Net gain (loss) on currency hedge contracts	Other, net	\$	30	\$ 11	\$	45	\$	2
Net gain (loss) on currency transaction exposures	Other, net		(31)	(20)		(51)		(26)
Net currency exchange gain (loss)		\$	(1)	\$ (9)	\$	(6)	\$	(24)

⁽²⁾ For our outstanding forward currency contracts designated as net investment hedges, the net gain or loss reclassified from AOCI to earnings as a reduction of *Interest expense* represents the straight-line amortization of the excluded component as calculated at the date of designation. This initial value of the excluded component has been excluded from the assessment of effectiveness in accordance with FASB ASC Topic 815. In the current and prior periods, we did not recognize any gains or losses on the components included in the assessment of hedge effectiveness in earnings.

(3) For our outstanding euro-denominated debt principal designated as a net investment hedge, the change in fair value attributable to changes in the spot rate is recorded in the CTA component of OCI. No amounts were reclassified from AOCI to current period earnings.

Fair Value Measurements

FASB ASC Topic 815 requires all derivative and nonderivative instruments to be recognized at their fair values as either assets or liabilities on the balance sheet. We determine the fair value of our derivative and nonderivative instruments using the framework prescribed by FASB ASC Topic 820, Fair Value Measurements and Disclosures (FASB ASC Topic 820), and considering the estimated amount we would receive or pay to transfer these instruments at the reporting date with respect to current currency exchange rates, interest rates, the creditworthiness of the counterparty for unrealized gain positions and our own creditworthiness for unrealized loss positions. In certain instances, we may utilize financial models to measure fair value of our derivative and nonderivative instruments. In doing so, we use inputs that include quoted prices for similar assets or liabilities in active markets, quoted prices for identical or similar assets or liabilities in markets that are not active, other observable inputs for the asset or liability and inputs derived principally from, or corroborated by, observable market data by correlation or other means. The following are the balances of our derivative and nonderivative assets and liabilities:

	Location on Unaudited		As of				
(in millions)	Consolidated Balance Sheets ⁽¹⁾	J	June 30, 2024		December 31, 2023		
Derivative and Nonderivative Assets:							
<u>Designated Hedging Instruments</u>							
Forward currency contracts	Other current assets	\$	236	\$	140		
Forward currency contracts	Other long-term assets		46		107		
			282		246		
Non-Designated Hedging Instruments							
Forward currency contracts	Other current assets		31		20		
Total Derivative and Nonderivative Assets		\$	313	\$	266		
Derivative and Nonderivative Liabilities:							
<u>Designated Hedging Instruments</u>							
Forward currency contracts	Other current liabilities	\$	3	\$	15		
Forward currency contracts	Other long-term liabilities		0		9		
Foreign currency-denominated debt ⁽²⁾	Long-term debt		957		988		
			960		1,012		
Non-Designated Hedging Instruments							
Forward currency contracts	Other current liabilities		14		38		
Total Derivative and Nonderivative Liabilities		\$	974	\$	1,050		

⁽¹⁾ We classify derivative and nonderivative assets and liabilities as current when the settlement date of the contract is one year or less.

Recurring Fair Value Measurements

On a recurring basis, we measure certain financial assets and financial liabilities at fair value based upon quoted market prices. Where quoted market prices or other observable inputs are not available, we apply valuation techniques to estimate fair value. FASB ASC Topic 820 establishes a three-level valuation hierarchy for disclosure of fair value measurements. The category of a financial asset or a financial liability within the valuation hierarchy is based upon the lowest level of input that is significant to the measurement of fair value. The three levels of the hierarchy are defined as follows:

- Level 1 Inputs to the valuation methodology are quoted market prices for identical assets or liabilities.
- Level 2 Inputs to the valuation methodology are other observable inputs, including quoted market prices for similar assets or liabilities and market-corroborated inputs.
- Level 3 Inputs to the valuation methodology are unobservable inputs based on management's best estimate of inputs market participants would use in pricing the asset or liability at the measurement date, including assumptions about risk.

⁽²⁾ Foreign currency-denominated debt is the €900 million debt principal associated with our 2027 Notes designated as a net investment hedge. A portion of this notional is subject to de-designation and re-designation based on changes in the underlying hedged item.

Assets and liabilities measured at fair value on a recurring basis consist of the following:

						As	as of									
	June 30, 2024					December 31, 2023										
(in millions)	Level 1		Level 2		L	Level 3		Total		evel 1	Level 2		Level 3		Total	
<u>Assets</u>																
Money market funds and time deposits	\$	2,153	\$	_	\$	_	\$	2,153	\$	454	\$	_	\$	_	\$	454
Publicly-held equity securities		19		_		_		19		18		_		_		18
Hedging instruments		_		313		_		313		_		266		_		266
Licensing arrangements		_		_		51		51		_		_		77		77
	\$	2,173	\$	313	\$	51	\$	2,536	\$	472	\$	266	\$	77	\$	816
<u>Liabilities</u>																
Hedging instruments	\$	_	\$	974	\$	_	\$	974	\$	_	\$	1,050	\$	_	\$	1,050
Contingent consideration liability		_		_		293		293		_		_		404		404
Licensing arrangements		_		_		60		60		_		_		90		90
	\$	_	\$	974	\$	353	\$	1,327	\$		\$	1,050	\$	494	\$	1,545

Our investments in money market funds and time deposits are classified within Level 1 of the fair value hierarchy because they are valued using quoted market prices. These investments are classified as *Cash and cash equivalents* or *Other current assets* within our accompanying unaudited consolidated balance sheets, in accordance with GAAP and our accounting policies. In addition to \$2.153 billion invested in money market funds and time deposits as of June 30, 2024 and \$454 million as of December 31, 2023, we held \$793 million in interest-bearing and non-interest-bearing bank accounts as of June 30, 2024 and \$411 million as of December 31, 2023.

Our recurring fair value measurements using Level 3 inputs include those related to our contingent consideration liability. Refer to *Note B – Acquisitions and Strategic Investments* for a discussion of the changes in the fair value of our contingent consideration liability. In addition, our recurring fair value measurements using Level 3 inputs related to our licensing arrangements, including the contractual right to receive future royalty payments related to the ZytigaTM Drug. We maintain a financial asset and associated liability for our licensing arrangements measured at fair value within our unaudited consolidated balance sheets in accordance with FASB ASC Topic 825, *Financial Instruments*. We elected the fair value option to measure the financial asset and associated liability as it provides for consistency and comparability of these financial instruments with others. Refer to *Note D – Hedging Activities and Fair Value Measurements* to our audited financial statements contained in Item 8. Financial Statements and Supplementary Data of our most recent Annual Report on Form 10-K for additional information.

The recurring Level 3 fair value measurements of our licensing arrangements recognized within our accompanying unaudited consolidated balance sheet as of June 30, 2024 include the following significant unobservable inputs:

Licensing Arrangements	Fair Value as of June 30, 2024	Valuation Technique	Unobservable Input	Range	Weighted Average ⁽¹⁾
Einanaial Assa4	¢51:11:	Discounted	Discount Rate	15%	15%
Financial Asset	\$51 million	Cash Flow	Projected Year of Payment	2024 - 2025	2025
Eineneiel Liebilie.	Discou		Discount Rate	12 % - 15%	13%
Financial Liability	\$60 million	Cash Flow	Projected Year of Payment	2024 - 2026	2025

⁽¹⁾ Unobservable inputs relate to a single financial asset and liability. As such, unobservable inputs were not weighted by the relative fair value of the instruments. For projected year of payment, the amount represents the median of the inputs and is not a weighted average.

Changes in the fair value of our licensing arrangements' financial asset were as follows:

(in millions)

Balance as of December 31, 2023	\$ 77
Proceeds from royalty rights	(21)
Fair value adjustment (expense) benefit	(5)
Balance as of June 30, 2024	\$ 51

Changes in the fair value of our licensing arrangements' financial liability were as follows:

(in millions)

Balance as of December 31, 2023	\$ 90
Payments for royalty rights	(26)
Fair value adjustment expense (benefit)	(4)
Balance as of June 30, 2024	\$ 60

Non-Recurring Fair Value Measurements

We hold certain assets and liabilities that are measured at fair value on a non-recurring basis in periods after initial recognition. The fair value of a measurement alternative investment is not estimated if there are no identified events or changes in circumstances that may have a significant adverse effect on the fair value of the investment. Refer to Note B – Acquisitions and Strategic Investments for a discussion of our strategic investments and Note C – Goodwill and Other Intangible Assets for a discussion of the fair values of our intangible assets including goodwill.

The fair value of our outstanding debt obligations, excluding finance leases, was \$10.073 billion as of June 30, 2024 and \$8.735 billion as of December 31, 2023. We determined fair value by using quoted market prices for our publicly registered senior notes, classified as Level 1 within the fair value hierarchy, and face value for commercial paper, term loans and credit facility borrowings outstanding. Refer to *Note E - Contractual Obligations and Commitments* for a discussion of our debt obligations.

NOTE E - CONTRACTUAL OBLIGATIONS AND COMMITMENTS

Borrowings and Credit Arrangements

We had total debt outstanding of \$10.571 billion as of June 30, 2024 and \$9.102 billion as of December 31, 2023, with current obligations of \$1.580 billion as of June 30, 2024 and \$531 million as of December 31, 2023. The debt maturity schedule for our long-term debt obligations is presented below:

				As of		
(in millions, except interest rates)	Issuance Date	Maturity Date		ne 30, 024	December 31, 2023	Coupon Rate ⁽¹⁾
March 2025 Senior Notes ⁽³⁾	March 2022	March 2025	-		1,105	0.750%
June 2025 Senior Notes	May 2020	June 2025		_	500	1.900%
March 2026 Senior Notes	February 2019	March 2026		255	255	3.750%
December 2027 Senior Notes ⁽³⁾	November 2019	December 2027		963	995	0.625%
March 2028 Senior Notes ⁽³⁾	March 2022	March 2028		802	829	1.375%
March 2028 Senior Notes	February 2018	March 2028		344	344	4.000%
March 2029 Senior Notes	February 2019	March 2029		272	272	4.000%
March 2029 Senior Notes ⁽³⁾	February 2024	March 2029		802	_	3.375%
June 2030 Senior Notes	May 2020	June 2030		1,200	1,200	2.650%
March 2031 Senior Notes ⁽³⁾	March 2022	March 2031		802	829	1.625%
March 2032 Senior Notes ⁽³⁾	February 2024	March 2032		1,337	_	3.500%
March 2034 Senior Notes ⁽³⁾	March 2022	March 2034		535	553	1.875%
November 2035 Senior Notes ⁽²⁾	November 2005	November 2035		350	350	6.500%
March 2039 Senior Notes	February 2019	March 2039		450	450	4.550%
January 2040 Senior Notes	December 2009	January 2040		300	300	7.375%
March 2049 Senior Notes	February 2019	March 2049		650	650	4.700%
Unamortized Debt Issuance Discount and Deferred Financing Costs		2024 - 2049		(77)	(65)	
Finance Lease Obligation		Various		4	5	_
Long-term debt			\$	8,991	\$ 8,571	_

⁽¹⁾ Coupon rates are semi-annual, except for the euro-denominated notes, which bear an annual coupon.

Revolving Credit Facility

On May 10, 2021, we entered into a \$2.750 billion revolving credit facility (as amended, supplemented or otherwise modified from time to time, the 2021 Revolving Credit Facility) with a global syndicate of commercial banks. On May 10, 2024, we entered into a third amendment to the 2021 Revolving Credit Facility credit agreement, which provided for, among other things, an extension of the scheduled maturity date to May 10, 2029, an amendment of the Ratings based pricing grid of the Applicable Margin, each as defined in the credit agreement, and reset the applicable date for purposes of determining the amounts of restructuring charges and restructuring-related expenses that may be excluded from consolidated Earnings Before Interest, Taxes, Depreciation and Amortization (EBITDA), as defined by the credit agreement, for purposes of our maximum leverage ratio covenant, from December 31, 2022 to March 31, 2024, as further discussed under *Financial Covenant* below. This facility provides backing for our commercial paper program, and outstanding commercial paper directly reduces borrowing capacity under the 2021 Revolving Credit Facility. We had no amounts outstanding under the 2021 Revolving Credit Facility as of June 30, 2024 or December 31, 2023.

⁽²⁾ Corporate credit rating improvements may result in a decrease in the adjusted interest rate on our November 2035 Notes to the extent that our lowest credit rating is above BBB- or Baa3. The interest rates on our November 2035 Notes will be permanently reinstated to the issuance rate of 6.25% if the lowest credit ratings assigned to these senior notes is either A- or A3 or higher.

⁽³⁾ These notes are euro-denominated and presented in U.S. dollars based on the exchange rate in effect as of June 30, 2024 and December 31, 2023, respectively.

Financial Covenant

As of June 30, 2024, we were in compliance with the financial covenant required by the 2021 Revolving Credit Facility.

	Covenant Requirement	Actual
	as of June 30, 2024	as of June 30, 2024
Maximum permitted leverage ratio(1)	3.75 times	1.97 times

⁽¹⁾ Ratio of total debt to deemed consolidated EBITDA, as defined by the 2021 Revolving Credit Facility credit agreement, as amended.

The 2021 Revolving Credit Facility includes the financial covenant requirement for all of our credit arrangements that we maintain the maximum permitted leverage ratio of 3.75 times for the remaining term. The credit agreement provides for higher leverage ratios, at our election, for the period following a Qualified Acquisition, as defined by the agreement, for which consideration exceeds \$1.000 billion. In the event of such an acquisition, for the four succeeding quarters immediately following, including the quarter in which the acquisition occurs, the maximum permitted leverage ratio is 4.75 times. It steps down for the fifth, sixth and seventh succeeding quarters to 4.50 times, 4.25 times and 4.00 times, respectively. Thereafter, a maximum leverage ratio of 3.75 times is required through the remaining term of the 2021 Revolving Credit Facility. We have elected to designate the Axonics acquisition as a Qualified Acquisition under the credit agreement, and upon closing, will increase the maximum permitted leverage ratio at that time. The agreement also provides for an exclusion of any debt incurred to fund a Qualified Acquisition, until the earlier of the acquisition close date or date of abandonment, termination or expiration of the acquisition agreement. As of June 30, 2024, we excluded from our leverage ratio calculation \$2.120 billion of debt incurred in connection with the Axonics acquisition.

The financial covenant requirement, as amended on May 10, 2024, provides for an exclusion from the calculation of consolidated EBITDA, as defined by the credit agreement, through maturity, of certain charges and expenses. The credit agreement amendment reset the starting date for purposes of calculating such permitted exclusions related to restructuring charges and restructuring-related expenses from December 31, 2022 to March 31, 2024. Permitted exclusions include up to \$500 million in cash and non-cash restructuring charges and restructuring-related expenses associated with our current or future restructuring plans. As of June 30, 2024, we had \$453 million of the restructuring charge exclusion remaining. In addition, any cash litigation payments (net of any cash litigation receipts), as defined by the agreement, are excluded from the calculation of consolidated EBITDA, as defined by the agreement, provided that the sum of any excluded net cash litigation payments do not exceed \$1.000 billion plus all accrued legal liabilities as of December 31, 2022. As of June 30, 2024, we had \$1.448 billion of the litigation exclusion remaining.

Any inability to maintain compliance with this covenant could require us to seek to renegotiate the terms of our credit arrangements or seek waivers from compliance with this covenant, both of which could result in additional borrowing costs. Further, there can be no assurance that our lenders would agree to such new terms or grant such waivers on terms acceptable to us. In this case, all 2021 Revolving Credit Facility commitments would terminate, and any amounts borrowed under the facility would become immediately due and payable. Furthermore, any termination of our 2021 Revolving Credit Facility may negatively impact the credit ratings assigned to our commercial paper program, which may impact our ability to refinance any then outstanding commercial paper as it becomes due and payable.

Commercial Paper

Our commercial paper program is backed by the 2021 Revolving Credit Facility. We did not have any commercial paper outstanding as of June 30, 2024 or December 31, 2023.

Senior Notes

We had senior notes outstanding of \$10.633 billion as of June 30, 2024 and \$9.136 billion as of December 31, 2023. Our senior notes were issued in public offerings, are redeemable prior to maturity and are not subject to sinking fund requirements. Our senior notes are unsecured, unsubordinated obligations and rank on parity with each other. These notes are effectively junior to liabilities of our subsidiaries (refer to *Other Arrangements* below).

In February 2024, American Medical Systems Europe B.V. (AMS Europe), an indirect, wholly owned subsidiary of Boston Scientific, completed a registered public offering (the Offering) of €2.000 billion in aggregate principal amount of euro-denominated senior notes comprised of €750 million of 3.375% Senior Notes due 2029 and €1.250 billion of 3.500% Senior Notes due 2032 (collectively, the 2024 Eurobonds). Boston Scientific has fully and unconditionally guaranteed all of AMS Europe's obligations under the 2024 Eurobonds, and no other subsidiary of Boston Scientific will guarantee these obligations.

AMS Europe is a "finance subsidiary" as defined in Rule 13-01(a)(4)(vi) of Regulation S-X. The financial condition, results of operations and cash flows of AMS Europe are consolidated in the financial statements of Boston Scientific. The Offering resulted in cash proceeds of \$2.145 billion, net of investor discounts and issuance costs.

We used the net proceeds from the Offering of the 2024 Eurobonds to fund the repayment at maturity of our 3.450% Senior Notes due March 2024 and to pay accrued and unpaid interest with respect to such notes. Additionally, we plan to use the remaining net proceeds from the Offering to fund a portion of the purchase price of the Axonics acquisition and to pay related fees and expenses, and for general corporate purposes. The transaction is expected to be completed in the second half of 2024, subject to the expiration or termination of the waiting period under the HSR Act and the satisfaction (or waiver) of other customary closing conditions. If (i) the Axonics acquisition is not consummated on or before the later of (x) January 8, 2025 (as such date may be extended in accordance with the merger agreement to no later than January 8, 2026) and (y) the date that is five business days after any later date to which we and Axonics may agree to extend the "Outside Date" in the merger agreement or (ii) AMS Europe notifies the trustee that we will not pursue consummation of the Axonics Acquisition, AMS Europe will be required to redeem each series of the notes at a special mandatory redemption price equal to 101% of the aggregate principal amount of such series of notes, plus accrued and unpaid interest, if any, to, but excluding, the date on which the notes will be redeemed. Refer to *Note B – Acquisitions and Strategic Investments* for more information on the Axonics acquisition.

Other Arrangements

We have accounts receivable factoring programs in certain European countries and with commercial banks in China and Japan which include promissory notes discounting programs. We account for our factoring programs as sales under FASB ASC Topic 860, *Transfers and Servicing*. We have no retained interest in the transferred receivables, other than collection and administration, and once sold, the accounts receivable are no longer available to satisfy creditors in the event of bankruptcy. Amounts de-recognized for accounts and notes receivable, which are excluded from *Trade accounts receivable*, *net* within our accompanying unaudited consolidated balance sheets, are aggregated by contract denominated currency below (in millions):

	 As of June	30, 2024	As of December 31, 2023			
Factoring Arrangements	 nount cognized	Weighted Average Amount Interest Rate De-recognized		Weighted Average Interest Rate		
Euro denominated	\$ 187	5.3 %	\$ 206	5.1 %		
Yen denominated	187	0.8 %	214	0.6 %		
Renminbi denominated	26	1.7 %	14	2.9 %		

Other Contractual Obligations and Commitments

We had outstanding letters of credit of \$189 million as of June 30, 2024 and \$174 million as of December 31, 2023, which consisted primarily of bank guarantees and collateral for workers' compensation insurance arrangements. As of June 30, 2024 and December 31, 2023 we had not recognized a related liability for our outstanding letters of credit within our accompanying unaudited consolidated balance sheets.

We have a supplier financing program offered primarily in the U.S. that enables our suppliers to opt to receive early payment at a nominal discount, while allowing us to lengthen our payment terms and optimize working capital. Our standard payment term in the U.S. is 90 days. All outstanding payables related to the supplier finance program are classified within *Accounts Payable* within our unaudited consolidated balance sheets and were \$131 million as of June 30, 2024 and \$152 million as of December 31, 2023.

Refer to *Note E - Contractual Obligations and Commitments* to our audited financial statements contained in Item 8. Financial Statements and Supplementary Data of our most recent Annual Report on Form 10-K for additional information on our borrowings and credit agreements.

NOTE F – SUPPLEMENTAL BALANCE SHEET INFORMATION

Components of selected captions within our accompanying unaudited consolidated balance sheets are as follows:

Trade accounts receivable, net

	<u> </u>	As of				
(in millions)	Ju	ne 30, 2024	December 31, 2023			
Trade accounts receivable	\$	2,505	\$ 2,338			
Allowance for credit losses		(105)	(110)			
	\$	2,400	\$ 2,228			

The following is a roll forward of our *Allowance for credit losses*:

	Three Months Ended June 30,					Six Months Ended June 30,				
(in millions)	2024 2023				2024	2023				
Beginning balance	\$	109	\$	118	\$	110	\$	109		
Credit loss expense		12		19		21		35		
Write-offs		(16)		(10)		(26)		(16)		
Ending balance	\$	105	\$	127	\$	105	\$	127		

Inventories

(in millions)		June 30, 2024		December 31, 2023
Finished goods	\$	1,558	\$	1,537
Work-in-process		201		174
Raw materials		849		773
	\$	2,608	\$	2,484

Other current assets

	As of						
(in millions)		June 30, 2024		December 31, 2023			
Restricted cash and restricted cash equivalents	\$	75	\$	130			
Derivative assets		267		159			
Licensing arrangements		38		47			
Other		376		285			
	\$	756	\$	621			

Property, plant and equipment, net

(in millions) Land		of	
	June	30, 2024	December 31, 2023
	\$	140	\$ 140
Buildings and improvements		1,876	1,843
Equipment, furniture and fixtures		3,598	3,503
Capital in progress		897	857
		6,512	6,343
Less: accumulated depreciation		3,561	3,484
	\$	2,951	\$ 2,859

Depreciation expense was \$98 million for the second quarter of 2024, \$88 million for the second quarter of 2023, \$188 million for the first six months of 2024 and \$170 million for the first six months of 2023.

Other long-term assets

	As of						
(in millions)		e 30, 2024	December 31, 2023				
Restricted cash equivalents	\$	74	\$	60			
Operating lease right-of-use assets		416		439			
Derivative assets		46		107			
Investments		450		413			
Licensing arrangements		13		30			
Indemnification asset		172		176			
Other		381		306			
	\$	1,551	\$	1,531			

Accrued expenses

	As of						
(in millions)		ne 30, 2024	D	ecember 31, 2023			
Legal reserves	\$	118	\$	206			
Payroll and related liabilities		948		1,051			
Rebates		412		389			
Contingent consideration		175		304			
Other		667		696			
	\$	2,320	\$	2,646			

Other current liabilities

	As of						
(in millions)	June	30, 2024	Decemb	er 31, 2023			
Deferred revenue	\$	284	\$	266			
Licensing arrangements		40		49			
Taxes payable		224		220			
Other		222		278			
	\$	770	\$	814			

Other long-term liabilities

	As of						
(in millions)	June	30, 2024	December 31, 2023				
Accrued income taxes	\$	356	3 470				
Legal reserves		133	172				
Contingent consideration		118	100				
Licensing arrangements		20	41				
Operating lease liabilities		365	390				
Deferred revenue		316	311				
Other		491	484				
	\$	1,800	5 1,967				

NOTE G - INCOME TAXES

Our effective tax rate from continuing operations is presented below:

	Three Months En	ded June 30,	Six Months En	ded June 30,	
	2024	2023	2024	2023	
Effective tax rate from continuing operations	23.4 %	36.6 %	20.7 %	32.9 %	

Our reported tax rate is affected by recurring items such as the amount of our earnings subject to differing tax rates in foreign jurisdictions and the impact of certain receipts and charges that are taxed at rates that differ from our effective tax rate.

The primary reasons for the difference between the statutory rate and our reported tax rate in the second quarter and first six months of 2024 relate to certain charges for acquisition expenses, as well as certain benefits for intangible asset impairment charges and integration costs, and discrete benefits primarily related to stock-based compensation. In the second quarter and first six months of 2023, the primary difference between the statutory tax rate and our reported tax rate relates to acquisition-related charges, as well as certain benefits for integration costs, and discrete tax items primarily related to unrecognized tax benefits.

As of June 30, 2024, we had \$478 million of gross unrecognized tax benefits, of which a net \$406 million, if recognized, would affect our effective tax rate. As of December 31, 2023, we had \$467 million of gross unrecognized tax benefits, of which a net \$395 million, if recognized, would affect our effective tax rate. The change in gross unrecognized tax benefit is primarily related to current year accruals for reserves and audits.

It is reasonably possible that within the next 12 months, we will resolve multiple issues with foreign, federal and state taxing authorities, resulting in a reduction in our balance of unrecognized tax benefits of up to \$9 million.

NOTE H - COMMITMENTS AND CONTINGENCIES

The medical device market in which we participate is largely technology driven. As a result, intellectual property rights, particularly patents and trade secrets, play a significant role in product development and differentiation. Over the years, there has been litigation initiated against us by others, including our competitors, claiming that our current or former product offerings infringe patents owned or licensed by them. Intellectual property litigation is inherently complex and unpredictable. In addition, competing parties frequently file multiple suits to leverage patent portfolios across product lines, technologies and geographies and to balance risk and exposure between the parties. In some cases, several competitors are parties in the same proceeding, or in a series of related proceedings, or litigate multiple features of a single class of devices. These dynamics frequently drive settlement not only for individual cases, but also for a series of pending and potentially related and unrelated cases. Although monetary and injunctive relief is typically sought, remedies and restitution are generally not determined until the conclusion of the trial court proceedings and can be modified on appeal. Accordingly, the outcomes of individual cases are difficult to time, predict or quantify and are often dependent upon the outcomes of other cases in other geographies.

During recent years, we successfully negotiated closure of several long-standing legal matters and have received favorable rulings in several other matters; however, there continues to be outstanding litigation. Adverse outcomes in one or more of these matters could have a material adverse effect on our ability to sell certain products and on our operating margins, financial position, results of operations and/or liquidity.

In addition, product liability, securities and commercial claims have been asserted against us and similar claims may be asserted against us in the future related to events not known to management at the present time. We maintain an insurance policy providing limited coverage against securities claims and we are substantially self-insured with respect to product liability claims and fully self-insured with respect to intellectual property infringement claims. The absence of significant third-party insurance coverage increases our potential exposure to unanticipated claims or adverse decisions. Product liability claims, securities and commercial litigation and other legal proceedings in the future, regardless of their outcome, could have a material adverse effect on our ability to sell certain products and on our operating margins, financial position, results of operations and/or liquidity.

In addition, like other companies in the medical device industry, we are subject to extensive regulation by national, state and local government agencies in the U.S. and other countries in which we operate. From time to time we are the subject of qui tam actions and governmental investigations often involving regulatory, marketing and other business practices. These qui tam actions and governmental investigations could result in the commencement of civil and criminal proceedings, substantial fines, penalties and administrative remedies and have a material adverse effect on our financial position, results of operations and/or liquidity. For additional information, refer to *Note I - Commitments and Contingencies* to our audited financial statements contained in Item 8 of our most recent Annual Report on Form 10-K.

In accordance with FASB ASC Topic 450, Contingencies, we accrue anticipated costs of settlement, damages, losses for product liability claims and, under certain conditions, costs of defense, based on historical experience or to the extent specific losses are probable and estimable. Otherwise, we expense these costs as incurred. If the estimate of a probable loss is a range and no amount within the range is more likely, we accrue the minimum amount of the range. We record certain legal and product liability charges, credits and costs of defense, which we consider to be unusual or infrequent and significant as Litigation-related net charges (credits) within our accompanying unaudited consolidated financial statements. All other legal and product liability charges, credits and costs are recorded within Selling, general and administrative expenses within our accompanying unaudited consolidated statements of operations.

Our accrual for legal matters that are probable and estimable was \$251 million as of June 30, 2024 and \$377 million as of December 31, 2023 and includes certain estimated costs of settlement, damages and defense primarily related to product liability cases or claims related to our transvaginal surgical mesh products. A portion of this accrual is already funded through our qualified settlement fund, which is included in restricted cash and restricted cash equivalents in *Other current assets* of \$75 million as of June 30, 2024 and \$130 million as of December 31, 2023. Refer to *Note F - Supplemental Balance Sheet Information* for additional information. We did not record any litigation-related net charges during the second quarter and first six months of 2024 or 2023.

We continue to assess certain litigation and claims to determine the amounts, if any, that management believes will be paid as a result of such claims and litigation and, therefore, additional losses may be accrued and paid in the future, which could materially adversely impact our operating results, cash flows and/or our ability to comply with our financial covenant.

In management's opinion, we are not currently involved in any legal proceedings other than those disclosed in our most recent Annual Report on Form 10-K and those specifically identified below, which, individually or in the aggregate, could have a material adverse effect on our financial condition, operations and/or cash flows. Unless included in our legal accrual or

otherwise indicated below, a range of loss associated with any individual material legal proceeding cannot be reasonably estimated.

Patent Litigation

On November 20, 2017, The Board of Regents, University of Texas System and TissueGen. Inc. (collectively, UT), served a lawsuit against us in the Western District of Texas. The complaint against the Company alleges patent infringement of two U.S. patents owned by UT, relating to "Drug Releasing Biodegradable Fiber Implant" and "Drug Releasing Biodegradable Fiber for Delivery of Therapeutics," and affects the manufacture, use and sale of our SynergyTM Stent System. UT primarily seeks a reasonable royalty. On March 12, 2018, the District Court for the Western District of Texas dismissed the action and transferred it to the United States District Court for the District of Delaware. On September 5, 2019, the Court of Appeals for the Federal Circuit affirmed the dismissal of the District Court for the Western District of Texas. In April 2020, the United States Supreme Court denied the UT's Petition for Certiorari. UT proceeded with its case against the Company in Delaware. In January 2023, a jury trial was held on the issue of whether the one UT patent still asserted in the case was valid and whether it was infringed by the Company. On January 31, 2023, a jury concluded that UT's patent was valid and willfully infringed by the Company, and awarded UT \$42 million in damages. Following the trial, UT filed a motion seeking prejudgment interest and enhanced damages. The Company filed a motion seeking judgment as a matter of law in its favor or alternatively a new trial. On June 5, 2024, the Court granted the Company's motion for enhanced damages, awarded approximately \$7.4 million in pre-judgment interest, and awarded post-judgment interest. On July 3, 2024, UT and the Company each filed a notice of appeal.

Product Liability Litigation

Multiple product liability cases or claims related to transvaginal surgical mesh products designed to treat stress urinary incontinence and pelvic organ prolapse have been asserted against us, predominantly in the United States, Canada, the United Kingdom, Scotland, Ireland, and Australia. Plaintiffs generally seek monetary damages based on allegations of personal injury associated with the use of our transvaginal surgical mesh products, including design and manufacturing claims, failure to warn, breach of warranty, fraud, violations of state consumer protection laws and loss of consortium claims. We have entered into individual and master settlement agreements or are in the final stages of entering agreements with certain plaintiffs' counsel, to resolve the majority of these cases and claims. All settlement agreements were entered into solely by way of compromise and without any admission or concession by us of any liability or wrongdoing. In addition, in April 2021 the Company's Board of Directors received a shareholder demand under section 220 of the Delaware General Corporation Law, for inspection of books and records related to mesh settlements. The Company has notified our insurer and retained counsel to respond to the demand.

We have established a product liability accrual for remaining claims asserted against us associated with our transvaginal surgical mesh products and the costs of defense thereof. We continue to engage in discussions with plaintiffs' counsel regarding potential resolution of pending cases and claims, which we continue to vigorously contest. The final resolution of the cases and claims is uncertain and could have a material impact on our results of operations, financial condition and/or liquidity. Trials involving our transvaginal surgical mesh products have resulted in both favorable and unfavorable judgments for us. We do not believe that the judgment in any one trial is representative of potential outcomes of all cases or claims related to our transvaginal surgical mesh products.

Governmental Investigations and Qui Tam Matters

On December 1, 2015, the Brazilian governmental entity known as CADE (the Administrative Council of Economic Defense), served a search warrant on the offices of our Brazilian subsidiary, as well as on the Brazilian offices of several other major medical device makers who do business in Brazil, in furtherance of an investigation into alleged anti-competitive activity with respect to certain tender offers for government contracts. On June 20, 2017, CADE, through the publication of a "technical note," announced that it was launching a formal administrative proceeding against Boston Scientific's Brazilian subsidiary, Boston Scientific do Brasil Ltda. (BSB), as well as against the Brazilian operations of Medtronic, Biotronik and St. Jude Medical, two Brazilian associations, ABIMED and AMBIMO and 29 individuals for alleged anti-competitive behavior. Under applicable guidance, BSB could be fined a percentage of BSB's 2016 gross revenues. In August 2021, the investigating commissioner issued a preliminary recommendation of liability against all of the involved companies, and also recommended that CADE impose fines and penalties. However, on October 25, 2021, the CADE Attorney General's office recommended dismissal of the charges and allegations against BSB and the individual BSB employees who were still individual defendants. Subsequently, on March 30, 2022, the Federal Prosecutor's office issued a non-binding recommendation that is contrary to the Attorney General's recommendation. The full Commission issued its decision on May 22, 2024, holding that BSB and a former employee should be held liable and setting a fine of \$17 million. The Commission held that the three other current or former Company employees should be acquitted. BSB has filed a motion to clarify the proper calculation of the fine and, to the extent

any fine is upheld, the Company intends to appeal the decision of liability by the full Commission to the Brazilian courts. During such an appeal, the decision would have no force and effect, and the Court would consider the case without being bound by CADE's decision.

Like many healthcare companies, the Company receives inquiries and has ongoing discussions with governmental agencies with respect to the Company's operations, such as the Securities and Exchange Commission (SEC), the Department of Justice (DOJ) and foreign regulators, including its operations in Vietnam with respect to alleged Foreign Corrupt Practices Act (FCPA) violations the Company received in March 2022. The Company has received related subpoenas for documents from the DOJ and the SEC with respect to the Vietnam matter, and is cooperating with the government while investigating these allegations. From time to time, the Company also self-discloses potential concerns to regulators. In the course of Vietnam-related discussions with the DOJ and SEC, the Company has disclosed that it is investigating other potential concerns in Vietnam and other countries.

From time to time, the Company also receives U.S.-based subpoenas and DOJ Civil Investigative Demands (CID), including the following matters: in April 2023, the Company received a DOJ subpoena that seeks documents and information related to its ambulatory electrocardiography monitoring business; in December 2023, the Company received a DOJ CID related to the provision of peripheral intervention services through office-based labs. The Company is cooperating with the DOJ in these matters.

Other Proceedings

On December 4, 2020, Enrique Jevons, individually and on behalf of all others similarly situated, filed a class action complaint against the Company, Michael F. Mahoney and Daniel J. Brennan, stemming from the recall and retirement of the LOTUS EdgeTM Aortic Valve System (LOTUS System) in United States District Court for the Eastern District of New York. On December 14, 2020, the parties agreed to transfer the case to the United States District Court for the District of Massachusetts. On December 16, 2020, Mariano Errichiello, individually and on behalf of all others similarly situated, filed a second, materially similar class action complaint against the Company, Michael F. Mahoney, Joseph M. Fitzgerald, and Daniel J. Brennan in the United States District Court for the District of Massachusetts. Subsequently, on March 30, 2021, the Court consolidated the two actions, and appointed Union Asset Management Holding AG as the lead plaintiff. The plaintiffs filed an Amended Complaint in June 2021 that seeks unspecified compensatory damages in favor of the alleged class as well as unspecified equitable relief. The Company filed a Motion to Dismiss in July 2021, which, in December 2022, the Court granted in part and denied in part. On October 23, 2023, the Company reached an agreement in principle with the lead plaintiff to settle the case. The Court granted the motion for preliminary approval of the proposed settlement on December 27, 2023, and approved the settlement and dismissed the case on April 23, 2024.

On February 8, 2021, the Company received a letter from The Vladimir Gusinsky Revocable Trust, a shareholder, demanding that the Company's Board of Directors conduct an investigation into actions by the Company's directors and executive officers regarding statements made about the effectiveness and commercial viability of the LOTUS System. The Trust subsequently agreed to stay its demand, pending the outcome of any dispositive motion against the Amended Complaint in the class action complaint described above. The Company received letters on behalf of the Union Excavators Local 731 Pension Fund, Diane Nachbaur, and Frank Tripson, three stockholders of the Company, on July 26, 2021, July 29, 2021, and February 13, 2023, respectively, each demanding access to certain books and records of the Company, pursuant to Section 220 of the Delaware General Corporation Law, regarding the business, operations, effectiveness and commercial viability of the LOTUS system, and related items. On April 7, 2023, Diane Nachbaur filed a shareholder derivative complaint in the United States District Court for the District of Massachusetts against the Company, Michael F. Mahoney, Nelda J. Connors, Charles J. Dockendorff, Yoshiaki Fujimori, Donna A. James, Edward J. Ludwig, David Roux, John E. Sununu, Ellen M. Zane, Joseph M. Fitzgerald, Daniel J. Brennan, Shawn McCarthy, Ian Meredith, Kevin Ballinger, and Susan Vissers Lisa. On May 8, 2023, the Court stayed the case until the conclusion of the consolidated class action case. On October 18, 2023, Frank Tripson filed a shareholder derivative complaint in the Court of Chancery of the State of Delaware against the Company, Michael F. Mahoney, Daniel J. Brennan, Joseph M. Fitzgerald, Shawn McCarthy, Kevin Ballinger, Ian Meredith, Susan Vissers Lisa, Nelda J. Connors, Charles J. Dockendorff, Yoshiaki Fujimori, Donna A. James, Edward J. Ludwig, Stephen P. MacMillan, David Roux, John E. Sununu, and Ellen M. Zane. On December 15, 2023, the Court stayed that case until March 31, 2024. On March 2

NOTE I – STOCKHOLDERS' EQUITY

Preferred Stock

We are authorized to issue 50 million shares of preferred stock in one or more series and to fix the powers, designations, preferences and relative participating, option or other rights thereof, including dividend rights, conversion rights, voting rights, redemption terms, liquidation preferences and the number of shares constituting any series, without any further vote or action by our stockholders.

On May 27, 2020, we completed an offering of 10,062,500 shares of 5.50% Mandatory Convertible Preferred Stock, Series A (MCPS) at a price to the public and liquidation preference of \$100 per share. The net proceeds from the MCPS offering were approximately \$975 million after deducting underwriting discounts and commissions and offering expenses.

On June 1, 2023, (the Mandatory Conversion Date), all outstanding shares of MCPS automatically converted into shares of common stock. The conversion rate for each share of MCPS was 2.3834 shares of common stock. No action by the holders of the MCPS was required in connection with the mandatory conversion. Cash was paid in lieu of fractional shares in accordance with the terms of the MCPS. An aggregate of approximately 24 million shares of common stock, including shares of common stock issued to holders of MCPS that elected to convert prior to the Mandatory Conversion Date, were issued upon conversion of the MCPS. Following the mandatory conversion of the MCPS, there were no outstanding shares of MCPS.

Refer to *Note J – Stockholders' Equity* to our audited financial statements contained in Item 8. Financial Statements and Supplementary Data of our most recent Annual Report on Form 10-K for information on the pertinent rights and privileges of our outstanding common stock.

NOTE J – WEIGHTED AVERAGE SHARES OUTSTANDING

	Three Months E	nded June 30,	Six Months En	ided June 30,	
(in millions)	2024	2023	2024	2023	
Weighted average shares outstanding — basic	1,470.6	1,446.2	1,469.5	1,441.0	
Net effect of common stock equivalents	13.6	10.1	13.5	10.1	
Weighted average shares outstanding - diluted	1,484.2	1,456.2	1,483.0	1,451.1	

The following securities were excluded from the calculation of weighted average shares outstanding - diluted because their effect in the periods presented below would have been antidilutive:

	Three Months E	nded June 30,	Six Months Ended June 30		
(in millions)	2024	2023	2024	2023	
Stock options outstanding ⁽¹⁾		_	_	0	
$MCPS^{(2)}$	_	16	_	2.0	

⁽¹⁾ Represents stock options outstanding pursuant to our employee stock-based compensation plans with exercise prices that were greater than the average fair market value of our common stock for the related periods.

We base *Net income (loss) per common share - diluted* upon the weighted-average number of common shares and common stock equivalents outstanding during each year. Potential common stock equivalents are determined using the treasury stock method. We exclude stock options, stock awards and, prior to the Mandatory Conversion Date, our MCPS, from the calculation if the effect would be anti-dilutive. The dilutive effect of MCPS is calculated using the if-converted method. The if-converted method assumes that these securities were converted to shares of common stock at the beginning of the reporting period to the extent that the effect is dilutive.

⁽²⁾ Represents common stock issuable upon the conversion of MCPS. Refer to Note I – Stockholders' Equity for additional information.

For the second quarter and first six months of 2023, the effect of assuming the conversion of MCPS into shares of common stock was anti-dilutive, and therefore excluded from the calculation of earnings per share (EPS). Accordingly, *Net income (loss)* was reduced by cumulative *Preferred stock dividends*, as presented within our accompanying unaudited consolidated statements of operations, for purposes of calculating *Net income attributable to Boston Scientific common stockholders*. On June 1, 2023, all outstanding shares of MCPS automatically converted into shares of common stock.

We issued approximately one million shares of our common stock in the second quarter of 2024, approximately five million shares in the first six months of 2024, approximately 25 million shares in the second quarter of 2023 and approximately 29 million shares in the first six months of 2023. Shares were issued following the exercise of stock options, vesting of restricted stock units or purchases under our employee stock purchase plan and, specific to 2023, following the automatic conversion of the MCPS. We did not repurchase any shares of our common stock in the first six months of 2024 or 2023. On December 14, 2020, our Board of Directors approved a stock repurchase program authorizing the repurchase of up to \$1.000 billion of our common stock. As of June 30, 2024, we had the full amount remaining available under the authorization.

NOTE K - SEGMENT REPORTING

We aggregate our core businesses into two reportable segments: MedSurg and Cardiovascular, each of which generates revenues from the sale of medical devices. In accordance with FASB ASC Topic 280, Segment Reporting, we identified our reportable segments based on the nature of our products, production processes, type of customer, selling and distribution methods and regulatory environment, as well as the economic characteristics of each of our operating segments.

We measure and evaluate our reportable segments based on their respective net sales, operating income, excluding intersegment profits, and operating income as a percentage of net sales, all based on internally-derived standard currency exchange rates to exclude the impact of foreign currency, which may be updated from year to year. We exclude from operating income of reportable segments certain corporate-related expenses and certain transactions or adjustments that our chief operating decision maker (CODM) considers to be non-operational, such as amounts related to amortization expense, goodwill and other intangible asset impairment charges, acquisition/divestiture-related net charges (credits), restructuring and restructuring-related net charges (credits), and certain litigation-related net charges (credits) and European Union (EU) Medical Device Regulation (MDR) implementation costs. Although we exclude these amounts from operating income of reportable segments, they are included in reported *Income (loss) before income taxes* within our accompanying unaudited consolidated statements of operations and are included in the reconciliation below. Refer to *Note L - Revenue* for net sales by reportable segment presented in accordance with GAAP.

A reconciliation of the totals reported for the reportable segments to the applicable line items within our accompanying unaudited consolidated statements of operations is as follows (in millions, except percentages). Prior period amounts have been restated at constant currency to conform to current year presentation.

	Three Months Ended June 30,					Six Months Ended June 30,			
Net Sales		2024		2023	2024			2023	
MedSurg	\$	1,480	\$	1,346	\$	2,882	\$	2,611	
Cardiovascular		2,632		2,204		5,058		4,275	
Total net sales of reportable segments		4,112		3,549		7,939		6,886	
Impact of foreign currency fluctuations		8		51		37		102	
	\$	4,120	\$	3,599	\$	7,977	\$	6,988	
Income (loss) before income taxes									
MedSurg	\$	511	\$	458	\$	986	\$	867	
Cardiovascular		761		582		1,414		1,135	
Total operating income of reportable segments		1,272		1,040		2,400		2,002	
Unallocated amounts:									
Corporate expenses, including hedging activities and impact of foreign currency fluctuations on operating income of reportable segments		(153)		(75)		(268)		(173)	
Goodwill and other intangible asset impairment charges, acquisition/divestiture-related net charges (credits), restructuring and restructuring-related net charges (credits), certain litigation-related net charges (credits) and EU MDR implementation costs	t	(386)		(241)		(510)		(351)	
Amortization expense		(213)		(210)		(427)		(412)	
Operating income (loss)		520		514		1,195		1,066	
Other income (expense), net		(100)		(88)		(167)		(195)	
Income (loss) before income taxes	\$	420	\$	426	\$	1,028	\$	870	

	Three Months Er	Six Months June 3		
Operating income margin of reportable segments	2024	2023	2024	2023
MedSurg	34.6 %	34.0 %	34.2 %	33.2 %
Cardiovascular	28.9 %	26.4 %	28.0 %	26.6 %

NOTE L-REVENUE

We generate revenue primarily from the sale of single-use medical devices and present revenue net of sales taxes within our accompanying unaudited consolidated statements of operations. Our business structure is organized into five operating segments. The following tables disaggregate our revenue from contracts with customers by business unit and geographic region (in millions). Generally, we allocate revenue from contracts with customers to geographic regions based on the location where the sale originated.

	Three Months Ended June 30,													
				2024		2023								
Businesses		U.S.		Int'l		Total		U.S.		Int'l		Total		
Endoscopy	\$	415	\$	261	\$	676	\$	384	\$	246	\$	631		
Urology		363		162		525		340		146		485		
Neuromodulation		214		68		282		183		61		244		
MedSurg		992		491		1,483		907		453		1,360		
Interventional Cardiology Therapies		201		464		665		189		440		629		
Watchman		342		36		379		286		30		317		
Cardiac Rhythm Management		352		224		576		356		209		566		
Electrophysiology		271		157		428		85		108		193		
Cardiology		1,166		881		2,047		917		787		1,704		
Peripheral Interventions		308		281		590		285		250		535		
Cardiovascular		1,474		1,163		2,637		1,202		1,037		2,239		
Total Net Sales	\$	2,466	\$	1,654	\$	4,120	\$	2,110	\$	1,490	\$	3,599		

	Six Months Ended June 30,												
	-			2024		2023							
Businesses		U.S.		Int'l		Total		U.S.		Int'l		Total	
Endoscopy	\$	810	\$	508	\$	1,318	\$	735	\$	472	\$	1,207	
Urology		719		319		1,038		666		289		954	
Neuromodulation		405		134		539		356		123		478	
MedSurg		1,933		961		2,895		1,757		883		2,640	
Interventional Cardiology Therapies		396		921		1,316		372		848		1,220	
Watchman		653		70		723		552		55		607	
Cardiac Rhythm Management		706		446		1,151		702		412		1,114	
Electrophysiology		429		300		729		170		199		370	
Cardiology		2,183		1,736		3,919		1,796		1,514		3,310	
Peripheral Interventions		608		555		1,163		560		478		1,039	
Cardiovascular		2,791		2,291		5,082		2,356		1,993		4,349	
Total Net Sales	\$	4,724	\$	3,252	\$	7,977	\$	4,113	\$	2,876	\$	6,988	

Refer to *Note K - Segment Reporting* for information on our reportable segments.

	Three Months Ended June 30,					Six Months Ended June 30,					
Geographic Regions		2024		2023		2024		2023			
U.S.	\$	2,466	\$	2,110	\$	4,724	\$	4,113			
Europe, Middle East and Africa		822		723		1,625		1,435			
Asia-Pacific		670		626		1,317		1,174			
Latin America and Canada		162		140		311		267			
Total Net Sales	\$	4,120	\$	3,599	\$	7,977	\$	6,988			
Emerging Markets ⁽¹⁾	\$	680	\$	592	\$	1,328	\$	1,121			

⁽¹⁾ Periodically, we assess our list of Emerging Markets countries, and effective January 1, 2023, modified our list to include all countries except the United States, Western and Central Europe, Japan, Australia, New Zealand and Canada.

Deferred Revenue

Contract liabilities are classified within *Other current liabilities* and *Other long-term liabilities* within our accompanying unaudited consolidated balance sheets. Our deferred revenue balance was \$599 million as of June 30, 2024 and \$577 million as of December 31, 2023. Our contract liabilities are primarily composed of deferred revenue related to the LATITUDE™ Patient Management System within our Cardiology business, for which revenue is recognized over the average service period based on device and patient longevity. Our contract liabilities also include deferred revenue related to the LUX-Dx™ Insertable Cardiac Monitor system, also within our Cardiology business, for which revenue is recognized over the average service period based on device longevity and usage. We recognized revenue of \$59 million in the second quarter and \$120 million in the first six months of 2024 that was included in the above contract liability balance as of December 31, 2023. We have elected not to disclose the transaction price allocated to unsatisfied performance obligations when the original expected contract duration is one year or less. In addition, we have not identified material unfulfilled performance obligations for which revenue is not currently deferred.

Variable Consideration

For additional information on variable consideration, refer to *Note A – Significant Accounting Policies* to our audited financial statements contained in Item 8. Financial Statements and Supplementary Data of our most recent Annual Report on Form 10-K.

NOTE M – CHANGES IN OTHER COMPREHENSIVE INCOME

The following tables provide the reclassifications out of Other comprehensive income (loss), net of tax attributable to Boston Scientific common stockholders:

(in millions)	Foreign Currency Translation Adjustment	Net Change in Derivative Financial Instruments	Net Change in Defined Benefit Pensions and Other Items	Total			
Balance as of March 31, 2024	\$ (34)	\$ 175	\$ (8)	\$ 132			
Other comprehensive income (loss) before reclassifications	50	24	0	75			
(Income) loss amounts reclassified from accumulated other comprehensive income	(3)	(40)	(0)	(44)			
Total other comprehensive income (loss)	47	(16)	0	31			
Balance as of June 30, 2024	\$ 13	\$ 159	\$ (8)	\$ 164			

(in millions)	Foreign Currency Derivative Defined Benefit Translation Financial Pensions and Adjustment Instruments Other Items			Derivative Financial		efined Benefit Pensions and	Total
Balance as of March 31, 2023	\$	(43)	\$	225	\$	(4)	\$ 178
Other comprehensive income (loss) before reclassifications		17		58		_	74
(Income) loss amounts reclassified from accumulated other comprehensive income		(2)		(42)		(0)	(44)
Total other comprehensive income (loss)		15		15		(0)	30
Balance as of June 30, 2023	\$	(28)	\$	241	\$	(4)	\$ 208

(in millions)	Tran	Currency slation stment	Net Change in Derivative Financial Instruments	Γ	Net Change in Defined Benefit Pensions and Other Items	Total
Balance as of December 31, 2023	\$	(96)	\$ 154	\$	(8)	\$ 49
Other comprehensive income (loss) before reclassifications		115	84		0	199
(Income) loss amounts reclassified from accumulated other comprehensive income		(6)	(78)		(0)	(84)
Total other comprehensive income (loss)		108	6		0	114
Balance as of June 30, 2024	\$	13	\$ 159	\$	(8)	\$ 164

(in millions)	Foreign Currency Translation Adjustment	Net Change in Derivative Financial Instruments	Net Change in Defined Benefit Pensions and Other Items	Total
Balance as of December 31, 2022	\$ (1)	\$ 269	\$ 1	\$ 269
Other comprehensive income (loss) before reclassifications	(24)	67	(5)	39
(Income) loss amounts reclassified from accumulated other comprehensive income	(4)	(95)	(1)	(100)
Total other comprehensive income (loss)	(28)	(28)	(5)	(61)
Balance as of June 30, 2023	\$ (28)	\$ 241	\$ (4)	\$ 208

Refer to Note D – Hedging Activities and Fair Value Measurements for further detail on our net investment hedges recorded in Foreign currency translation adjustment and our cash flow hedges recorded in Net change in derivative financial instruments.

NOTE N – NEW ACCOUNTING PRONOUNCEMENTS

Periodically, new accounting pronouncements are issued by the FASB or other standard setting bodies. Recently issued standards typically do not require adoption until a future effective date. Prior to their effective date, we evaluate the pronouncements to determine the potential effects of adoption on our accompanying unaudited consolidated financial statements. During the first six months of 2024, we implemented the following standard on a prospective basis, which did not have a material impact on our unaudited consolidated financial statements:

ASC Update No. 2022-03

ASC Update No. 2022-03, Fair Value Measurement (Topic 820): Fair Value Measurement of Equity Securities Subject to Contractual Sale Restrictions clarifies the guidance in Topic 820 related to measuring the fair value of an equity security subject to contractual restrictions that prohibit the sale of an equity security, as well as introduces new disclosure requirements for these types of equity securities.

Standards to be Implemented

In November 2023, the FASB issued ASC Update No. 2023-07, Segment Reporting (Topic 280): Improvements to Reportable Segment Disclosures. Update No. 2023-07 requires disclosure, on an annual and interim basis, of significant segment expenses that are regularly provided to the CODM and included within each reported measure of segment profit or loss in addition to disclosure of amounts for other segment items and a description of its composition. Update No. 2023-07 is effective for fiscal years beginning after December 15, 2023, and interim periods within fiscal years beginning after December 15, 2024. As this accounting standard update impacts disclosures only, we do not expect the adoption to have a material impact on our unaudited consolidated financial statements.

In December 2023, the FASB issued ASC Update No. 2023-09, Income Taxes (Topic 740): Improvements to Income Tax Disclosures. Update No. 2023-09 aims to enhance the transparency and decision usefulness of income tax disclosures. Update No. 2023-09 modifies the rules on income tax disclosures to require entities to disclose (1) specific categories in the rate reconciliation, (2) the income or loss from continuing operations before income tax expense or benefit (separated between domestic and foreign) and (3) income tax expense or benefit from continuing operations (separated by federal, state, and foreign). ASU 2023-09 also requires entities to disclose their income tax payments to international, federal, state and local jurisdictions, among other changes. Update No. 2023-09 is effective for fiscal years beginning after December 15, 2024. We expect to adopt Update No. 2023-09 prospectively. As this accounting standard update impacts disclosures only, we do not expect the adoption to have a material impact on our unaudited consolidated financial statements.

No other new accounting pronouncements issued or effective in the period had or are expected to have a material impact on our accompanying unaudited consolidated financial statements.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Introduction

Boston Scientific Corporation is a global developer, manufacturer and marketer of medical devices that are used in a broad range of interventional medical specialties. Our mission is to transform lives through innovative medical solutions that improve the health of patients around the world. As a medical technology leader for more than 40 years, we have advanced the practice of less-invasive medicine by helping physicians and other medical professionals diagnose and treat a wide range of diseases and medical conditions and improve patients' quality of life by providing alternatives to surgery and other medical procedures that are typically traumatic to the body. We advance science for life by providing a broad range of high performance solutions to address unmet patient needs and reduce the cost of health care. When used in this report, the terms "we," "us," "our" and "the Company" mean Boston Scientific Corporation and its divisions and subsidiaries.

Financial Summary

Three Months Ended June 30, 2024

Our net sales for the second quarter of 2024 were \$4.120 billion, compared to \$3.599 billion for the second quarter of 2023. This increase of \$521 million, or 14.5 percent, included operational net sales growth of 16.1 percent and the negative impact of 160 basis points from foreign currency fluctuations. Operational net sales growth included organic² net sales growth of 14.7 percent and the positive impact of 140 basis points from our majority stake investment in Acotec Scientific Holdings Limited (Acotec) and the acquisitions of Relievant Medsystems, Inc. (Relievant) and the endoluminal vacuum therapy portfolio of B. Braun Medical Inc. (Braun) during the first and fourth quarters of 2023 and first quarter of 2024, respectively, for which there is less than a full period of comparable net sales. The increase in our net sales was primarily driven by strong commercial execution across our businesses, led by the rapid adoption of our FarapulseTM Pulsed Field Ablation System following U.S. launch in early 2024. Refer to *Quarterly Results and Business Overview* for a discussion of our net sales by business.

Our reported net income attributable to Boston Scientific common stockholders for the second quarter of 2024 was \$324 million, or \$0.22 per diluted share. Our reported results for the second quarter of 2024 included certain charges and/or credits totaling \$589 million (after-tax), or \$0.40 per diluted share. Excluding these items, adjusted net income attributable to Boston Scientific common stockholders³ was \$914 million, or \$0.62 per diluted share.

Our reported net income attributable to Boston Scientific common stockholders for the second quarter of 2023 was \$261 million, or \$0.18 per diluted share. Our reported results for the second quarter of 2023 included certain charges and/or credits totaling \$516 million (after-tax), or \$0.35 per diluted share. Excluding these items, adjusted net income attributable to Boston Scientific common stockholders³ was \$777 million, or \$0.53 per diluted share.

 $^{^{1}}$ Operational net sales growth excludes the impact of foreign currency fluctuations.

²Organic net sales growth excludes the impact of foreign currency fluctuations and net sales attributable to acquisitions and divestitures for which there are less than a full period of comparable net sales.

³Adjusted measures, including operational and organic net sales growth and adjusted net income attributable to Boston Scientific common stockholders, exclude certain items required by generally accepted accounting principles in the United States (GAAP), are not prepared in accordance with GAAP and should not be considered in isolation from, or as a replacement for, the most directly comparable GAAP measure. Refer to *Additional Information* for a discussion of management's use of these non-GAAP financial measures.

The following is a reconciliation of our results of operations prepared in accordance with GAAP to those adjusted results considered by management. Refer to *Quarterly Results and Business Overview* and *Additional Information* for a discussion of these reconciling items:

2 ,	J		Three M	onths Ended June	2 30, 2024		
(in millions, except per share data)	Income (Loss) Before Income Taxes	Income Tax Expense (Benefit)	Net Income (Loss)	Preferred Stock Dividends	Net Income (Loss) Attributable to Noncontrolling Interests	Net Income (Loss) Attributable to Boston Scientific Common Stockholders	Impact per Share
Reported	\$ 420	\$ 98 \$	322	s — s	(2)	\$ 324	\$ 0.22
Non-GAAP adjustments:							
Amortization expense	213	29	184	_	2	182	0.12
Goodwill and other intangible asset impairment charges	276	33	243	_	_	243	0.16
Acquisition/divestiture-related net charges (credits)	49	11	38	_	_	38	0.03
Restructuring and restructuring-related net charges (credits)	50	7	44	_	_	44	0.03
Investment portfolio net losses (gains) and impairments	31	3	29	_	_	29	0.02
European Union (EU) Medical device regulation (MDR) implementation costs	12	2	10	_	_	10	0.01
Deferred tax expenses (benefits)	_	(44)	44	_	_	44	0.03
Adjusted	\$ 1,051	\$ 138 \$	913	s — s	(0)	\$ 914	\$ 0.62

		Three Months Ended June 30, 2023							
(in millions, except per share data)	Income (Before II Taxo	icomé	Income Tax Expense (Benefit)	Net Income (Loss)	Preferred Stock Dividends	Net Income (Loss) Attributable to Noncontrolling Interests	Net Income (Loss) Attributable to Boston Scientific Common Stockholders	Impact per Share ⁽⁴⁾	
Reported	\$	426	\$ 156 \$	270	\$ (9) \$	_	\$ 261	\$ 0.18	
Non-GAAP adjustments:									
Amortization expense		210	28	182	_	_	182	0.12	
Goodwill and other intangible asset impairment charges		57	3	54	_	_	54	0.04	
Acquisition/divestiture-related net charges (credits)		118	(57)	175	_	_	175	0.12	
Restructuring and restructuring-related net charges (credits)		42	7	35	_	_	35	0.02	
Investment portfolio net losses (gains) and impairments		(2)	(6)	5	_	_	5	0.00	
European Union (EU) Medical device regulation (MDR) implementation costs		20	3	17	_	_	17	0.01	
Deferred tax expenses (benefits)		_	(47)	47	_	_	47	0.03	
Discrete tax items		_	(1)	1	_	_	1	0.00	
Adjusted	\$	871	\$ 86 \$	786	\$ (9) \$	_	\$ 777	\$ 0.53	

⁴ For the three months ended June 30, 2023, the effect of assuming the conversion of our 5.50% Mandatory Convertible Preferred Stock, Series A (MCPS) into shares of common stock was anti-dilutive, and therefore excluded from the calculation of *Net income (loss) per common share - diluted* earnings per share (EPS). Accordingly, GAAP *Net income (loss)* and Adjusted net income were reduced by cumulative *Preferred stock dividends*, as presented within our unaudited consolidated statements of operations, for purposes of calculating GAAP *Net income attributable to Boston Scientific common stockholders*. On June 1, 2023, all outstanding shares of MCPS automatically converted into shares of common stock.

Six Months Ended June 30, 2024

Our net sales for the first six months of 2024 were \$7.977 billion, compared to \$6.988 billion for the first six months of 2023. This increase of \$988 million, or 14.1 percent, included operational¹ net sales growth of 15.5 percent and the negative impact of 140 basis points from foreign currency fluctuations. Operational net sales growth included organic² net sales growth of 13.9 percent and the positive impact of 160 basis points from our majority stake investment in Acotec and the acquisitions of Apollo Endosurgery, Inc. (Apollo), Relievant and the endoluminal vacuum therapy portfolio of Braun during the first, second and fourth quarters of 2023 and first quarter of 2024, respectively, for which there is less than a full period of comparable sales. The increase in our net sales was primarily driven by strong commercial execution across our businesses, led by the rapid adoption of our FarapulseTM Pulsed Field Ablation System following U.S. launch in early 2024. Refer to *Quarterly Results and Business Overview* for a discussion of our net sales by business.

Our reported net income attributable to Boston Scientific common stockholders for the first six months of 2024 was \$819 million, or \$0.55 per diluted share. Our reported results for the first six months of 2024 included certain charges and/or credits totaling \$926 million (after-tax), or \$0.62 per diluted share. Excluding these items, adjusted net income attributable to Boston Scientific common stockholders³ for the first six months of 2024 was \$1.745 billion, or \$1.18 per diluted share.

Our reported net income attributable to Boston Scientific common stockholders for the first six months of 2023 was \$561 million, or \$0.39 per diluted share. Our reported results for the first six months of 2023 included certain charges and/or credits totaling \$889 million (after-tax), or \$0.61 per diluted share. Excluding these items, adjusted net income attributable to Boston Scientific common stockholders³ for the first six months of 2023 was \$1.449 billion, or \$1.00 per diluted share.

¹Operational net sales growth excludes the impact of foreign currency fluctuations.

²Organic net sales growth excludes the impact of foreign currency fluctuations and net sales attributable to acquisitions and divestitures for which there are less than a full period of comparable net sales.

³Adjusted measures, including operational and organic net sales growth and adjusted net income attributable to Boston Scientific common stockholders, exclude certain items required by generally accepted accounting principles in the United States (GAAP), are not prepared in accordance with GAAP and should not be considered in isolation from, or as a replacement for, the most directly comparable GAAP measure. Refer to *Additional Information* for a discussion of management's use of these non-GAAP financial measures.

The following is a reconciliation of our results of operations prepared in accordance with GAAP to those adjusted results considered by management. Refer to *Quarterly Results and Business Overview* and *Additional Information* for a discussion of these reconciling items:

	Six Months Ended June 30, 2024								
(in millions, except per share data)	ome (Loss) ore Income Taxes	Income Tax Expense (Benefit)	Net Income (Loss)	Preferred Stock Dividends	Net Income (Loss) Attributable to Noncontrolling Interests	Net Income (Loss) Attributable to Boston Scientific Common Stockholders	Impact per Share		
Reported	\$ 1,028	\$ 213 \$	815	s — s	(4)	\$ 819	\$ 0.55		
Non-GAAP adjustments:									
Amortization expense	427	58	369	_	4	364	0.25		
Goodwill and other intangible asset impairment charges	276	33	243	_	_	243	0.16		
Acquisition/divestiture-related net charges (credits)	112	(2)	115	_	_	115	0.08		
Restructuring and restructuring-related net charges (credits)	97	13	84	_	_	84	0.06		
Investment portfolio net losses (gains) and impairments	18	(0)	18	_	_	18	0.01		
European Union (EU) Medical device regulation (MDR) implementation costs	26	3	22	_	_	22	0.01		
Deferred tax expenses (benefits)	_	(81)	81	_	_	81	0.05		
Adjusted	\$ 1,983	\$ 237 \$	1,746	s — s	1	\$ 1,745	\$ 1.18		

			Six Mo	nths Ended June 3	30, 2023		
(in millions, except per share data)	Income (Loss) Before Income Taxes	Income Tax Expense (Benefit)	Net Income (Loss)	Preferred Stock Dividends	Net Income (Loss) Attributable to Noncontrolling Interests	Net Income (Loss) Attributable to Boston Scientific Common Stockholders	Impact per Share ⁽⁴⁾
Reported	\$ 870	\$ 287 \$	584	\$ (23) \$	_	\$ 561	\$ 0.39
Non-GAAP adjustments:							
Amortization expense	412	56	357	_	_	357	0.25
Goodwill and other intangible asset impairment charges	57	3	54	_	_	54	0.04
Acquisition/divestiture-related net charges (credits)	178	(64)	242	_	_	242	0.17
Restructuring and restructuring-related net charges (credits)	86	14	71	_	_	71	0.05
Investment portfolio net losses (gains) and impairments	19	(2)	20	_	_	20	0.01
European Union (EU) Medical device regulation (MDR) implementation costs	36	5	31	_	_	31	0.02
Deferred tax expenses (benefits)	_	(88)	88	_	_	88	0.06
Discrete tax items	_	(26)	26	_		26	0.02
Adjusted	\$ 1,659	\$ 186 \$	3 1,472	\$ (23) \$	_	\$ 1,449	\$ 1.00

⁽⁴⁾ For the first six months ended June 30, 2023, the effect of assuming the conversion of MCPS into shares of common stock was anti-dilutive, and therefore excluded from the calculation of *Net income (loss) per common share - diluted* EPS. Accordingly, GAAP *Net income (loss)* and Adjusted net income were reduced by cumulative *Preferred stock dividends*, as presented within our unaudited consolidated statements of operations, for purposes of calculating GAAP *Net income attributable to Boston Scientific common stockholders*. On June 1, 2023, all outstanding shares of MCPS automatically converted into shares of common stock.

Quarterly Results and Business Overview

The following section describes our net sales and results of operations by reportable segment and business. For additional information on our businesses and product offerings, refer to *Item 1. Business* of our most recent Annual Report on Form 10-K.

		Three Months Ended June 30,					
(in millions)	2	2024		2023	Increase/(Decrease)		
Endoscopy	\$	676	\$	631	7.1%		
Urology		525		485	8.2%		
Neuromodulation		282		244	15.5%		
MedSurg		1,483		1,360	9.0%		
Cardiology		2,047		1,704	20.1%		
Peripheral Interventions		590		535	10.2%		
Cardiovascular		2,637		2,239	17.8%		
Net Sales	\$	4,120	\$	3,599	14.5%		

	Six Months Ended J	June 30,	
(in millions)	2024	2023	Increase/(Decrease)
Endoscopy	\$ 1,318 \$	1,207	9.2%
Urology	1,038	954	8.7%
Neuromodulation	 539	478	12.6%
MedSurg	 2,895	2,640	9.6%
Cardiology	3,919	3,310	18.4%
Peripheral Interventions	1,163	1,039	12.0%
Cardiovascular	 5,082	4,349	16.9%
Net Sales	\$ 7,977 \$	6,988	14.1%

MedSurg

Endoscopy

Our Endoscopy business develops and manufactures devices to diagnose and treat a broad range of gastrointestinal (GI) and pulmonary conditions with innovative, less-invasive technologies. Net sales of Endoscopy products of \$676 million during the second quarter and \$1.318 billion during the first six months of 2024 represented 16 percent and 17 percent of our consolidated net sales, respectively. Endoscopy net sales increased \$45 million, or 7.1 percent, during the second quarter and \$111 million, or 9.2 percent, during the first six months of 2024, compared to the prior year periods. During the second quarter of 2024, this increase included operational net sales growth of 8.4 percent and a negative impact of 130 basis points from foreign currency fluctuations, compared to the prior year period. During the first six months of 2024, this increase included operational net sales growth of 10.2 percent and a negative impact of 100 basis points from foreign currency fluctuations, compared to the prior year period.

Operational net sales growth included organic net sales growth of 7.9 percent during the second quarter of 2024 and 8.8 percent for the first six months of 2024, and the positive impact of 50 and 140 basis points, respectively, from our acquisition of Apollo and divestiture of our pathology business in the second quarter of 2023, and our acquisition of the endoluminal vacuum therapy portfolio of Braun in the first quarter of 2024. Organic net sales growth in both periods was primarily driven by our biliary franchise led by our AXIOSTM Stent and Delivery System and our endoluminal surgery franchise.

Urology

Our Urology business develops and manufactures devices to treat various urological conditions for both male and female anatomies, including kidney stones, benign prostatic hyperplasia (BPH), prostate cancer, erectile dysfunction and incontinence. Net sales of Urology products of \$525 million during the second quarter and \$1.038 billion during the first six months of 2024 represented 13 percent of our consolidated net sales in both periods. Urology net sales increased \$40 million, or 8.2 percent, during the second quarter and \$83 million, or 8.7 percent, during the first six months of 2024, compared to the prior year periods. During the second quarter of 2024, this increase included operational net sales growth of 9.1 percent and a negative impact of 90 basis points from foreign currency fluctuations, compared to the prior year period. During the first six months of 2024, this increase included operational net sales growth of 9.5 percent and a negative impact of 70 basis points from foreign currency fluctuations, compared to the prior year period.

Operational net sales growth in both periods was primarily driven by our stone management franchise, led by our Lumenis PulseTM Holmium Laser Systems with MOSESTM Technology, as well as our core stone and prosthetic urology franchises.

Neuromodulation

Our Neuromodulation business develops and manufactures devices to treat various neurological movement disorders and manage chronic pain. Net sales of Neuromodulation products of \$282 million during the second quarter and \$539 million during the first six months of 2024 represented 7 percent of our consolidated net sales in both periods. Neuromodulation net sales increased \$38 million, or 15.5 percent during the second quarter and \$60 million, or 12.6 percent during the first six months of 2024, compared to the prior year periods. During the second quarter of 2024, this increase included operational net sales growth of 16.2 percent and a negative impact of 70 basis points from foreign currency fluctuations, compared to the prior year period. During the first six months of 2024, this increase included operational net sales growth of 13.1 percent and a negative impact of 50 basis points from foreign currency fluctuations, compared to the prior year period.

Operational net sales growth included organic net sales growth of 3.7 percent during the second quarter of 2024 and 1.2 percent during the first six months of 2024, and the positive impact of 1,250 and 1,180 basis points, respectively, from our acquisition of Relievant in the fourth quarter of 2023. Organic net sales growth in the second quarter of 2024 was primarily driven by improved performance in our U.S. pain franchise from our radiofrequency ablation portfolio. Organic net sales growth in the first six months of 2024 was primarily driven by improved performance in our U.S. pain franchise from our radiofrequency ablation portfolio and our deep brain stimulation franchise.

Cardiovascular

Cardiology

Our Cardiology business develops and manufactures devices and medical technologies for diagnosing and treating a variety of diseases and abnormalities of the heart. Net sales of Cardiology products of \$2.047 billion during the second quarter and \$3.919 billion for the first six months of 2024 represented 50 percent and 49 percent of our consolidated net sales, respectively. Cardiology net sales increased \$343 million, or 20.1 percent, during the second quarter and \$609 million, or 18.4 percent, during the first six months of 2024, compared to the prior year periods. During the second quarter of 2024, this increase included operational net sales growth of 22.0 percent and a negative impact of 190 basis points from foreign currency fluctuations, compared to the prior year period. During the first six months of 2024, this increase included operational net sales growth of 20.0 percent and a negative impact of 160 basis points from foreign currency fluctuations, compared to the prior year period.

Operational net sales growth in both periods was primarily driven by growth of our electrophysiology business, led by our FarapulseTM Pulsed Field Ablation System, continued market penetration of Left Atrial Appendage Closure (LAAC) procedures with our WATCHMAN FLXTM LAAC Device and our WATCHMAN FLXTM Pro LAAC Device, as well as our access solutions portfolio and our percutaneous coronary intervention guidance franchises.

Peripheral Interventions

Our Peripheral Interventions business develops and manufactures products to diagnose and treat peripheral arterial and venous diseases, as well as products to diagnose, treat and ease various forms of cancer. Net sales of Peripheral Interventions products of \$590 million during the second quarter and \$1.163 billion for the first six months of 2024 represented 14 percent and 15 percent of our consolidated net sales, respectively. Peripheral Interventions net sales increased \$55 million, or 10.2 percent, during the second quarter and \$125 million, or 12.0 percent, during the first six months of 2024, compared to the prior year periods. During the second quarter of 2024, this increase included operational net sales growth of 12.3 percent and a negative impact of 210 basis points from foreign currency fluctuations, compared to the prior year period. During the first six months of 2024, this increase included operational net sales growth of 14.1 percent and a negative impact of 210 basis points from foreign currency fluctuations, compared to the prior year period.

Operational net sales growth included organic net sales growth of 9.4 percent during the second quarter of 2024 and 10.4 percent during the first six months of 2024, and the positive impact of 280 and 370 basis points, respectively, from our majority stake investment in Acotec which we acquired in the first quarter of 2023. Organic net sales growth in both periods was primarily driven by our interventional oncology franchise led by our TherasphereTM Y-90 Radioactive Glass Microspheres and EMBOLDTM Fibered Coil, as well as our drug-eluting portfolio within our vascular franchise led by our RangerTM Drug Coated Balloon.

Emerging Markets

As part of our strategic imperative to drive global expansion, we are seeking to grow net sales and market share by expanding our global presence, including in Emerging Markets. Periodically, we assess our list of Emerging Markets countries, and effective January 1, 2023, modified our list to include all countries except the United States, Western and Central Europe, Japan, Australia, New Zealand and Canada.

Our Emerging Markets' net sales represented 16 percent and 17 percent of our consolidated net sales during the second quarter and first six months of 2024, respectively, and 16 percent during the second quarter and first six months of 2023. During the second quarter of 2024, our Emerging Markets net sales grew 14.8 percent on a reported basis, which included operational net sales growth of 19.3 percent and a negative impact of 460 basis points from foreign currency fluctuations, compared to the prior year period. During the first six months of 2024, our Emerging Markets net sales grew 18.5 percent on a reported basis, which included operational net sales growth of 23.5 percent and a negative impact of 500 basis points from foreign currency fluctuations, compared to the prior year period. Operational growth in both periods was driven primarily by growth in China, fueled by the breadth of our portfolio and focus on innovation and strong commercial execution.

Economic Environment

Our business has been impacted by global supply chain disruptions, which have continued to improve in recent quarters, however challenges still exist. We have experienced, and may continue to experience, increases in cost and limited availability of certain raw materials, components, and other inputs necessary to manufacture and distribute our products due to constraints and continued inflation within the global supply chain, as well as increases in wage costs and the cost and time to distribute our products. Uncertainty around inflationary pressures, interest rates, monetary policy and changes in tax laws could potentially cause new, or exacerbate existing, economic challenges that we may face, including the impact of foreign currency fluctuations on our results of operations, or result in an economic downtown or recession, which could negatively impact our business operations and results. Existing and future potential geopolitical dynamics, including matters related to the Russia/Ukraine war, Israel/Hamas war, as well as tension in the Taiwan strait, may create economic, supply chain, energy, and other challenges, including disruptions to business operations, which impact, and may in the future negatively impact, our business. In particular, international conflicts could create instability, have and may further result in sanctions, tariffs, and other measures that restrict international trade and may negatively affect our business operations and results.

Gross Profit

Our *Gross profit* was \$2.850 billion for the second quarter of 2024, \$2.542 billion for the second quarter of 2023, \$5.498 billion for the first six months of 2024 and \$4.891 billion for the first six months of 2023. The following is a reconciliation of our gross profit margin and a description of the drivers of the changes from period to period:

	Percentage of	of Net Sales
	Three Months	Six Months
Gross profit margin - period ended June 30, 2023	70.6%	70.0%
Sales pricing, volume and mix	(0.3)	0.4
Net impact of foreign currency fluctuations	0.2	0.0
All other, including inventory charges and other period expenses	(1.3)	(1.4)
Gross profit margin - period ended June 30, 2024	69.2%	68.9%

The primary factors contributing to the decrease in our gross profit margin in the second quarter of 2024, as compared to the same period in the prior year, were inventory charges, including related to the POLARxTM cryoablation system given the strong commercial adoption of our FarapulseTM Pulsed Field Ablation System in the U.S., and other period expenses. These impacts were partially offset by favorable foreign currency fluctuations. The decrease in our gross profit margin in the first six months of 2024, as compared to the same period in the prior year, resulted primarily from inventory charges, including related to the POLARxTM cryoablation system given the strong commercial adoption of our FarapulseTM Pulsed Field Ablation System, and other period expenses, partially offset by increased sales of higher margin products.

Operating Expenses

The following table provides a summary of our key operating expenses:

	Three Months Ended June 30,					Six Months Ended June 30,							
		2024			2(2023		20	2024		20	2023	
			% of Net			% of Net			% of Net			% of Net	
(in millions)		\$	Sales		\$	Sales		\$	Sales		\$	Sales	
Selling, general and administrative expenses	\$	1,446	35.1 %	\$	1,354	37.6 %	\$	2,810	35.2 %	\$	2,570	36.8 %	
Research and development expenses		383	9.3 %		359	10.0 %		749	9.4 %		695	10.0 %	

Selling, General and Administrative expenses (SG&A Expenses)

During the second quarter of 2024, SG&A expenses increased \$92 million, or 7 percent, compared to the prior year period and were 250 basis points lower as a percentage of net sales. During the first six months of 2024, SG&A expenses increased \$241 million, or 9 percent, compared to the prior year period and were 160 basis points lower as a percentage of net sales. The increase in SG&A expenses in both periods was primarily due to higher selling costs driven by higher global net sales and costs to support recent and upcoming product launches, including the FarapulseTM Pulsed Field Ablation System.

Research and Development expenses (R&D Expenses)

We remain committed to advancing medical technologies and investing in meaningful R&D projects across our businesses. During the second quarter of 2024, *R&D expenses* increased \$24 million, or 7 percent, compared to the prior year period and were 70 basis points lower as a percentage of net sales. During the first six months of 2024, *R&D expenses* increased \$53 million, or 8 percent, compared to the prior year period, and were 60 basis points lower as a percentage of net sales. *R&D expenses* increased in both periods as a result of investments across our businesses in order to maintain a pipeline of new products that we believe will contribute to profitable sales growth.

Other Operating Expenses

The following provides a summary of certain of our other operating expenses, which are excluded by management for purposes of evaluating operating performance; refer to *Additional Information* for a further description.

Amortization Expense

During the second quarter of 2024, *Amortization expense* increased \$3 million, or 2 percent, compared to the prior year period. In the first six months of 2024, *Amortization expense* increased \$14 million, or 3 percent, compared to the prior year period. The increase in *Amortization expense* during both periods was driven by the addition of amortizable intangible assets associated with our recent acquisitions.

Intangible Asset Impairment Charges

We recorded *Intangible asset impairment charges* of \$276 million in the second quarter and first six months of 2024 and \$57 million in the second quarter and first six months of 2023. The impairment charges recorded in 2024 were associated with amortizable intangible assets established in connection with our acquisitions of Cryterion Medical, Inc. (Cryterion) and Devoro Medical, Inc. (Devoro), which were integrated into our Electrophysiology and Peripheral Interventions business units, respectively. Intangible assets acquired from Cryterion were impaired due to strong commercial adoption of our FarapulseTM Pulsed Field Ablation System and the resulting lower revenue projections and cannibalization of our cryoablation business in major markets like the U.S. Intangible assets acquired from Devoro were impaired following management's decision to cancel the related program in the second quarter of 2024. Refer to *Note C – Goodwill and Other Intangible Assets* to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q and *Critical Accounting Policies and Estimates* contained in Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations of our most recent Annual Report on Form 10-K for additional details and a discussion of key assumptions used in our intangible asset impairment testing and future events that could have a negative impact on the recoverability of our intangible assets.

Contingent Consideration Net Expense (Benefit)

To recognize changes in the fair value of our contingent consideration liability, we recorded net charges of \$2 million in the second quarter of 2024, \$19 million in the second quarter of 2023, \$18 million in the first six months of 2024 and \$31 million in the first six months of 2023. The net charges recorded in the second quarter and first six months of 2024 related to an increase in expected payments for achievement of revenue-based earn outs as a result of overperformance. In addition, we made payments of \$133 million and \$73 million associated with prior acquisitions during the first six months of 2024 and 2023, respectively, following the achievement of revenue-based earnouts. Refer to *Note B – Acquisitions and Strategic Investments* to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q for additional details related to our contingent consideration arrangements.

Restructuring and Restructuring-related Net Charges (Credits)

On February 22, 2023, our Board of Directors approved, and we committed to, a new global restructuring program (the 2023 Restructuring Plan). The 2023 Restructuring Plan will advance our Global Supply Chain Optimization strategy, which is intended to simplify our manufacturing and distribution network by transferring certain production lines among facilities and drive operational efficiencies and resiliency. Key activities under the 2023 Restructuring Plan will also include optimizing certain functional capabilities to achieve cost synergies and better support business growth. For more information, refer to 2023 Restructuring Plan contained in Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations of our most recent Annual Report on Form 10-K.

Pursuant to the 2023 Restructuring Plan, we recorded restructuring charges in accordance with FASB ASC Topic 420, Exit or Disposal Cost Obligations of \$1 million and \$5 million in the second quarter and first six months of 2024, respectively. The restructuring reserve balance was \$29 million as of June 30, 2024. In addition, we recorded restructuring-related charges of \$49 million and \$92 million in the second quarter and first six months of 2024, respectively, primarily within Cost of products sold and SG&A Expenses. During the second quarter and first six months of 2023, we recorded restructuring charges of \$16 million and \$36 million, respectively, and restructuring-related charges of \$26 million and \$50 million, respectively, and the restructuring reserve balance as of December 31, 2023 was \$41 million.

Litigation-related Net Charges (Credits)

We did not record any litigation-related net charges (credits) during the second quarter and first six months of 2024 or 2023. We record certain legal and product liability charges, credits and costs of defense, which we consider to be unusual or infrequent and significant as *Litigation-related net charges* (credits) within our accompanying unaudited consolidated financial statements. All other legal and product liability charges, credits and costs are recorded within SG&A expenses.

We continue to assess certain litigation and claims to determine the amounts, if any, that management believes will be paid as a result of such claims and litigation, and therefore, additional losses may be accrued and paid in the future, which could materially adversely impact our operating results, cash flows and/or our ability to comply with the financial covenant required by our credit arrangements. Refer to *Note H - Commitments and Contingencies* to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q for discussion of our material legal proceedings.

Interest Expense

The following table provides a summary of our *Interest expense* and average borrowing rate:

	Three Months Ended June 30,			Six Months Ended June 30,			
	 2024		2023	2024		2023	
Interest expense (in millions)	\$ (77)	\$	(70)	\$ (146)	\$	(135)	
Average borrowing rate	2.8 %		2.9 %	2.8 %		2.8 %	

Interest expense increased during the second quarter and first six months of 2024 compared to the prior year period primarily due to increased debt from the registered public offering of \in 2.000 billion in aggregate principal amount of euro-denominated senior notes during the first quarter of 2024. Our average borrowing rate remained relatively flat during the second quarter and first six months of 2024 compared to the prior year period. Refer to Liquidity and Capital Resources and Note E-Contractual Obligations and Commitments to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q for more information regarding our debt obligations.

Tax Rate

Our effective tax rate from continuing operations is presented below:

	Three Months End	led June 30,	Six Months Ended June 30,		
	2024	2023	2024	2023	
Effective tax rate from continuing operations	23.4 %	36.6 %	20.7 %	32.9 %	

Our reported tax rate is affected by recurring items such as the amount of our earnings subject to differing tax rates in foreign jurisdictions and the impact of certain receipts and charges that are taxed at rates that differ from our effective tax rate.

The primary reasons for the difference between the statutory rate and our reported tax rate in the second quarter and first six months of 2024 relate to certain charges for acquisition expenses, as well as certain benefits for intangible asset impairment charges and integration costs, and discrete benefits primarily related to stock-based compensation. In the second quarter and first six months of 2023, the primary difference between the statutory tax rate and our reported tax rate relates to acquisition-related charges, as well as certain benefits for integration costs, and discrete tax items primarily related to unrecognized tax benefits.

Effective January 1, 2024, many countries where we do business, including the United Kingdom, Japan, South Korea, Canada and many EU member states, adopted a global minimum effective tax rate of 15% based on the Pillar Two framework issued by the Organization for Economic Cooperation and Development (OECD). Other countries where we do business are also actively considering adopting the framework or are in various stages of enacting the framework into their country's laws. While the Company continues to monitor legislative adoption of the Pillar Two rules by country, as well as for additional guidance from the OECD, there is significant uncertainty that exists regarding the interpretation of the detailed Pillar Two rules, whether such rules will be implemented consistently across taxing jurisdictions, how such rules interact with existing national tax laws and whether such rules are consistent with existing tax treaty obligations. Although the current impact of the adoption of a global minimum effective tax on our financial statements is not material, it is possible that the final adoption, implementation, and

interpretation of Pillar Two across all jurisdictions where we do business could have a material adverse impact on our financial position, results of operations, and cash flows.

Our operations presently benefit from various tax provisions of the Tax Cuts and Jobs act which are set to expire in 2025. If future legislation is unable to extend or modify these provisions, this could have a material adverse impact on our overall effective tax rate, financial condition, results of operations, and cash flows.

Critical Accounting Policies and Estimates

Our financial results are affected by the selection and application of accounting policies and methods. During the second quarter and first six months of 2024, there were no material changes to the application of critical accounting policies previously disclosed in our most recent Annual Report on Form 10-K.

Liquidity and Capital Resources

Based on our current business plan, we believe our existing balance of *Cash and cash equivalents*, future cash generated from operations, access to capital markets and existing credit facilities will be sufficient to fund our operations, invest in our infrastructure, pay our legal-related liabilities, pay taxes due, service and repay our existing debt and fund possible acquisitions for the next 12 months and for the foreseeable future.

As of June 30, 2024, we had \$2.913 billion of unrestricted *Cash and cash equivalents* on hand, including approximately \$83 million held by Acotec, a less than wholly owned entity of which we acquired a majority stake investment during the first quarter of 2023. The balance is comprised of \$2.153 billion invested in money market funds and time deposits and \$793 million in interest bearing and non-interest-bearing bank accounts. We invest excess cash on hand in short-term financial instruments that earn at market interest rates while mitigating principal risk through instrument and counterparty diversification, as well as what we believe to be prudent instrument selection. We limit our direct exposure to securities in any one industry or issuer.

In 2021, we entered into our \$2.750 billion revolving credit facility (as amended, supplemented or otherwise modified from time to time, the 2021 Revolving Credit Facility) with a global syndicate of commercial banks. On May 10, 2024, we entered into a third amendment to the 2021 Revolving Credit Facility credit agreement, which provided for, among other things, an extension of the scheduled maturity date to May 10, 2029, an amendment of the Ratings based pricing grid of the Applicable Margin, each as defined in the credit agreement, and reset the applicable date for purposes of determining the amounts of restructuring charges and restructuring-related expenses that may be excluded from consolidated Earnings Before Interest, Taxes, Depreciation and Amortization (EBITDA), as defined by the credit agreement, for purposes of our maximum leverage ratio covenant, from December 31, 2022 to March 31, 2024, as further discussed under *Financial Covenant* below. This facility provides backing for our commercial paper program, and outstanding commercial paper directly reduces borrowing capacity under the 2021 Revolving Credit Facility. There were no amounts outstanding under the 2021 Revolving Credit Facility or our commercial paper program as of June 30, 2024, resulting in an additional \$2.750 billion of available liquidity.

For additional details related to our debt obligations, including our financial covenant requirement, refer to *Note E - Contractual Obligations and Commitments* to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q.

The following provides a summary and description of our net cash inflows (outflows):

		Six Months Ended June 30,					
(in millions)	20	24	2023				
Cash provided by (used for) operating activities	\$	977 \$	848				
Cash provided by (used for) investing activities		(556)	(1,324)				
Cash provided by (used for) financing activities		1,593	(26)				

Operating Activities

During the first six months of 2024, cash provided by (used for) operating activities increased \$130 million compared to the prior year period primarily due to comparatively higher sales, operating income and slower inventory buildup due to improved macroeconomic supply chain conditions, offset by higher income tax and employee related payments.

Investing Activities

During the first six months of 2024, cash provided by (used for) investing activities included *purchases of property, plant and equipment and internal use* software of \$334 million as well as payments for investments and acquisitions of certain technologies, net of investment proceeds of \$139 million. During the first six months of 2023, cash used for investing activities included cash payments of \$1.018 billion, net of cash acquired, for the acquisition of Apollo and a majority stake investment in Acotec, as well as purchases of property, plant and equipment and internal use software of \$254 million. For more information on our acquisitions, refer to Note B – Acquisitions and Strategic Investments to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q.

Financing Activities

During the first six months of 2024, cash provided by (used for) financing activities included a registered public offering (the Offering) of €2.000 billion in aggregate principal amount of euro-denominated senior notes (the 2024 Eurobonds). The Offering resulted in cash proceeds of \$2.145 billion, net of investor discounts and issuance costs. We used the net proceeds from the Offering of the 2024 Eurobonds to fund the repayment at maturity of our 3.450% Senior Notes due March 2024 and to pay accrued and unpaid interest with respect to such notes. Additionally, we plan to use the remaining net proceeds from the Offering to fund a portion of the purchase price of our announced agreement to acquire Axonics, Inc. (Axonics) and to pay related fees and expenses, and for general corporate purposes. If the Axonics acquisition is not consummated by the applicable outside date pursuant to the merger agreement or we choose to not pursue consummation of the acquisition, we will be required to redeem each series of the notes at a special mandatory redemption price equal to 101% of the aggregate principal amount of such series of notes, plus accrued and unpaid interest, if any, to, but excluding, the date on which the notes will be redeemed. For more information, refer to *Note E − Contractual Obligations and Commitments* to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q. Cash provided by (used for) financing activities in the first six months of 2023 included *proceeds from issuances of common stock pursuant to employee stock compensation and purchase plans* of \$90 million, *cash used to net share settle employee equity awards* of \$52 million and *payments of contingent consideration previously established in purchase accounting of* \$39 million.

Financial Covenant

As of June 30, 2024, we were in compliance with the financial covenant required by the 2021 Revolving Credit Facility.

The 2021 Revolving Credit Facility includes the financial covenant requirement for all of our credit arrangements that we maintain the maximum permitted leverage ratio of 3.75 times for the remaining term. The credit agreement provides for higher leverage ratios, at our election, for the period following a Qualified Acquisition, as defined by the agreement, for which consideration exceeds \$1.000 billion. In the event of such an acquisition, for the four succeeding quarters immediately following, including the quarter in which the acquisition occurs, the maximum permitted leverage ratio is 4.75 times. It steps down for the fifth, sixth and seventh succeeding quarters to 4.50 times, 4.25 times and 4.00 times, respectively. Thereafter, a maximum leverage ratio of 3.75 times is required through the remaining term of the 2021 Revolving Credit Facility. We have elected to designate the Axonics acquisition as a Qualified Acquisition under the credit agreement, and upon closing, will increase the maximum permitted leverage ratio at that time. The agreement also provides for an exclusion of any debt incurred to fund a Qualified Acquisition, until the earlier of the acquisition close date or date of abandonment, termination or expiration of the acquisition agreement. As of June 30, 2024, we excluded from our leverage ratio calculation \$2.120 billion of debt incurred in connection with the Axonics acquisition. We believe that we have the ability to comply with the financial covenant for the next 12 months.

The financial covenant requirement, as amended on May 10, 2024, provides for an exclusion from the calculation of consolidated EBITDA, as defined by the credit agreement, through maturity, of certain charges and expenses. The credit agreement amendment reset the starting date for purposes of calculating such permitted exclusions related to restructuring charges and restructuring-related expenses from December 31, 2022 to March 31, 2024. Permitted exclusions include up to \$500 million in cash and non-cash restructuring charges and restructuring-related expenses associated with our current or future restructuring plans. As of June 30, 2024, we had \$453 million of the restructuring charge exclusion remaining. In addition, any cash litigation payments (net of any cash litigation receipts), as defined by the agreement, are excluded from the calculation of consolidated EBITDA, as defined by the agreement, provided that the sum of any excluded net cash litigation payments do not exceed \$1.000 billion plus all accrued legal liabilities as of December 31, 2022. As of June 30, 2024, we had \$1.448 billion of the litigation exclusion remaining.

Contractual Obligations and Commitments

On June 18, 2024, we announced our entry into a definitive agreement to acquire 100 percent of Silk Road Medical, Inc. (Silk Road Medical), a publicly traded medical device company that has developed an innovative platform of products to prevent stroke in patients with carotid artery disease through a minimally invasive procedure called transcarotid artery revascularization (TCAR). The purchase price is \$27.50 in cash per share, or approximately \$1.260 billion. The transaction is expected to be completed in the second half of 2024, subject to customary closing conditions. We plan to fund the acquisition through a mix of cash on hand and commercial paper, if needed. The Silk Road Medical business will be integrated into our Peripheral Interventions division.

On January 8, 2024, we announced our entry into a definitive agreement to acquire 100 percent of Axonics, a publicly traded medical technology company primarily focused on the development and commercialization of devices to treat urinary and bowel dysfunction. The purchase price is \$71.00 in cash per share, or approximately \$3.670 billion. On April 3, 2024, we and Axonics each received a request for additional information (Second Request) from the United States Federal Trade Commission (FTC) in connection with the FTC's review of the transaction. The issuance of the Second Request extends the waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (HSR Act), until 30 days after both we and Axonics have substantially complied with the Second Request, unless the waiting period is extended voluntarily by the parties or terminated earlier by the FTC. We and Axonics are responding to the Second Request and continue to work cooperatively with the FTC in its review. The transaction is expected to be completed in the second half of 2024, subject to the expiration or termination of the waiting period under the HSR Act and the satisfaction (or waiver) of other customary closing conditions. We plan to fund the acquisition through a mix of cash on hand and net proceeds from the Offering. The Axonics business will be integrated into our Urology division.

Certain of our acquisitions involve the payment of contingent consideration. Refer to *Note B – Acquisitions and Strategic Investments* to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q for further details regarding the estimated potential amount of future contingent consideration we could be required to pay associated with our acquisitions. There have been no other material changes to our contractual obligations and commitments as of June 30, 2024.

Equity

On June 1, 2023, in accordance with the terms of our MCPS, all outstanding shares of MCPS automatically converted into shares of common stock. No action by the holders of the MCPS was required in connection with the mandatory conversion. The conversion rate for each share of MCPS was 2.3834 shares of common stock. Cash was paid in lieu of fractional shares in accordance with the terms of the MCPS. An aggregate of approximately 24 million shares of common stock, including shares of common stock issued to holders of MCPS that elected to convert prior to the Mandatory Conversion Date, were issued upon conversion of the MCPS. Following the mandatory conversion of the MCPS, there were no outstanding shares of MCPS, resulting in the retirement of the annualized approximately \$55 million cash dividend payment on the MCPS.

We received \$106 million during the first six months of 2024 and \$90 million during the first six months of 2023 in proceeds from stock issuances related to our stock option and employee stock purchase plans. Proceeds from the exercise of employee stock options and employee stock purchases vary from period to period based upon, among other factors, fluctuations in the trading price of our common stock and in the exercise and stock purchase patterns of our employees.

We did not repurchase any shares of our common stock during the first six months of 2024 or 2023. On December 14, 2020, our Board of Directors approved a new stock repurchase program authorizing the repurchase of up to \$1.000 billion of our common stock. As of June 30, 2024, we had the full amount remaining available under the authorization.

Legal Matters

For a discussion of our material legal proceedings refer to *Note H - Commitments and Contingencies* to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q and *Note I - Commitments and Contingencies* to our audited financial statements contained in Item 8 of our most recent Annual Report on Form 10-K.

Recent Accounting Pronouncements

Information regarding new accounting pronouncements implemented since December 31, 2023, and relevant accounting pronouncements to be implemented in the future are included in *Note* N-New *Accounting Pronouncements* to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q.

Additional Information

Corporate Responsibility

Our sustainable environmental, social and governance (ESG) practices underpin all aspects of our global business. Our approach is aligned with the United Nations Sustainable Development Goals and our material topics and practices are informed by a broad range of internal and external stakeholders – locally, nationally and globally. Our employees around the world work with suppliers and other organizations that share our commitment to these practices that help address issues related to health inequity, economic disparity, climate change and environmental protection. Our global ESG vision and strategy is led by our ESG Executive Steering Committee and our vice president of ESG, who provides regular updates to our Board of Directors or committees thereof as appropriate. Our ESG team works closely with subject matter experts and key advisors from across the business to implement our ESG practices and determine how we measure and share progress. The importance of our ESG efforts is reinforced by a company wide scorecard that is part of our annual bonus program. For additional information on our sustainability efforts, as well as our Diversity, Equity and Inclusion initiatives, refer to our most recent Annual Report on Form 10-K. For additional information on our annual bonus plan, refer to our Proxy Statement for the 2024 Annual Meeting of Shareholders.

Cybersecurity

We have established controls and procedures to escalate enterprise level issues, including cybersecurity matters, to the appropriate management levels within our organization and our Board of Directors, or members or committees thereof, as appropriate. Under our framework, cybersecurity issues, including those involving vulnerabilities introduced by our use of third-party software, are analyzed by subject matter experts, including a crisis committee as needed in accordance with our incident response plans, for potential financial, operational, and reputational risks, based on, among other factors, the nature of the matter and breadth of impact. Matters determined to present potential material impacts to our financial results, operations, and/or reputation are immediately reported by management to the Board of Directors, or individual members or committees thereof, as appropriate, in accordance with our established escalation framework. In addition, we have established procedures to help ensure that members of management responsible for overseeing the effectiveness of disclosure controls are informed in a timely manner of known cybersecurity risks and incidents that may materially impact our operations and that timely public disclosure is made, as appropriate. For additional information on our risk management, strategy and governance around cybersecurity, refer to Part I, Item 1C. Cybersecurity in our most recent Annual Report on Form 10-K.

Stock Trading Policy

Our directors and executive officers are subject to our Stock Trading Policy, which is designed to facilitate compliance with insider trading laws and governs transactions in our common stock and related derivative securities. Our policy designates certain regular periods, dictated by release of financial results, in which trading is restricted for individuals in information-sensitive positions, including directors and executive officers. In addition, additional periods of trading restriction may be imposed as determined by the President and Chief Executive Officer, General Counsel, or Chief Financial Officer in light of material pending developments. Further, during permitted windows, certain individuals in information-sensitive positions are required to seek pre-clearance for trades from the General Counsel, who assesses whether there are any important pending developments which need to be made public before the individual may participate in the market.

Periodically, certain of our executive officers adopt written stock trading plans in accordance with Rule 10b5-1 under the Exchange Act and our own Stock Trading Policy. A Rule 10b5-1 trading plan is a written document that pre-establishes the amount, prices and dates (or formulas for determining the amounts, prices and dates) of future purchases or sales of our stock, including shares issued upon exercise of stock options or vesting of restricted stock units. These plans are entered into at a time when the person is not in possession of material non-public information about the Company. In addition to any plans described in Part II, Item 5 of this Quarterly Report on Form 10-Q, we disclose details regarding individual Rule 10b5-1 trading plans on the Investor Relations section of our website.

Use of Non-GAAP Financial Measures

To supplement our unaudited consolidated financial statements presented on a GAAP basis, we disclose certain non-GAAP financial measures, including adjusted net income (loss), adjusted net income (loss) attributable to Boston Scientific common stockholders and adjusted net income (loss) per share (EPS) that exclude certain charges (credits); operational net sales, which exclude the impact of foreign currency fluctuations; and organic net sales, which exclude the impact of foreign currency fluctuations as well as the impact of acquisitions and divestitures with less than a full period of comparable net sales. These non-GAAP financial measures are not in accordance with GAAP and should not be considered in isolation from or as a replacement for the most directly comparable GAAP financial measures. Further, other companies may calculate these non-GAAP financial measures differently than we do, which may limit the usefulness of those measures for comparative purposes.

To calculate adjusted net income (loss), adjusted net income (loss) attributable to Boston Scientific common stockholders and adjusted net income (loss) per share, we exclude certain charges (credits) from GAAP net income and GAAP net income attributable to Boston Scientific common stockholders, which include amortization expense, goodwill and other intangible asset impairment charges, acquisition/divestiture-related net charges (credits), investment portfolio net losses (gains) and impairments, restructuring and restructuring-related net charges (credits), certain litigation-related net charges (credits), EU MDR implementation costs, debt extinguishment net charges, deferred tax expenses (benefits) and certain discrete tax items. Amounts are presented after-tax using our effective tax rate, unless the amount is a significant unusual or infrequently occurring item in accordance with Financial Accounting Standards Board Accounting Standards Codification Topic 740-270-30, "General Methodology and Use of Estimated Annual Effective Tax Rate." Please refer to Part II, Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations in our most recent Annual Report on Form 10-K filed with the Securities and Exchange Commission for an explanation of each of these adjustments and the reasons for excluding each item.

The GAAP financial measures most directly comparable to adjusted net income (loss), adjusted net income (loss) attributable to Boston Scientific common stockholders and adjusted net income (loss) per share are GAAP net income (loss), GAAP net income (loss) attributable to Boston Scientific common stockholders and GAAP net income (loss) per common share – diluted, respectively.

To calculate operational net sales growth rates, which exclude the impact of foreign currency fluctuations, we convert actual net sales from local currency to U.S. dollars using constant foreign currency exchange rates in the current and prior periods. To calculate organic net sales growth rates, we also remove the impact of acquisitions and divestitures with less than a full period of comparable net sales. The GAAP financial measure most directly comparable to operational net sales and organic net sales is net sales reported on a GAAP basis.

Reconciliations of each of these non-GAAP financial measures to the corresponding GAAP financial measure are included in the relevant sections of this Quarterly Report on Form 10-Q.

Management uses these supplemental non-GAAP financial measures to evaluate performance period over period, to analyze the underlying trends in our business, to assess our performance relative to our competitors and to establish operational goals and forecasts that are used in allocating resources. In addition, management uses these non-GAAP financial measures to further its understanding of the performance of our operating segments. The adjustments excluded from our non-GAAP financial measures are consistent with those excluded from our operating segments' measures of net sales and profit or loss. These adjustments are excluded from the segment measures reported to our chief operating decision maker that are used to make operating decisions and assess performance.

We believe that presenting adjusted net income (loss), adjusted net income (loss) attributable to Boston Scientific common stockholders, adjusted net income (loss) per share, operational net sales and organic net sales growth rates, in addition to the corresponding GAAP financial measures, provides investors greater transparency to the information used by management for its operational decision-making and allows investors to see our results "through the eyes" of management. We further believe that providing this information assists our investors in understanding our operating performance and the methodology used by management to evaluate and measure such performance.

Safe Harbor for Forward-Looking Statements

Certain statements that we may make from time to time, including statements contained in this Quarterly Report on Form 10-Q and information incorporated by reference herein, constitute "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. Forward-looking statements may be identified by words like "anticipate," "expect," "project," "believe," "plan," "estimate," "intend," "aim," "goal," "target," "continue," "hope," "may" and similar words. These forward-looking statements are based on our beliefs, assumptions and estimates using information available to us at the time and are not intended to be guarantees of future events or performance. If our underlying assumptions turn out to be incorrect, or if certain risks or uncertainties materialize, actual results could vary materially from the expectations and projections expressed or implied by our forward-looking statements.

The forward-looking statements in this Quarterly Report on Form 10-Q are based on certain risks and uncertainties, including the risk factors described in Part I, Item 1A. Risk Factors in our most recent Annual Report on Form 10-K and the specific risk factors discussed herein and in connection with forward-looking statements throughout this Quarterly Report on Form 10-Q, which could cause actual results to vary materially from the expectations and projections expressed or implied by our forward-looking statements. These risks and uncertainties, in some cases, have affected and in the future could affect our ability to implement our business strategy and may cause actual results to differ materially from those contemplated by the statements expressed in this Quarterly Report on Form 10-Q. As a result, readers are cautioned not to place undue reliance on any of our forward-looking statements. Risks and uncertainties that may cause such differences include, among other things: economic conditions, including the impact of foreign currency fluctuations; future U.S. and global political, competitive, reimbursement and regulatory conditions; geopolitical events; manufacturing, distribution and supply chain disruptions and cost increases; disruptions caused by cybersecurity events; disruptions caused by public health emergencies or extreme weather or other climate change-related events; labor shortages and increases in labor costs; variations in outcomes of ongoing and future clinical trials and market studies; new product introductions and the market acceptance of those products; market competition for our products; expected pricing environment; expected procedural volumes; the closing and integration of acquisitions; demographic trends; intellectual property rights; litigation; financial market conditions; the execution and effect of our restructuring program; the execution and effect of our business strategy, including our cost-savings and growth initiatives; our ability to achieve environmental, social and governance goals and commitments; and future business decisions made by us and our competitors. New risks and uncertainties may arise from time to time and are difficult to predict. All of these factors are difficult or impossible to predict accurately and many of them are beyond our control. For a further list and description of these and other important risks and uncertainties that may affect our future operations, refer to Part I, Item 1A. Risk Factors in our most recent Annual Report on Form 10-K filed with the Securities and Exchange Commission, which we may update in Part II, Item 1A. Risk Factors in subsequent Quarterly Reports on Form 10-Q that we will file hereafter. We disclaim any intention or obligation to publicly update or revise any forward-looking statement to reflect any change in our expectations or in events, conditions, or circumstances on which those expectations may be based, or that may affect the likelihood that actual results will differ from those contained in the forward-looking statements, except as required by law. This cautionary statement is applicable to all forward-looking statements contained in this Quarterly Report on Form 10-Q.

The following are some of the important risk factors that could cause our actual results to differ materially from our expectations in any forward-looking statements. For further discussion of these and other risk factors, refer to Part I, Item 1A. *Risk Factors* in our most recent Annual Report on Form 10-K.

Our Business

- Risks associated with challenging or uncertain domestic and international economic conditions, including those related to interest rates, inflation, supply chain disruptions and constraints, adverse developments and volatility in the banking industry, currency devaluations or economies entering into periods of recession,
- The impact of disruptions in the supply of the materials and components used in manufacturing our products or the sterilization of our products,
- Labor shortages and the impact of inflation on the cost of raw materials and direct labor,
- The impact of any future pandemics or other public health crises on worldwide economies, financial markets, manufacturing and distribution systems, including disruption in the manufacture or supply of certain components, materials or products, and business operations,
- The impact of natural disasters, climate change or other catastrophic events on our ability to manufacture, distribute and sell our products,

- The impact of competitive offerings, value-based procurement practices, government-imposed payback provisions and changes in reimbursement practices and policies on average selling prices for our products,
- The ongoing impact on our business of physician alignment to hospitals, governmental investigations and audits of hospitals and other market and economic conditions on the overall number of procedures performed,
- The performance of, and physician and patient confidence in, our products and technologies or those of our competitors,
- The impact and outcome of ongoing and future clinical trials and market studies undertaken by us, our competitors or other third parties or perceived product performance of our or our competitors' products,
- Variations in clinical results, reliability or product performance of our and our competitors' products,
- Our ability to acquire or develop, launch and supply new or next-generation products and technologies worldwide and in line with our
 commercialization strategies in a timely and successful manner and with respect to our recent acquisitions,
- The effect of consolidation and competition in the markets in which we do business or plan to do business,
- · Our ability to achieve our projected level or mix of product sales, as some of our products are more profitable than others,
- Our ability to attract and retain talent, including key personnel associated with acquisitions, and to maintain our corporate culture in a hybrid work environment,
- The impact of enhanced requirements to obtain and maintain regulatory approval in the U.S. and around the world, including EU MDR and the associated timing and cost of product approval,
- The impact of increased pressure on the availability and rate of third-party reimbursement for our products and procedures in the U.S. and around the world, including with respect to the timing and costs of creating and expanding markets for new products and technologies,
- · The issuance of new or revised accounting standards by the Financial Accounting Standards Board or the Securities and Exchange Commission, and
- The impact of potential goodwill and intangible asset impairment charges on our results of operations.

Regulatory Compliance, Litigation and Data Protection

- The impact of health care policy changes and legislative or regulatory efforts in the U.S., the EU and around the world to modify product approval or reimbursement processes, including a trend toward demonstrating clinical outcomes, comparative effectiveness and cost efficiency, as well as the impact of other health care reform legislation,
- Risks associated with our regulatory compliance and quality systems and activities in the U.S., the EU and around the world, including meeting regulatory standards applicable to manufacturing and quality processes,
- The effect of global legal, regulatory or market responses to climate change and sustainability matters, including increased compliance burdens and costs to meet regulatory obligations,
- Our ability to minimize or avoid future field actions or FDA warning letters, or similar actions by regulatory agencies around the world, relating to our products and processes and the ongoing inherent risk of potential physician advisories related to our or our competitors' products,
- The impact of increased scrutiny of and heightened global regulatory enforcement facing the medical device industry arising from political and regulatory changes, economic pressures or otherwise, including under U.S. Anti-Kickback Statute, U.S. False Claims Act and similar laws in other jurisdictions, U.S. Foreign Corrupt Practices Act (FCPA) and similar laws in other jurisdictions, and U.S. and foreign export control, trade embargo and customs laws,

- · Costs and risks associated with current and future asserted litigation,
- The effect of our litigation and risk management practices, including self-insurance and compliance activities on our loss contingencies, legal provisions and cash flows,
- The impact of, diversion of management attention as a result of, and costs to cooperate with, litigate and/or resolve governmental investigations and our class action, product liability, contract and other legal proceedings,
- The possibility of failure to protect our intellectual property rights and the outcome of patent litigation,
- Our ability to secure our information technology and operational technology systems that support our business operations and protect our data
 integrity and products from a cyber-attack, other breach or other malicious actors that may have a material adverse effect on our business, reputation
 or results of operations including increased risks as an indirect result of the ongoing Russia/Ukraine war and Israel/Hamas war, and
- The potential impact to internal control over financial reporting relating to potential restrictions to access to consigned inventory at customer locations for our inventory count procedures.

Innovation and Certain Growth Initiatives

- The timing, size and nature of our strategic growth initiatives and market opportunities, including with respect to our internal research and development platforms and externally available research and development platforms and technologies and the ultimate cost and success of those initiatives and opportunities,
- Our ability to complete planned clinical trials successfully, obtain regulatory approvals and launch new and next generation products in a timely manner consistent with cost estimates, including the successful completion of projects from in-process research and development,
- Our ability to identify and prioritize our internal research and development project portfolio and our external investment portfolio on profitable net sales growth opportunities as well as to maintain the estimated timing and costs of such projects and expected revenue levels for the resulting products and technologies,
- Our ability to develop, manufacture and market new products and technologies successfully and in a timely manner and the ability of our competitors and other third parties to develop products or technologies that render our products or technologies noncompetitive or obsolete,
- Our ability to execute appropriate decisions to discontinue, write-down or reduce the funding of any of our research and development projects, including projects from in-process research and development from our acquisitions, in our growth adjacencies or otherwise,
- Our dependence on acquisitions, alliances or investments to introduce new products or technologies and to enter new or adjacent growth markets and our ability to fund them or to fund contingent payments with respect to those acquisitions, alliances and investments, and
- The potential failure to successfully integrate, collaborate or realize the expected benefits, including cost synergies, from strategic acquisitions, alliances and investments we have consummated or may consummate in the future.

International Markets

- Our dependency on international net sales to achieve growth, and our ability to maintain or expand our worldwide market positions in the various markets in which we compete or seek to compete, including through investments in China and other Emerging Markets countries,
- The timing and collectability of customer payments, as well as our ability to continue factoring customer receivables where we have factoring arrangements, or to enter new factoring arrangements with favorable terms,

- The impact on pricing due to national and regional tenders, including value-based procurement practices and government-imposed payback provisions,
- Geopolitical and economic conditions, including civil unrest, terrorist activity, governmental changes, restrictions on the ability to transfer capital
 across borders, tariffs and other protectionist measures,
- The impact of the Russia/Ukraine war, Israel/Hamas war and tension in the Taiwan strait, and related, downstream effects thereof, including disruptions to operations or the impact of sanctions on U.S. manufacturers doing business in these regions,
- Protection of our intellectual property,
- Our ability to comply with established and developing U.S. and foreign legal and regulatory requirements, including FCPA, EU MDR and similar laws in other jurisdictions,
- Our ability to comply with U.S. and foreign export control, trade embargo and customs laws,
- The impact of significant developments or uncertainties stemming from changes in the U.S. government following the 2024 presidential and congressional elections, including changes in U.S. trade policies, tariffs and the reaction of other countries thereto, particularly China, and
- The potential effect of foreign currency fluctuations and interest rate fluctuations on our net sales, operating expenses and resulting profit margins.

Liquidity

- Our ability to generate sufficient cash flow to fund operations, capital expenditures, global expansion initiatives, any litigation settlements and
 judgments, share repurchases and strategic investments and acquisitions as well as maintaining our investment grade ratings and managing our debt
 levels and financial covenant compliance,
- Our ability to access the public and private capital markets when desired and to issue debt or equity securities on terms reasonably acceptable to us,
- The unfavorable resolution of open tax matters, exposure to additional tax liabilities and the impact of changes in U.S. and international tax laws,
- The unfavorable resolution of open litigation matters, exposure to additional loss contingencies and legal provisions,
- The impact of examinations and assessments by domestic and international taxing authorities on our tax provisions, financial condition or results of operations,
- The possibility of counterparty default on our derivative financial instruments, and
- Our ability to collect outstanding and future receivables and/or sell receivables under our factoring programs.

Cost Reduction and Optimization Initiatives

- Risks associated with changes made or expected to be made to our organizational and operational structure, pursuant to our restructuring plans as well
 as any further restructuring or optimization plans we may undertake in the future and our ability to recognize benefits and cost reductions from such
 programs, and
- Business disruption and employee distraction as we execute our global compliance program, restructuring and optimization plans and any divestitures
 of assets or businesses and implement our other strategic and cost reduction initiatives.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

We develop, manufacture and sell medical devices globally and our earnings and cash flows are exposed to market risk from changes in currency exchange rates and interest rates. We address these risks through a risk management program that includes the use of derivative financial instruments. We operate the program pursuant to documented corporate risk management policies. We do not enter derivative transactions for speculative purposes. Gains and losses on derivative financial instruments substantially offset losses and gains on underlying hedged exposures. Furthermore, we manage our exposure to counterparty risk on derivative instruments by entering into contracts with a diversified group of major financial institutions and by actively monitoring outstanding positions.

Our currency risk consists primarily of foreign currency denominated firm commitments, forecasted foreign currency denominated intercompany and third-party transactions and net investments in certain subsidiaries. We use both nonderivative (primarily European manufacturing operations) and derivative instruments to manage our earnings and cash flow exposure to changes in currency exchange rates. We had currency derivative instruments outstanding in the contract amount of \$5.898 billion as of June 30, 2024 and \$5.899 billion as of December 31, 2023. A ten percent appreciation in the U.S. dollar's value relative to the hedged currencies would increase the derivative instruments' fair value by \$256 million as of June 30, 2024 compared to \$236 million as of December 31, 2023. A ten percent depreciation in the U.S. dollar's value relative to the hedged currencies would decrease the derivative instruments' fair value by \$313 million as of June 30, 2024 compared to \$288 million as of December 31, 2023. Any increase or decrease in the fair value of our currency exchange rate sensitive derivative instruments would be substantially offset by a corresponding decrease or increase in the fair value of the hedged underlying asset, liability or forecasted transaction, resulting in minimal impacts on our unaudited consolidated statements of operations.

Our interest rate risk relates primarily to U.S. dollar and euro-denominated borrowings partially offset by U.S. dollar cash investments. We have historically used interest rate derivative instruments to manage our earnings and cash flow exposure to changes in interest rates. We had no interest rate derivative instruments outstanding as of June 30, 2024 or December 31, 2023. As of June 30, 2024, \$10.633 billion in aggregate principal amount of our outstanding debt obligations was at fixed interest rates, representing approximately 100 percent of our total debt, on an amortized cost basis. As of June 30, 2024, our outstanding debt obligations at fixed interest rates were comprised of senior notes.

Refer to Note D – Hedging Activities and Fair Value Measurements to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q for further information regarding our derivative financial instruments.

ITEM 4. CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures

Our management, with the participation of our President and Chief Executive Officer (CEO) and our Executive Vice President and Chief Financial Officer (CFO), evaluated the effectiveness of our disclosure controls and procedures as of June 30, 2024 pursuant to Rule 13a-15(b) of the Securities Exchange Act of 1934, as amended. Disclosure controls and procedures are designed to ensure that material information required to be disclosed by us in the reports that we file or submit under the Securities Exchange Act of 1934, as amended, is recorded, processed, summarized and reported within the time periods specified in the Securities and Exchange Commission's rules and forms and ensure that such material information is accumulated and communicated to our management, including our CEO and CFO, as appropriate, to allow timely decisions regarding required disclosure. Based on their evaluation, our CEO and CFO concluded that, as of June 30, 2024, our disclosure controls and procedures were effective.

Changes in Internal Control over Financial Reporting

During 2022, we began a multi-year implementation of a new global enterprise resource planning (ERP) system, which will replace our existing system. The implementation is expected to occur in phases over the next several years. The portion of the transition to the new ERP system which we have completed to date resulted in changes in our internal control over financial reporting in 2023. No changes occurred during the first six months of 2024. As future phases are implemented, we expect the changes to have a material impact on our internal controls over financial reporting and we will evaluate whether these process changes necessitate further changes in the design of and testing for effectiveness of internal controls over financial reporting.

PART II OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

Refer to Note H - Commitments and Contingencies to our unaudited consolidated financial statements contained in Item 1 of Part I of this Quarterly Report on Form 10-Q, which is incorporated herein by reference.

ITEM 1A. RISK FACTORS

In addition to other information contained elsewhere in this report, you should carefully consider the factors discussed in Part I, Item 1A. Risk Factors in our most recent Annual Report on Form 10-K, which could materially affect our business, financial condition or future results.

ITEM 5. OTHER INFORMATION

(c)

On May 23, 2024, Joseph M. Fitzgerald, our Executive Vice President and Group President, Cardiology, entered into a trading plan intended to satisfy the affirmative defense conditions of Rule 10b5-1(c). Mr. Fitzgerald's plan covers the sale of 553,370 shares of our common stock, including 492,532 shares to be acquired upon exercise of stock options. Transactions under Mr. Fitzgerald's plan are based upon pre-established dates and stock price thresholds and will only occur upon the expiration of the applicable mandatory cooling-off period. Mr. Fitzgerald's plan will terminate on the earlier of May 20, 2026, or the date all shares subject to the plan have been sold.

ITEM 6. EXHIBITS (* documents filed or furnished with this report; # compensatory plans or arrangements)

3.1	Amended and Restated By-Laws of Boston Scientific Corporation (incorporated by reference to Exhibit 3.1 to the Company's Current Report on Form 8-K, filed on May 10, 2024, File No. 1-110183).
10.1*	Third Amendment, dated as of May 10, 2024, to the Credit Agreement, dated as of May 10, 2021, by and among Boston Scientific Corporation, the lenders party thereto and Wells Fargo Bank, National Association, as administrative agent.
22	Subsidiary Issuer of Guaranteed Securities (incorporated herein by reference to Exhibit 22 to the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2024, filed on May 1, 2024, File No. 1-11083).
31.1*	Certification of the Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
31.2*	Certification of the Chief Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
32.1*	Certification of the Chief Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
32.2*	Certification of the Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
101.SCH*	Inline XBRL Taxonomy Extension Schema Document.
101.CAL*	Inline XBRL Taxonomy Extension Calculation Linkbase Document.
101.DEF*	Inline XBRL Taxonomy Extension Definition Linkbase Document.
101.LAB*	Inline XBRL Taxonomy Extension Label Linkbase Document.

101.PRE* Inline XBRL Taxonomy Extension Presentation Linkbase Document.

104 Cover Page Interactive Data File (embedded within the Inline XBRL document and contained in Exhibit 101).

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized on August 1, 2024.

BOSTON SCIENTIFIC CORPORATION

By: /s/ Daniel J. Brennan

> Name: Daniel J. Brennan

Executive Vice President and Chief Financial Officer Title:

THIRD AMENDMENT

THIRD AMENDMENT, dated as of May 10, 2024 (this "Amendment"), to the Credit Agreement, dated as of May 10, 2021 (as amended by that certain Amendment dated as of December 21, 2022 and the Second Amendment, dated as of March 1, 2023, and as further amended, supplemented or otherwise modified from time to time, the "Credit Agreement"), among (i) BOSTON SCIENTIFIC CORPORATION, a Delaware corporation (the "Borrower"), (ii) the several banks and other financial institutions from time to time parties thereto (the "Lenders") and (iii) WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent for the Lenders thereunder (in such capacity, the "Administrative Agent").

WITNESSETH:

WHEREAS, the Borrower, the Lenders and the Administrative Agent are parties to the Credit Agreement; and

WHEREAS, the Borrower has requested that the Lenders otherwise amend the Credit Agreement to refinance the Borrower's existing \$2,750,000,000 Revolving Credit Commitments as set forth herein; and

WHEREAS, the Lenders and the Administrative Agent are willing to agree to such amendments to the Credit Agreement and such other agreements, subject to the terms and conditions set forth herein;

NOW, THEREFORE, in consideration of the premises and mutual covenants contained herein, the Borrower, the Lenders and the Administrative Agent hereby agree as follows:

- SECTION 1. <u>Defined Terms</u>. Terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement.
- SECTION 2. Amendments to the Credit Agreement. Subject to the satisfaction of the conditions set forth in Section 3, (i) the Credit Agreement (including the schedules thereto) is hereby amended in accordance with Exhibit A hereto by deleting the stricken text (indicated textually in the same manner as the following example: stricken text) and by inserting the double-underlined text (indicated textually in the same manner as the following example: double-underlined text) and (ii) the Exhibits to the Credit Agreement are hereby amended and restated as set forth in Schedule I hereto.
- SECTION 3. <u>Conditions to Effectiveness</u>. This Amendment shall become effective on the date (the "<u>Amendment Effective</u> <u>Date</u>") on which:
- (a) the Borrower, the Administrative Agent and the Lenders shall have executed and delivered to the Administrative Agent this Amendment, with a counterpart for each Lender and original Notes executed by the Borrower (in the case of Revolving Credit Notes, Multicurrency Notes or CAF Advance Notes), in favor of each Lender requesting a Note,
- (b) The Administrative Agent shall have received a certificate of the Borrower, dated the Amendment Effective Date, substantially in the form of Exhibit F, with appropriate insertions and

attachments, satisfactory in form and substance to the Administrative Agent, executed by the President or any Vice President and the Secretary or any Assistant Secretary of the Borrower,

- (c) The Administrative Agent shall have received an executed legal opinion (addressed to the Administrative Agent and the Lenders) from Shearman & Sterling LLP, counsel for the Borrower, in form and substance reasonably satisfactory to the Administrative Agent. The Borrower hereby requests such counsel to deliver such opinion,
- (d) Each of the representations and warranties made by the Borrower in or pursuant to the Loan Documents shall be true and correct in all material respects on and as of the Amendment Effective Date as if made on and as of the Amendment Effective Date,
- (e) all accrued fees and reasonable out-of-pocket expenses (including the reasonable fees and expenses of counsel to the Arrangers) of the Arrangers through the Amendment Effective Date invoiced with reasonable detail at least three Business Days prior to the Amendment Effective Date in connection with the Loan Documents shall have been paid; <u>provided</u> that the Arrangers shall have provided an estimate and available reasonable detail five business days prior to the Amendment Effective Date,
- (f) The Administrative Agent shall have received an unaudited consolidated balance sheet of the Borrower and its consolidated Subsidiaries and the related unaudited consolidated statements of operations and of cash flows for each fiscal quarter ended after December 31, 2023 (so long as such fiscal quarters have ended at least 40 days prior to the Amendment Effective Date). The Borrower's filing of any required unaudited financial statements with respect to the Borrower on Form 10-Q will satisfy the requirements under this paragraph,
- (g) No Default or Event of Default shall have occurred and be continuing on such date or after giving effect to the Loans requested to be made, or Letters of Credit requested to be issued, and
- (i) The Administrative Agent shall have received at least three Business Days prior to the Amendment Effective Date all documentation and other information about the Borrower as has been reasonably requested by the Administrative Agent at least 10 Business Days prior to the Amendment Effective Date that is required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including without limitation the USA PATRIOT Act.
- SECTION 4. <u>Foreign Currencies</u>. The parties hereto acknowledge that notwithstanding anything to the contrary set forth in this Amendment, the Suspension of Rights Agreement, dated as of November 30, 2021 and the Suspension of Rights Agreement, dated as of May 10, 2024, between the Borrower and the Administrative Agent remains in full force and effect.
- SECTION 5. <u>Representation and Warranties</u>. To induce the Administrative Agent to enter into this Amendment, the Borrower hereby represents and warrants to the Administrative Agent and all of the Lenders as of the Amendment Effective Date that:
 - (a) <u>Corporate Power; Authorization; Enforceable Obligations.</u>

- (i) The Borrower has the corporate power and authority, and the legal right, to make and deliver this Amendment and to perform the Loan Documents, as amended by this Amendment, to which it is a party and has taken all necessary corporate action to authorize the execution, delivery and performance of this Amendment and the performance of the Loan Documents, as so amended, to which it is a party.
- (ii) No consent or authorization of, filing with (other than the Borrower's public filing of the Amendment on Form 8-K, if applicable), or notice to or other act by or in respect of, any Governmental Authority or any other Person is required with respect to the Borrower or any of its Subsidiaries in connection with the execution and delivery of this Amendment or with the performance, validity or enforceability of the Loan Documents, as amended by this Amendment, to which the Borrower is party.
 - (iii) This Amendment has been duly executed and delivered on behalf of the Borrower.
- (iv) This Amendment and each Loan Document, as amended by this Amendment, to which the Borrower is a party constitutes a legal, valid and binding obligation of the Borrower enforceable against the Borrower in accordance with its terms, subject to the effects of bankruptcy, examination, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, general equitable principles (whether considered in a proceeding in equity or at law) and an implied covenant of good faith and fair dealing.
- (b) <u>Representations and Warranties</u>. (i) The representations and warranties made by the Borrower in or pursuant to the Loan Documents are true and correct in all material respects on and as of the Amendment Effective Date, after giving effect to the effectiveness of this Amendment, as if made on and as of the Amendment Effective Date and (ii) after giving effect to this Amendment, no Default or Event of Default has occurred and is continuing.
- SECTION 6. <u>Payment of Expenses</u>. The Borrower agrees to pay or reimburse the Administrative Agent for all of its reasonable and documented out-of-pocket costs and expenses incurred in connection with this Amendment, any other documents prepared in connection herewith and the transactions contemplated hereby, including, without limitation, the reasonable and documented fees and disbursements of counsel to the Administrative Agent.
- SECTION 7. <u>No Other Amendments; Confirmation</u>. Except as expressly amended, modified and supplemented hereby, the provisions of the Credit Agreement and the other Loan Documents are and shall remain in full force and effect.

SECTION 8. Governing Law; Counterparts.

(a) This Amendment and the rights and obligations of the parties hereto and any claim, controversy, dispute, proceeding or cause of action (whether in contract, tort or otherwise and whether at law or in equity) based upon, arising out of or relating to this Amendment and the transactions contemplated hereby shall be governed by, and construed and interpreted in accordance with, the laws of

the State of New York. The provisions of <u>subsections 13.12</u> and <u>13.17</u> of the Credit Agreement are incorporated herein, *mutatis mutandis*.

(b) This Amendment may be executed by one or more of the parties to this Amendment on any number of separate counterparts (including by facsimile transmission or in electronic (i.e., "pdf" or "tif") format), and all of said counterparts taken together shall be deemed to constitute one and the same instrument. The words "execution," "signed," "signature," "delivery," and words of like import in or relating to this Amendment shall be deemed to include Electronic Signatures (as defined below), deliveries or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature, physical delivery thereof or the use of a paper-based recordkeeping system, as the case may be. "Electronic Signatures" means any electronic symbol or process attached to, or associated with, any contract or other record and adopted by a person with the intent to sign, authenticate or accept such contract or record.

SECTION 9. Miscellaneous.

- (a) Upon and after the Amendment Effective Date, each reference in the Credit Agreement to "this Agreement", "hereunder", "hereof" or words of like import referring to the Credit Agreement, and each reference in the other Loan Documents to "the Credit Agreement", "thereunder", "thereof" or words of like import referring to the Credit Agreement, shall mean and be a reference to the Credit Agreement as modified hereby. This Amendment shall constitute a Loan Document.
- (b) The execution, delivery and effectiveness of this Amendment shall not, except as expressly provided herein, operate as a waiver of any right, power or remedy of any Lender or the Agent under any of the Loan Documents, nor, except as expressly provided herein, constitute a waiver or amendment of any provision of any of the Loan Documents. It is the intent of the parties hereto, and the parties hereto agree, that this Amendment shall not constitute a novation of the Credit Agreement, any other Loan Document or any of the rights, obligations or liabilities thereunder.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed and delivered by their respective proper and duly authorized officers as of the day and year first above written.

BOSTON SCIENTIFIC CORPORATION

By: /s/ Robert J. Castagna_

Name: Robert J. Castagna

Title: Vice President and Treasurer

Signature Page to Third Amendment to Credit Agreement

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent and as a Lender

By: /s/ Darin Mullis Name: Darin Mullis Title: Managing Director
BANK OF AMERICA, N.A. , as a Lender
By: /s/ Darren Merten Name: Darren Merten Title: Director
BARCLAYS BANK PLC, as a Lender
By: /s/ Edward Pan Name: Edward Pan Title: Vice President
CITIBANK, N.A. , as a Lender
By: /s/ Richard Rivera Name: Richard Rivera Title: Vice President
<u>JPMORGAN CHASE BANK, N.A.</u> , as a Lender
By: /s/ Marcelo Nicolas Osovi Conti Name: Marcelo Nicolas Osovi Conti Title: Vice President



Société Générale, as a Lender By: /s/ Shelly Yu Name: Shelly Yu Title: Director BNP Paribas, as a Lender By: /s/ John Bosco Name: John Bosco Title: Managing Director By: /s/ Claudia Zarate Name: Claudia Zarate Title: Managing Director DEUTSCHE BANK AG NEW YORK BRANCH, as a Lender By: /s/ Ming K. Chu_ Name: Ming K. Chu Title: Director By: /s/ Marko Lukin Name: Marko Lukin Title: Vice President **GOLDMAN SACHS BANK USA** as a Lender By: /s/ Nicholas Merino Name: Nicholas Merino

Title: Authorized Signatory

Royal Bank of Canada, as a Lender By: /s/ Scott MacVicar Name: Scott MacVicar Title: Authorized Signatory STANDARD CHARTERED BANK, as a Lender By: /s/ Kristopher Tracy Name: Kristopher Tracy Title: Director, Financing Solutions THE BANK OF NOVA SCOTIA, as a Lender By: /s/ Iain Stewart Name: Iain Stewart Title: Managing Director The Toronto-Dominion Bank, New York Branch, as a Lender By: /s/ Mike Tkach Name: Mike Tkach

MUFG Bank LTD.

as a Lender

By: /s/ Reema Sharma

Name: Reema Sharma Title: Authorized Signatory

Title: Authorized Signatory

Exhibit A

AMENDED CREDIT AGREEMENT

[See attached]

\$2,750,000,000

CREDIT AGREEMENT

among

BOSTON SCIENTIFIC CORPORATION, as Borrower,

The Several Lenders from Time to Time Parties Hereto,

BANK OF AMERICA, N.A.,
BARCLAYS BANK PLC,
CITIBANK, N.A.,
JPMORGAN CHASE BANK, N.A.,
and
SOCIÉTÉ GÉNÉRALE,
as Documentation Agents,
and

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

Dated as of May 10, 2021

WELLS FARGO SECURITIES, LLC,

as Joint Lead Arranger and Bookrunner,

BofA SECURITIES, INC.,
BARCLAYS BANK PLC,
CITIBANK, N.A.,
JPMORGAN CHASE BANK, N.A.,
and
SOCIÉTÉ GÉNÉRALE
as Joint Lead Arrangers

TABLE OF CONTENTS

<u>Page</u>

SECTION 1 DEFINITIONS 1
1.1. Defined Terms 1
1.2. Other Definitional Provisions 32
1.3. Rates 32
SECTION 2 AMOUNT AND TERMS OF COMMITMENTS 33
2.1. Revolving Credit Commitments 33
2.2. Procedure for Revolving Credit Borrowing 33
2.3. Reserved 34
2.4. Reserved 34
2.5. Fees 34
2.6. Termination or Reduction of Revolving Credit Commitments 34
2.7. Repayment of Loans 35
2.8. CAF Advances 35
2.9. Procedure for CAF Advance Borrowing 35
2.10. Repayment of CAF Advances 38
2.11. Certain Restrictions with Respect to CAF Advances 38
2.12. Multicurrency Commitments 38
2.13. Repayment of Multicurrency Loans 38
2.14. Procedure for Multicurrency Borrowing 39
2.15. Termination or Reduction of Multicurrency Commitments 39
2.16. Borrowings of Revolving Credit Loans and Refunding of Loans 39
2.17. Increase of Revolving Credit Commitments 41
2.18. Cash Collateral 41
2.19. Defaulting Lenders 42
2.20. Extension of Termination Date 44
SECTION 3 CERTAIN PROVISIONS APPLICABLE TO THE LOANS 46
3.1. Optional and Mandatory Prepayments 46
3.2. Conversion and Continuation Options 47
3.3. Minimum Amounts and Maximum Number of Tranches 47
3.4. Interest Rates and Payment Dates 48
3.5. Computation of Interest and Fees 48
3.6. Changed Circumstances 49
3.7. Pro Rata Treatment and Payments 51
3.8. Illegality 52

3.9. Requirements of Law 53

TABLE OF CONTENTS

/		1)
LCO	ntin	ned
(00	111111	ucu,

Page

3.	10.	Taxes	54

- 3.11. Indemnity 57
- 3.12. Change of Lending Office; Removal of Lender 58
- 3.13. Evidence of Debt 58

SECTION 4 LOCAL CURRENCY FACILITIES 59

- 4.1. Terms of Local Currency Facilities. 59
- 4.2. Reporting of Local Currency Outstandings 60

SECTION 5 LETTERS OF CREDIT 61

- 5.1. L/C Commitment 61
- 5.2. Procedure for Issuance of Letter of Credit 61
- 5.3. Fees and Other Charges 62
- 5.4. L/C Participations 63
- 5.5. Reimbursement Obligation of the Borrower 63
- 5.6. Obligations Absolute 64
- 5.7. Letter of Credit Payments 64
- 5.8. Applications 64
- 5.9. Reimbursement Obligations for Certain Letters of Credit Denominated in Currencies Other Than Dollars 65
- 5.10. Replacement of Issuing Lenders 65

SECTION 6 REPRESENTATIONS AND WARRANTIES 65

- 6.1. Financial Condition 65
- 6.2. Corporate Existence; Compliance with Law 66
- 6.3. Corporate Power; Consents and Authorization; Enforceable Obligations 66
- 6.4. No Legal Bar 66
- 6.5. No Default 67
- 6.6. Taxes 67
- 6.7. Federal Regulations 67
- 6.8. ERISA 67
- 6.9. Investment Company Act; Other Regulations 67
- 6.10. Purpose of Loans 67
- 6.11. Environmental Matters 67
- 6.12. Disclosure 68
- 6.13. No Change 68
- 6.14. No Material Litigation 69
- 6.15. Anti-Corruption Laws and Sanctions 69

SECTION 7 CONDITIONS PRECEDENT 69

7.1. Conditions to Closing 69

TABLE OF CONTENTS

		1)
- (continu	ലവ
٠,	Commin	ou,

Page

•	7	2	Conditions	tο	Fach	Loan	and	Letter	αf	Credit	70
	/ • .	∠.	Continuons	w	Lacii	Loan	anu.	LCIICI	OI.	Cicuit	/ U

SECTION 8 AFFIRMATIVE COVENANTS 71

- 8.1. Financial Statements 71
- 8.2. Certificates; Other Information 72
- 8.3. Payment of Obligations 72
- 8.4. Conduct of Business and Maintenance of Existence 72
- 8.5. Maintenance of Property; Insurance 73
- 8.6. Inspection of Property; Books and Records; Discussions 73
- 8.7. Notices 73

SECTION 9 NEGATIVE COVENANTS 74

- 9.1. Financial Covenant 74
- 9.2. Limitation on Liens 74
- 9.3. Limitation on Indebtedness pursuant to Receivables Transactions 75
- 9.4. Limitation on Fundamental Changes 75
- 9.5. Limitation on Indebtedness of Subsidiaries 76

SECTION 10 EVENTS OF DEFAULT 76

SECTION 11 THE AGENTS 78

- 11.1. Appointment 78
- 11.2. Delegation of Duties 78
- 11.3. Exculpatory Provisions 79
- 11.4. Reliance by Administrative Agent 79
- 11.5. Notice of Default 79
- 11.6. Non-Reliance on Administrative Agent and Other Lenders 79
- 11.7. Indemnification 80
- 11.8. Administrative Agent in Its Individual Capacity 80
- 11.9. Successor Administrative Agent 80
- 11.10. The Arrangers, the Bookrunner and the Documentation Agents 81
- 11.11. Certain ERISA Matters 81
- 11.12. Accidental Payments 82

SECTION 12 GUARANTEE 84

- 12.1. Guarantee 84
- 12.2. No Subrogation 84
- 12.3. Amendments, etc. with respect to the Obligations; Waiver of Rights 85
- 12.4. Guarantee Absolute and Unconditional 85
- 12.5. Reinstatement 86
- 12.6. Payments 86

TABLE OF CONTENTS

(continued)

Page Page

12.7.	"Lenders"	, ,	86
17	Lenders		w

SECTION 13 MISCELLANEOUS 87

- 13.1. Amendments and Waivers 87
- 13.2. Notices 88
- 13.3. No Waiver; Cumulative Remedies 90
- 13.4. Survival of Representations and Warranties 91
- 13.5. Payment of Expenses and Taxes 91
- 13.6. Successors and Assigns; Participations and Assignments 91
- 13.7. Adjustments; Set-off 94
- 13.8. Counterparts 95
- 13.9. Severability 95
- 13.10. Integration 95
- 13.11. GOVERNING LAW 95
- 13.12. Submission To Jurisdiction and Waivers 95
- 13.13. Acknowledgements 96
- 13.14. Confidentiality 96
- 13.15. Loan Conversion/Participations 97
- 13.16. Judgment 98
- 13.17. WAIVERS OF JURY TRIAL 98
- 13.18. USA Patriot Act Notice 98
- 13.19. No Advisory or Fiduciary Responsibility 99
- 13.20. Acknowledgement and Consent to Bail-In of Affected Financial Institutions 99
- 13.21. Acknowledgement Regarding Any Supported QFCs 100

SCHEDULES

- Schedule I Names and Commitments of Lenders
- Schedule II Information Concerning Local Currency Loans
- Schedule 9.2 Existing Liens
- Schedule 9.5 Existing Subsidiary Indebtedness

EXHIBITS

- Exhibit A Form of Revolving Credit Note
- Exhibit B Form of CAF Advance Note
- Exhibit C Form of CAF Advance Request
- Exhibit D Form of CAF Advance Offer
- Exhibit E Form of CAF Advance Confirmation
- Exhibit F Form of Closing Certificate
- Exhibit G Form of Opinion of Counsel to Borrower
- Exhibit H Form of Assignment and Assumption
- Exhibit I Form of Local Currency Facility Addendum

TABLE OF CONTENTS (continued)

<u>Page</u>

Exhibit J-1 – J-4 U.S. Tax Compliance Certificate

Exhibit K [Reserved]

Exhibit L Form of Multicurrency Note

CREDIT AGREEMENT, dated as of May 10, 2021, among (i) BOSTON SCIENTIFIC CORPORATION, a Delaware corporation (the "Borrower"), (ii) the several banks and other financial institutions or entities from time to time parties hereto (the "Lenders"), (iii) Bank of America, N.A., Barclays Bank PLC, Citibank, N.A., JPMorgan Chase Bank, N.A. and Société Générale, as Documentation Agents (each, in such capacity, a "Documentation Agent", and collectively, the "Documentation Agents") and (iv) WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders hereunder (in such capacity, the "Administrative Agent").

WITNESSETH:

WHEREAS, the Borrower has requested that the Lenders make Revolving Credit Loans (as defined below), CAF Advances (as defined below), Multicurrency Loans (as defined below) and Local Currency Loans (as defined below) to the Borrower on the terms and conditions set forth herein;

WHEREAS, the Lenders have agreed to make the credit facilities available upon the terms and subject to the conditions set forth herein; and

WHEREAS, the new credit facilities will be used to refinance the existing credit facilities under the Credit Agreement, dated as of December 19, 2018, among the Borrower, the lenders and the administrative agent party thereto (the "Existing Credit Facilities);

NOW, THEREFORE, in consideration of the premises, and of the mutual covenants and agreements herein contained and other good and valuable consideration, receipt of which is hereby acknowledged, the parties hereto hereby agree as follows:

SECTION 1

DEFINITIONS

1.1. Defined Terms

•

As used in this Agreement, the following terms shall have the following meanings:

"ABR": for any day, a rate per annum equal to the greatest of (a) the Prime Rate in effect on such day, (b) the NYFRB Rate in effect on such day plus 1/2 of 1% per annum and (c) Adjusted Term SOFR for a one month Interest Period on such day (or if such day is not a Business Day, the immediately preceding Business Day) plus 1% per annum. For purposes hereof: "Prime Rate" shall mean the rate of interest per annum publicly announced from time to time by the Administrative Agent as its prime rate in effect (the Prime Rate not being intended to be the lowest rate of interest charged by the Administrative Agent in connection with extensions of credit to debtors); and "NYFRB Rate" shall mean, for any day, the greater of (a) the Federal Funds Effective Rate in effect on such day and (b) the Overnight Bank Funding Rate in effect on such day (or for any day that is not a Business Day, for the immediately preceding Business Day); provided that if none of such rates are published for any day that is a Business Day, the term "NYFRB Rate" means the rate for a federal funds transaction quoted at 11:00 A.M. on such day received by the Administrative Agent from a Federal funds broker of recognized standing selected by it (it being agreed the Prime Rate and NYFRB Rate will not be less than zero). Any change in the ABR due to a change in the Prime Rate, the NYFRB Rate or Adjusted Term SOFR shall be effective as of the

opening of business on the effective day of such change in the Prime Rate, the NYFRB Rate or Adjusted Term SOFR, respectively.

- "ABR Loans": Revolving Credit Loans bearing interest based upon the ABR.
- "<u>Accrued Legal Liabilities</u>": any cash litigation costs, including judgments, orders, awards, settlements and related legal costs, that are accrued prior to, and unpaid as of, December 31, 2022.
 - "Additional Lender": as defined in subsection 2.20(c).
 - "Additional Revolving Credit Commitment": as defined in subsection 2.17.
- "Adjusted Aggregate Committed Outstandings": with respect to each Lender, the Aggregate Committed Outstandings of such Lender, plus the amount of any participating interests purchased by such Lender pursuant to subsection 13.15, minus the amount of any participating interests sold by such Lender pursuant to subsection 13.15.
- "Adjusted Term SOFR" means, for purposes of any calculation, the rate per annum equal to (a) Term SOFR for such calculation <u>plus</u> (b) the Term SOFR Adjustment; <u>provided</u> that if Adjusted Term SOFR as so determined shall ever be less than the Floor, then Adjusted Term SOFR shall be deemed to be the Floor.
 - "Administrative Agent": as defined in the preamble hereto.
 - "Affected Financial Institution": (a) any EEA Financial Institution or (b) any UK Financial Institution.
- "Affiliate": as to any Person, any other Person (other than a Subsidiary) which, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" of a Person means the power, directly or indirectly, either to (a) vote 10% or more of the securities having ordinary voting power for the election of directors of such Person or (b) direct or cause the direction of the management and policies of such Person, whether by contract or otherwise.
 - "Agency Fee Letter": the agency fee letter, dated April 1410, 2021 2024, between Wells Fargo and the Borrower.
 - "Agent Parties": as defined in subsection 13.2(c).
 - "Agents": the collective reference to the Administrative Agent, the Documentation Agents, the Arrangers and the Bookrunner.
 - "Aggregate Exposure Percentage": as defined in the definition of "Majority Lenders."
- "<u>Aggregate Available Multicurrency Commitments</u>": as at any date of determination with respect to all Multicurrency Lenders, an amount in Dollars equal to the sum of the Available Multicurrency Commitments of all Multicurrency Lenders on such date.

- "Aggregate Available Revolving Credit Commitments": as at any date of determination with respect to all Lenders, an amount in Dollars equal to the sum of the Available Revolving Credit Commitments of all Lenders on such date.
- "Aggregate Committed Outstandings": as at any date of determination with respect to any Lender, an amount in Dollars equal to the sum of (a) the Aggregate Revolving Credit Outstandings of such Lender on such date, (b) the Dollar Equivalent of the Aggregate Multicurrency Outstandings of such Lender on such date, and (c) the Dollar Equivalent of the Aggregate Local Currency Outstandings of such Lender on such date.
- "Aggregate Local Currency Outstandings": as at any date of determination with respect to any Lender, an amount in the applicable Local Currencies equal to the aggregate unpaid principal amount of such Lender's Local Currency Loans.
- "Aggregate Multicurrency Outstandings": as at any date of determination with respect to any Lender, an amount in the applicable Available Foreign Currencies equal to the aggregate unpaid principal amount of such Lender's Multicurrency Loans.
 - "Aggregate Revolving Credit Commitments": the aggregate amount of the Revolving Credit Commitments of all of the Lenders.
- "Aggregate Revolving Credit Outstandings": as at any date of determination with respect to any Lender, an amount in Dollars equal to the sum of (a) the aggregate unpaid principal amount of such Lender's Revolving Credit Loans on such date <u>plus</u> (b) such Lender's Revolving Credit Commitment Percentage of the L/C Obligations.
- "Aggregate Total Outstandings": as at any date of determination with respect to any Lender, an amount in Dollars equal to the sum of (a) the Aggregate Revolving Credit Outstandings of such Lender on such date, (b) the Dollar Equivalent of the aggregate unpaid principal amount of such Lender's CAF Advances on such date, (c) the Dollar Equivalent of the Aggregate Multicurrency Outstandings of such Lender on such date and (d) the Dollar Equivalent of the Aggregate Local Currency Outstandings of such Lender on such date.
 - "Agreement": this Credit Agreement, as amended, amended and restated, supplemented or otherwise modified from time to time.
 - "Agreement Currency": as defined in subsection 13.16(b).
- "Anti-Corruption Laws": the United States Foreign Corrupt Practices Act of 1977, as amended, and all similar laws, rules, and regulations of any jurisdiction applicable to the Borrower or its Subsidiaries prohibiting bribery or corruption.
- "Applicable Margin": with respect to each day for each Type of Loan, the rate per annum based on the Ratings in effect on such day, as set forth under the relevant column heading below:

SOFR Loans/				
Rating	Multicurrency Loans	ABR Loans		
Rating I	<u>0.680%</u>	<u>0.0%</u>		
Rating I <mark>I</mark>	0.795%	0.0%		
Rating II <mark>I</mark>	0.910%	0.0%		
Rating I <mark>HV</mark>	1.015%	0.015%		
Rating I V	1.100%	0.100%		
Rating V	1.400%	0.400%		

[&]quot;Applicable Screen Rate": as defined in the definition of "Impacted Interest Period".

"Arrangers": Wells Fargo Securities, LLC, BofA Securities, Inc. (or any other registered broker-dealer wholly-owned by Bank of America Corporation to which all or substantially all of Bank of America Corporation's or any of its subsidiaries' investment banking, commercial lending services or related businesses may be transferred following the date of this Agreement), Barclays Bank PLC, Citibank, N.A., JPMorgan Chase Bank, N.A. and Société Générale as Joint Lead Arrangers for this Agreement.

"Available Foreign Currencies": (a) Euro, Japanese Yen, Australian Dollar, Canadian Dollar, Singapore Dollar, Swiss Franc and Pound Sterling; provided that (i) as of the Second Amendment Effective Date, Japanese Yen, Swiss Franc and Pound Sterling are subject to that certain Suspension of Rights Agreement, dated as of November 30, 2021, between the Borrower and the Administrative Agent and (ii) as of the Third Amendment Effective Date, Canadian Dollar and Singapore Dollar are subject to that certain Suspension of Rights Agreement, dated as of May 10, 2024, between the Borrower and the Administrative Agent and (b) any other available and freely convertible non-Dollar currency selected by the Borrower and approved by the Administrative Agent and the Multicurrency Lenders.

"Available Foreign Currency Equivalent": for any amount of any Available Foreign Currency, at the time of determination thereof, (a) if such amount is expressed in such Available Foreign Currency, such amount and (b) if such amount is expressed in Dollars, the equivalent of such amount in such Available Foreign Currency determined by using the rate of exchange for the purchase of such Available Foreign Currency with Dollars last provided (either by publication or otherwise provided to the Administrative Agent) by the applicable Reuters source on the Business Day (New York City time) immediately preceding the date of determination or if such service ceases to be available or ceases to provide a rate of exchange for the purchase of such Available Foreign Currency with Dollars, as provided by such other publicly available information service which provides that rate of exchange at such time in place of Reuters chosen by the Administrative Agent in its sole discretion (or if such service ceases to be available or ceases to provide such rate of exchange, the equivalent of such amount in Dollars as determined by the Administrative Agent using any method of determination it deems appropriate in its sole discretion).

[&]quot;Application": an application, in such form as the Issuing Lender may specify from time to time, requesting the Issuing Lender to issue a Letter of Credit.

[&]quot;Assignee": as defined in subsection 13.6(c).

[&]quot;Australian Dollar": the lawful currency of the Commonwealth of Australia.

"Available Multicurrency Commitment": as at any date of determination with respect to any Multicurrency Lender (after giving effect to the making and payment of any Revolving Credit Loans required to be made on such date pursuant to subsection 2.16), an amount in Dollars equal to the lesser of (a) the excess, if any, of (i) the amount of such Multicurrency Lender's Multicurrency Commitment in effect on such date over (ii) the Dollar Equivalent of the Aggregate Multicurrency Outstandings of such Multicurrency Lender on such date, and (b) the excess, if any, of (i) the amount of such Lender's Revolving Credit Commitment in effect on such date over (ii) the Aggregate Committed Outstandings of such Lender on such date.

"Available Revolving Credit Commitment": as at any date of determination with respect to any Lender (after giving effect to the making and payment of any Revolving Credit Loans required to be made on such date pursuant to subsection 2.16), an amount in Dollars equal to the excess, if any, of (a) the amount of such Lender's Revolving Credit Commitment in effect on such date over (b) the Aggregate Committed Outstandings of such Lender on such date.

"Available Tenor" means, as of any date of determination and with respect to any then-current Benchmark for any Currency, as applicable, (a) if such Benchmark is a term rate, any tenor for such Benchmark (or component thereof) that is or may be used for determining the length of an interest period pursuant to this Agreement or (b) otherwise, any payment period for interest calculated with reference to such Benchmark (or component thereof) that is or may be used for determining any frequency of making payments of interest calculated with reference to such Benchmark, in each case, as of such date and not including, for the avoidance of doubt, any tenor for such Benchmark that is then-removed from the definition of "Interest Period" pursuant to Section 3.6(b)(iv).

"Bail-In Action": the exercise of any Write-Down and Conversion Powers by the applicable Resolution Authority in respect of any liability of an Affected Financial Institution.

"Bail-In Legislation": (a) with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law, regulation rule or requirement for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule and (b) with respect to the United Kingdom, Part I of the United Kingdom Banking Act 2009 (as amended from time to time) and any other law, regulation or rule applicable in the United Kingdom relating to the resolution of unsound or failing banks, investment firms or other financial institutions or their affiliates (other than through liquidation, administration or other insolvency proceedings).

"Bank of America": Bank of America, N.A.

"Bankruptcy Code": the Bankruptcy Code in Title 11 of the United States Code, as amended, modified, succeeded or replaced from time to time.

"Benchmark" means, initially, with respect to any (a) Obligations, interest, fees, commissions or other amounts denominated in, or calculated with respect to, Dollars, the Term SOFR Reference Rate; provided that if a Benchmark Transition Event has occurred with respect to the Term SOFR Reference Rate or then-current Benchmark for Dollars, then "Benchmark" means, with respect to such Obligations, interest, fees, commissions or other amounts, the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to Section 3.6(b)(i) and (b) Obligations, interest, fees, commissions or other amounts denominated in, or calculated with respect to any Available Foreign Currency, the applicable benchmark rate for Multicurrency Loans denominated

in such Currency; <u>provided</u> that if a Benchmark Transition Event or an Early Opt-In Election, as applicable, has occurred with respect to the then-current Benchmark for such Currency, then "Benchmark" means, with respect to such Obligations, interest, fees, commissions or other amounts, the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to <u>Section 3.6(b)(i)</u>.

"Benchmark Replacement" means, with respect to any Benchmark Transition Event for any then-current Benchmark, the sum of: (a) the alternate benchmark rate that has been selected by the Administrative Agent and the Borrower as the replacement for such Benchmark giving due consideration to (i) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (ii) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement for such Benchmark for syndicated credit facilities denominated in the applicable Currency at such time and (b) the related Benchmark Replacement Adjustment; provided that, if such Benchmark Replacement as so determined would be less than the Floor, such Benchmark Replacement will be deemed to be the Floor for the purposes of this Agreement and the other Loan Documents.

"Benchmark Replacement Adjustment": with respect to any replacement of any then-current Benchmark with an Unadjusted Benchmark Replacement for any applicable Available Tenor, the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by the Administrative Agent and the Borrower giving due consideration to (a) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body or (b) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement for syndicated credit facilities at such time denominated in the applicable Currency.

"Benchmark Replacement Conforming Changes": means, with respect to the use or administration of an initial Benchmark or the use, administration, adoption or implementation of any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Base Rate", the definition of "Business Day," the definition of "London Banking Day," the definition of "U.S. Government Securities Business Day," the definition of "Interest Period" or any similar or analogous definition (or the addition of a concept of "interest period"), timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, the applicability and length of lookback periods, the applicability of Section 3.11 and other technical, administrative or operational matters) that the Administrative Agent decides (in consultation with the Borrower) may be appropriate to reflect the adoption and implementation of any such rate or to permit the use and administration thereof by the Administrative Agent in a manner substantially consistent with market practice (or, if the Administrative Agent decides that adoption of any portion of such market practice is not administratively feasible or if the Administrative Agent determines that no market practice for the administration of any such rate exists, in such other manner of administration as the Administrative Agent decides is reasonably necessary in connection with the administration of this Agreement and the other Loan Documents).

"Benchmark Replacement Date" means the earliest to occur of the following events with respect to the then-current Benchmark for any Currency:

- (a) in the case of clause (a) or (b) of the definition of "Benchmark Transition Event," the later of (i) the date of the public statement or publication of information referenced therein and (ii) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide all Available Tenors of such Benchmark (or such component thereof); or
- (b) in the case of clause (c) of the definition of "Benchmark Transition Event," the first date on which such Benchmark (or the published component used in the calculation thereof) has been determined and announced by the regulatory supervisor for the administrator of such Benchmark (or such component thereof) to be non-representative; <u>provided</u> that such non-representativeness will be determined by reference to the most recent statement or publication referenced in such clause (c) and even if any Available Tenor of such Benchmark (or such component thereof) continues to be provided on such date.

For the avoidance of doubt, the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (a) or (b) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Event" means, with respect to the then-current Benchmark for any Currency, the occurrence of one or more of the following events with respect to such Benchmark:

- (a) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely; <u>provided</u> that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof);
- (b) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the Board, the Federal Reserve Bank of New York, the central bank for the Currency applicable to such Benchmark, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component), which states that the administrator of such Benchmark (or such component) has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof); or
- (c) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that all Available Tenors of such Benchmark (or such component thereof) are not, or as of a specified future date will not be, representative.

For the avoidance of doubt, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Start Date" means, with respect to any Benchmark for any Currency, (a) in the case of a Benchmark Transition Event, the earlier of (i) the applicable Benchmark Replacement Date and (ii) if such Benchmark Transition Event is a public statement or publication of information of a prospective event, the 90th day prior to the expected date of such event as of such public statement or publication of information (or if the expected date of such prospective event is fewer than 90 days after such statement or publication, the date of such statement or publication) and (b) in the case of an Early Opt-in Election, the date specified by the Administrative Agent or the Majority Lenders, as applicable, by notice to the Borrower, the Administrative Agent (in the case of such notice by the Majority Lenders) and the Lenders.

"Benchmark Unavailability Period" means, with respect to any then-current Benchmark for any Currency, the period (if any) (x) beginning at the time that a Benchmark Replacement Date with respect to such Benchmark pursuant to clauses (a) or (b) of that definition has occurred if, at such time, no Benchmark Replacement has replaced such Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 3.6(b) and (y) ending at the time that a Benchmark Replacement has replaced such Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 3.6(b).

"BBSY": with respect to Australian Dollars for any Interest Period (i) the rate of interest per annum equal to the per annum rate of interest which appears as "BID" on the page designated as "BBSY" on the Reuters Monitor System (or such other comparable page as may, in the reasonable opinion of the Administrative Agent and in consultation with the Borrower, replace such BBSY page on such system for the purpose of displaying the bank bill swap rates) with maturities comparable to such Interest Period at approximately 10:30 am (Sydney, Australia time) on the first day of such Interest Period or (ii) if such rate is not available at such time for any reason, the rate per annum determined by the Administrative Agent as the average of the buying rates quoted to the Administrative Agent at or around 10:30 am (Sydney, Australia time) on the first day of such Interest Period for bills of exchange accepted by leading Australian banks which have a tenor equal to such Interest Period. If any Interest Period is an Impacted Interest Period, then BBSY shall be the Interpolated Rate for such Interest Period. BBSY shall not be less than zero.

"Beneficial Ownership Regulation": 31 C.F.R. § 1010.230.

"Benefit Plan": any of (a) an "employee benefit plan" (as defined in Section 3(3) of ERISA) that is subject to Title I of ERISA, (b) a "plan" as defined in Section 4975 of the Code to which Section 4975 of the Code applies and (c) any Person whose assets include (for purposes of the Plan Asset Regulations or otherwise for purposes of Title I of ERISA or Section 4975 of the Code) the assets of any such "employee benefit plan" or "plan".

"benefited Lender": as defined in subsection 13.7(a).

"Board": the Board of Governors of the Federal Reserve System.

"Bookrunner": Wells Fargo Securities, LLC as the sole Bookrunner for this Agreement.

"Borrower": as defined in the preamble hereto.

"Borrower Materials": as defined in subsection 8.2.

"Borrowing Date": any Business Day specified in a notice pursuant to <u>subsection 2.2, 2.9</u> or <u>2.14</u> as a date on which the Borrower requests the Lenders to make Loans hereunder or, with respect to Local Currency Loans, the date on which a Foreign Subsidiary Borrower requests Local Currency Lenders to make Local Currency Loans to such Foreign Subsidiary Borrower pursuant to the Local Currency Facility to which such Foreign Subsidiary Borrower and Local Currency Lenders are parties.

"Business": as defined in subsection 6.11(b).

"Business Day": a day other than a Saturday, Sunday or other day on which commercial banks in New York are authorized or required by law to close; <u>provided</u> that when such term is used for the purpose of determining the date on which the Eurocurrency Base Rate is determined under this Agreement for any Loan denominated in euro for any Interest Period therefor and for purposes of determining the first and last day of any such Interest Period, references in this Agreement to Business Days shall be deemed to be references to Target Operating Days; provided, further,

- (a) if such day relates to any interest rate settings as to a SOFR Loan, any fundings, disbursements, settlements and payments in Dollars in respect of any such SOFR Loan, or any other dealings in Dollars to be carried out pursuant to this Agreement in respect of any such SOFR Loan, means any such day that is also a U.S. Government Securities Business Day;
- (b) if such day relates to any interest rate settings as to a Multicurrency Loan or CAF Advance denominated in a currency other than Dollars or euro, means any such day on which dealings in deposits in the relevant currency are conducted by and between banks in the London or other applicable offshore interbank market for such currency; and
- (c) if such day relates to any fundings, disbursements, settlements and payments in a currency other than Dollars or euro in respect of a Multicurrency Loan or a CAF Advance denominated in a currency other than Dollars or euro, or any other dealings in any currency other than Dollars or euro to be carried out pursuant to this Agreement in respect of any such Multicurrency Loan or CAF Advance (other than any interest rate settings), means any such day on which banks are open for foreign exchange business in the principal financial center of the country of such currency.

"CAF Advance": each competitive advance facility advance made pursuant to subsection 2.8.

"CAF Advance Availability Period": the period from and including the Closing Date to and including the date which is seven days prior to the Termination Date.

"CAF Advance Confirmation": each confirmation by the Borrower of its acceptance of CAF Advance Offers, which confirmation shall be substantially in the form of Exhibit E and shall be delivered to the Administrative Agent by facsimile transmission.

"CAF Advance Interest Payment Date": as to each CAF Advance, each interest payment date specified by the Borrower for such CAF Advance in the related CAF Advance Request.

"CAF Advance Maturity Date": as to any CAF Advance, the date specified by the Borrower pursuant to subsection 2.9(d)(ii) in its acceptance of the related CAF Advance Offer.

"CAF Advance Note": as defined in subsection 3.13(e).

"CAF Advance Offer": each offer by a Lender to make CAF Advances pursuant to a CAF Advance Request, which offer shall contain the information specified in Exhibit D and shall be delivered to the Administrative Agent by telephone, immediately confirmed by facsimile transmission.

"CAF Advance Request": each request by the Borrower for Lenders to submit bids to make CAF Advances, which request shall contain the information in respect of such requested CAF Advances specified in Exhibit C and shall be delivered to the Administrative Agent in writing, by facsimile transmission, or by telephone, immediately confirmed by facsimile transmission.

"Canadian Dollar": the lawful currency of Canada.

"Capital Lease Obligations": as to any Person, the obligations of such Person to pay rent or other amounts under any lease of (or other arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases on a balance sheet of such Person under GAAP and, for the purposes of this Agreement, the amount of such obligations at any time shall be the capitalized amount thereof at such time determined in accordance with GAAP.

"Capital Stock": any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all equivalent ownership interests in a Person (other than a corporation) and any and all warrants or options to purchase any of the foregoing.

"Cash Collateralize": to pledge and deposit with or deliver to the Administrative Agent, for the benefit of the Administrative Agent, an Issuing Lender (as applicable) and the Lenders, as collateral for L/C Obligations or obligations of Lenders to fund participations in respect thereof (as the context may require), cash or deposit account balances or, if an Issuing Lender benefiting from such collateral shall agree in its sole discretion, other credit support, in each case pursuant to documentation in form and substance reasonably satisfactory to (a) the Administrative Agent and (b) the Issuing Lender (as applicable). "Cash Collateral" shall have a meaning correlative to the foregoing and shall include the proceeds of such cash collateral and other credit support.

"CDOR": with respect to Canadian Dollars for any Interest Period, the rate per annum, equal to the average of the annual yield rates applicable to Canadian banker's acceptances at or about 10:00 a.m. (Toronto, Canada time) on the first day of such Interest Period on the "CDOR Page" (or any display substituted therefor) of Reuters Monitor Money Rates Service (or such other page or commercially available source displaying Canadian interbank bid rates for Canadian Dollar bankers' acceptances as may be designated by the Administrative Agent from time to time in consultation with the Borrower) for a term equivalent to such Interest Period (or if such Interest Period is not equal to a number of months, for a term equivalent to the number of months closest to such Interest Period). If any Interest Period is an Impacted Interest Period, then CDOR shall be the Interpolated Rate for such Interest Period. CDOR shall not be less than zero.

"Closing Date": May 10, 2021.

"Code": the Internal Revenue Code of 1986, as amended from time to time.

"Commitment Period": the period from and including the Closing Date to but excluding the Termination Date or such earlier date on which the Commitments shall terminate as provided herein.

"Commitments": the collective reference to the Revolving Credit Commitments, Multicurrency Commitments and L/C Commitments.

"Committed Outstandings Percentage": on any date with respect to any Lender, the percentage which the Adjusted Aggregate Committed Outstandings of such Lender constitutes of the Adjusted Aggregate Committed Outstandings of all Lenders.

"Commonly Controlled Entity": an entity, whether or not incorporated, which is under common control with the Borrower within the meaning of Section 4001(a)(14)(A)-(B) of ERISA or is part of a group which includes the Borrower and which is treated as a single employer under Sections 414(b), 414(c), 414(m) and 414(o) of the Code.

"Conduit Lender": any special purpose corporation organized and administered by any Lender for the purpose of making Loans otherwise required to be made by such Lender and designated by such Lender in a written instrument; provided, that the designation by any Lender of a Conduit Lender shall not relieve the designating Lender of any of its obligations to fund a Loan under this Agreement if, for any reason, its Conduit Lender fails to fund any such Loan, and the designating Lender (and not the Conduit Lender) shall have the sole right and responsibility to deliver all consents and waivers required or requested under this Agreement with respect to its Conduit Lender; and provided, further, that no Conduit Lender shall (a) be entitled to receive any greater amount pursuant to subsection 3.9, 3.10, 3.11 or 13.5 than the designating Lender would have been entitled to receive in respect of the extensions of credit made by such Conduit Lender or (b) be deemed to have any Commitment.

"Consolidated EBITDA": of any Person for any period, without duplication, Consolidated Net Income of such Person and its Subsidiaries for such period plus, to the extent reflected as a charge in the statement of such Consolidated Net Income for such period, the sum of:

- (a) income tax expense, including any expenses resulting from income tax disputes with a Governmental Authority,
- (b) Consolidated Interest Expense of such Person and its Subsidiaries, amortization or writeoff of debt discount and debt issuance costs and commissions, discounts and other fees and charges associated with Indebtedness,
 - (c) depreciation expense,
 - (d) amortization or write-down of intangibles (including, but not limited to, goodwill) and organization costs,
- (e) any extraordinary, unusual or nonrecurring expenses or losses (to the extent any of the foregoing are non-cash items) (including, whether or not otherwise includable as a separate item in the statement of such Consolidated Net Income for such period, losses on sales of assets outside of the ordinary course of business (including as a result of write downs of goodwill or net intangible assets) and including special charges and purchased research and development charges in connection with acquisitions and other strategic alliances, inventory step-up charges, fair value adjustments, and unrealized investment impairments),
 - (f) any non-cash stock compensation expense in accordance with GAAP,

- (g) any cash litigation costs, including judgments, orders, awards, settlements and related legal costs paid during such period (net of any cash litigation or settlement payments received during such period) ("<u>Cash Litigation Payments</u>"), provided that, solely for the purposes of this definition, the aggregate amount of Cash Litigation Payments under this clause (g) since December 31, 2022 shall not exceed \$1,000,000,000 plus all Accrued Legal Liabilities as set forth in the financial statements of the Borrower required to be delivered pursuant to Section 8.1 for the period ended December 31, 2022,
- (h) any cash or non-cash charges in respect of restructurings, plant closings, staff reductions, distributor network optimization initiatives, distribution technology optimization initiatives or other similar charges, provided that, solely for the purposes of this definition, the aggregate amount of all charges under this clause (h) shall not exceed \$500,000,000 since December March 31, 2022 2024,
- (i) any income or expense associated with business combinations following the adoption of FASB Statement No. 141(R), "Business Combinations a replacement of FASB Statement No. 141", which would have been treated as a cost of the acquisition (e.g., as goodwill) under FASB Statement No. 141, "Business Combinations" including income or expense relating to contingent consideration, and
- (j) any Non-Cash Charges, including those attributable to litigation, intangible asset impairment, intellectual property research and development charges;

and minus, to the extent included in the statement of such Consolidated Net Income for such period, the sum of

- (a) interest income (except to the extent deducted in determining Consolidated Interest Expense), and
- (b) any extraordinary, unusual or nonrecurring income or gains (to the extent any of the foregoing are non-cash items) (including, whether or not otherwise includable as a separate item in the statement of such Consolidated Net Income for such period, gains on the sales of assets outside of the ordinary course of business, inventory step-up charges, fair value adjustments, and unrealized investment impairments).

"Consolidated Interest Expense": of any Person for any period, total interest expense of such Person and its Subsidiaries for such period with respect to all outstanding Indebtedness of such Person and its Subsidiaries determined in accordance with GAAP (including, all net costs that are allocable to such period in accordance with GAAP).

"Consolidated Leverage Ratio": as at the last day of any period of four consecutive fiscal quarters of the Borrower, the ratio of (a) Consolidated Total Debt on such day to (b) Consolidated EBITDA of the Borrower and its Subsidiaries for such period.

"Consolidated Net Income": of any Person for any period, the consolidated net income (or loss) of such Person and its Subsidiaries for such period, determined on a consolidated basis in accordance with GAAP plus cash receipts received in connection with litigation related Non-Cash Charges and minus cash payments made in connection with such litigation related Non-Cash Charges.

"Consolidated Tangible Assets": at any date, Consolidated Total Assets minus (without duplication) the net book value of all assets which would be treated as intangible assets, as determined on a consolidated basis in accordance with GAAP.

"Consolidated Total Assets": at any date, the net book value of all assets of the Borrower and its Subsidiaries as determined on a consolidated basis in accordance with GAAP.

"Consolidated Total Debt": at any date, an amount equal to the aggregate principal amount of all Indebtedness (excluding, for the avoidance of doubt, any operating leases) of the Borrower and its Subsidiaries at such date, determined on a consolidated basis in accordance with GAAP as in effect on the date of this Agreement and with subsection 1.2(e), provided that Indebtedness incurred to prefund a Qualified Acquisition shall not constitute Consolidated Total Debt for purposes of compliance with Section 9.1 until the earlier of (i) the Qualified Acquisition has closed or (ii) the date the consummation of such Qualified Acquisition has been abandoned, terminated or expired pursuant to the terms of the relevant acquisition documentation.

"Continuing Directors": as defined in subsection 10(j).

"Contractual Obligation": as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its property is bound.

"Conversion Date": any date on which either (a) an Event of Default under <u>subsection 10(g)</u> has occurred or (b) the Commitments shall have been terminated and/or the Loans shall have been declared immediately due and payable pursuant to <u>Section 10</u>.

"Conversion Sharing Percentage": on any date with respect to any Lender and any Loans of such Lender outstanding in any currency other than Dollars, the percentage of such Loans such that, after giving effect to the conversion of such Loans to Dollars and the purchase and sale by such Lender of participating interests as contemplated by subsection 13.15, the Committed Outstandings Percentage of such Lender will equal such Lender's Revolving Credit Commitment Percentage on such date (calculated immediately prior to giving effect to any termination or expiration of the Commitments on the Conversion Date).

"Converted Loans": as defined in subsection 13.15(a).

"Covered Party": as defined in subsection 13.21(a).

"Currencies" means Dollars and each Available Foreign Currency, and "Currency" means any of such Currencies.

"<u>Debtor Relief Laws</u>": the Bankruptcy Code, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"<u>Default</u>": any of the events specified in <u>Section 10</u>, whether or not any requirement for the giving of notice, the lapse of time, or both, or any other condition, has been satisfied.

"Defaulting Lender": subject to subsection 2.19(b), any Lender that, as reasonably determined by the Administrative Agent, (a) has failed to perform any of its funding obligations hereunder, including in respect of its Loans or participations in respect of Letters of Credit, within two Business Days of the date required to be funded by it hereunder (other than as a result of a good faith dispute with respect to amount), (b) has notified the Borrower, the Administrative Agent or any Lender that it does not intend to comply with its funding obligations or has made a public statement to that effect with respect to its funding obligations hereunder or generally under other agreements in which it commits to extend credit, (c) has failed, within three Business Days after request by the Administrative Agent, to provide a certification in writing from an authorized officer of such Lender that it will comply with its funding obligations (and is financially able to meet such obligations), provided that such Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon the Administrative Agent's receipt of such certification in form and substance satisfactory to it and the Borrower, or (d) has (i) become the subject of a proceeding under any Debtor Relief Law or a Bail-In Action, (ii) had a receiver, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or a custodian appointed for it, or (iii) taken any action in furtherance of, or indicated its consent to, approval of or acquiescence in any such proceeding or appointment; provided that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority as long as such ownership or acquisition interest does not result in or provide such Lender with immunity from the jurisdiction of courts within the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Authority) to reject, repudiate, disayow or disaffirm any contracts or agreements made by such Lender.

"<u>Disposition</u>" or "<u>Dispose</u>": the sale, transfer, license, lease or other disposition, whether effected pursuant to a Division or otherwise (including any sale and leaseback transaction), of any property by any Person, including any sale, assignment, transfer or other disposal, with or without recourse, of any notes or accounts receivable or any rights and claims associated therewith.

"Dividing Person": as defined in the definition of "Division".

"<u>Division</u>": the division of assets, liabilities and/or obligations of a Person (the "<u>Dividing Person</u>") among two or more Persons (whether pursuant to a "plan of division" or similar arrangement), which may or may not include the Dividing Person and pursuant to which the Dividing Person may or may not survive.

"<u>Documentation Agents</u>": Bank of America, N.A., Barclays Bank PLC, Citibank, N.A., JPMorgan Chase Bank, N.A. and Société Générale, as Documentation Agents for this Agreement.

"<u>Dollar Equivalent</u>": at the time of determination thereof, (a) if such amount is expressed in Dollars, such amount, (b) if such amount is expressed in an Available Foreign Currency, the equivalent of such amount in Dollars determined by using the rate of exchange for the purchase of Dollars with such Available Foreign Currency in the London foreign exchange market at or about 11:00 a.m. London time (or New York time, as applicable) on a particular day as displayed by ICE Data Services as the "ask price", or as displayed on such other information service which publishes that rate of exchange from time to time in place of ICE Data Services (or if such service ceases to be available, the equivalent of such amount in Dollars as determined by the Administrative Agent based on the rate published or quoted by Bloomberg) and (c) if such amount is denominated in any other currency, the equivalent of such amount in Dollars as determined by the Administrative Agent based on the rate published or quoted by Bloomberg.

"Dollars" and "\$": dollars in lawful currency of the United States of America.

"Early Opt-in Election": with respect to any then-current Benchmark (other than any Benchmark in respect of Dollars), the occurrence of:

- (1) (i) a determination by the Administrative Agent or (ii) a notification by the Majority Lenders to the Administrative Agent (with a copy to the Borrower) that the Majority Lenders have determined that syndicated credit facilities being executed at such time, or that include language similar to that contained in this Section are being executed or amended, as applicable, to incorporate or adopt a new benchmark interest rate to replace such Benchmark, and
- (2) (i) the election by the Administrative Agent or (ii) the election by the Majority Lenders to declare that an Early Opt-in Election with respect to such Benchmark has occurred and the provision, as applicable, by the Administrative Agent of written notice of such election to the Borrower and the Lenders or by the Majority Lenders of written notice of such election to the Administrative Agent.

"EEA Financial Institution": (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country": any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"EEA Resolution Authority": any public administrative authority or any Person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution.

"Environmental Laws": any and all applicable foreign, Federal, state, local or municipal laws, rules, regulations, statutes, ordinances, codes, decrees or other enforceable requirements or orders of any Governmental Authority or other Requirements of Law regulating, relating to or imposing liability or standards of conduct concerning protection of human health or the environment, as now or may at any time hereafter be in effect.

"ERISA": the Employee Retirement Income Security Act of 1974, as amended from time to time.

"Erroneous Payment": as defined in subsection 11.12(a).

"Erroneous Payment Deficiency Assignment": as defined in subsection 11.12(d)

"Erroneous Payment Return Deficiency": as defined in subsection 11.12(d).

"EU Bail-In Legislation Schedule": the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor Person), as in effect from time to time.

"EURIBOR": for any Interest Period for each Loan denominated in Euro comprising part of the same borrowing, an interest rate per annum equal to (a) the Euro interbank offered rate administered by the Banking Federation of the European monetary union (or any other person which takes over administration of that rate) for the relevant period displayed on page EURIBOR01 of the Reuters screen at or about 11:00 A.M. (Central European time) two Target Operating Days before the first day of such Interest Period or, if such page or such service shall cease to be available, such other page or such other service for the purpose of displaying an average rate of the Banking Federation of the European monetary union as the Administrative Agent, after consultation with the Lenders and the Borrower, shall reasonably select; provided that any comparable or successor rate shall be applied in a manner consistent with market practice or (b) if no quotation for the Euro for the relevant period is displayed and the Administrative Agent has not selected an alternative service on which a quotation is displayed, the rate per annum determined by the Administrative Agent as the average at which deposits in Euro for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Loan being made, continued or converted and with a term equivalent to such Interest Period are offered to the Administrative Agent's London branch (or other branch or Affiliate) by leading banks in the European interbank market at or about 11:00 A.M. (Central European time) two Target Operating Days before the first day of such Interest Period. EURIBOR shall not be less than zero. If any Interest Period is an Impacted Interest Period, then EURIBOR shall be the Interpolated Rate for such Interest Period.

"Euros": the single currency of the participating member states of the European monetary union.

"Eurocurrency Rate":

for any Interest Period with respect to a Loan, a rate per annum determined in accordance with the following formula:

Eurocurrency Base Rate
1.00 - Eurocurrency Reserve Requirements

where,

"Eurocurrency Base Rate" means (i) with respect to each day during each Interest Period pertaining to a Loan or CAF Advance denominated in Japanese Yen or Swiss Franc, the rate per annum determined by the Administrative Agent to be the rate administered by the ICE Benchmark Administration (or any successor thereto) as the offered rate for deposits in Dollars with a term comparable to such Interest Period appearing on the Reuters Screen LIBOR01 Page at approximately 11:00 A.M., London time, two Business Days prior to the beginning of such Interest Period; provided, however, that if at any time for any reason such offered rate for any such currency does not appear on a Reuters page, "Eurocurrency Base Rate" shall mean, with respect to each day during each Interest Period pertaining to a Loan denominated in such Currency, the rate per annum determined by the Administrative Agent to be the average rate at which the Administrative Agent's London branch is offered deposits in such currency at or about 11:00 A.M., London time, two Business Days prior to the beginning of such Interest Period in the London interbank market for delivery on the first day of such Interest Period for the number of days comprised therein, (ii) with respect to each day during each Interest Period pertaining to a Multicurrency Loan or CAF Advance denominated in Pounds Sterling, the rate per annum determined by the Administrative Agent to be the offered rate for deposits in the applicable currency with a term comparable to such Interest Period appearing on the Reuters Screen LIBOR01 Page at approximately 11:00 A.M., London time, on the first day of such Interest

Period; <u>provided</u>, <u>however</u>, that if at any time for any reason such offered rate for Pounds Sterling does not appear on a Reuters page, "<u>Eurocurrency Base Rate</u>" shall mean, with respect to each day during each Interest Period pertaining to a Loan denominated in Pounds Sterling, the rate per annum determined by the Administrative Agent to be the average rate at which the Administrative Agent is offered deposits in Pounds Sterling at or about 11:00 A.M., London time, on the first day of such Interest Period in the London interbank market for delivery on the first day of such Interest Period for the number of days comprised therein, (iii) with respect to each day during each Interest Period pertaining to a Multicurrency Loan or CAF Advance denominated in Euros, EURIBOR, (iv) with respect to each day during each Interest Period pertaining to a Multicurrency Loan or CAF Advance denominated in Australian Dollars, BBSY, (v) with respect to each day during each Interest Period pertaining to a Multicurrency Loan or CAF Advance denominated in Canadian Dollars, CDOR, and (vi) with respect to each day during each Interest Period pertaining to a Multicurrency Loan or CAF Advance denominated in Singapore Dollars, SGD; <u>provided</u> that if any Interest Period with respect to any currency is an Impacted Interest Period, then the Eurocurrency Base Rate shall be the Interpolated Rate with respect to the applicable currency for such Interest Period; and

The Eurocurrency Rate and Eurocurrency Base Rate shall not be less than zero.

"Eurocurrency Reserve Requirements": for any day as applied to a Loan, the aggregate (without duplication) of the rates (expressed as a decimal) of reserve requirements in effect on such day (including, without limitation, basic, supplemental, marginal and emergency reserves) under any regulations of the Board or other Governmental Authority having jurisdiction with respect thereto dealing with reserve requirements prescribed for eurocurrency funding (currently referred to as "Eurocurrency Liabilities" in Regulation D of such Board) maintained by a member bank of such System.

"Event of Default": any of the events specified in Section 10, provided that any requirement for the giving of notice, the lapse of time, or both, or any other condition, has been satisfied.

"Exchange Rate": with respect to any non-Dollar currency on any date, the rate at which such currency may be exchanged into Dollars, as set forth on such date on the relevant Reuters currency page at or about 11:00 A.M., London time, on the date that is (x) in the case of Pounds Sterling and Canadian Dollars, on the date as of which the foreign exchange computation is made and (y) in the case of any other non-Dollar currency, two Business Days prior to the date as of which the foreign exchange computation is made. In the event that such rate does not appear on any Reuters currency page, the "Exchange Rate" with respect to such non-Dollar currency shall be determined by reference to such other publicly available service for displaying exchange rates as may be agreed upon by the Administrative Agent and the Borrower or, in the absence of such agreement, such "Exchange Rate" shall instead be the Administrative Agent's spot rate of exchange in the interbank market where its foreign currency exchange operations in respect of such non-Dollar currency are then being conducted, at or about 10:00 A.M., London time, on such applicable date for the purchase of Dollars with such non-Dollar currency, for delivery (x) in the case of Pounds Sterling and Canadian Dollars, on such date and (y) in the case of any other non-Dollar currency, two Business Days later; provided, that if at the time of any such determination, no such spot rate can reasonably be quoted, the Administrative Agent may use any reasonable method as it deems applicable to determine such rate, and such determination shall be conclusive absent manifest error. The Administrative Agent shall determine the Exchange Rate as of each Revaluation Date in accordance with the foregoing.

"Excluded Taxes": as defined in subsection 3.10(a).

"Existing Credit Facilities": as defined in the recitals hereto.

"Extensions of Credit": as to any Lender at any time, (a) an amount equal to the sum of (i) the aggregate principal amount of all Revolving Credit Loans made by such Lender then outstanding and (ii) such Lender's Revolving Credit Commitment Percentage of the L/C Obligations then outstanding, or (b) the making of any Loan or participation in any Letter of Credit by such Lender, as the context requires.

"Extension Request": a written request from the Borrower to the Administrative Agent requesting an extension of the Termination Date.

"<u>Facility Fee Rate</u>": for each day during each calculation period, (x) until the <u>Third Amendment Effective Date</u>, 0.090% per annum and (y) on and following the <u>Third Amendment Effective Date</u>, the rate per annum based on the Ratings in effect on such day, as set forth below:

Rating	Facility Fee Rate
Rating I	0.080 0.070%
Rating II	0.090 0.080%
Rating III	0.110 <u>0.090</u> %
Rating IV	0.150 <u>0.110</u> %
Rating V	0.225 <u>0.150</u> %

"FATCA": Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations or official interpretations thereof, any applicable intergovernmental agreements with respect thereto, any law, regulations, or other official guidance enacted in any other jurisdiction relating to such intergovernmental agreement, and any agreement entered into pursuant to Section 1471(b)(1) of the Code.

"Federal Funds Effective Rate": for any day, the rate calculated by the NYFRB based on such day's federal funds transactions by depositary institutions (as determined in such manner as the NYFRB shall set forth on its public website from time to time) and published on the next succeeding Business Day by the NYFRB as the federal funds effective rate, provided that if the Federal Funds Effective Rate shall be less than zero, such rate shall be deemed to zero for the purposes of this Agreement.

"Fee Commencement Date": the Closing Date.

"Financing Lease": any lease of property, real or personal, the obligations of the lessee in respect of which are required in accordance with GAAP to be capitalized on a balance sheet of the lessee.

"Fixed Rate CAF Advance": any CAF Advance made pursuant to a Fixed Rate CAF Advance Request.

"<u>Fixed Rate CAF Advance Request</u>": any CAF Advance Request requesting the Lenders to offer to make CAF Advances at a fixed rate (as opposed to a rate composed of the Eurocurrency Rate or Adjusted Term SOFR plus (or minus) a margin).

"Floor" means a rate of interest equal to 0.00%.

"Foreign Subsidiary Borrower": each Subsidiary of the Borrower organized under the laws of a jurisdiction outside the United States that the Borrower designates as a "Foreign Subsidiary Borrower" in a Local Currency Facility Addendum.

"Fronting Exposure": at any time there is a Defaulting Lender with respect to the Issuing Lender, such Defaulting Lender's Revolving Credit Commitment Percentage of the outstanding L/C Obligations other than L/C Obligations as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof.

"Funding Commitment Percentage": as at any date of determination (after giving effect to the making and payment of any Loans made on such date pursuant to subsection 2.16), with respect to any Lender, that percentage which the Available Revolving Credit Commitment of such Lender then constitutes of the Aggregate Available Revolving Credit Commitments.

"GAAP": generally accepted accounting principles in the United States of America consistent with those utilized in preparing the audited financial statements referred to in subsection 6.1.

"Governmental Authority": any nation or government, any state or other political subdivision thereof and any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government.

"Guarantee Obligation": as to any Person (the "guaranteeing person"), any obligation of (a) the guaranteeing person or (b) another Person (including, without limitation, any bank under any letter of credit) to induce the creation of which the guaranteeing person has issued a reimbursement, counterindemnity or similar obligation, in either case guaranteeing or in effect guaranteeing any Indebtedness, leases, dividends or other obligations (the "primary obligations") of any other unrelated third Person (the "primary obligor") in any manner, whether directly or indirectly, including, without limitation, any obligation of the guaranteeing person, whether or not contingent, (i) to purchase any such primary obligation or any property constituting direct or indirect security therefor, (ii) to advance or supply funds (1) for the purchase or payment of any such primary obligation or (2) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (iii) to purchase property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation or (iv) otherwise to assure or hold harmless the owner of any such primary obligation against loss in respect thereof; provided, however, that the term Guarantee Obligation shall not include endorsements of instruments for deposit or collection in the ordinary course of business. The amount of any Guarantee Obligation of any guaranteeing person shall be deemed to be the lower of (a) an amount equal to the stated or determinable amount of the primary obligation in respect of which such Guarantee Obligation is made and (b) the maximum amount for which such guaranteeing person may be liable pursuant to the terms of the instrument embodying such Guarantee Obligation, unless such primary obligation and the maximum amount for which such guaranteeing person may be liable are not stated or determinable, in which case the amount of such Guarantee Obligation shall be such guaranteeing person's reasonably anticipated liability in respect thereof as determined by the Borrower in good faith.

"Hedge Agreements": all agreements with non-related third parties of the Borrower with respect to any swap, forward, future or derivative transaction or option or similar agreements involving, or settled by reference to, one or more rates, currencies, commodities, equity or debt instruments or securities, or

economic, financial or pricing indices or measures of economic, financial or pricing risk or value or any similar transaction or any combination of these transactions; <u>provided</u> that no employee benefit plan of the Borrower or any of its Subsidiaries shall be a "Hedge Agreement."

"Impacted Interest Period": any Interest Period for which the applicable rate or screen used to determine the applicable rate (the "Applicable Screen Rate") shall not be available at the applicable time for such Interest Period.

"Indebtedness": of any Person at any date, without duplication, (a) all obligations of such Person for borrowed money, (b) all obligations of such Person for the deferred purchase price of property or services (other than current trade liabilities incurred in the ordinary course of such Person's business and payable in accordance with customary practices and earn-outs and other similar obligations in respect of acquisition and other similar agreements), (c) all obligations of such Person evidenced by notes, bonds, debentures or other similar instruments, (d) all indebtedness created or arising under any conditional sale or other title retention agreement with respect to property acquired by such Person (even though the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such property), (e) all Capital Lease Obligations of such Person, (f) all obligations of such Person, contingent or otherwise, as an account party or applicant under or in respect of acceptances, letters of credit, surety bonds or similar arrangements, (g) the liquidation value of all redeemable preferred Capital Stock of such Person, (h) all indebtedness of such Person, determined in accordance with GAAP, arising out of a Receivables Transaction, (i) all Guarantee Obligations of such Person in respect of obligations of the kind referred to in clauses (a) through (h) above, (j) all obligations of the kind referred to in clauses (a) through (i) above secured by (or for which the holder of such obligation has an existing right, contingent or otherwise, to be secured by) any Lien on property (including accounts and contract rights) owned by such Person, whether or not such Person has assumed or become liable for the payment of such obligation, and (k) for the purposes of subsection 10(f) only, all obligations of such Person in respect of Hedge Agreements. The Indebtedness of any Person shall, for the avoidance of doubt, exclude any operating leases (as defined in GAAP as in effect on the date of this Agreement) and the Indebtedness of any Person shall include the Indebtedness of any other entity (including any partnership in which such Person is a general partner) to the extent such Person is liable therefor as a result of such Person's ownership interest in or other relationship with such entity, except to the extent the terms of such Indebtedness expressly provide that such Person is not liable therefor.

"indemnified liabilities" as defined in subsection 13.5.

"indemnified party: as defined in subsection 13.5.

"Information" as defined in subsection 13.14.

"Insolvency": with respect to any Multiemployer Plan, the condition that such Multiemployer Plan is insolvent within the meaning of Section 4245 of ERISA.

"Interest Payment Date": (a) as to any ABR Loan, the last day of each March, June, September and December and the Termination Date, (b) as to any SOFR Loan or Multicurrency Loan having an Interest Period of three months or less, the last day of such Interest Period, and (c) as to any SOFR Loan or Multicurrency Loan having an Interest Period longer than three months, each day which is three months, or a whole multiple thereof, after the first day of such Interest Period and the last day of such Interest Period.

"Interest Period": (a) with respect to any SOFR Loan or Multicurrency Loan:

- (i) initially, the period commencing on the Borrowing Date or conversion date, as the case may be, with respect to such SOFR Loan or Multicurrency Loan and ending, with respect to such SOFR Loan, one or three months, or with respect to such Multicurrency Loan, one, three or six (or if available to all Lenders, twelve) months thereafter, as selected by the Borrower in its notice of borrowing or notice of conversion or continuation, as the case may be, given with respect thereto; and
- (ii) thereafter, each period commencing on the last day of the next preceding Interest Period applicable to such SOFR Loan or Multicurrency Loan and ending, with respect to such SOFR Loan, one or three months, or with respect to such Multicurrency Loan, one, three or six (or if available to all Lenders, twelve) months thereafter, as selected by the Borrower by irrevocable notice to the Administrative Agent not less than three Business Days prior to the last day of the then current Interest Period with respect thereto;

provided that, all of the foregoing provisions relating to Interest Periods are subject to the following:

- (1) if any Interest Period would otherwise end on a day that is not a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless the result of such extension would be to carry such Interest Period into another calendar month in which event such Interest Period shall end on the immediately preceding Business Day;
- (2) any Interest Period in respect of any Loan made by any Lender that would otherwise extend beyond the Termination Date applicable to such Lender shall end on such Termination Date;
- (3) any Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of a calendar month;
- (4) no Multicurrency Loan or CAF Advance denominated in Singapore Dollars shall have an Interest Period of one week or two months;
- (5) for any Interest Period, the durations available above shall only be available in a currency if a rate based on a corresponding duration (one, three, six or twelve months) is available from the applicable Screen Rate for the requested currency at the such time;
- (6) no tenor that has been removed from this definition pursuant to Section 3.6(b)(iv) shall be available for specification in a notice of borrowing or a notice of conversion or continuation, unless such tenor has been reinstated in accordance with Section 3.6(b)(iv), and
- (b) with respect to any Term Rate CAF Advance, the period beginning on the Borrowing Date with respect thereto and ending on the CAF Advance Maturity Date with respect thereto.

"Interpolated Rate": at any time, for any Interest Period, the rate per annum (rounded to the same number of decimal places as the Applicable Screen Rate) determined by the Administrative Agent (which determination shall be conclusive and binding absent manifest error) to be equal to the rate that results from interpolating on a linear basis between: (a) the Applicable Screen Rate for the longest period for which the Applicable Screen Rate is available) that is shorter than the Impacted Interest Period; and (b) the Applicable Screen Rate for the shortest period (for which that Applicable Screen Rate is available) that exceeds the Impacted Interest Period, in each case, at such time. The Interpolated Rate shall not be less than zero.

"Issuing Lender": Wells Fargo, Bank of America, Barclays Bank PLC, Citibank, N.A., JPMorgan Chase Bank, N.A., Société Générale and each other Lender which agrees to act as Issuing Lender hereunder and which is approved (such approval not to be unreasonably withheld) by the Administrative Agent and the Borrower, in its capacity as issuer of any Letter of Credit.

"Issuing Lender Commitment": as to any Issuing Lender at any time, the amount set forth opposite such Issuing Lender's name in Schedule I under the heading "Issuing Lender Commitment", as such amount may be reduced or increased from time to time in accordance with the provisions of this Agreement.

"Japanese Yen": the lawful currency of Japan.

"Judgment Currency": as defined in subsection 13.16(b).

"<u>L/C Commitment</u>": the lesser of (i) \$500,000,000 and (ii) the aggregate amount of the Revolving Credit Commitments then in effect (or, if the Revolving Credit Commitments have terminated, the Aggregate Revolving Credit Commitments in effect immediately prior to such termination).

"L/C Fee Payment Date": the last day of each March, June, September and December and the last day of the Commitment Period.

"L/C Obligations": at any time, an amount equal to the sum of (a) the aggregate then undrawn amount of the then outstanding Letters of Credit and (b) the aggregate amount of drawings under Letters of Credit that have not then been reimbursed pursuant to subsection 5.5.

"<u>L/C Participants</u>": the collective reference to all the Lenders other than the Issuing Lender.

"Lender Affiliate": (a) any Affiliate of any Lender, (b) any Person that is administered or managed by any Lender and that is engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business and (c) with respect to any Lender which is a fund that invests in commercial loans and similar extensions of credit, any other fund that invests in commercial loans and similar extensions of credit and is managed or advised by the same investment advisor as such Lender or by an Affiliate of such Lender or investment advisor.

"Lenders": as defined in the preamble hereto together with any affiliate of any such Person through which such Person elects, by notice to the Administrative Agent and the Borrower, to make any Loans available to the Borrower or any Foreign Subsidiary Borrower; <u>provided</u> that, for all purposes of voting or consenting with respect to (a) any amendment, supplementation or modification of any Loan Document, (b) any waiver of any requirements of any Loan Document or any Default or Event of Default and its consequences, or (c) any other matter as to which a Lender may vote or consent pursuant to

subsection 13.1 of this Agreement, the Lender making such election shall be deemed the "Person" rather than such affiliate, which shall not be entitled to vote or consent; and <u>provided further</u>, that unless the context otherwise requires, each reference herein to the Lenders shall be deemed to include any Conduit Lender.

"Lending Office": with respect to any Lender, the office of such Lender maintaining such Lender's Extensions of Credit, which office may, to the extent the applicable Lender notifies the Administrative Agent in writing, include an office of any Affiliate of such Lender or any domestic or foreign branch of such Lender or Affiliate.

"Letters of Credit": as defined in subsection 5.1(a).

"Lien": any mortgage, pledge, hypothecation, assignment, deposit arrangement, encumbrance, lien (statutory or other), charge or other security interest or any preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever (including, without limitation, any conditional sale or other title retention agreement and any Financing Lease having substantially the same economic effect as any of the foregoing but excluding for all purposes hereunder any letter of credit).

"Loan": any Revolving Credit Loan, CAF Advance, Multicurrency Loan or Local Currency Loan, as the case may be.

"Loan Documents": this Agreement, any Notes, the Applications, any Letters of Credit and any document or instrument evidencing or governing any Local Currency Facility.

"Loans to be Converted": as defined in subsection 13.15(a).

"Local Currency": any available and freely convertible non-Dollar currency selected by a Foreign Subsidiary Borrower and approved by the Administrative Agent.

"Local Currency Facility": any Qualified Credit Facility that the Borrower designates as a "Local Currency Facility" pursuant to a Local Currency Facility Addendum.

"Local Currency Facility Addendum": a Local Currency Facility Addendum received by the Administrative Agent, substantially in the form of Exhibit I, and conforming to the requirements of Section 4.

"Local Currency Facility Agent": with respect to each Local Currency Facility, the Local Currency Lender acting as agent for the Local Currency Lenders parties thereto (and, in the case of any Local Currency Facility to which only one Lender is a party, such Lender).

"Local Currency Facility Maximum Borrowing Amount": as defined in subsection 4.1(b).

"Local Currency Lender": any Lender (or, if applicable, any Affiliate, branch or agency thereof) party to a Local Currency Facility.

"Local Currency Lender Maximum Borrowing Amount": as defined in subsection 4.1(b).

"Local Currency Loan": any loan made pursuant to a Local Currency Facility.

"London Banking Day": any day on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"Majority Lenders": Lenders whose Aggregate Exposure Percentage in the aggregate is more than 50%. "Aggregate Exposure Percentage": as at any date of determination with respect to any Lender (a) at any time prior to the termination of the Revolving Credit Commitments, the aggregate Revolving Credit Commitments of such Lender divided by the aggregate Revolving Credit Commitments of all Lenders and (b) at any time after the termination of the Revolving Credit Commitments, the Aggregate Total Outstandings of such Lender divided by the Aggregate Total Outstandings of all Lenders; provided that for purposes of this definition, the Aggregate Exposure Percentage of each Lender shall be adjusted upward or downward so as to give effect to any participations or assignments effected pursuant to subsection 13.15; and provided further that the Revolving Credit Commitment of, and the Aggregate Total Outstandings held or deemed to be held by, any Defaulting Lender shall be excluded for purposes of making a determination of Majority Lenders.

"Majority Multicurrency Lenders": at any time, Multicurrency Lenders the Multicurrency Commitment Percentages of which aggregate more than 50%; provided that the Multicurrency Commitment of, and the Aggregate Multicurrency Outstandings held or deemed to be held by, any Defaulting Lender shall be excluded for purposes of making a determination of Majority Multicurrency Lenders.

"Material Adverse Effect": a material adverse effect on (a) the business, operations, property or condition (financial or otherwise) of the Borrower and its Subsidiaries taken as a whole or (b) the validity or enforceability of this Agreement or any of the other Loan Documents or the rights or remedies of the Administrative Agent or the Lenders hereunder or thereunder.

"Materials of Environmental Concern": any gasoline or petroleum (including crude oil or any fraction thereof) or petroleum products or any hazardous or toxic substances, materials or wastes, defined or regulated as such in or under any Environmental Law, including, without limitation, asbestos, polychlorinated biphenyls and urea-formaldehyde insulation.

"Maximum Leverage Ratio": 3.75 to 1.00; provided that (a) for the four consecutive fiscal quarters ended immediately following the consummation of any Qualified Acquisition (including the fiscal quarter in which such Qualified Acquisition occurs), the Maximum Leverage Ratio shall be 4.75 to 1.00, (b) for the fiscal quarter ended immediately after such four fiscal quarters referred to in clause (a), the Maximum Leverage Ratio shall be 4.50 to 1.00, (c) for the fiscal quarter ended immediately after the fiscal quarter referred to in clause (b), the Maximum Leverage Ratio shall be 4.25 to 1.00 and (d) for the fiscal quarter ended immediately after the fiscal quarter referred to in clause (c), the Maximum Leverage Ratio shall be 4.00 to 1.00 (and, for the avoidance of doubt, for each fiscal quarter ended after the fiscal quarter referred to in clause (d), the Maximum Leverage Ratio shall be 3.75 to 1.00).

"Moody's": Moody's Investors Service, Inc. or any successor thereto.

"Multicurrency Commitment": as to any Multicurrency Lender at any time, its obligation to make Multicurrency Loans to the Borrower in an aggregate amount in Available Foreign Currencies the Dollar Equivalent of which does not exceed at any time outstanding the amount set forth opposite such Multicurrency Lender's name in Schedule I under the heading "Multicurrency Commitment", as such amount may be reduced from time to time as provided in subsection 2.15 and the other applicable provisions hereof.

- "Multicurrency Commitment Percentage": as to any Multicurrency Lender at any time, the percentage which such Multicurrency Lender's Multicurrency Commitment at such time constitutes of the aggregate Multicurrency Commitments of all Multicurrency Lenders at such time (or, if the Multicurrency Commitments have terminated or expired, the percentage which (a) the Dollar Equivalent of the Aggregate Multicurrency Outstandings of such Multicurrency Lender at such time constitutes of (b) the Dollar Equivalent of the Aggregate Multicurrency Outstandings of all Multicurrency Lenders at such time).
- "Multicurrency Lender": each Lender having an amount greater than zero set forth opposite such Lender's name in Schedule I under the heading "Multicurrency Commitment."
 - "Multicurrency Loans": as defined in subsection 2.12.
 - "Multicurrency Note": as defined in subsection 3.13(f).
- "Multicurrency Sublimit": at any time, the lesser of (i) \$800,000,000 and (ii) the aggregate amount of the Revolving Credit Commitments then in effect (or, if the Revolving Credit Commitments have terminated, the aggregate Revolving Credit Commitments in effect immediately prior to such termination).
 - "Multiemployer Plan": a Plan which is a multiemployer plan as defined in Section 4001(a)(3) of ERISA.
- "Non-Cash Charges": (a) losses on asset sales, disposals or abandonments, (b) any impairment charge or asset write-off related to intangible assets, long-lived assets, and investments in debt and equity securities pursuant to GAAP, (c) all losses from investments recorded using the equity method, (d) stock-based awards compensation expense, and (e) other non-cash charges (provided that if any non-cash charges referred to in this clause (e) represent an accrual or reserve for potential cash items in any future period, the cash payment in respect thereof in such future period shall be subtracted from Consolidated EBITDA to such extent, and excluding amortization of a prepaid cash item that was paid in a prior period).
- "Non-Consenting Lender": any Lender that does not approve any consent, waiver or amendment that (i) requires the approval of all affected Lenders in accordance with the terms of subsection 13.1 and (ii) has been approved by the Majority Lenders.
 - "Non-Excluded Taxes": as defined in subsection 3.10(a).
 - "Non-extending Lender": as defined in subsection 2.20(a).
 - "Non-Multicurrency Lender": each Revolving Credit Lender which is not a Multicurrency Lender.
 - "Non-U.S. Lender": a Lender that is not a U.S. Person.
 - "Notes": the collective reference to any Revolving Credit Notes, any Multicurrency Notes and any CAF Advance Notes.
- "Notice of Local Currency Outstandings": with respect to each Local Currency Facility Agreement, a notice from the relevant Local Currency Facility Agent containing the information,

delivered to the Administrative Agent and any other Person, in the manner and by the time, specified for a Notice of Local Currency Outstandings in Schedule II.

"Notice of Multicurrency Loan Borrowing": with respect to a Multicurrency Loan, a notice from the Borrower containing the information in respect of such Loan, delivered to the Administrative Agent and any other Person, in the manner and by the time, specified for a Notice of Multicurrency Loan Borrowing in respect of the currency of such Loan in Schedule II.

"Notice of Multicurrency Loan Continuation": with respect to a Multicurrency Loan, a notice from the Borrower containing the information in respect of such Loan, delivered to the Person, in the manner and by the time, specified for a Notice of Multicurrency Loan Continuation in respect of the currency of such Loan in Schedule II.

"NYFRB": the Federal Reserve Bank of New York.

"Obligations": collectively, the unpaid principal of and interest on the Loans and all other obligations and liabilities of the Borrower and each Foreign Subsidiary Borrower under this Agreement and any Local Currency Facility and other Loan Documents to which it is a party (including, without limitation, interest accruing at the then applicable rate provided in this Agreement or any other applicable Loan Document after the maturity of the Loans and interest accruing at the then applicable rate provided in this Agreement or any other applicable Loan Document after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding, relating to the Borrower or such Foreign Subsidiary Borrower, whether or not a claim for post-filing or post-petition interest is allowed in such proceeding), whether direct or indirect, absolute or contingent, due or to become due, or now existing or hereafter incurred, which may arise under, out of, or in connection with, this Agreement, the Notes, the other Loan Documents, Hedge Agreements entered into with Lenders or any Affiliates of a Lender or any other document made, delivered or given in connection therewith, in each case whether on account of principal, interest, reimbursement obligations, fees, indemnities, costs, expenses or otherwise (including, without limitation, all fees and disbursements of counsel to the Administrative Agent or to the Lenders that are required to be paid by the Borrower or any Foreign Subsidiary Borrower pursuant to the terms of this Agreement or any other Loan Document).

"Overnight Bank Funding Rate": for any day, the rate comprised of both overnight federal funds and overnight Eurodollar borrowings by U.S.-managed banking offices of depository institutions, as such composite rate shall be determined by the NYFRB as set forth on its public website from time to time, and published on the next succeeding Business Day by the NYFRB as an overnight bank funding rate (from and after such date as the NYFRB shall commence to publish such composite rate).

"Participant": as defined in subsection 13.6(b).

"Participant Register": as defined in subsection 13.6(b).

"Patriot Act": as defined in subsection 13.18.

"Payment Receipt": as defined in subsection 11.12(a).

"PBGC": the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA.

"Person": an individual, partnership, corporation, business trust, joint stock company, trust, unincorporated association, joint venture, Governmental Authority or other entity of whatever nature.

"Plan": at a particular time, any employee benefit plan which is covered by ERISA and in respect of which the Borrower or a Commonly Controlled Entity is (or, if such plan were terminated at such time, would under Section 4069 of ERISA be deemed to be) an "employer" as defined in Section 3(5) of ERISA.

"Plan Asset Regulations": 29 C.F.R. §2510.3-101, et seq., as modified by Section 3(42) of ERISA.

"Platform": as defined in subsection 8.2.

"Pounds Sterling": the lawful currency of the United Kingdom of Great Britain and Northern Ireland.

"Properties": as defined in subsection 6.11(a).

"PTE": a prohibited transaction class exemption issued by the U.S. Department of Labor, as any such exemption may be amended from time to time.

"Public Lenders": as defined in subsection 8.2.

"QFC Credit Support": as defined in subsection 13.21.

"Qualified Acquisition": any transaction permitted under this Agreement and consummated on or after the Closing Date, (a) by which the Borrower or any of its Subsidiaries (i) acquires any going concern or business or all or substantially all of the assets of any firm, corporation or limited liability company, or division or business unit thereof, whether through purchase of assets, merger or otherwise or (ii) directly or indirectly acquires at least a majority (in number of votes) of the Capital Stock of a Person if the aggregate amount of Indebtedness incurred by the Borrower and its Subsidiaries to finance the purchase price and other consideration for such transaction, plus the amount of Indebtedness assumed by the Borrower and its Subsidiaries in connection with such transaction, is at least \$1,000,000,000 and (b) for which the Borrower notifies the Administrative Agent in writing prior to or promptly upon consummation of such transaction that such transaction shall be a "Qualified Acquisition" for purposes of this Agreement.

"Qualified Credit Facility": a credit facility (a) providing for one or more Local Currency Lenders to make unsecured loans denominated in a Local Currency to a Foreign Subsidiary Borrower, (b) providing for such loans to bear interest at a rate or rates determined by the Borrower and such Local Currency Lender or Local Currency Lenders and (c) otherwise conforming to the requirements of Section 4.

"Rating": the respective rating of each of the Rating Agencies applicable to the long-term senior unsecured non-credit enhanced debt of the Borrower, as announced by the Rating Agencies from time to time.

"Rating Agencies": collectively, Moody's and S&P.

"Rating Category": each of Rating I, Rating II, Rating III, Rating IV and Rating V.

"Rating I, Rating II, Rating III, Rating IV and Rating V": the respective Ratings set forth below:

Rating Category	Moody's	S&P
Rating I	greater than or equal to A2	greater than or equal to A
Rating I	greater than or equal to A3	greater than or equal to A-
Rating II <u>I</u>	greater than or equal to Baa1	greater than or equal to BBB+
Rating IHV	greater than or equal to Baa2	greater than or equal to BBB
Rating IV	greater than or equal to Baa3	greater than or equal to BBB
Rating V	lower than or equal to BalBaa3	lower than or equal to BBB+-

If different Ratings Categories are applicable, the higher Ratings Category shall apply unless one of the two Ratings is two or more levels lower than the other, in which case the Ratings Category shall be the level immediately below that of the higher of the two Ratings.

"Receivables": any accounts receivable of any Person, including, without limitation, any thereof constituting or evidenced by chattel paper, instruments or general intangibles (as defined in the Uniform Commercial Code of the State of New York), and all proceeds thereof and rights (contractual and other) and collateral related thereto.

"Receivables Transaction": any transactions or series of related transactions providing for the financing of Receivables of the Borrower or any of its Subsidiaries.

"Register": as defined in subsection 13.6(d).

"Reimbursement Obligation": the obligation of the Borrower to reimburse the Issuing Lender pursuant to <u>subsection 5.5</u> for amounts drawn under Letters of Credit.

"Related Parties": with respect to any Person, such Person's Affiliates and partners, officers, employees, agents and advisors of such Person and such Person's Affiliates.

"Relevant Governmental Body" means (a) with respect to a Benchmark Replacement in respect of Obligations, interest, fees, commissions or other amounts denominated in, or calculated with respect to, Dollars, the Board or the Federal Reserve Bank of New York, or any successor thereto and (b) with respect to a Benchmark Replacement in respect of Obligations, interest, fees, commissions or other amounts denominated in, or calculated with respect to, any Available Foreign Currency, (i) the central bank for the Currency in which such Obligations, interest, fees, commissions or other amounts are denominated, or calculated with respect to, or any central bank or other supervisor which is responsible for supervising either (A) such Benchmark Replacement or (B) the administrator of such Benchmark Replacement or (ii) any working group or committee officially endorsed or convened by (A) the central bank for the Currency in which such Obligations, interest, fees, commissions or other amounts are denominated, or calculated with respect to, (B) any central bank or other supervisor that is responsible for

supervising either (1) such Benchmark Replacement or (2) the administrator of such Benchmark Replacement, (C) a group of those central banks or other supervisors or (D) the Financial Stability Board or any part thereof.

"Reportable Event": any of the events set forth in Section 4043(c) of ERISA, other than those events as to which the thirty day notice period is waived under regulations issued under PBGC Reg. § 4043 or which notice is otherwise waived by PBGC.

"Requested Local Currency Loans": as defined in subsection 2.16(b).

"Requested Multicurrency Loans": as defined in subsection 2.16(a).

"Requirement of Law": as to any Person, the Certificate of Incorporation and By-Laws or other organizational or governing documents of such Person, and any law, treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.

"Resolution Authority": an EEA Resolution Authority or, with respect to any UK Financial Institution, a UK Resolution Authority.

"Response Date": as defined in subsection 2.20(a).

"Responsible Officer": the chief executive officer or the president of the Borrower or, with respect to financial matters, the chief financial officer of the Borrower.

"Revaluation Date": (a) with respect to any Loan, each of the following: (i) each date of a borrowing of a Multicurrency Loan or a Local Currency Loan or the making of a CAF Advance, (ii) each date of a continuation of a Multicurrency Loan pursuant to subsection 3.2 and (iii) such additional dates as the Administrative Agent shall determine or the Majority Lenders shall require; and (b) with respect to any Letter of Credit, each of the following: (i) each date of issuance of a Letter of Credit denominated in a currency other than Dollars, (ii) each date of an amendment of any such Letter of Credit having the effect of increasing the amount thereof (solely with respect to the increased amount), (iii) each date of any payment by the Issuing Lender under any Letter of Credit denominated in a currency other than Dollars and (iv) such additional dates as the Administrative Agent or the Issuing Lender shall determine or the Majority Lenders shall require.

"Revolving Credit Commitment": as to any Lender, the obligation of such Lender to make Revolving Credit Loans to the Borrower hereunder in an aggregate principal amount at any one time outstanding not to exceed the amount set forth opposite such Lender's name on Schedule I under the heading "Revolving Credit Commitment," as such amount may be reduced or increased from time to time in accordance with the provisions of this Agreement.

"Revolving Credit Commitment Percentage": as to any Lender at any time, the percentage which such Lender's Revolving Credit Commitment at such time constitutes of the Aggregate Revolving Credit Commitments at such time (or, if the Revolving Credit Commitments have terminated or expired, the percentage which (a) the Aggregate Revolving Credit Outstandings of such Lender at such time then constitutes of (b) the Aggregate Revolving Credit Outstandings of all Lenders at such time), subject to adjustment as provided in subsection 2.19.

- "Revolving Credit Loans": as defined in subsection 2.1.
- "Revolving Credit Note": as defined in subsection 3.13(d).
- "Revolving Lender": each Lender that has a Revolving Credit Commitment hereunder or that holds Revolving Credit Loans.
- "S&P": Standard & Poor's Ratings Services or any successor thereto.
- "Sanctioned Country": a country or territory which is the subject or target of any Sanctions (as of the Second Amendment Effective Date, the so-called Donetsk People's Republic, so-called Luhansk People's Republic, Crimea, Kherson, and Zaporizhzhia regions of Ukraine, Cuba, Iran, North Korea and Syria).
- "Sanctioned Person": (a) any Person described or listed in any Sanctions-related executive order or list of designated Persons maintained by the Office of Foreign Assets Control of the U.S. Department of the Treasury, the U.S. Department of State, the United Nations Security Council, the European Union, any European Union member state, His Majesty's Treasury of the United Kingdom or other relevant sanctions authority, (b) any Person operating, organized or resident in a Sanctioned Country or (c) any Person 50 percent or more owned or controlled by any such Person.
- "Sanctions": economic or financial sanctions or trade embargoes imposed, administered or enforced from time to time by (a) the U.S. government, including those administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury or the U.S. Department of State or (b) the United Nations Security Council, the European Union, any European Union member state, His Majesty's Treasury of the United Kingdom or other relevant sanctions authority.
 - "Second Amendment": that certain Second Amendment, dated as of March 1, 2023, to this Agreement.
 - "Second Amendment Consenting Lenders": those Lenders party to the Second Amendment.
 - "Second Amendment Effective Date": March 1, 2023.
- "SGD": with respect to Singapore Dollars for any Interest Period, the rate per annum equal to the rate administered by the Association of Banks in Singapore (or any other person which takes over the administration of that rate) for Singapore Dollars displayed on page SIBOR of the Reuters screen (or any replacement Reuters page which displays that rate) at approximately the Singapore time equivalent of 12:00 p.m. (London time) two Business Days prior to the first day of such Interest Period for a term comparable to such Interest Period. If any Interest Period is an Impacted Interest Period, then SGD shall be the Interpolated Rate for such Interest Period. SGD shall not be less than zero.
 - "Singapore Dollar": the lawful currency of the Republic of Singapore.
 - "Single Employer Plan": any Plan that is covered by Title IV of ERISA, but that is not a Multiemployer Plan.
 - "SOFR" means a rate equal to the secured overnight financing rate as administered by the SOFR Administrator.

"SOFR Administrator" means the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate).

"SOFR Loan" means any Loan bearing interest at a rate based on Adjusted Term SOFR other than pursuant to clause (c) of the definition of "ABR".

"Subsidiary": as to any Person, a corporation, limited liability company, partnership or other entity of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person. Unless otherwise qualified, all references to a "Subsidiary" or to "Subsidiaries" in this Agreement shall refer to a Subsidiary or Subsidiaries of the Borrower.

"Supported OFC": as defined in subsection 13.21.

"Swiss Franc": the lawful currency of the Swiss Confederation.

"Target Operating Day": any day that is not (a) a Saturday or Sunday, (b) Christmas Day or New Year's Day or (c) any other day on which the Trans-European Real-time Gross Settlement Operating System (or any successor settlement system) is not operating (as determined by the Administrative Agent).

"Taxes": all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments or other charges in the nature of a tax imposed by any Governmental Authority, including any interest additions or penalties applicable thereto.

"Term Rate CAF Advance": any CAF Advance made pursuant to a Term Rate CAF Advance Request.

"Term Rate CAF Advance Request": any CAF Advance Request requesting the Lenders to offer to make CAF Advances at an interest rate equal to (a) in respect of CAF Advances denominated in Dollars, Adjusted Term SOFR plus (or minus) a margin and (b) in respect of CAF Advances denominated in an Available Foreign Currency, the Eurocurrency Rate for the currency of such CAF Advance plus (or minus) a margin.

"Term SOFR" means,

• (a) for any calculation with respect to a SOFR Loan, the Term SOFR Reference Rate for a tenor comparable to the applicable Interest Period on the day (such day, the "Periodic Term SOFR Determination Day") that is two (2) U.S. Government Securities Business Days prior to the first day of such Interest Period, as such rate is published by the Term SOFR Administrator; provided, however, that if as of 5:00 p.m. (Eastern time) on any Periodic Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as published by the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such

first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such Periodic Term SOFR Determination Day, and

• (b) for any calculation with respect to an ABR Loan on any day, the Term SOFR Reference Rate for a tenor of one month on the day (such day, the "ABR Term SOFR Determination Day") that is two (2) U.S. Government Securities Business Days prior to such day, as such rate is published by the Term SOFR Administrator; provided, however, that if as of 5:00 p.m. (Eastern time) on any ABR Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as published by the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such first preceding U.S. Government Securities Business Days prior to such ABR SOFR Determination Day.

"Term SOFR Adjustment" means a percentage equal to 0.10% per annum.

"Term SOFR Administrator" means CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the Term SOFR Reference Rate selected by the Administrative Agent in its reasonable discretion).

"Term SOFR Reference Rate" means the forward-looking term rate based on SOFR.

"Termination Date": the later of (a) May 10, 20262029 and (b) if the maturity date is extended pursuant to subsection 2.20, such extended maturity date as determined pursuant to such subsection; provided, however, that, in each case, if such date is not a Business Day, the Termination Date shall be the next preceding Business Day. For the avoidance of doubt, pursuant to the Second Amendment, clause (a) of the above was extended to May 10, 2027 in respect of the Commitments and Loans of the Second Amendment Consenting Lenders (and may be similarly extended in respect of the Commitments and Loans of other Lenders in accordance with Section 2.20(e)).

"Third Amendment": that certain Third Amendment, dated as of May 10, 2024, to this Agreement.

"Third Amendment Effective Date": May 10, 2024.

"Tranche": the collective reference to SOFR Loans or Multicurrency Loans of the same Currency the then current Interest Periods with respect to all of which begin on the same date and end on the same later date (whether or not such Loans shall originally have been made on the same day); Tranches may be identified as "SOFR Tranches" or "Multicurrency Tranches".

"Transactions": the execution and delivery of this Agreement and the delivery of a notice to terminate the Existing Credit Facilities.

"Transferee": as defined in subsection 13.6(f).

"Type": as to any Revolving Credit Loan, its nature as an ABR Loan or SOFR Loan.

- "<u>UK Financial Institutions</u>": any BRRD Undertaking (as such term is defined under the PRA Rulebook (as amended from time to time) promulgated by the United Kingdom Prudential Regulation Authority) or any person falling within IFPRU 11.6 of the FCA Handbook (as amended from time to time) promulgated by the United Kingdom Financial Conduct Authority, which includes certain credit institutions and investment firms, and certain affiliates of such credit institutions or investment firms.
- "UK Resolution Authority": the Bank of England or any other public administrative authority having responsibility for the resolution of any UK Financial Institution.
- "<u>Unadjusted Benchmark Replacement</u>": with respect to a given Benchmark Replacement, such Benchmark Replacement excluding the Benchmark Replacement Adjustment for such Benchmark Replacement.
 - "United States": the United States of America (including the states, commonwealths and territories thereof and the District of Columbia).
- "<u>U.S. Government Securities Business Day</u>" means any day except for (a) a Saturday, (b) a Sunday or (c) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities; <u>provided</u>, that for purposes of notice requirements in <u>Section 2.2</u>, 3.1(a) and 3.2, in each case, such day is also a Business Day.
 - "<u>U.S. Person</u>": a "United States person" within the meaning of Section 7701(a)(30) of the Code.
 - "U.S. Special Resolution Regimes": as defined in subsection 13.21.
 - "U.S. Tax Compliance Certificate": as defined in subsection 3.10(c)(B)(iii).
 - "Wells Fargo": Wells Fargo Bank, National Association.
- "Write-Down and Conversion Powers": (a) with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule, and (b) with respect to the United Kingdom, any powers of the applicable Resolution Authority under the Bail-In Legislation to cancel, reduce, modify or change the form of a liability of any UK Financial Institution or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability or any of the powers under that Bail-In Legislation that are related to or ancillary to any of those powers.

1.2. Other Definitional Provisions

(a) Unless otherwise specified therein, all terms defined in this Agreement shall have the defined meanings when used in any Notes or any certificate or other document made or delivered pursuant hereto.

- (b) As used herein and in any Notes, and any certificate or other document made or delivered pursuant hereto, accounting terms relating to the Borrower and its Subsidiaries not defined in <u>subsection 1.1</u> and accounting terms partly defined in <u>subsection 1.1</u>, to the extent not defined, shall have the respective meanings given to them under GAAP.
- (c) The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section, subsection, Schedule and Exhibit references are to this Agreement unless otherwise specified.
 - (d) The meanings given to terms defined herein shall be equally applicable to both the singular and plural forms of such terms.
- (e) Computations of amounts and ratios referred to herein shall be made without giving effect to any change in accounting for leases pursuant to GAAP resulting from the implementation of Accounting Standards Update 2016-02, Leases (Topic 842) issued February 2016, as amended from time to time.

1.3. Rates

. The Administrative Agent does not warrant or accept any responsibility for, and shall not have any liability with respect to, (a) the continuation of, administration of, submission of, calculation of or any other matter related to the Term SOFR Reference Rate, Adjusted Term SOFR, Term SOFR, the Eurocurrency Rate or any other Benchmark, or any component definition thereof or rates referred to in the definition thereof, or with respect to any alternative, successor or replacement rate thereto (including any Benchmark Replacement), including whether the composition or characteristics of any such alternative, successor or replacement rate (including any Benchmark Replacement), as it may or may not be adjusted pursuant to Section 3.6(b), will be similar to, or produce the same value or economic equivalence of, or have the same volume or liquidity as, the Term SOFR Reference Rate, Term SOFR, Adjusted Term SOFR, the Eurocurrency Rate, such Benchmark or any other Benchmark prior to its discontinuance or unavailability, or (b) the effect, implementation or composition of any Benchmark Replacement Conforming Changes. The Administrative Agent and its Affiliates or other related entities may engage in transactions that affect the calculation of a Benchmark, any alternative, successor or replacement rate (including any Benchmark Replacement) or any relevant adjustments thereto and such transactions may be adverse to the Borrower. The Administrative Agent may select information sources or services in its reasonable discretion to ascertain any Benchmark, any component definition thereof or rates referred to in the definition thereof, in each case pursuant to the terms of this Agreement, and shall have no liability to the Borrower, any Lender or any other person or entity for damages of any kind, including direct or indirect, special, punitive, incidental or consequential damages, costs, losses or expenses (whether in tort, contract or otherwise and whether at law or in equity), for any error or calculation of any such rate (or component thereof) provided by any such information source or service.

SECTION 2

AMOUNT AND TERMS OF COMMITMENTS

2.1. Revolving Credit Commitments

- (a) Subject to the terms and conditions hereof, each Revolving Lender severally agrees to make revolving credit loans ("Revolving Credit Loans") in Dollars to the Borrower from time to time during the Commitment Period so long as after giving effect thereto (i) the Available Revolving Credit Commitment of each Revolving Lender is greater than or equal to zero and (ii) the Aggregate Total Outstandings of all Lenders do not exceed the Aggregate Revolving Credit Commitments. During the Commitment Period, the Borrower may use the Revolving Credit Commitments by borrowing, prepaying the Revolving Credit Loans in whole or in part, and reborrowing, all in accordance with the terms and conditions hereof.
- (b) The Revolving Credit Loans may from time to time be (i) SOFR Loans, (ii) ABR Loans or (iii) a combination thereof, as determined by the Borrower and notified to the Administrative Agent in accordance with <u>subsections 2.2</u> and <u>3.2</u>, <u>provided</u> that no Revolving Credit Loan shall be made as a SOFR Loan after the day that is one month prior to the Termination Date.

2.2. Procedure for Revolving Credit Borrowing

. The Borrower may borrow under the Revolving Credit Commitments during the Commitment Period on any Business Day, provided that the Borrower shall give the Administrative Agent irrevocable notice (which notice must be received by the Administrative Agent prior to (x) with respect to SOFR Loans, 10:00 A.M., New York City time, or (y) with respect to ABR Loans, 1:00 P.M., New York City time: (a) three U.S. Government Securities Business Days prior to the requested Borrowing Date, if all or any part of the requested Revolving Credit Loans are to be initially SOFR Loans or (b) on the requested Borrowing Date, otherwise), in each case specifying (i) the amount to be borrowed, (ii) the requested Borrowing Date, (iii) whether the borrowing is to be of SOFR Loans, ABR Loans or a combination thereof and (iv) if the borrowing is to be entirely or partly of SOFR Loans, the amount of such Type of Loan and the length of the initial Interest Period therefor. Each borrowing under the Revolving Credit Commitments (other than a borrowing under subsections 2.16 and 5.5) shall be in an amount equal to (x) in the case of ABR Loans, \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof (or, if the Aggregate Available Revolving Credit Commitments are less than \$1,000,000, such lesser amount) and (y) in the case of SOFR Loans, \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof. Upon receipt of any such notice from the Borrower, the Administrative Agent shall promptly notify each Revolving Lender thereof. Prior to (x) with respect to SOFR Loans, 11:00 A.M., New York City time, or (y) with respect to ABR Loans, 2:00 P.M., New York City time, on the Borrowing Date requested by the Borrower, each Revolving Lender will make an amount equal to its Funding Commitment Percentage of the principal amount of the Revolving Credit Loans requested to be made on such Borrowing Date available to the Administrative Agent for the account of the Borrower at the office of the Administrative Agent specified in writing by the Administrative Agent to the Lenders in funds immediately available to the Administrative Agent. Except as otherwise provided in subsection 2.16, such borrowing will then be made available to the Borrower by the Administrative Agent crediting the account of the Borrower with the aggregate of the amounts made available to the Administrative Agent by the Revolving Lenders and in like funds as received by the Administrative Agent.

2.3. Reserved

.

2.4. Reserved

•

2.5. Fees

•

- (a) <u>Facility Fee</u>. The Borrower agrees to pay to the Administrative Agent for the account of each Revolving Lender a facility fee for the period from and including the Fee Commencement Date to the Termination Date computed at the Facility Fee Rate on the actual daily amount of the Revolving Credit Commitment of such Revolving Lender (regardless of usage and, following terminating the Revolving Credit Commitments, on the actual daily amount of the Aggregate Total Outstandings of such Revolving Lender) during the period for which payment is made, subject to adjustment as provided in <u>subsection 2.19</u> and payable quarterly in arrears on the last day of each March, June, September and December, subject to the provisions of <u>subsection 3.7</u>, and on the Termination Date or such earlier date on which the Revolving Credit Commitments shall terminate as provided herein, commencing on the first of such dates to occur after the date hereof.
- (b) <u>Agency Fee Letter</u>. The Borrower agrees to pay to the Administrative Agent the fees in the amounts and on the dates set forth in the Agency Fee Letter.

2.6. Termination or Reduction of Revolving Credit Commitments

. The Borrower shall have the right, upon not less than three Business Days' notice to the Administrative Agent, to terminate the Revolving Credit Commitments or, from time to time, to reduce the amount of the Revolving Credit Commitments; provided that no such termination or reduction shall be permitted if, after giving effect thereto and to any prepayments of the Loans made on the effective date thereof, either (a) the Aggregate Available Revolving Credit Commitments would not be greater than or equal to zero or (b) the Available Revolving Credit Commitments of any Revolving Lender would not be greater than or equal to zero; provided further that such notice delivered by the Borrower may state that such notice is conditioned upon the effectiveness of other credit facilities, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified effective date) if such condition is not satisfied. Any such reduction shall be in an amount equal to \$5,000,000 or a whole multiple thereof and shall reduce permanently the Revolving Credit Commitments then in effect. The Administrative Agent shall give each Revolving Lender prompt notice of any notice received from the Borrower pursuant to this subsection 2.6.

2.7. Repayment of Loans

. The Borrower hereby unconditionally promises to pay to the Administrative Agent for the account of each Revolving Lender the then unpaid principal amount of each Revolving Credit Loan of such Lender on the Termination Date (or such earlier date on which the Revolving Credit Loans become due and payable pursuant to Section 10). The Borrower hereby further agrees to pay interest on the unpaid principal amount of the Revolving Credit Loans from time to time outstanding from the date hereof until payment in full thereof at the rates per annum, and on the dates, set forth in subsection 3.4.

2.8. CAF Advances

. Subject to the terms and conditions of this Agreement, the Borrower may borrow CAF Advances from time to time on any Business Day during the CAF Advance Availability Period. Term Rate CAF Advances may be denominated in Dollars or a lawful eurocurrency that is (x) readily available

and freely transferable and convertible into dollars, (y) available in the London interbank deposit market and (z) agreed to by the Administrative Agent and the bidding Lender. CAF Advances may be borrowed in amounts such that the amount of Aggregate Total Outstandings of all Lenders at any time shall not exceed the Aggregate Revolving Credit Commitments at such time. Within the limits and on the conditions hereinafter set forth with respect to CAF Advances, the Borrower from time to time may borrow, repay and request new CAF Advances.

2.9. Procedure for CAF Advance Borrowing

.

- (a) The Borrower shall request CAF Advances by delivering a CAF Advance Request to the Administrative Agent, not later than 11:00 AM (New York City time, or in the case of a Term CAF Advance Request for any non-Dollar currency, London time) five Business Days prior to the proposed Borrowing Date (in the case of a Term Rate CAF Advance Request), and not later than 10:00 AM. (New York City time) one Business Day prior to the proposed Borrowing Date (in the case of a Fixed Rate CAF Advance Request). Each CAF Advance Request in respect of any Borrowing Date may solicit bids for CAF Advances on such Borrowing Date in an aggregate principal amount of \$5,000,000 or an integral multiple of \$1,000,000 in excess thereof (or, in the case of CAF Advances to be denominated in a currency other than Dollars, an amount in such currency the Dollar Equivalent of which is equal to \$5,000,000 or \$1,000,000, as the case may be) and having not more than three alternative CAF Advance Maturity Dates. The CAF Advance Maturity Date for each CAF Advance shall be the date set forth therefor in the relevant CAF Advance Request, which date shall be (i) not less than 7 days nor more than 360 days after the Borrowing Date therefor, in the case of a Fixed Rate CAF Advance, (ii) not less than 30 days nor more than 360 days after the Borrowing Date therefor, in the case of a Term Rate CAF Advance and (iii) not later than the Termination Date, in the case of any CAF Advance. The Administrative Agent shall notify each Revolving Lender promptly by facsimile transmission of the contents of each CAF Advance Request received by the Administrative Agent.
- (b) In the case of a Term Rate CAF Advance Request, upon receipt of notice from the Administrative Agent of the contents of such CAF Advance Request, each Revolving Lender may elect, in its sole discretion, to offer irrevocably to make one or more CAF Advances at the applicable Eurocurrency Rate or (in the case of CAF Advances denominated in Dollars, Adjusted Term SOFR) plus (or minus) a margin determined by such Revolving Lender in its sole discretion for each such CAF Advance. Any such irrevocable offer shall be made by delivering a CAF Advance Offer to the Administrative Agent, before 10:30 A.M. (New York City time) on the day that is three Business Days (or, in the case of CAF Advances denominated in Dollars, three U.S. Government Securities Business Days) before the proposed Borrowing Date, setting forth:
 - (i) the maximum amount of CAF Advances for each CAF Advance Maturity Date and the aggregate maximum amount of CAF Advances for all CAF Advance Maturity Dates which such Revolving Lender would be willing to make (which amounts may, subject to subsection 2.8, exceed such Lender's Revolving Credit Commitment); and
 - (ii) the margin above or below the applicable Eurocurrency Rate or Adjusted Term SOFR, as applicable, at which such Revolving Lender is willing to make each such CAF Advance.

The Administrative Agent shall advise the Borrower before 11:00 A.M. (New York City time) on the date which is three Business Days before the proposed Borrowing Date of the contents of each such CAF Advance Offer received by it. If the Administrative Agent, in its capacity as a Revolving Lender, shall elect, in its sole discretion, to make any such CAF Advance Offer, it shall advise the Borrower of the contents of its CAF Advance Offer before 10:15 A.M. (New York City time) on the date which is three Business Days before the proposed Borrowing Date.

- (c) In the case of a Fixed Rate CAF Advance Request, upon receipt of notice from the Administrative Agent of the contents of such CAF Advance Request, each Revolving Lender may elect, in its sole discretion, to offer irrevocably to make one or more CAF Advances at a rate of interest determined by such Revolving Lender in its sole discretion for each such CAF Advance. Any such irrevocable offer shall be made by delivering a CAF Advance Offer to the Administrative Agent before 9:30 A.M. (New York City time) on the Borrowing Date, setting forth:
 - (i) the maximum amount of CAF Advances for each CAF Advance Maturity Date, and the aggregate maximum amount for all CAF Advance Maturity Dates, which such Lender would be willing to make (which amounts may, subject to subsection 2.8, exceed such Revolving Lender's Revolving Credit Commitment); and
 - (ii) the rate of interest at which such Revolving Lender is willing to make each such CAF Advance.

The Administrative Agent shall advise the Borrower before 10:00 A.M. (New York City time) on the proposed Borrowing Date of the contents of each such CAF Advance Offer received by it. If the Administrative Agent, in its capacity as a Revolving Lender, shall elect, in its sole discretion, to make any such CAF Advance Offer, it shall advise the Borrower of the contents of its CAF Advance Offer before 9:15 A.M. (New York City time) on the proposed Borrowing Date.

- (d) Before 11:30 A.M. (New York City time, or in the case of a CAF Advance requested by a Term Rate CAF Advance Request for any non-Dollar currency, London time) three Business Days (or in the case of a CAF Advance requested in Dollars, three U.S. Government Securities Business Days) before the proposed Borrowing Date (in the case of CAF Advances requested by a Term Rate CAF Advance Request) and before 10:30 A.M. (New York City time) on the proposed Borrowing Date (in the case of CAF Advances requested by a Fixed Rate CAF Advance Request), the Borrower, in its absolute discretion, shall:
 - (i) cancel such CAF Advance Request by giving the Administrative Agent telephone notice to that effect, or
 - (ii) by giving telephone notice to the Administrative Agent (immediately confirmed by delivery to the Administrative Agent of a CAF Advance Confirmation by facsimile transmission) (A) subject to the provisions of <u>subsection 2.9(e)</u>, accept one or more of the offers made by any Revolving Lender or Revolving Lenders pursuant to <u>subsection 2.9(b)</u> or <u>subsection 2.9(c)</u>, as the case may be, and (B) reject any remaining offers made by Revolving Lenders pursuant to <u>subsection 2.9(b)</u> or <u>subsection 2.9(c)</u>, as the case may be.
 - (e) The Borrower's acceptance of CAF Advances in response to any CAF Advance Offer shall be subject to the following limitations:

- (i) the amount of CAF Advances accepted for each CAF Advance Maturity Date specified by any Revolving Lender in its CAF Advance Offer shall not exceed the maximum amount for such CAF Advance Maturity Date specified in such CAF Advance Offer;
- (ii) the aggregate amount of CAF Advances accepted for all CAF Advance Maturity Dates specified by any Revolving Lender in its CAF Advance Offer shall not exceed the aggregate maximum amount specified in such CAF Advance Offer for all such CAF Advance Maturity Dates;
- (iii) the Borrower may not accept offers for CAF Advances for any CAF Advance Maturity Date in an aggregate principal amount in excess of the maximum principal amount requested in the related CAF Advance Request; and
- (iv) if the Borrower accepts any of such offers, it must accept offers based solely upon pricing for each relevant CAF Advance Maturity Date and upon no other criteria whatsoever, and if two or more Revolving Lenders submit offers for any CAF Advance Maturity Date at identical pricing and the Borrower accepts any of such offers but does not wish to (or, by reason of the limitations set forth in subsection 2.8, cannot) borrow the total amount offered by such Revolving Lenders with such identical pricing, the Borrower shall accept offers from all of such Revolving Lenders in amounts allocated among them pro rata according to the amounts offered by such Lenders (with appropriate rounding, in the sole discretion of the Borrower, to assure that each accepted CAF Advance is an integral multiple of \$1,000,000 or, in the case of CAF Advances to be denominated in a currency other than Dollars, an amount in such currency the Dollar Equivalent of which is approximately equal to \$1,000,000); provided that if the number of Revolving Lenders that submit offers for any CAF Advance Maturity Date at identical pricing is such that, after the Borrower accepts such offers pro rata in accordance with the foregoing provisions of this paragraph, the CAF Advance to be made by any such Revolving Lender would be less than \$5,000,000 (or, in the case of CAF Advances to be denominated in a currency other than Dollars, an amount in such currency the Dollar Equivalent of which is approximately equal to \$5,000,000) principal amount, the number of such Revolving Lenders shall be reduced by the Administrative Agent by lot until the CAF Advances to be made by each such remaining Revolving Lender would be in a principal amount of \$5,000,000 or an integral multiple of \$1,000,000 in excess thereof (or, in the case of CAF Advances to be denominated in a currency other than Dollars, an amount in such currency the Dollar Equivalent of which is approximately equal to \$5,000,000 or an integral multiple of \$1,000,000 in excess thereof).
- (f) If the Borrower notifies the Administrative Agent that a CAF Advance Request is cancelled pursuant to <u>subsection 2.9(d)(i)</u>, the Administrative Agent shall give prompt telephonic or written notice thereof to the Revolving Lenders.
- (g) If the Borrower accepts pursuant to <u>subsection 2.9(d)(ii)</u> one or more of the offers made by any Revolving Lender or Revolving Lenders, the Administrative Agent promptly shall notify each Revolving Lender which has made such an offer of (i) the aggregate amount of such CAF Advances to be made on such Borrowing Date for each CAF Advance Maturity Date and (ii) the acceptance or rejection of any offers to make such CAF Advances made by such Revolving Lender. Before 12:00 Noon (New York City time) on the Borrowing Date specified in the applicable CAF Advance Request (in the case of CAF Advances denominated in Dollars) and before the funding time for the relevant currency from time to time specified by the Administrative Agent by notice to the Revolving Lenders (in the case of CAF Advances denominated in any currency other than Dollars), each Revolving Lender whose CAF Advance

Offer has been accepted shall make available to the Administrative Agent the amount of CAF Advances to be made by such Revolving Lender, in immediately available funds, at the funding office for the relevant currency specified from time to time by the Administrative Agent by notice to the Revolving Lenders. The Administrative Agent will make such funds available to the Borrower as soon as practicable on such date at such office of the Administrative Agent. As soon as practicable after each Borrowing Date, the Administrative Agent shall notify each Revolving Lender of the aggregate amount of CAF Advances advanced on such Borrowing Date and the respective CAF Advance Maturity Dates thereof.

2.10. Repayment of CAF Advances

. The Borrower hereby unconditionally promises to pay to the Administrative Agent, for the account of each Revolving Lender which has made a CAF Advance, on the applicable CAF Advance Maturity Date the then unpaid principal amount of such CAF Advance. The Borrower shall have the right to prepay any principal amount of any CAF Advance only with the consent of the Revolving Lender to which such CAF Advance is owed. The Borrower hereby further agrees to pay interest on the unpaid principal amount of each CAF Advance from the Borrowing Date to the applicable CAF Advance Maturity Date at the rate of interest specified in the CAF Advance Offer accepted by the Borrower in connection with such CAF Advance (calculated on the basis of a 360-day year for actual days elapsed), payable on each applicable CAF Advance Interest Payment Date.

2.11. Certain Restrictions with Respect to CAF Advances

. A CAF Advance Request may request offers for CAF Advances to be made on not more than one Borrowing Date and to mature on not more than three CAF Advance Maturity Dates. No CAF Advance Request may be submitted earlier than five Business Days after submission of any other CAF Advance Request.

2.12. Multicurrency Commitments

. Subject to the terms and conditions hereof, each Multicurrency Lender severally agrees to make revolving credit loans (each, a "Multicurrency Loan") in any Available Foreign Currency to the Borrower from time to time during the Commitment Period so long as after giving effect thereto (a) the Available Multicurrency Commitment of each Multicurrency Lender is greater than or equal to zero, (b) the aggregate outstanding principal amount of Multicurrency Loans, plus (i) the aggregate outstanding principal amount of Local Currency Loans and (ii) the aggregate outstanding amount of L/C Obligations attributable to Letters of Credit denominated in any currency other than Dollars, does not exceed an amount the Dollar Equivalent of which is the Multicurrency Sublimit and (c) the Aggregate Total Outstandings of all Lenders do not exceed the Aggregate Revolving Credit Commitments. During the Commitment Period, the Borrower may use the Multicurrency Commitments by borrowing, repaying the Multicurrency Loans in whole or in part, and reborrowing, all in accordance with the terms and conditions hereof. Any Multicurrency Lender may cause its Multicurrency Loans to be made by any branch, affiliate or international banking facility of such Multicurrency Lender, provided, that such Multicurrency Lender shall remain responsible for all of its obligations hereunder and no additional taxes, costs or other burdens shall be imposed upon the Borrower or the Administrative Agent as a result thereof.

2.13. Repayment of Multicurrency Loans

. The Borrower hereby unconditionally promises to pay to the Administrative Agent for the account of each Multicurrency Lender the then unpaid principal amount of each Multicurrency Loan of such Multicurrency Lender on the Termination Date and on such other date(s) and in such other amounts

- 40 -

as may be required from time to time pursuant to this Agreement. The Borrower hereby further agrees to pay interest on the unpaid principal amount of the Multicurrency Loans advanced to it and from time to time outstanding until payment thereof in full at the rates per annum, and on the dates, set forth in <u>subsection 3.4</u>.

2.14. Procedure for Multicurrency Borrowing

. The Borrower may request the Multicurrency Lenders to make Multicurrency Loans during the Commitment Period on any Business Day by delivering a Notice of Multicurrency Loan Borrowing. Each borrowing under the Multicurrency Commitments shall be in an amount in an Available Foreign Currency the Dollar Equivalent of which is equal to at least \$1,000,000 (or, if the then Aggregate Available Multicurrency Commitments are less than \$1,000,000, such lesser amount). Upon receipt of any such Notice of Multicurrency Borrowing from the Borrower, the Administrative Agent shall promptly notify each Multicurrency Lender thereof. Not later than the funding time, on the applicable Borrowing Date, for the relevant Available Foreign Currency specified from time to time by the Administrative Agent by notice to the Borrower and the Multicurrency Lenders each Multicurrency Lender shall make an amount equal to its Multicurrency Commitment Percentage of the principal amount of Multicurrency Loans requested to be made on such Borrowing Date available to the Administrative Agent at the appropriate funding office for the relevant Available Foreign Currency specified from time to time by the Administrative Agent by notice to the Borrower and the Multicurrency Lenders in the relevant Available Foreign Currency and in immediately available funds; provided that, unless and until the Administrative Agent shall otherwise direct, any Multicurrency Lender may provide funds in the then applicable Dollar Equivalent (including an exchange fee and other normal and customary fees for providing this service as determined by the Administrative Agent in its sole discretion) thereof in lieu of the requested currency, so long as the Multicurrency Lender has given reasonable notice to the Administrative Agent of its desire and intent to so provide funds therefor promptly after (but in any event within one hour of) its receipt of any such notice for a Multicurrency Loan. The amounts made available by each Multicurrency Lender will then be made available to the Borrower at such funding office and in like funds as received by the Administrative Agent; provided, in the case of Multicurrency Loans for which any Lender has provided funds in Dollars, the Administrative Agent shall provide such funds to the applicable Borrower in the requested currency.

2.15. Termination or Reduction of Multicurrency Commitments

. The Borrower shall have the right, upon not less than three Business Days' notice to the Administrative Agent (which shall give prompt notice thereof to each Multicurrency Lender), to terminate the Multicurrency Commitments or, from time to time, to reduce the amount of the Multicurrency Commitments; provided that no such termination or reduction shall be permitted if, after giving effect thereto and to any prepayments of the Loans made on the effective date thereof, the Available Multicurrency Commitment of any Multicurrency Lender would be less than zero. Any such reduction shall be in an amount equal to U.S. \$1,000,000 or a whole multiple of U.S. \$100,000 in excess thereof and shall reduce permanently the Multicurrency Commitments then in effect; provided further that such notice delivered by the Borrower may state that such notice is conditioned upon the effectiveness of other credit facilities, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified effective date) if such condition is not satisfied.

2.16. Borrowings of Revolving Credit Loans and Refunding of Loans

•

- (a) If on any Borrowing Date on which the Borrower has requested the Multicurrency Lenders to make Multicurrency Loans (the "Requested Multicurrency Loans"),
 - (i) the aggregate principal amount of the Requested Multicurrency Loans exceeds the Aggregate Available Multicurrency Commitments on such Borrowing Date (before giving effect to the making and payment of any Loans required to be made pursuant to this <u>subsection 2.16</u> on such Borrowing Date) and,
 - (ii) the Dollar Equivalent of the amount of such excess is less than or equal to the aggregate Available Revolving Credit Commitments of all Non-Multicurrency Lenders (before giving effect to the making and payment of any Loans pursuant to this subsection 2.16 on such Borrowing Date),

each Non-Multicurrency Lender shall make a Revolving Credit Loan to the Borrower on such Borrowing Date, and the proceeds of such Revolving Credit Loans shall be simultaneously applied to repay outstanding Revolving Credit Loans, Local Currency Loans and/or Multicurrency Loans of the Multicurrency Lenders (as directed by the Borrower) in each case in amounts such that, after giving effect to (1) such borrowings and repayments and (2) the borrowing from the Multicurrency Lenders of the Requested Multicurrency Loans, the Committed Outstanding Percentage of each Lender will equal (as nearly as possible) its Revolving Credit Commitment Percentage. To effect such borrowings and repayments, (x) not later than 12:00 Noon, New York City time, on such Borrowing Date, the proceeds of such Revolving Credit Loans shall be made available by each Non-Multicurrency Lender to the Administrative Agent at its office specified in subsection 13.2 in Dollars and in immediately available funds and the Administrative Agent shall apply the proceeds of such Revolving Credit Loans toward repayment of outstanding Revolving Credit Loans, Multicurrency Loans and/or Local Currency Loans of the Multicurrency Lenders (as directed by the Borrower) and (y) concurrently with the repayment of such Loans on such Borrowing Date, (I) the Multicurrency Lenders shall, in accordance with the applicable provisions hereof, make the Requested Multicurrency Loans in an aggregate amount equal to the amount so requested by the Borrower (but not in any event greater than the Aggregate Available Multicurrency Commitments after giving effect to the making of such repayment of any Loans on such Borrowing Date) and (II) the Borrower shall pay to the Administrative Agent for the account of the Lenders whose Loans to the Borrower are repaid on such Borrowing Date pursuant to this subsection 2.16 all interest accrued on the amounts repaid to the date of repayment, together with any amounts payable pursuant to subsection 3.11 in connection with such repayment.

(b) Subject to the limitations on borrowings contained in a given Local Currency Facility, if on any Borrowing Date on which a Foreign Subsidiary Borrower has requested Local Currency Lenders to make Local Currency Loans (the "Requested Local Currency Loans") under a Local Currency Facility to which such Foreign Subsidiary Borrower and Local Currency Lenders are parties, (i) the aggregate principal amount of the Requested Local Currency Loans (A) exceeds the aggregate available amount of the commitments of such Local Currency Lenders under such Local Currency Facility on such Borrowing Date (before giving effect to the making and payment of any Revolving Credit Loans required to be made pursuant to this subsection 2.16 on such Borrowing Date) or (B) together with the aggregate then outstanding principal amount of Multicurrency Loans and the aggregate outstanding amount of L/C Obligations attributable to Letters of Credit denominated in any currency other than Dollars, would exceed an amount of which the Dollar Equivalent is the Multicurrency Sublimit, (ii) after giving effect to the Requested Local Currency Loans, the Dollar Equivalent of the aggregate outstanding principal amount of Local Currency Loans of such Foreign Subsidiary Borrower will be less than or equal to the aggregate commitments of such Local Currency Lenders under such Local Currency Facility and (iii) the

Dollar Equivalent of the amount of the excess described in clause (i) above is less than or equal to the Aggregate Available Revolving Credit Commitments of all Revolving Lenders other than such Local Currency Lenders (before giving effect to the making and payment of any Revolving Credit Loans pursuant to this subsection 2.16 on such Borrowing Date), each such other Revolving Lender shall make a Revolving Credit Loan to the Borrower, on such Borrowing Date, and the proceeds of such Revolving Credit Loans shall be simultaneously applied to repay outstanding Revolving Credit Loans, Multicurrency Loans and/or Local Currency Loans of such Local Currency Lenders (as directed by the Borrower) in each case in amounts such that, after giving effect to (1) such borrowings and repayments and (2) the borrowing from such Local Currency Lenders of the Requested Local Currency Loans, the Committed Outstandings Percentage of each Lender will equal (as nearly as possible) its Revolving Credit Commitment Percentage and the Dollar Equivalent of the aggregate outstanding principal amount of Multicurrency Loans and Local Currency Loans will not exceed the Multicurrency Sublimit. To effect such borrowings and repayments, (x) not later than 12:00 Noon, New York City time, on such Borrowing Date, the proceeds of such Revolving Credit Loans shall be made available by each such other Revolving Lender to the Administrative Agent at its office specified in subsection 13.2 in Dollars and in immediately available funds and the Administrative Agent shall apply the proceeds of such Revolving Credit Loans toward the repayment of outstanding Revolving Credit Loans, Multicurrency Loans and/or Local Currency Loans of such Local Currency Lenders (as directed by the Borrower) and (y) concurrently with the repayment of such Revolving Credit Loans on such Borrowing Date, (I) such Local Currency Lenders shall, in accordance with the applicable provisions hereof, make the Requested Local Currency Loans in an aggregate amount equal to the amount so requested by such Foreign Subsidiary Borrower and (II) the relevant Foreign Subsidiary Borrower shall pay to the Administrative Agent for the account of the Revolving Lenders whose Loans to such Foreign Subsidiary Borrower are repaid on such Borrowing Date pursuant to this subsection 2.16 all interest accrued on the amounts repaid to the date of repayment, together with any amounts payable pursuant to subsection 3.11 in connection with such repayment.

(c) If any borrowing of Revolving Credit Loans is required pursuant to this <u>subsection 2.16</u>, the Borrower shall notify the Administrative Agent in the manner provided for Revolving Credit Loans in <u>subsection 2.2</u>, except that the minimum borrowing amounts set forth in <u>subsection 2.2</u> shall not be applicable to the extent that such minimum borrowing amounts exceed the amounts of Revolving Credit Loans required to be made pursuant to this subsection 2.16.

2.17. Increase of Revolving Credit Commitments

. The Borrower shall have the right to increase the Revolving Credit Commitments from time to time in an aggregate principal amount not to exceed \$500,000,000 upon prior notice to the Administrative Agent and pursuant to procedures established by the Administrative Agent and acceptable to the Borrower. Each such increase (each, an "Additional Revolving Credit Commitment") shall be in a principal amount of at least \$25,000,000, and no such increase shall become effective (i) if a Default or Event of Default then exists or would exist immediately after giving effect thereto and (ii) unless all representations and warranties made by the Borrower required to be made pursuant to subsection 7.2 in this Agreement and the other Loan Documents are true in all material respects as of the date of such increase. Each bank or other financial institution (other than an existing Lender or Lender Affiliate providing any Additional Revolving Credit Commitments) shall be reasonably acceptable to the Administrative Agent and the Borrower and shall be a "Lender" for all purposes under this Agreement. No Lender shall have any obligation to provide any such Additional Revolving Credit Commitments unless and until it shall expressly agree under the applicable procedures to do so, at which time such Lender shall be deemed to have made an Additional Revolving Credit Commitment. The Lenders agree that this Agreement and the other Loan Documents may be amended from time to time with the consent

- 43 -

of the Administrative Agent and the Borrower to reflect such Additional Revolving Credit Commitments in connection with this subsection 2.17.

2.18. Cash Collateral

.

- (a) At any time that (i) there shall exist a Defaulting Lender and (ii) the Rating provided by (x) S&P is less than BBB or (y) Moody's is less than Baa2, within twenty Business Days after receipt of the written request of the Administrative Agent or the Issuing Lender, the Borrower shall deliver to the Administrative Agent Cash Collateral in an amount sufficient to cover all Fronting Exposure (after giving effect to subsection 2.19(a)(iv) and any Cash Collateral provided by the Defaulting Lender); provided however, notwithstanding the foregoing, in lieu of providing Cash Collateral to cover all or a portion of the Fronting Exposure, at any time within twenty Business Days after the receipt of written request of the Administrative Agent or the Issuing Lender, the Borrower (but only to the extent necessary to cover Fronting Exposure which is not covered by Cash Collateral) (i) may request that all or a portion of the Revolving Credit Commitments of such Defaulting Lender be re-allocated to another Lender or Lenders (as acceptable to each Lender in its sole discretion) or (ii) shall have the option to exercise its right to increase the Revolving Credit Commitments under subsection 2.17 by obtaining Additional Revolving Credit Commitments from one or more financial institutions (which may be a Lender) in an amount to cover all or a portion of the Fronting Exposure; provided further that the only limitation applicable to such exercise shall be that the aggregate amount of such increases pursuant to Section 2.17 shall not exceed \$500,000,000.
- (b) All Cash Collateral (other than credit support not constituting funds subject to deposit) shall be maintained in blocked, interest bearing deposit accounts at the Administrative Agent. The Borrower, and to the extent provided by any Lender, such Lender, hereby grants to (and subjects to the control of) the Administrative Agent, for the benefit of the Administrative Agent, the Issuing Lender and the Lenders, and agrees to maintain, a first priority security interest in all such cash, deposit accounts and all balances therein, and all other property so provided as collateral pursuant hereto, and in all proceeds of the foregoing, all as security for the obligations to which such Cash Collateral may be applied pursuant to subsection 2.18(c). If at any time the Administrative Agent reasonably determines that Cash Collateral is subject to any right or claim of any Person that is prior to the claim of the Administrative Agent as herein provided, or that the total amount of such Cash Collateral is less than the applicable Fronting Exposure and other obligations secured thereby, the Borrower or the relevant Defaulting Lender will, promptly upon demand by the Administrative Agent, pay or provide to the Administrative Agent additional Cash Collateral in an amount sufficient to eliminate such deficiency.
- (c) Notwithstanding anything to the contrary contained in this Agreement, Cash Collateral provided under any of this <u>subsection 2.18</u>, <u>subsection 5.1</u> or <u>2.19</u> or <u>Section 10</u> in respect of Letters of Credit shall be held and applied to the satisfaction of the specific L/C Obligations, obligations to fund participations therein (including, as to Cash Collateral provided by a Defaulting Lender, any interest accrued on such obligation) and other obligations for which the Cash Collateral was so provided, prior to any other application of such property as may be provided for herein.
- (d) Cash Collateral (or the appropriate portion thereof) provided to reduce Fronting Exposure or other obligations shall be released promptly following (i) the elimination of the applicable Fronting Exposure or other obligations giving rise thereto (including by the termination of Defaulting Lender status of the applicable Lender (or, as appropriate, its assignee following compliance with subsection 13.6)) or

(ii) the Administrative Agent's good faith determination that there exists excess Cash Collateral; <u>provided</u>, <u>however</u>, (x) that Cash Collateral furnished by or on behalf the Borrower shall not be released during the continuance of an Event of Default (and following application as provided in this <u>subsection 2.18</u> may be otherwise applied in accordance with <u>Section 10</u>), and (y) the Person providing Cash Collateral and the Issuing Lender, as applicable, may agree that Cash Collateral shall not be released but instead held to support Fronting Exposure or other obligations with respect to requested but unfunded extensions of credit.

2.19. Defaulting Lenders

•

- (a) Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as that Lender is no longer a Defaulting Lender, to the extent permitted by applicable law:
 - (i) That Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in <u>subsection 13.1</u>.
 - (ii) Any payment of principal, interest, fees or other amounts received by the Administrative Agent for the account of that Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Section 10 or otherwise, and including any amounts made available to the Administrative Agent by that Defaulting Lender pursuant to subsection 13.7), shall be applied at such time or times as may be determined by the Administrative Agent as follows: first, to the payment of any amounts owing by that Defaulting Lender to the Administrative Agent hereunder; second, to the payment on a pro rata basis of any amounts owing by that Defaulting Lender to the Issuing Lender hereunder; third, if so determined by the Administrative Agent or requested by the Issuing Lender, to be held as Cash Collateral for future funding obligations of that Defaulting Lender of any participation in any Letter of Credit; fourth, as the Borrower may request (so long as no Event of Default exists), to the funding of any Loan in respect of which that Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; fifth, if so determined by the Administrative Agent and the Borrower, to be held in an interest bearing deposit account and released in order to satisfy obligations of that Defaulting Lender to fund Loans under this Agreement; sixth, to the payment of any amounts owing to the Lenders or the Issuing Lender as a result of any judgment of a court of competent jurisdiction obtained by any Lender or the Issuing Lender against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; seventh, so long as no Event of Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; and eighth, to that Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (x) such payment is a payment of the principal amount of any Loans or L/C Obligations in respect of which that Defaulting Lender has not fully funded its appropriate share and (y) such Loans or L/C Obligations were made at a time when the conditions set forth in subsection 7.2 were satisfied or waived, such payment shall be applied solely to pay the Loans of, and L/C Obligations owed to, all non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of, or L/C Obligations owed to, that Defaulting Lender. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender or to post Cash Collateral

pursuant to this <u>subsection 2.19(a)(ii)</u> shall be deemed paid to and redirected by that Defaulting Lender, and each Lender irrevocably consents to the foregoing.

- (iii) That Defaulting Lender (x) shall be entitled to receive any facility fee pursuant to <u>subsection 2.5(a)</u> for any period during which that Lender is a Defaulting Lender only to extent allocable to the sum of (1) the Dollar Equivalent of the principal amount of the Loans funded by it and (2) its Revolving Credit Commitment Percentage of the stated amount of Letters of Credit for which it has provided Cash Collateral pursuant to <u>subsection 2.14</u>, 2.19(a)(ii) or 5.1, as applicable (and the Borrower shall (A) be required to pay to the Issuing Lender, as applicable, the amount of such fee allocable to its Fronting Exposure arising from that Defaulting Lender and (B) not be required to pay the remaining amount of such fee that otherwise would have been required to have been paid to that Defaulting Lender) and (y) shall be limited in its right to receive Letter of Credit fees as provided in <u>subsection 5.3(a)</u>.
- (iv) During any period in which there is a Defaulting Lender, for purposes of computing the amount of the obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit pursuant to <u>subsections 5.1</u> and <u>5.4(a)</u>, the "Revolving Credit Commitment Percentage" of each non-Defaulting Lender shall be computed without giving effect to the Revolving Credit Commitment of that Defaulting Lender; <u>provided</u>, that, (i) each such reallocation shall be given effect only if, at the date the applicable Lender becomes a Defaulting Lender, no Event of Default exists; and (ii) the aggregate obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit shall not exceed the positive difference, if any, of (1) the Revolving Credit Commitment of that non-Defaulting Lender *minus* (2) the Dollar Equivalent of the principal amount of the Loans of that Lender.
- (b) If the Borrower, the Administrative Agent and the Issuing Lender agree in writing that a Defaulting Lender should no longer be deemed to be a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein (which may include arrangements with respect to any Cash Collateral), that Lender will, to the extent applicable, purchase that portion of outstanding Revolving Credit Loans of the other Lenders or take such other actions as the Administrative Agent may determine to be necessary to cause the Revolving Credit Loans and funded and unfunded participations in Letters of Credit to be held on a pro rata basis by the Revolving Lenders in accordance with their Revolving Credit Commitment Percentages (without giving effect to subsection 2.19(a)(iv)), whereupon that Lender will cease to be a Defaulting Lender; provided that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrower while that Lender was a Defaulting Lender; and provided, further, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.
- (c) If any Lender is a Defaulting Lender, then the Borrower may, at its sole expense and effort, upon notice to such Defaulting Lender and the Administrative Agent, require such Defaulting Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, subsection 13.6), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an assignee that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment), provided that:
 - (i) the Borrower shall have paid to the Administrative Agent the assignment fee specified in <u>subsection 13.6</u>;

- (ii) such Defaulting Lender shall have received payment of an amount equal to 100% of the outstanding principal of its Loans, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any amounts under <u>subsection 3.11</u>) from the assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts);
 - (iii) such assignment does not conflict with applicable laws; and
- (iv) such Defaulting Lender shall not be required to sign and deliver any assignment form in order for such assignment to become effective.

A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.

2.20. Extension of Termination Date

•

- (a) The Borrower may, by sending an Extension Request to the Administrative Agent (in which case the Administrative Agent shall promptly deliver a copy to each of the Lenders), at least three months in advance of the Termination Date in effect at such time (including at any time prior thereto on or after any anniversary of the Closing Date), request that the Lenders extend such Termination Date then in effect to the first anniversary of the Termination Date then in effect. Each Lender, acting in its sole discretion, shall, by notice to the Administrative Agent given not later than the date that is the 20th day after the date of the Extension Request, or if such day is not a Business Day, the immediately following Business Day (the "Response Date") advise the Administrative Agent in writing whether or not such Lender agrees to such extension (each Lender that so advises the Administrative Agent that it will not extend the Termination Date, being referred to herein as a "Non-extending Lender"); provided that any Lender that does not advise the Administrative Agent by the Response Date and any Defaulting Lender shall be deemed to be a Non-extending Lender. The election of any Lender to agree to such extension shall not obligate any other Lender to agree.
- (b) (i) If, on the Response Date, Lenders holding Revolving Credit Commitments that aggregate to 50% or more of the total Revolving Credit Commitments shall not have agreed to extend the Termination Date, then such Termination Date shall not be so extended and the outstanding principal balance of all Loans and other amounts payable hereunder shall be payable on such Termination Date.
 - (ii) If (and only if), on the Response Date, Lenders holding Revolving Credit Commitments that aggregate to more than 50% of the total Revolving Credit Commitments shall have agreed to extend the Termination Date, then the Termination Date that shall so have agreed shall be the first anniversary of the current Termination Date (subject to satisfaction of the conditions set forth in subsection 2.20(d)). In the event of such extension, the Revolving Credit Commitment of each Non-extending Lender shall terminate on the Termination Date in effect prior to such extension, all Loans and other amounts payable hereunder to such Non-extending Lenders shall become due and payable on such Termination Date and the total Revolving Credit Commitments of the Lenders hereunder shall be reduced by the Revolving Credit Commitments of the Non-extending Lenders so terminated on such Termination Date.

- (c) In the event that the conditions of clause (ii) of paragraph (b) above have been satisfied, the Borrower shall have the right on or before the Termination Date in effect prior to the requested extension, at its own expense, to require any Non-extending Lender to transfer and assign without recourse (in accordance with and subject to the restrictions contained in <u>subsection 13.6</u>) all its interests, rights and obligations under this Agreement to one or more banks or other financial institutions identified to the Non-extending Lender, which may include any Lender (each an "<u>Additional Lender</u>"), <u>provided</u> that (w) such Additional Lender, if not already a Lender hereunder, shall be subject to the approval of the Administrative Agent, the Issuing Lender and the Borrower (such approvals not to be unreasonably withheld), (x) such assignment shall become effective as of a date specified by the Borrower (which shall not be later than the Termination Date in effect prior to the requested extension); (y) the Additional Lender shall pay to such Non-extending Lender in immediately available funds on the effective date of such assignment the principal of and interest accrued to the date of payment on the Loans made by it hereunder and all other amounts accrued for its account or owed to it hereunder and (z) such Non-extending Lender shall not be required to sign and deliver any assignment form in order for such assignment to become effective.
- (d) As a condition precedent to each such extension, the Borrower shall deliver to the Administrative Agent a certificate of the Borrower dated as of the Termination Date then in effect signed by a Responsible Officer of the Borrower certifying that, before and after giving effect to such extension, (A) the representations and warranties contained in Section 6 (including subsections 6.13 and 6.14) and the other Loan Documents are true and correct on and as of the Termination Date, and except that for purposes of this subsection 2.20, the representations and warranties contained in subsection 6.1 shall be deemed to refer to the most recent statements furnished pursuant to paragraphs (a) and (b), respectively, of subsection 8.1, and (B) no Default or Event of Default exists. In addition, no such extension of the Revolving Credit Facility shall be effective if on the Termination Date then in effect, the Aggregate Total Outstandings would exceed the Revolving Credit Commitments then being extended.

SECTION 3

CERTAIN PROVISIONS APPLICABLE TO THE LOANS

3.1. Optional and Mandatory Prepayments

•

(a) The Borrower may at any time and from time to time prepay the Loans, in whole or in part, without premium or penalty (other than any amounts payable pursuant to <u>subsection 3.11</u> if such prepayment is of SOFR Loans and is made on a day other than the last day of the Interest Period with respect thereto), upon at least three U.S. Government Securities Business Days' irrevocable notice to the Administrative Agent, specifying the date and amount of prepayment and whether the prepayment is of SOFR Loans, ABR Loans or a combination thereof, and, if of a combination thereof, the amount allocable to each; <u>provided further</u> that such notice delivered by the Borrower may state that such notice is conditioned upon the effectiveness of other credit facilities, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified effective date) if such condition is not satisfied. Upon receipt of any such notice the Administrative Agent shall promptly notify each Lender thereof. If any such notice is given, the amount specified in such notice shall be due and payable on the date specified therein. Partial prepayments of Loans shall be in an aggregate principal amount of at least \$1,000,000 or an integral multiple of \$100,000 in excess thereof.

- (b) The Borrower may at any time and from time to time prepay, without premium or penalty (other than any amounts payable pursuant to subsection 3.11 if such prepayment is of Multicurrency Loans and is made on a day other than the last day of the Interest Period with respect thereto), the Multicurrency Loans, in whole or in part, upon at least three Business Days' irrevocable notice to the Administrative Agent specifying the date and amount of prepayment; provided further that such notice delivered by the Borrower may state that such notice is conditioned upon the effectiveness of other credit facilities, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified effective date) if such condition is not satisfied. Upon the receipt of any such notice, the Administrative Agent shall promptly notify each Multicurrency Lender thereof. If any such notice is given, the amount specified in such notice shall be due and payable on the date specified therein. Partial prepayments of Multicurrency Loans shall be in an aggregate principal amount the Dollar Equivalent of which is at least \$1,000,000 or an integral multiple of \$100,000 in excess thereof.
- (c) If at any time during the Commitment Period, for any reason the Aggregate Total Outstandings of all Lenders exceed the Aggregate Revolving Credit Commitments then in effect, the Borrower shall, without notice or demand, immediately prepay the Revolving Credit Loans and/or the Multicurrency Loans in amounts such that the sum of (A) the aggregate principal amount of the Revolving Credit Loans so prepaid and (B) the Dollar Equivalent of the aggregate principal amount of the Multicurrency Loans so prepaid, equals or exceeds the amount of such excess.
 - (i) If, at any time during the Commitment Period, for any reason either (1) the Aggregate Total Outstandings of all Multicurrency Lenders exceed the Aggregate Revolving Credit Commitments of the Multicurrency Lenders by 5% or more, (2) the Aggregate Multicurrency Outstandings exceed the aggregate Multicurrency Commitments by 5% or more, (3) the sum of the Aggregate Multicurrency Outstandings plus the Dollar Equivalent of (x) the aggregate outstanding principal amount of Local Currency Loans and (y) the aggregate outstanding amount of L/C Obligations attributable to Letters of Credit denominated in currencies other than Dollars, exceeds the aggregate Multicurrency Commitments by 5% or more or (4) the Dollar Equivalent of all L/C Obligations attributable to Letters of Credit denominated in currencies other than Dollars exceeds, in the aggregate, the Multicurrency Sublimit by 5% or more, the Borrower shall, without notice or demand, immediately prepay the Revolving Credit Loans and/or the Multicurrency Loans and/or Local Currency Loans and/or cash collateralize the L/C Obligations in amounts such that any such excess is eliminated.
 - (ii) Each prepayment of Loans pursuant to this <u>subsection 3.1(c)</u> shall be accompanied by the payment of unpaid accrued interest on the principal amount so prepaid and any amounts payable under <u>subsection 3.11</u> in connection with such prepayment.
 - (iii) Notwithstanding the foregoing, mandatory prepayments of Revolving Credit Loans or Multicurrency Loans that would otherwise be required pursuant to this <u>subsection 3.1(c)</u> solely as a result of fluctuations in Exchange Rates from time to time shall only be required to be made pursuant to this <u>subsection 3.1(c)</u> on the last Business Day of each month on the basis of the Exchange Rate in effect on such Business Day.

3.2. Conversion and Continuation Options

•

- (a) The Borrower may elect from time to time to convert SOFR Loans to ABR Loans by giving the Administrative Agent at least two Business Days' prior irrevocable notice of such election. The Borrower may elect from time to time to convert ABR Loans to SOFR Loans by giving the Administrative Agent at least three U.S. Government Securities Business Days' prior irrevocable notice of such election. Any such notice of conversion to SOFR Loans shall specify the length of the initial Interest Period therefor. Upon receipt of any such notice the Administrative Agent shall promptly notify each Lender thereof. All or any part of outstanding SOFR Loans and ABR Loans may be converted as provided herein, provided that (i) no Loan may be converted into a SOFR Loan when any Event of Default has occurred and is continuing and the Administrative Agent has or the Majority Lenders have determined that such a conversion is not appropriate and (ii) no Loan may be converted into a SOFR Loan after the date that is one month prior to the Termination Date. No Multicurrency Loan or CAF Advance may be converted into or continued as a Loan denominated in a different currency, but instead must be prepaid in the original currency of such Loan and reborrowed in the other currency.
- (b) Any SOFR Loans may be continued as such upon the expiration of the then current Interest Period with respect thereto by the Borrower giving notice to the Administrative Agent, in accordance with the applicable provisions of the term "Interest Period" set forth in subsection 1.1, of the length of the next Interest Period to be applicable to such Loans, provided that no SOFR Loan may be continued as such (i) when any Event of Default has occurred and is continuing and the Administrative Agent has or the Majority Lenders have determined that such a continuation is not appropriate or (ii) after the date that is one month prior to the Termination Date, and provided, further, that if the Borrower shall fail to give such notice or if such continuation is not permitted, SOFR Loans shall be automatically converted to ABR Loans on the last day of such then expiring Interest Period.
- (c) Any Multicurrency Loans may be continued as such upon the expiration of the then current Interest Period with respect thereto by the Borrower giving a Notice of Multicurrency Loan Continuation, <u>provided</u>, that if the Borrower shall fail to give such Notice of Multicurrency Loan Continuation by the deadline specified therefor in <u>Schedule II</u>, such Multicurrency Loans shall automatically be continued for an Interest Period of one month.

3.3. Minimum Amounts and Maximum Number of Tranches

. All borrowings, conversions and continuations of Revolving Credit Loans and Multicurrency Loans hereunder and all selections of Interest Periods hereunder shall be in such amounts and be made pursuant to such elections so that, after giving effect thereto, (i) the aggregate principal amount of the SOFR Loans comprising each SOFR Tranche shall be equal to \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof and (ii) the aggregate principal amount of the Multicurrency Loans comprising each Multicurrency Tranche shall be in an amount the Dollar Equivalent of which is at least \$5,000,000. In no event shall there be more than ten Tranches outstanding at any time, unless the Borrower and the Administrative Agent shall so agree.

3.4. Interest Rates and Payment Dates

•

(a) Each SOFR Loan shall bear interest for each day during each Interest Period with respect thereto at a rate per annum equal to Adjusted Term SOFR determined for such Interest Period <u>plus</u> the Applicable Margin in effect for such day.

- (b) Each ABR Loan shall bear interest at a rate per annum equal to the ABR plus the Applicable Margin.
- (c) Each Multicurrency Loan shall bear interest at a rate per annum equal to the Eurocurrency Rate <u>plus</u> the Applicable Margin in effect on the day such Multicurrency Loan is made.
 - (d) Each CAF Advance shall bear interest at the rate determined in accordance with subsection 2.9.
- (e) If all or a portion of (i) any principal of any Loan, (ii) any interest payable thereon, (iii) any facility fee or (iv) any other amount payable hereunder shall not be paid when due (whether at the stated maturity, by acceleration or otherwise), the principal of the Loans and/or any such overdue interest, facility fee or other amount shall bear interest at a rate per annum which is (x) in the case of principal, the rate that would otherwise be applicable thereto pursuant to the foregoing provisions of this subsection <u>plus</u> 2% or (y) in the case of any such overdue interest, facility fee or other amount, the rate described in paragraph (b) of this subsection <u>plus</u> 2%, in each case from the date of such non-payment until such overdue principal, interest, facility fee or other amount is paid in full (as well after as before judgment).
- (f) Interest pursuant to this subsection shall be payable in arrears on each Interest Payment Date or CAF Advance Interest Payment Date, as the case may be, <u>provided</u> that interest accruing pursuant to paragraph (e) of this subsection shall be payable from time to time on demand.

3.5. Computation of Interest and Fees

.

- (a) All interest and fees hereunder shall be computed on the basis of a year of 360 days, except that (i) interest computed by reference to the ABR when it is based on the Prime Rate shall be computed on the basis of a year of 365 days (or 366 days in a leap year) and (ii) interest computed with respect to Loans denominated in Australian Dollar, Canadian Dollar and Pounds Sterling shall be computed on the basis of a year of 365 days, and in each case shall be payable for the actual number of days elapsed (including the first day but excluding the last day). The applicable ABR, Adjusted Term SOFR or Eurocurrency Rate shall be determined by the Administrative Agent, and such determination shall be conclusive absent manifest error. The Administrative Agent shall as soon as practicable notify the Borrower and the Lenders of each determination of a Eurocurrency Rate or Adjusted Term SOFR. Any change in the interest rate on a Loan resulting from a change in the ABR, Adjusted Term SOFR or the Eurocurrency Reserve Requirements shall become effective as of the opening of business on the day on which such change becomes effective. The Administrative Agent shall as soon as practicable notify the Borrower and the relevant Lenders of the effective date and the amount of each such change in interest rate.
- (b) Each determination of an interest rate by the Administrative Agent pursuant to any provision of this Agreement shall be conclusive and binding on the Borrower and the Lenders in the absence of manifest error. The Administrative Agent shall, at the request of either of the Borrower, deliver to the Borrower a statement showing the quotations used by the Administrative Agent in determining any interest rate pursuant to subsections 3.4(a), (b) or (d).

3.6. Changed Circumstances

.

- (a) If prior to the first day of any Interest Period:
 - (i) the Administrative Agent shall have determined (which determination shall be conclusive and binding upon the Borrower) that, by reason of circumstances affecting the relevant market, adequate and reasonable means do not exist for ascertaining Adjusted Term SOFR or the Eurocurrency Rate for any applicable Currency for such Interest Period, or
 - (ii) the Administrative Agent shall have received notice from the Majority Lenders or the Majority Multicurrency Lenders, as the case may be, that Adjusted Term SOFR or the Eurocurrency Rate for any applicable Currency determined or to be determined for such Interest Period will not adequately and fairly reflect the cost to such Lenders (as conclusively certified by such Lenders) of making or maintaining their affected Loans during such Interest Period,

the Administrative Agent shall give telecopy or telephonic notice thereof to the Borrower and the Lenders as soon as practicable thereafter. If such notice is given (w) any SOFR Loans or Multicurrency Loans, as the case may be, in the affected Currency requested to be made on the first day of such Interest Period shall be made as ABR Loans (in the case of Multicurrency Loans, in an amount equal to the Dollar Equivalent of such requested Multicurrency Loans), provided, that, notwithstanding the provisions of subsection 2.2 or 2.14, the Borrower may cancel the request for such SOFR Loan or Multicurrency Loan, as the case may be, by written notice to the Administrative Agent one Business Day prior to the first day of such Interest Period and the Borrower shall not be subject to any liability pursuant to subsection 3.11 with respect to such cancelled request, (x) if the affected Currency is Dollars, any Loans that were to have been converted on the first day of such Interest Period to SOFR Loans shall be continued as ABR Loans, (y) if the affected Currency is Dollars, any outstanding SOFR Loans shall be converted, on the first day of such Interest Period, to ABR Loans and (z) any Multicurrency Loans in the affected Currency to which such Interest Period relates shall be repaid on the first day of such Interest Period. Until such notice has been withdrawn by the Administrative Agent, no further SOFR Loans or Multicurrency Loans in the applicable affected Currency shall be made or continued as such, nor (if the affected Currency is Dollars) shall the Borrower have the right to convert ABR Loans to SOFR Loans.

(b)

(i) Notwithstanding anything to the contrary herein or in any other Loan Document, upon the occurrence of a Benchmark Transition Event or an Early Opt-in Election with respect to any applicable then-current Benchmark, as applicable, the Administrative Agent and the Borrower may amend this Agreement to replace such Benchmark with one or more Benchmark Replacements (it being understood that all amounts denominated in a given currency for which a Benchmark is being replaced shall be subject to the same Benchmark Replacement). Any such amendment with respect to a Benchmark Transition Event will become effective at 5:00 p.m. on the fifth (5th) Business Day after the Administrative Agent has posted such proposed amendment to all affected Lenders and the Borrower so long as the Administrative Agent has not received, by such time, written notice of objection to such amendment from Lenders comprising the Majority Lenders. Any such amendment with respect to an Early Opt-in Election will become effective on the date that Lenders comprising the Majority Lenders have delivered to the Administrative Agent written notice that such Majority Lenders accept such amendment. No replacement of a Benchmark with a Benchmark Replacement pursuant to this Section 3.6(b)(i) will occur prior to the applicable Benchmark Transition Start Date.

- (ii) In connection with the use, administration, adoption or implementation of a Benchmark Replacement, the Administrative Agent will have the right (in consultation with the Borrower) to make Benchmark Replacement Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Benchmark Replacement Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document.
- (iii) The Administrative Agent will promptly notify the Borrower and the Lenders of (i) any occurrence of a Benchmark Transition Event or an Early Opt-in Election, as applicable, and its related Benchmark Replacement Date and Benchmark Transition Start Date, (ii) the implementation of any Benchmark Replacement, (iii) the effectiveness of any Benchmark Replacement Conforming Changes in connection with the use, administration, adoption or implementation of a Benchmark Replacement and (iv) the commencement or conclusion of any Benchmark Unavailability Period. The Administrative Agent will promptly notify the Borrower of the removal or reinstatement of any tenor of a Benchmark pursuant to Section 3.6(b)(iv). Any determination, decision or election that may be made by the Administrative Agent or Lenders pursuant to this Section 3.6(b), including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made in its or their sole discretion and without consent from any other party to this Agreement or any other Loan Document, except, in each case, as expressly required pursuant to this Section 3.6(b).
- (iv) Notwithstanding anything to the contrary herein or in any other Loan Document, at any time (including in connection with the implementation of a Benchmark Replacement), (A) if any then-current Benchmark is a term rate (including the Term SOFR Reference Rate, EURIBOR, BBSY or CDOR) and either (1) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by the Administrative Agent in its reasonable discretion or (2) the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor for such Benchmark is not or will not be representative, then the Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for any Benchmark settings at or after such time to remove such unavailable or non-representative tenor and (B) if a tenor that was removed pursuant to clause (A) above either (1) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (2) is not, or is no longer, subject to an announcement that it is not or will not be representative for a Benchmark (including a Benchmark Replacement), then the Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for all Benchmark settings at or after such time to reinstate such previously removed tenor
- (v) Upon the Borrower's receipt of notice of the commencement of a Benchmark Unavailability Period with respect to a given Benchmark, (A) the Borrower may revoke any pending request for a borrowing of, conversion to or continuation of SOFR Loans or Multicurrency Loans, in each case, to be made, converted or continued during any Benchmark Unavailability Period denominated in the applicable Currency and, failing that, (I) in the case of any request for any affected SOFR Loans, if applicable, the Borrower will be deemed to have converted any such request into a request for a borrowing of or conversion to ABR Loans in the amount specified therein and (II) in the case of any request for any affected Multicurrency Loan,

then such request shall be ineffective and (B)(I) any outstanding affected SOFR Loans, if applicable, will be deemed to have been converted into ABR Loans at the end of the applicable Interest Period and (II) any outstanding affected Multicurrency Loans, at the Borrower's election, shall either (1) be converted into ABR Loans denominated in Dollars (in an amount equal to the Dollar Equivalent of such Available Foreign Currency) at the end of the applicable Interest Period or (2) be prepaid in full at the end of the applicable Interest Period; provided that if no election is made by the Borrower by the earlier of (x) the date that is three (3) Business Days after receipt by the Borrower of such notice and (y) the last day of the current Interest Period for the applicable Multicurrency Loan, the Borrower shall be deemed to have elected clause (1) above. Upon any such prepayment or conversion, the Borrower shall also pay accrued interest on the amount so prepaid or converted, together with any additional amounts required pursuant to Section 3.11. During a Benchmark Unavailability Period with respect to any Benchmark or at any time that a tenor for any then-current Benchmark is not an Available Tenor, the component of ABR based upon the then-current Benchmark that is the subject of such Benchmark Unavailability Period or such Benchmark, as applicable, will not be used in any determination of Base Rate.

3.7. Pro Rata Treatment and Payments

.

- (a) Each payment of principal and interest in respect of any optional prepayment of Revolving Credit Loans or Multicurrency Loans shall be made <u>pro rata</u> according to the amounts then due and owing to the respective Revolving Lender or Multicurrency Lender, as the case may be.
- (b) Except as provided in subsection 2.16, each borrowing by the Borrower of Revolving Credit Loans from the Revolving Lenders hereunder shall be made pro rata according to the Funding Commitment Percentages of the Revolving Lenders in effect on the date of such borrowing. Each payment by the Borrower on account of any facility fee hereunder and any reduction of the Revolving Credit Commitments of the Revolving Lenders shall be allocated by the Administrative Agent among the Revolving Lenders pro rata according to the Revolving Credit Commitment Percentages of the Lenders. Except as provided in subsection 2.16, each payment (including each prepayment) by the Borrower on account of principal on the Revolving Credit Loans shall be made pro rata according to the respective outstanding principal amounts of the Revolving Credit Loans then due and owing to the Revolving Lenders. Each payment by the Borrower of interest in respect of the Revolving Credit Loans shall be applied to the amounts of such obligations owing to the Revolving Lenders pro rata according to the respective amounts then due and owing to the Revolving Lenders. All payments (including prepayments) to be made by the Borrower hereunder in respect of amounts denominated in Dollars, whether on account of principal, interest, fees or otherwise, shall be made without set off or counterclaim and shall be made prior to 12:00 Noon, New York City time, on the due date thereof to the Administrative Agent, for the account of the relevant Lenders, at the Administrative Agent's office specified in subsection 13.2, in Dollars and in immediately available funds. The Administrative Agent shall distribute such payments to the relevant Lenders promptly upon receipt in like funds as received. If any payment hereunder (other than payments on the SOFR Loans or Multicurrency Loans) becomes due and payable on a day other than a Business Day, such payment shall be extended to the next succeeding Business Day, and, with respect to payments of principal, interest thereon shall be payable at the then applicable rate during such extension. If any payment on a SOFR Loan or a Multicurrency Loan becomes due and payable on a day other than a Business Day, the maturity of such payment shall be extended to the next succeeding Business Day (and, with respect to payments of principal, interest thereon shall be payable at the then

- 54 -

applicable rate during such extension) unless the result of such extension would be to extend such payment into another calendar month, in which event such payment shall be made on the immediately preceding Business Day.

- (c) Each borrowing of Multicurrency Loans by the Borrower shall be made, and any reduction of the Multicurrency Commitments shall be allocated by the Administrative Agent, pro rata according to the Multicurrency Commitment Percentages of the Multicurrency Lenders. Each payment (including each prepayment) by the Borrower on account of principal on Multicurrency Loans shall be allocated by the Administrative Agent pro rata according to the respective principal amounts of the Multicurrency Loans then due and owing by the Borrower to each Multicurrency Lender. Each payment by the Borrower of interest in respect of the Multicurrency Loans shall be allocated by the Administrative Agent to the amounts of such obligations owing to the Multicurrency Lenders pro rata according to the respective amounts then due and owing to the Multicurrency Lenders. All payments (including prepayments) to be made by the Borrower on account of Multicurrency Loans hereunder, whether on account of principal, interest, fees or otherwise, shall be made without set-off or counterclaim and shall be made at or before the payment time for the currency of such Multicurrency Loan prior to 3:00 PM (in the local time of the Borrower), on the due date thereof to the Administrative Agent, for the account of the Multicurrency Lenders, at the payment office for the currency of such Multicurrency Loan from time to time specified by the Administrative Agent by notice to the Multicurrency Lenders and the Borrower, in the currency of such Multicurrency Loan and in immediately available funds. The Administrative Agent shall distribute such payments to the Multicurrency Lenders entitled to receive the same promptly upon receipt in like funds as received; provided that the Administrative Agent will provide funds in the then applicable Dollar Equivalent thereof (including an exchange fee and other normal and customary fees for providing this service as determined by the Administrative Agent in its sole discretion) to those Multicurrency Lenders that had funded the Revolving Credit Loans with the Dollar Equivalent as provided in subsection 2.14.
- (d) Notwithstanding the foregoing, payments and Commitment reductions may be made on a non <u>pro rata</u> basis under this Agreement in order to give effect to subsections 2.19 and 2.20.
- (e) Unless the Administrative Agent shall have been notified in writing by any Lender prior to a borrowing that such Lender will not make the amount that would constitute its share of such borrowing available to the Administrative Agent, the Administrative Agent may assume that such Lender is making such amount available to the Administrative Agent, and the Administrative Agent may, in reliance upon such assumption, make available to the Borrower a corresponding amount. If such amount is not made available to the Administrative Agent by the required time on the Borrowing Date therefor, such Lender shall pay to the Administrative Agent, on demand, such amount with interest thereon at a rate equal to (i) the daily average NYFRB Rate (in the case of a borrowing of Revolving Credit Loans or CAF Advances denominated in Dollars) and (ii) the Administrative Agent's reasonable estimate of its average daily cost of funds (in the case of a borrowing of Multicurrency Loans or CAF Advances denominated in a currency other than Dollars), in each case for the period until such Lender makes such amount immediately available to the Administrative Agent. A certificate of the Administrative Agent submitted to any Lender with respect to any amounts owing under this subsection 3.7 shall be conclusive in the absence of manifest error. If such Lender's share of such borrowing is not made available to the Administrative Agent by such Lender within three Business Days of such Borrowing Date, the Administrative Agent shall also be entitled to recover such amount with interest thereon equal to (i) the rate per annum applicable to ABR Loans hereunder (in the case of a borrowing of Revolving Credit Loans or CAF Advances or CAF Advances denominated in Dollars) and (ii) the Administrative Agent's reasonable estimate of its average daily cost of funds <u>plus</u> the Applicable Margin applicable to

- 55 -

Multicurrency Loans (in the case of a borrowing of Multicurrency Loans or CAF Advances denominated in a currency other than Dollars), on demand, from the Borrower.

3.8. Illegality

Notwithstanding any other provision herein, if after the date hereof the adoption of or any change in any Requirement of Law or in the interpretation or application thereof shall make it unlawful for any Lender to make or maintain SOFR Loans or Multicurrency Loans as contemplated by this Agreement, (a) the commitment of such Lender hereunder to make SOFR Loans or Multicurrency Loans, continue SOFR Loans or Multicurrency Loans as such and convert ABR Loans to SOFR Loans shall forthwith be cancelled, (b) such Lender's Loans then outstanding as SOFR Loans, if any, shall be converted automatically to ABR Loans on the respective last days of the then current Interest Periods with respect to such Loans or within such earlier period as required by law and (c) such Lender's Multicurrency Loans shall be prepaid on the last day of the then current Interest Period with respect thereto. If any such conversion of a SOFR Loan occurs on a day which is not the last day of the then current Interest Period with respect thereto, the Borrower shall pay to such Lender such amounts, if any, as may be required pursuant to subsection 3.11. For all purposes of this Section 3.8, the term "Lenders" shall be deemed to include Issuing Lenders.

3.9. Requirements of Law

•

- (a) If, due to either (i) the introduction of or any change in any law or regulation or in the interpretation or administration of any law or regulation by any Governmental Authority charged with the interpretation or administration thereof or (ii) the compliance with any guideline or request from any central bank or other Governmental Authority that would be complied with generally by similarly situated banks or lenders acting reasonably (whether or not having the force of law and for the avoidance of doubt, including any changes resulting from requests, rules, guidelines or directives concerning capital adequacy issued after the date hereof in connection with the Dodd-Frank Wall Street Reform and Consumer Protection Act or promulgated after the date hereof by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III), there shall be any increase in the cost to any Lender of agreeing to make or making, funding or maintaining SOFR Loans, Multicurrency Loans or Term Rate CAF Advances or issuing or participating in Letters of Credit (except any reserve or other requirement contemplated by subsection 3.9(b) or (c) other than as set forth below) by an amount deemed by such Lender to be material (except, for the avoidance of doubt, for Non-Excluded Taxes indemnified under subsection 3.10 and Excluded Taxes), then the Borrower shall from time to time, upon demand by such Lender (with a copy of such demand to the Administrative Agent), pay to the Administrative Agent for the account of such Lender additional amounts sufficient to compensate such Lender for such increased cost. A certificate as to the amount of such increased cost, submitted to the Borrower and the Administrative Agent by such Lender, shall be conclusive and binding for all purposes, absent manifest error.
- (b) If, due to either (i) the introduction of or any change in or interpretation of any law or regulation or (ii) compliance with any guideline or request from any central bank or other governmental or regulatory authority which becomes effective after the date hereof (for the avoidance of doubt, including any changes resulting from requests, rules, guidelines or directives concerning capital adequacy issued after the date hereof in connection with the Dodd-Frank Wall Street Reform and Consumer

- 56 -

Protection Act or promulgated after the date hereof by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III), there shall be any increase in the amount of capital required or expected to be maintained by any Lender or any corporation controlling such Lender and the amount of such capital is increased by or based upon the existence of such Lender's Loans or commitment to extend credit and other commitments of this type by an amount deemed by such Lender to be material, then, upon demand by such Lender (with a copy of such demand to the Administrative Agent), the Borrower shall pay to the Administrative Agent for the account of such Lender, from time to time as specified by such Lender, additional amounts sufficient to compensate such Lender or such corporation in the light of such circumstances, to the extent that such Lender reasonably determines such increase in capital to be allocable to the existence of such Lender's Loans or commitment to extend credit hereunder. A certificate as to such amounts submitted to the Borrower and the Administrative Agent by such Lender shall be conclusive and binding for all purposes as to the calculations therein, absent manifest error. Such certificate shall be in reasonable detail and shall certify that the claim for additional amounts referred to therein is generally consistent with such Lender's treatment of similarly situated customers of such Lender whose transactions with such Lender are similarly affected by the change in circumstances giving rise to such payment, but such Lender shall not be required to disclose any confidential or proprietary information therein. For all purposes of this Section 3.9(b), the term "Lenders" shall be deemed to include Issuing Lenders.

(c) For purposes of the foregoing paragraphs (a) and (b), the amendments to 12 C.F.R. Part 327 set forth in the final rule attached to the Federal Deposit Insurance Corporation Financial Institution Letter FIL-8-2011, dated February 9, 2011, shall be deemed to have been introduced and adopted after the date of this Agreement.

3.10. Taxes

.

(a) All payments made by the Borrower under any Loan Document shall be made free and clear of, and without deduction or withholding for or on account of, any Taxes, excluding (i) all net income Taxes, franchise Taxes or other Taxes, including branch profits Taxes, in each case imposed on the Administrative Agent or any Lender as a result of a present or former connection between the Administrative Agent or such Lender and the jurisdiction of the Governmental Authority imposing such Tax or any political subdivision or taxing authority thereof or therein (other than any such connection arising solely from the Administrative Agent or such Lender having executed, delivered or performed its obligations or received a payment under, or enforced, any Loan Document), (ii) in the case of a Lender other than an assignee pursuant to a request by a Borrower under subsection 3.12, any U.S. federal withholding Tax that is imposed under a law in effect at the time such Lender becomes a party hereto (or designates a new lending office), except to the extent that such Lender (or its assignor, if any) was entitled, immediately prior to the time of designation of a new lending office (or assignment), to receive additional amounts from the Borrower with respect to such withholding Tax pursuant to this Section, (iii) any withholding Tax imposed as a result of a Lender failing to comply with subsection 3.10(c) and (iv) any U.S. federal Taxes that are imposed by reason of FATCA (Taxes in clauses (i) to (iv) being "Excluded Taxes"). If any Taxes are required to be withheld from any amounts payable to the Administrative Agent or any Lender hereunder or under any Loan Document as determined in good faith by the applicable withholding agent, (x) such amounts shall be paid to the relevant Governmental Authority in accordance with applicable law and (y) if such Taxes are not Excluded Taxes ("Non-Excluded Taxes") the amounts so payable by the Borrower to the Administrative Agent or such Lender

- 57 -

shall be increased to the extent necessary to yield to the Administrative Agent or such Lender (after payment of all Non-Excluded Taxes) interest or any such other amounts payable hereunder at the rates or in the amounts specified in such Loan Document as if such withholding or deduction had not been made. Whenever any Taxes are payable by the Borrower pursuant to this Section, as promptly as possible thereafter the Borrower shall send to the Administrative Agent for its own account or for the account of such Lender, as the case may be, a certified copy of an original official receipt received by the Borrower showing payment thereof or other evidence of such payment satisfactory to the Administrative Agent (in its reasonable discretion). If the Borrower fails to pay any Non-Excluded Taxes when due to the appropriate taxing authority or fails to remit to the Administrative Agent the required receipts or other required documentary evidence, the Borrower shall indemnify the Administrative Agent and the Lenders for any incremental Taxes that may become payable by the Administrative Agent or any Lender as a result of any such failure. The agreements in this subsection 3.10 shall survive the termination of this Agreement and each other Loan Document and the payment of the Loans and all other amounts payable hereunder and thereunder.

- (b) The Borrower shall indemnify and hold harmless the Administrative Agent and each Lender within 20 days after demand therefor, for the full amount of any Non-Excluded Taxes (including Non-Excluded Taxes imposed or asserted on or attributable to amounts payable under this subsection 3.10) payable by the Administrative Agent or such Lender and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Non-Excluded Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority; provided that the Borrower and the Foreign Subsidiary Borrower, as applicable, shall not be obligated to indemnify the Administrative Agent or any such Lender pursuant to this subsection 3.10(b) in respect of penalties, interest or reasonable expenses if such penalties, interest or reasonable expenses are attributable to the gross negligence or willful misconduct of the person seeking indemnification.
- (c) Any Lender that is entitled to an exemption from or reduction of withholding Tax with respect to payments made under any Loan Document shall deliver to the Borrower and the Administrative Agent, at the time or times reasonably requested by the Borrower or the Administrative Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if reasonably requested by the Borrower or the Administrative Agent, shall deliver such other documentation prescribed by applicable law or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in subsections 3.10(c)(A), (B) and (D) below) shall not be required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.

Without limiting the generality of the foregoing, in the event the Borrower is a U.S. Person,

(i) any Lender that is a U.S. Person shall deliver to the Borrower and the Administrative Agent on or prior to the date on which such Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), executed originals of IRS Form W-9 certifying that such Lender is exempt from U.S. Federal backup withholding tax;

- 58 -

- (ii) any Non-U.S. Lender shall, to the extent it is legally entitled to do so, deliver to the Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Non-U.S. Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), whichever of the following is applicable:
 - (A) in the case of a Non-U.S. Lender claiming the benefits of an income tax treaty to which the United States is a party (x) with respect to payments of interest under any Loan Document, executed originals of IRS Form W-8BEN or W-8BEN-E, as applicable, establishing an exemption from, or reduction of, U.S. Federal withholding Tax pursuant to the "interest" article of such tax treaty and (y) with respect to any other applicable payments under any Loan Document, IRS Form W-8BEN or W-8BEN-E, as applicable, establishing an exemption from, or reduction of, U.S. Federal withholding Tax pursuant to the "business profits" or "other income" article of such tax treaty;
 - (B) executed originals of IRS Form W-8ECI;
 - (C) in the case of a Non-U.S. Lender claiming the benefits of the exemption for portfolio interest under Section 871(h) or Section 881(c) of the Code, (x) a certificate substantially in the form of Exhibit J-1 to the effect that such Non-U.S. Lender is not a "bank" within the meaning of Section 881(c)(3)(A) of the Code, a "10 percent shareholder" of the Borrower within the meaning of Section 881(c)(3)(B) of the Code, or a "controlled foreign corporation" related to the Borrower as described in Section 881(c)(3)(C) of the Code (a "U.S. Tax Compliance Certificate") and (y) executed originals of IRS Form W-8BEN or W-8BEN-E, as applicable; or
 - (D) to the extent a Non-U.S. Lender is not the beneficial owner, executed originals of IRS Form W-8IMY, accompanied by IRS Form W-8ECI, IRS Form W-8BEN or W-8BEN-E, as applicable, a U.S. Tax Compliance Certificate substantially in the form of Exhibit J-2 or Exhibit J-3, IRS Form W-9, and/or other certification documents from each beneficial owner, as applicable; provided that if the Non-U.S. Lender is a partnership and one or more direct or indirect partners of such Non-U.S. Lender are claiming the portfolio interest exemption, such Non-U.S. Lender may provide a U.S. Tax Compliance Certificate substantially in the form of Exhibit J-4 on behalf of each such direct and indirect partner;
- (iii) any Non-U.S. Lender shall, to the extent it is legally entitled to do so, deliver to the Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Non-U.S. Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the Administrative Agent), executed originals of any other form prescribed by applicable law as a basis for claiming exemption from or a reduction in U.S. Federal withholding Tax, duly completed, together with such supplementary documentation as may be prescribed by applicable

law to permit the Borrower or the Administrative Agent to determine the withholding or deduction required to be made; and

(iv) If a payment made to a Lender under any Loan Document would be subject to U.S. federal withholding Tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable), such Lender shall deliver to the Borrower and Administrative Agent, at the time or times prescribed by law and at such time or times reasonably requested by the Borrower or Administrative Agent, such documentation prescribed by applicable law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and such additional documentation reasonably requested by the Borrower or Administrative Agent as may be necessary for the Borrower and Administrative Agent to comply with their obligations under FATCA, to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this clause (D), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

Each Lender agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify the Borrower and the Administrative Agent in writing of its legal inability to do so. Notwithstanding any other provision of this Section, a Lender shall not be required to deliver any form pursuant to this Section that such Lender is not legally able to deliver.

- (d) If the Borrower pays any additional amounts or makes an indemnity payment under this <u>subsection 3.10</u> to any Lender or the Administrative Agent, and such Lender or the Administrative Agent determines in its sole discretion exercised in good faith that it has actually received in connection therewith any refund of the underlying Non-Excluded Taxes, such Lender or the Administrative Agent shall pay to the Borrower an amount equal to such refund which was obtained by such Lender or Administrative Agent (but only to the extent of indemnity payments made, or Additional Amounts paid by the Borrower under this <u>subsection 3.10</u> with respect to the Non-Excluded Taxes giving rise to such refund) net of all reasonable out-of-pocket expenses of the Lender or the Administrative Agent with respect to such refund, and without interest (other than any interest paid by the relevant taxation authority); <u>provided, however</u>, that the Borrower, upon the request of the Lender or the Administrative Agent, agrees to repay the amount paid over to the Borrower to any Lender or the Administrative Agent in the event any Lender or the Administrative Agent is required to repay such refund, plus interest and penalties (excluding interest and penalties attributable to the negligence or willful misconduct of such Lender or the Administrative Agent). This paragraph shall not be construed to require any Lender or the Administrative Agent to disclose any confidential information to the Borrower or any other Person (including its Tax returns).
- (e) Each Lender shall indemnify the Administrative Agent for the full amount of any Taxes imposed by any Governmental Authority that are attributable to such Lender and that are payable or paid by the Administrative Agent, together with all interest, penalties, reasonable costs and expenses arising therefrom or with respect thereto (but only to the extent not already paid by the Borrower), as determined by the Administrative Agent in good faith. A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent shall be conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under any Loan Document or otherwise payable by the Administrative Agent to the

Lender from any other source against any amount due to the Administrative Agent under this <u>subsection 3.10(e)</u>.

For all purposes of this Section 3.10, the term "Lenders" shall be deemed to include Issuing Lenders.

3.11. Indemnity

. The Borrower agrees to indemnify each Lender and to hold each such Lender harmless from any loss or expense which such Lender may sustain or incur as a consequence of (a) default by the Borrower in making a borrowing of, conversion into or continuation of SOFR Loans, Local Currency Loans, Multicurrency Loans or CAF Advances after the Borrower has given a notice requesting the same in accordance with the provisions of this Agreement, (b) default by the Borrower or any Foreign Subsidiary Borrower in making any prepayment after the Borrower or such Foreign Subsidiary Borrower has given a notice thereof in accordance with the provisions of this Agreement or any other Loan Document or (c) the making of a prepayment of SOFR Loans, Local Currency Loans, Multicurrency Loans or CAF Advances or the conversion of SOFR Loans to ABR Loans on a day which is not the last day of an Interest Period with respect thereto. Such indemnification may include an amount equal to the excess, if any, of (i) the amount of interest which would have accrued on the amount so prepaid, or not so borrowed, converted or continued, for the period from the date of such prepayment or of such failure to borrow, convert or continue to the last day of such Interest Period (or, in the case of a failure to borrow, convert or continue, the Interest Period that would have commenced on the date of such failure) or, in the case of CAF Advances, the applicable CAF Advance Maturity Date (or proposed CAF Advance Maturity Date), in each case at the applicable rate of interest for such Loans provided for herein (excluding, however, the Applicable Margin or any positive margin applicable to CAF Advances included therein, if any) over (ii) the amount of interest (as reasonably determined by such Lender) which would have accrued to such Lender on such amount by placing such amount on deposit for a comparable period with leading banks in the interbank eurodollar market. This covenant shall survive the termination of this Agreement and each other Loan Document and the payment of the Loans and all other amounts payable hereunder and thereunder. For all purposes of this Section 3.11, the term "Lenders" shall be deemed to include Issuing Lenders.

3.12. Change of Lending Office; Removal of Lender

Each Lender agrees that if it makes any demand for payment under <u>subsection 3.9</u> or <u>3.10(a)</u>, or if any adoption or change of the type described in <u>subsection 3.8</u> shall occur with respect to it, (i) it will use reasonable efforts (consistent with its internal policy and legal and regulatory restrictions and so long as such efforts would not be disadvantageous to it, as determined in its sole discretion) to designate a different lending office if the making of such a designation would reduce or obviate the need for the Borrower to make payments under <u>subsection 3.9</u> or <u>3.10(a)</u>, or would eliminate or reduce the effect of any adoption or change described in <u>subsection 3.8</u> or (ii) it will, upon at least five Business Days' notice from the Borrower to such Lender and the Administrative Agent, assign, pursuant to and in accordance with the provisions of <u>subsection 13.6(c)</u> and <u>13.19</u>, to one or more Assignees designated by the Borrower all, but not less than all, of such Lender's rights and obligations hereunder (other than rights in respect of such Lender's outstanding CAF Advance), without recourse to or warranty by, or expense to, such Lender, for a purchase price equal to the outstanding principal amount of each Loan (other than a CAF Advance) then owing to such Lender <u>plus</u> any accrued but unpaid interest thereon and any accrued but unpaid facility fees and utilization fees owing thereto and, in addition, all additional costs and reimbursements, expense reimbursements and indemnities, if any, owing in respect of such Lender's Commitment hereunder at such time (including any amount that would be payable under subsection 3.11

if such assignment were, instead, a prepayment in full of all amounts owing to such Lender) shall be paid to such Lender.

3.13. Evidence of Debt

•

- (a) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing indebtedness of the Borrower to such Lender resulting from each Loan of such Lender from time to time, including the amounts of principal and interest payable and paid to such Lender from time to time under this Agreement.
- (b) The Administrative Agent shall maintain the Register pursuant to <u>subsection 13.6(d)</u>, and a subaccount therein for each Lender, in which shall be recorded (i) in the case of Revolving Credit Loans, the amount of each Revolving Credit Loan made hereunder, the Type thereof and each Interest Period applicable thereto, (ii) in the case of Multicurrency Loans, the amount and currency of each Multicurrency Loans and each Interest Period applicable thereto, (iii) in the case of CAF Advances, the amount and currency of each CAF Advance made hereunder, the CAF Advance Maturity Date thereof, the interest rate applicable thereto and each CAF Advance Interest Payment Date applicable thereto, (iv) the amount of any principal or interest due and payable or to become due and payable from the Borrower to each Lender hereunder and (v) both the amount of any sum received by the Administrative Agent hereunder from the Borrower and each Lender's share thereof.
- (c) The entries made in the Register and the accounts of each Lender maintained pursuant to <u>subsection 3.13(a)</u> shall, to the extent permitted by applicable law, be prima facie evidence of the existence and amounts of the obligations of the Borrower therein recorded; <u>provided, however</u>, that the failure of any Lender or the Administrative Agent to maintain the Register or any such account, or any error therein, shall not in any manner affect the obligation of the Borrower to repay (with applicable interest) the Loans made to the Borrower by such Lender in accordance with the terms of this Agreement.
- (d) The Borrower agrees that, upon the request to the Administrative Agent by any Revolving Lender, the Borrower will execute and deliver to such Revolving Lender a promissory note of the Borrower evidencing the Revolving Credit Loans of such Revolving Lender, substantially in the form of Exhibit A with appropriate insertions as to date and principal amount (a "Revolving Credit Note").
- (e) The Borrower agrees that, upon the request to the Administrative Agent by any Lender, the Borrower will execute and deliver to such Lender a promissory note of the Borrower evidencing the CAF Advances of such Lender, substantially in the form of Exhibit B with appropriate insertions (a "CAF Advance Note").
- (f) The Borrower agrees that, upon the request to the Administrative Agent by any Multicurrency Lender, the Borrower will execute and deliver to such Multicurrency Lender a promissory note of the Borrower evidencing the Multicurrency Loans of such Multicurrency Lender, substantially in the form of Exhibit L with appropriate insertions as to date and principal amount (a "Multicurrency Note").

SECTION 4

LOCAL CURRENCY FACILITIES

4.1. Terms of Local Currency Facilities.

- (a) Subject to the provisions of this Section 4, the Borrower may in its discretion from time to time designate any Subsidiary of the Borrower organized under the laws of any jurisdiction outside the United States as a "Foreign Subsidiary Borrower" and any Qualified Credit Facility to which such Foreign Subsidiary Borrower and any one or more Lenders (or its Affiliates, agencies or branches) is a party as a "Local Currency Facility", with the consent of the Administrative Agent and each such Lender in its sole discretion, by delivering a Local Currency Facility Addendum to the Administrative Agent and the Lenders (through the Administrative Agent) executed by the Borrower, each such Foreign Subsidiary Borrower and each such Lender, provided, that on the effective date of such designation no Default or Event of Default shall have occurred and be continuing. Concurrently with the delivery of a Local Currency Facility Addendum, the Borrower or the relevant Foreign Subsidiary Borrower shall furnish to the Administrative Agent copies of all documentation executed and delivered by such Foreign Subsidiary Borrower in connection therewith, together with, if applicable, an English translation thereof, and any other information as any Lender may reasonably request for purposes of compliance with applicable "know your customer" and anti-money laundering rules and regulations, including the Patriot Act and the Beneficial Ownership Regulation, as reasonably requested by the Lenders. Except as otherwise provided in this Section 4 or in the definition of "Qualified Credit Facility" in subsection 1.1, the terms and conditions of each Local Currency Facility shall be determined by mutual agreement of the relevant Foreign Subsidiary Borrower(s), Local Currency Lender(s) and the Administrative Agent. The documentation governing each Local Currency Facility shall (i) contain an express acknowledgement that such Local Currency Facility shall be subject to the provisions of this Section 4, (ii) if more than one Lender is a party thereto, designate a Local Currency Facility Agent for such Local Currency Facility and (iii) include an opinion of counsel reasonably satisfactory to the Administrative Agent from the jurisdiction in which such Local Currency Facility is established that the documentation governing such Local Currency Facility is enforceable in accordance with its terms. Each of the Borrower and, by agreeing to any Local Currency Facility designation as contemplated hereby, each relevant Local Currency Lender (if any) party thereto which is an Affiliate, branch or agency of a Lender, acknowledges and agrees that each reference in this Agreement to any Lender shall, to the extent applicable, be deemed to be a reference to such Local Currency Lender. In the event of any inconsistency between the terms of this Agreement and the terms of any Local Currency Facility, the terms of this Agreement shall prevail.
- (b) The documentation governing each Local Currency Facility shall set forth (i) the maximum amount (expressed in Dollars) available to be borrowed from all Local Currency Lenders under such Local Currency Facility (as the same may be reduced from time to time, a "Local Currency Facility Maximum Borrowing Amount") and (ii) with respect to each Local Currency Lender party to such Local Currency Facility, the maximum amount (expressed in Dollars) available to be borrowed from such Local Currency Lender thereunder (as the same may be reduced from time to time, a "Local Currency Lender Maximum Borrowing Amount").
- (c) Except as otherwise required by applicable law, in no event shall the Local Currency Lenders party to a Local Currency Facility have the right to accelerate the Local Currency Loans outstanding thereunder, or to terminate their commitments (if any) to make such Local Currency Loans prior to the earlier of the stated termination date in respect thereof or the Termination Date, except, in each case, in connection with an acceleration of the Loans or a termination of the Commitments pursuant

to Section 10 hereof, provided, that nothing in this paragraph (c) shall be deemed to require any Local Currency Lender to make a Local Currency Loan if the applicable conditions precedent to the making of such Local Currency Loan set forth in the documentation governing the relevant Local Currency Facility have not been satisfied. No Local Currency Loan may be made under a Local Currency Facility if (i) after giving effect thereto, the conditions precedent in subsection 7.2 hereof would not be satisfied or (ii) after giving effect to the making of such Local Currency Loan and the simultaneous application of the proceeds thereof, (A) the Aggregate Total Outstandings of all Lenders at any time exceeds the Aggregate Revolving Credit Commitments or (B) the Dollar Equivalent of the aggregate outstanding principal amount of Multicurrency Loans, Local Currency Loans and L/C Obligations attributable to Letters of Credit denominated in any currency other than Dollars would exceed the Multicurrency Sublimit.

- (d) The relevant Foreign Subsidiary Borrower shall furnish to the Administrative Agent copies of any amendment, supplement or other modification (including any change in commitment amounts or in the Local Currency Lenders participating in any Local Currency Facility) to the terms of any Local Currency Facility promptly after the effectiveness thereof (together with, if applicable, an English translation thereof). If any such amendment, supplement or other modification to a Local Currency Facility shall (i) add a Local Currency Lender as a Local Currency Lender thereunder or (ii) change the Local Currency Facility Maximum Borrowing Amount or any Local Currency Lender Maximum Borrowing Amount with respect thereto, the Borrower shall promptly furnish an appropriately revised Local Currency Facility Addendum, executed by the Borrower, the relevant Foreign Subsidiary Borrower and the affected Local Currency Lenders (or any agent acting on their behalf), to the Administrative Agent and the Lenders (through the Administrative Agent).
- (e) The Borrower may terminate its designation of a facility as a Local Currency Facility, with the consent of each Local Currency Lender party thereto in its sole discretion, by written notice to the Administrative Agent, which notice shall be executed by the Borrower, the relevant Foreign Subsidiary Borrower and each Local Currency Lender party to such Local Currency Facility (or any agent acting on their behalf). Once notice of such termination is received by the Administrative Agent, such Local Currency Facility and the loans and other obligations outstanding thereunder shall immediately cease to be subject to the terms of this Agreement.

4.2. Reporting of Local Currency Outstandings

•

- (a) On the date of the making of any Local Currency Loan having a maturity of 30 or more days to a Foreign Subsidiary Borrower and on the last Business Day of each month on which a Foreign Subsidiary Borrower has any outstanding Local Currency Loans, the Local Currency Facility Agent for such Foreign Subsidiary Borrower, shall deliver to the Administrative Agent a Notice of Local Currency Outstandings and the Administrative Agent shall deliver a copy of such Notice of Local Currency Outstandings to the Lenders. The Administrative Agent will, at the request of any Local Currency Facility Agent, advise such Local Currency Facility Agent of the Exchange Rate used by the Administrative Agent in calculating the Dollar Equivalent of Local Currency Loans under the related Local Currency Facility on any date.
- (b) For purposes of any calculation under this Agreement in which the amount of the Aggregate Local Currency Outstandings of any Lender is a component, the Administrative Agent shall make such calculation on the basis of the Notices of Local Currency Outstanding received by it at least two Business Days prior to the date of such calculation.

- 64 -

SECTION 5

LETTERS OF CREDIT

5.1. L/C Commitment

.

- (a) Subject to the terms and conditions hereof, each Issuing Lender, in reliance on the agreements set forth in this Agreement, agrees to issue letters of credit (provided that Barclays shall only be required to issue standby letters of credit) ("Letters of Credit") for the account of the Borrower on any Business Day during the Commitment Period in such form as may be approved from time to time by the Issuing Lender; provided that the Issuing Lender shall not issue any Letter of Credit if, after giving effect to such issuance, (i) the L/C Obligations would exceed the L/C Commitment, (ii) the aggregate amount of the Available Revolving Credit Commitments would be less than zero, (iii) the Dollar Equivalent of the aggregate outstanding principal amount of Multicurrency Loans, Local Currency Loans and L/C Obligations attributable to Letters of Credit denominated in any currency other than Dollars would exceed the Multicurrency Sublimit, (iv) the L/C Obligations in respect of all Letters of Credit issued by such Issuing Lender would exceed its Issuing Lender Commitment or (v) any Lender is at that time a Defaulting Lender, unless the Issuing Lender has entered into arrangements, including the delivery of Cash Collateral, reasonably satisfactory to the Issuing Lender with the Borrower or such Lender to eliminate the Issuing Lender's actual or potential Fronting Exposure (after giving effect to subsection 2.19(a)(iv)) with respect to the Defaulting Lender arising from either the Letter of Credit then proposed to be issued or that Letter of Credit and all other L/C Obligations as to which the Issuing Lender has actual or potential Fronting Exposure. Each Letter of Credit shall (i) be denominated in Dollars, any Available Foreign Currency, or any other currency acceptable to the Issuing Lender, (ii) have a face amount of at least \$1,000,000 (unless otherwise agreed by the Issuing Lender) and (iii) expire (or be subject to termination by notice from the Issuing Lender to the beneficiary thereof) no later than the date t
- (b) The Issuing Lender shall not at any time be obligated to issue any Letter of Credit if such issuance would conflict with, or cause the Issuing Lender or any L/C Participant to exceed any limits imposed by, any applicable Requirement of Law.
- (c) Letters of Credit issued under the Existing Credit Facilities and outstanding on the Closing Date shall be deemed to be issued under this Agreement on the Closing Date.

5.2. Procedure for Issuance of Letter of Credit

. The Borrower may from time to time request that the Issuing Lender issue a Letter of Credit by delivering to the Issuing Lender at its address for notices specified herein an Application therefor, completed to the satisfaction of the Issuing Lender, and such other certificates, documents and other papers and information as the Issuing Lender may request. Upon receipt of any Application, the Issuing Lender will process such Application and the certificates, documents and other papers and information delivered to it in connection therewith in accordance with its customary policies and procedures and shall promptly issue the Letter of Credit requested thereby (but in no event shall the Issuing Lender be required to issue any Letter of Credit earlier than three Business Days after its receipt of the Application therefor and all such other certificates, documents and other papers and information relating thereto) by issuing the original of such Letter of Credit to the beneficiary thereof or as otherwise may be agreed to by the Issuing

Lender and the Borrower. The Issuing Lender shall furnish a copy of such Letter of Credit to the Borrower promptly following the issuance thereof. The Issuing Lender shall promptly furnish to the Administrative Agent, which shall in turn promptly furnish to the Lenders, notice of the issuance of each Letter of Credit (including the amount thereof). If the Borrower so requests, the applicable Issuing Lender may, in its sole and absolute discretion, agree to issue a Letter of Credit that has automatic extension provisions (each, an "Auto-Extension Letter of Credit"); provided, that any such Auto-Extension Letter of Credit must permit such Issuing Lender to prevent any such extension at least once in each 12month period (or such longer period in the case of a Letter of Credit with an expiration in excess of 12 months) (commencing with the date of issuance of such Letter of Credit) by giving prior notice to the beneficiary thereof not later than a day (the "Non-Extension Notice Date") in each such 12-month period (or such longer period in the case of a Letter of Credit with an expiration in excess of 12 months) to be agreed upon at the time such Letter of Credit is issued. Unless otherwise directed by the applicable Issuing Lender, the Borrower shall not be required to make a specific request to such Issuing Lender for any such extension. Once an Auto-Extension Letter of Credit has been issued, the Lenders shall be deemed to have authorized (but may not require) the applicable Issuing Lender to permit the extension of such Letter of Credit at any time to an expiry date not later than the Termination Date; provided, however, that such Issuing Lender shall not permit any such extension if (i) such Issuing Lender has determined that it would not be permitted, at such time to issue such Letter of Credit in its revised form (as extended) under the terms hereof, or (ii) it has received notice (which may be by telephone or in writing) on or before the day that is five Business Days before the Non-Extension Notice Date (A) from the Administrative Agent that the Majority Lenders have elected not to permit such extension or (B) from the Administrative Agent, any Lender or the Borrower that one or more of the applicable conditions specified in Section 7 are not then satisfied, and in each such case directing such Issuing Lender not to permit such extension.

5.3. Fees and Other Charges

.

(a) The Borrower will pay a fee on the actual daily undrawn amount of all outstanding Letters of Credit at a per annum rate equal to the Applicable Margin then in effect with respect to SOFR Loans, shared ratably among the Revolving Lenders and payable quarterly in arrears on each L/C Fee Payment Date after the issuance date; such fee with respect to each Letter of Credit denominated in any currency other than Dollars shall be payable in Dollars, and for purposes of calculating the amount of such fee applicable to each Letter of Credit denominated in any currency other than Dollars, the actual daily undrawn, outstanding amount of such Letter of Credit shall be the Dollar Equivalent of such amount calculated at the Exchange Rate as of the relevant L/C Fee Payment Date; provided, however, that any such fees otherwise payable for the account of a Defaulting Lender with respect to any Letter of Credit as to which such Defaulting Lender has not provided Cash Collateral satisfactory to the Issuing Lender pursuant to subsection 5.1 shall be payable, to the maximum extent permitted by applicable law, to the other Lenders in accordance with the upward adjustments in their respective Revolving Credit Commitment Percentages allocable to such Letter of Credit pursuant to subsection 2.19(a)(iv), with the balance of such fee, if any, payable to the Issuing Lender for its own account. In addition, the Borrower shall pay to the Issuing Lender for its own account a fronting fee in the amount separately agreed to by the Issuing Lender and the Borrower on the actual daily undrawn, outstanding amount of each Letter of Credit denominated in any currency other than Dollars shall be payable in Dollars, and for purposes of calculating the amount of such fee applicable to each Letter of Credit denominated in any currency other than Dollars, the actual daily undrawn, outstanding amount of such

- 66 -

Letter of Credit shall be the Dollar Equivalent of such amount calculated at the Exchange Rate as of the relevant L/C Fee Payment Date.

(b) In addition to the foregoing fees, the Borrower shall pay or reimburse the Issuing Lender for such normal and customary costs and expenses as are incurred or charged by the Issuing Lender in issuing, negotiating, examining documents under, effecting payment under, amending or otherwise administering any Letter of Credit.

5.4. L/C Participations

.

- (a) The Issuing Lender irrevocably agrees to grant and hereby grants to each L/C Participant, and, to induce the Issuing Lender to issue Letters of Credit, each L/C Participant irrevocably agrees to accept and purchase and hereby accepts and purchases from the Issuing Lender, on the terms and conditions set forth below, for such L/C Participant's own account and risk an undivided interest equal to such L/C Participant's Revolving Credit Commitment Percentage in the Issuing Lender's obligations and rights under and in respect of each Letter of Credit and the amount of each drawing paid by the Issuing Lender thereunder. Each L/C Participant unconditionally and irrevocably agrees with the Issuing Lender that, if a drawing is paid under any Letter of Credit for which the Issuing Lender is not reimbursed in full by the Borrower in accordance with the terms of this Agreement, such L/C Participant shall pay to the Issuing Lender through the Administrative Agent (or the Administrative Agent shall apply Cash Collateral provided for this purpose) upon demand at the Administrative Agent's address for notices specified herein an amount equal to such L/C Participant's Revolving Credit Commitment Percentage of the amount of such drawing, or any part thereof, that is not so reimbursed.
- (b) If any amount required to be paid by any L/C Participant to the Issuing Lender pursuant to <u>subsection 5.4(a)</u> in respect of any unreimbursed portion of any payment made by the Issuing Lender under any Letter of Credit is paid to the Issuing Lender within three Business Days after the date such payment is due, such L/C Participant shall pay to the Issuing Lender on demand an amount equal to the product of (i) such amount, times (ii) the daily average NYFRB Rate during the period from and including the date such payment is required to the date on which such payment is immediately available to the Issuing Lender, times (iii) a fraction the numerator of which is the number of days that elapse during such period and the denominator of which is 360. If any such amount required to be paid by any L/C Participant pursuant to <u>subsection 5.4(a)</u> is not made available to the Issuing Lender by such L/C Participant within three Business Days after the date such payment is due, the Issuing Lender shall be entitled to recover from such L/C Participant, on demand, such amount with interest thereon calculated from such due date at the rate per annum applicable to ABR Loans. A certificate of the Issuing Lender submitted through the Administrative Agent to any L/C Participant with respect to any amounts owing under this Section shall be conclusive in the absence of manifest error.
- (c) Whenever, at any time after the Issuing Lender has made payment under any Letter of Credit and has received from any L/C Participant its <u>pro rata</u> share of such payment in accordance with <u>subsection 5.4(a)</u>, the Administrative Agent receives any payment related to such Letter of Credit (whether directly from the Borrower or otherwise, including proceeds of collateral applied thereto by the Issuing Lender), or any payment of interest on account thereof, the Administrative Agent will distribute to such L/C Participant its <u>pro rata</u> share thereof; <u>provided</u>, <u>however</u>, that in the event that any such payment received by the Issuing Lender shall be required to be returned by the Issuing Lender, such L/C

Participant shall return to the Issuing Lender through the Administrative Agent the portion thereof previously distributed by the Issuing Lender to it.

5.5. Reimbursement Obligation of the Borrower

. The Borrower agrees to reimburse the Issuing Lender through the Administrative Agent on the Business Day next succeeding the Business Day on which the Issuing Lender notifies the Borrower and the Administrative Agent of the date and amount of a drawing under any Letter of Credit paid by the Issuing Lender for the amount of (a) such drawing so paid and (b) any taxes, fees, charges or other costs or expenses incurred by the Issuing Lender in connection with such payment. Each such payment shall be made to the Issuing Lender in Dollars and in immediately available funds. Interest shall be payable on any such amounts from the date on which the relevant drawing is paid until payment in full at the rate set forth in (i) until the Business Day next succeeding the date of the relevant notice, subsection 3.4(b) and (ii) thereafter, subsection 3.4(e). Each drawing under any Letter of Credit shall (unless an event of the type described in clause (i) or (ii) of subsection 10(g) shall have occurred and be continuing with respect to the Borrower, in which case the procedures specified in subsection 5.4 for funding by L/C Participants shall apply) constitute a request by the Borrower to the Administrative Agent for a borrowing pursuant to subsection 2.2 of ABR Loans in the amount of such drawing (and the minimum borrowing amount in such subsection shall not apply to such borrowing). The Borrowing Date with respect to such borrowing shall be the first date on which a borrowing of Revolving Credit Loans could be made, pursuant to subsection 2.2, if the Administrative Agent had received a notice of such borrowing at the time the Administrative Agent receives notice from the relevant Issuing Lender of such drawing under such Letter of Credit.

5.6. Obligations Absolute

. The Borrower's obligations under this Section 5 shall be absolute and unconditional under any and all circumstances and irrespective of any setoff, counterclaim or defense to payment that the Borrower may have or have had against the Issuing Lender, any beneficiary of a Letter of Credit or any other Person. The Borrower also agrees with the Issuing Lender that the Issuing Lender shall not be responsible for, and the Borrower's Reimbursement Obligations under subsection 5.5 shall not be affected by, among other things, the validity or genuineness of documents or of any endorsements thereon, even though such documents shall in fact prove to be invalid, fraudulent or forged, or any dispute between or among the Borrower and any beneficiary of any Letter of Credit or any other party to which such Letter of Credit may be transferred or any claims whatsoever of the Borrower against any beneficiary of such Letter of Credit or any such transferee. The Issuing Lender shall not be liable for any error, omission, interruption or delay in transmission, dispatch or delivery of any message or advice, however transmitted, in connection with any Letter of Credit, except for errors or omissions found by a final and nonappealable decision of a court of competent jurisdiction to have resulted from the gross negligence or willful misconduct of the Issuing Lender. The Borrower agrees that any action taken or omitted by the Issuing Lender under or in connection with any Letter of Credit or the related drafts or documents, if done in the absence of gross negligence or willful misconduct and in accordance with the standards of care specified in the Uniform Commercial Code of the State of New York, shall be binding on the Borrower and shall not result in any liability of the Issuing Lender to the Borrower.

5.7. Letter of Credit Payments

. If any drawing shall be presented for payment under any Letter of Credit, the Issuing Lender shall, promptly following its receipt thereof, examine all documents purporting to represent a demand for

payment under such Letter of Credit. Such Issuing Lender shall promptly after such examination notify the Administrative Agent and the Borrower in writing of such demand for payment and whether such Issuing Lender has made or will make payment thereunder; provided that any failure to give or delay in giving such notice shall not relieve the Borrower of its obligation to reimburse such Issuing Lender and L/C Participant with respect to any such payment under Letter of Credit. The responsibility of the Issuing Lender to the Borrower in connection with any drawing presented for payment under any Letter of Credit shall, in addition to any payment obligation expressly provided for in such Letter of Credit, be limited to determining that the documents (including each drawing) delivered under such Letter of Credit in connection with such presentment are substantially in conformity with such Letter of Credit.

5.8. Applications

. To the extent that any provision of any Application related to any Letter of Credit is inconsistent with the provisions of this <u>Section 5</u>, the provisions of this <u>Section 5</u> shall apply.

5.9. Reimbursement Obligations for Certain Letters of Credit Denominated in Currencies Other Than Dollars

. Notwithstanding any other provision of this Section 5, in the event that any Letter of Credit is denominated in any currency other than Dollars, the amount of the Reimbursement Obligation of the Borrower pursuant to subsection 5.5 in respect of such Letter of Credit shall bear interest as provided in subsection 5.5 with respect to amounts owing in Dollars; provided, that (i) the interest rate on such amounts shall be the rate reasonably determined by the Issuing Lender to be the equivalent rate, in respect of the relevant non-Dollar currency, to the applicable rate provided in subsection 5.5 with respect to amounts denominated in Dollars and (ii) if the Borrower fails to pay any such Reimbursement Obligation required by subsection 5.5 on or prior to the third Business Day following the date of the drawing to which such Reimbursement Obligation relates, then, on the fourth Business Day following such date of drawing, the relevant Issuing Lender, in cooperation with the Administrative Agent, shall determine the Dollar Equivalent of the amount of such Reimbursement Obligation, and the Borrower's obligation in respect of such Reimbursement Obligation shall be converted to such Dollar Equivalent, with interest thereon as provided in subsection 5.5 (provided, that if the Application in respect of such Letter of Credit provides for conversion of such amount into Dollars on any earlier date or at any other conversion rate, the provisions hereunder shall control with respect to such conversion).

5.10. Replacement of Issuing Lenders

An Issuing Lender may be replaced at any time by written agreement among the Borrower, the Administrative Agent, the replaced Issuing Lender and the successor Issuing Lender. The Administrative Agent shall notify the Lenders of any such replacement of an Issuing Lender. At the time any such replacement shall become effective, the Borrower shall pay all unpaid fees accrued for the account of the replaced Issuing Lender. From and after the effective date of any such replacement, (x) the successor Issuing Lender shall have all the rights and obligations of Issuing Lenders under this Agreement with respect to Letters of Credit to be issued thereafter and (y) references herein to the term "Issuing Lender" shall be deemed to refer to such successor or to any previous Issuing Lenders, or to such successor and all previous Issuing Lenders, as the context shall require. After the replacement of an Issuing Lender hereunder, the replaced Issuing Lender shall remain a party hereto and shall continue to have all the rights and obligations of an Issuing Lender under this Agreement with respect to Letters of Credit issued by it prior to such replacement, but shall not be required to issue additional Letters of Credit.

- 69 -

SECTION 6

REPRESENTATIONS AND WARRANTIES

To induce the Administrative Agent and the Lenders to enter into this Agreement and to make the Loans and issue or participate in the Letters of Credit, the Borrower hereby represents and warrants to the Administrative Agent, each Issuing Lender and each Lender that:

6.1. Financial Condition

. The consolidated balance sheet of the Borrower and its consolidated Subsidiaries as at each of December 31, 2019 and December 31, 20202023 and the related consolidated statements of operations and of cash flows for the fiscal years ended on such dates, reported on by Ernst & Young LLP, copies of which have heretofore been furnished to each Lender, are complete and correct and present fairly the consolidated financial condition of the Borrower and its consolidated Subsidiaries as at such dates, and the consolidated results of their operations and their consolidated cash flows for the fiscal years then ended. The unaudited consolidated balance sheet of the Borrower and its consolidated Subsidiaries as at the date of the Borrower's most recent publicly available Form 10-Q and the related unaudited consolidated statements of operations and of cash flows for the fiscal period ended on such date, certified by a Responsible Officer, copies of which have heretofore been furnished to each Lender, are complete and materially correct and present fairly (subject to normal year-end audit adjustments) the consolidated financial condition of the Borrower and its consolidated Subsidiaries as at such date, and the consolidated results of their operations and their consolidated cash flows for the fiscal period then ended. All such annual financial statements, including the related schedules and notes thereto, were, as of the date prepared, prepared in accordance with GAAP applied consistently throughout the periods involved (except as approved by such accountants or Responsible Officer, as the case may be, and as disclosed therein). The quarterly financial statements have been prepared in accordance with generally accepted accounting principles for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X under the Securities Act of 1933. Accordingly, such quarterly financial statements do not include all of the information and footnotes required by GAAP for complete financial statements. In the opinion of the Borrower, all adjustments (consisting only of normal recurring accruals) considered necessary for a fair presentation have been included. Neither the Borrower nor any of its consolidated Subsidiaries had, at the date of the most recent balance sheet referred to above, any of the following except as disclosed in the Borrower's Form 10-K and 10-O filings: any material Guarantee Obligation, material contingent liability or material liability for taxes, or any material long-term lease or material unusual forward or long-term commitment, including, without limitation, any interest rate or foreign currency swap or exchange transaction, which is not reflected in the foregoing financial statements or in the notes thereto.

6.2. Corporate Existence; Compliance with Law

. Each of the Borrower and its Subsidiaries (a) is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization, (b) has the corporate power and authority, and the legal right, to own and operate its property, to lease the property it operates as lessee and to conduct the business in which it is currently engaged, (c) is duly qualified as a foreign corporation and in good standing under the laws of each jurisdiction where its ownership, lease or operation of property or the conduct of its business requires such qualification and (d) is in compliance with all Requirements of Law, except to the extent that the failure of the foregoing clauses (a) and (b) (in each such case, only with

respect to Subsidiaries of the Borrower), (c) and (d) to be true and correct could not, in the aggregate, reasonably be expected to have a Material Adverse Effect.

6.3. Corporate Power; Consents and Authorization; Enforceable Obligations

. The Borrower has the corporate power and authority, and the legal right, to make, deliver and perform the Loan Documents to which it is a party and to borrow hereunder and has taken all necessary corporate action to (i) authorize the borrowings on the terms and conditions of this Agreement, any Notes and any Applications, (ii) to authorize the execution, delivery and performance of the Loan Documents to which it is a party and (iii) in the case of the Borrower, to guarantee all Obligations of any Foreign Subsidiary Borrower hereunder. No consent or authorization of, filing with, notice to or other act by or in respect of, any Governmental Authority (including, without limitation, exchange control) or any other Person is required with respect to the Borrower or any of its Subsidiaries in connection with the borrowings or guarantee hereunder or with the execution, delivery, performance, validity or enforceability of the Loan Documents to which the Borrower is party. This Agreement and each other Loan Document to which the Borrower is, or is to become, a party has been or will be, duly executed and delivered on behalf of the Borrower. This Agreement and each other Loan Document to which the Borrower is, or is to become, a party constitutes or will constitute, a legal, valid and binding obligation of the Borrower, enforceable against the Borrower in accordance with its terms, subject to the effects of bankruptcy, examination, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, general equitable principles (whether considered in a proceeding in equity or at law) and an implied covenant of good faith and fair dealing.

6.4. No Legal Bar

. The execution, delivery and performance of the Loan Documents, the borrowings hereunder and the use of the proceeds thereof will not violate any Requirement of Law or Contractual Obligation of the Borrower or of any of its Subsidiaries which could reasonably be expected to have a Material Adverse Effect and will not result in, or require, the creation or imposition of any Lien on any of its or their respective properties or revenues pursuant to any such Requirement of Law or Contractual Obligation which could reasonably be expected to have a Material Adverse Effect.

6.5. No Default

. Neither the Borrower nor any of its Subsidiaries is in default under or with respect to any of its Contractual Obligations in any respect which could reasonably be expected to have a Material Adverse Effect. No Default or Event of Default has occurred and is continuing.

6.6. Taxes

. Each of the Borrower and its Subsidiaries has filed or caused to be filed all tax returns which, to the knowledge of the Borrower, are required to be filed and has paid all taxes shown to be due and payable on said returns or on any assessments made against it (other than any the amount or validity of which are currently being contested in good faith by appropriate proceedings and with respect to which reserves in conformity with GAAP have been provided on the books of the Borrower or its Subsidiaries, as the case may be), except to the extent that the failure to do so could not reasonably be expected to result in a Material Adverse Effect.

6.7. Federal Regulations

. No part of the proceeds of any Loans will be used in any manner that would violate Regulation U of the Board as now and from time to time hereafter in effect.

6.8. ERISA

Neither a Reportable Event nor a failure to meet the minimum funding standards (within the meaning of Section 302 of ERISA), whether or not waived, has occurred during the five-year period prior to the date on which this representation is made or deemed made with respect to any Plan other than a Multiemployer Plan, and eachno Plan has failed to complied in all respects with the applicable provisions of ERISA and the Code, where the liability for such occurrence or failure to comply could be reasonably expected to result in a Material Adverse Effect; provided, however, that with respect to any Multiemployer Plan, such representation is made only to the knowledge of the Borrower. No termination of a Single Employer Plan pursuant to Section 4041(c) or 4042 of ERISA has occurred, and no Lien with respect to any Single Employer Plan in favor of the PBGC or a Plan has arisen, during such five-year period. There has been no determination that any Single Employer Plan is, or is reasonably expected to be, in "at risk" status (within the meaning of Section 430 of the Code or Section 303 of ERISA). Neither the Borrower nor any Commonly Controlled Entity has had a complete or partial withdrawal from any Multiemployer Plan that has resulted in liability and to the knowledge of the Borrower, neither the Borrower nor any Commonly Controlled Entity would become subject to any liability under ERISA if the Borrower or any such Commonly Controlled Entity were to withdraw completely from all Multiemployer Plans as of the valuation date most closely preceding the date on which this representation is made or deemed made which liability could be reasonably expected to result in a Material Adverse Effect. To the Borrower's knowledge, no Multiemployer Plan with respect to which the Borrower or any Commonly Controlled Entity has any liability is in Insolvency or in "endangered" or "critical" status (within the meaning of Section 432 of the Code or Section 305 of ERISA).

6.9. Investment Company Act; Other Regulations

. The Borrower is not required to be registered as an "investment company" within the meaning of the Investment Company Act of 1940, as amended. The Borrower is not subject to regulation under any Federal or State statute or regulation (other than Regulation X of the Board) which limits its ability to incur Indebtedness.

6.10. Purpose of Loans

. The proceeds of the Loans shall be used to finance working capital and general corporate needs of the Borrower and its Subsidiaries, including but not limited to acquisitions.

6.11. Environmental Matters

. Except to the extent that the failure of the following statements to be true and correct could not reasonably be expected to have a Material Adverse Effect:

- (a) The facilities and properties owned, leased or operated by the Borrower or any of its Subsidiaries (the "<u>Properties</u>") do not contain, and have not previously contained, any Materials of Environmental Concern in amounts or concentrations which (i) constitute or constituted a violation of, or (ii) could reasonably be expected to give rise to liability under, any Environmental Law.
- (b) The Properties and all operations at the Properties are in compliance, and have in the last five years been in compliance, in all material respects with all applicable Environmental Laws, and there

is no contamination at, under or about the Properties or violation of any Environmental Law with respect to the Properties or the business operated by the Borrower or any of its Subsidiaries (the "Business") which could reasonably be expected to materially interfere with the continued operation of the Properties or Business or materially impair the fair saleable value thereof.

- (c) Neither the Borrower nor any of its Subsidiaries has received any notice of violation, alleged violation, non-compliance, liability or potential liability regarding environmental matters or compliance with Environmental Laws with regard to any of the Properties or the Business, nor does the Borrower have knowledge or reason to believe that any such notice will be received or is being threatened.
- (d) Materials of Environmental Concern have not been transported or disposed of from the Properties in violation of, or in a manner or to a location which could reasonably be expected to give rise to liability under, any Environmental Law, nor have any Materials of Environmental Concern been generated, treated, stored or disposed of at, on or under any of the Properties in violation of, or in a manner that could reasonably be expected to give rise to liability under, any applicable Environmental Law.
- (e) No judicial proceeding or governmental or administrative action is pending or, to the knowledge of the Borrower, threatened, under any Environmental Law to which the Borrower or any Subsidiary is or will be named as a party with respect to the Properties or the Business, nor are there any consent decrees or other decrees, consent orders, administrative orders or other orders, or other administrative or judicial requirements outstanding under any Environmental Law with respect to the Properties or the Business.
- (f) There has been no release or threat of release of Materials of Environmental Concern at or from the Properties, or arising from or related to the operations of the Borrower or any Subsidiary in connection with the Properties or otherwise in connection with the Business, in violation of, or in amounts or in a manner that could reasonably be expected to give rise to liability under, Environmental Laws.

6.12. Disclosure

. The statements and information contained herein and in any of the information provided to the Administrative Agent or the Lenders in writing in connection with this Agreement, taken as a whole, do not contain any untrue statement of any material fact, or omit to state a fact necessary in order to make such statements or information not misleading in any material respect, in each case in light of the circumstances under which such statements were made or information provided as of the date so provided.

6.13. No Change

. There has been no change, effect, event, occurrence, state of facts or development which individually or in the aggregate has had or would reasonably be expected to result in a Material Adverse Effect since the Form 10-Q or Form 10-K most recently filed by the Borrower with the Securities and Exchange Commission.

6.14. No Material Litigation

. There are no actions, suits, proceedings, claims or disputes pending at law, in equity, in arbitration or before any Governmental Authority, by or against the Borrower or any of its Subsidiaries or

against any of their properties or revenues that (a) purport to affect or pertain to this Agreement or any other Loan Document, or any of the transactions contemplated hereby, or (b) except as specifically disclosed on the Form 10-Q or Form 10-K most recently filed by the Borrower with the Securities and Exchange Commission, either individually or in the aggregate could reasonably be expected to have a Material Adverse Effect.

6.15. Anti-Corruption Laws and Sanctions

. The Borrower has implemented and maintains in effect policies and procedures reasonably designed to promote compliance by the Borrower, its Subsidiaries and their respective directors, officers and employees with Anti-Corruption Laws and Sanctions applicable to the Borrower, its Subsidiaries and their respective directors, officers and employees, and the Borrower, its Subsidiaries and their respective officers and directors and to the knowledge of the Borrower its employees are in compliance with Anti-Corruption Laws and applicable Sanctions in all material respects. None of (a) the Borrower, any Subsidiary or, to the Borrower's knowledge, any of their respective directors or officers or (b) to the Borrower's knowledge, any of the Borrower's or such Subsidiary's respective employees is a Sanctioned Person or organized or resident in a Sanctioned Country. No Borrowing or Letter of Credit or use of proceeds contemplated by this Agreement will be used by the Borrower or any of its Subsidiaries directly or to its knowledge indirectly in violation of applicable Anti-Corruption Laws or applicable Sanctions.

SECTION 7 CONDITIONS PRECEDENT

7.1. Conditions to Closing

- . The effectiveness of this Agreement is subject to the satisfaction on the Closing Date of the following conditions precedent:
- (a) <u>Credit Agreement</u>. The Administrative Agent shall have received this Agreement, executed and delivered by a duly authorized officer of each Lender and the Borrower, with a counterpart for each Lender and original Notes executed by the Borrower (in the case of Revolving Credit Notes, Multicurrency Notes or CAF Advance Notes), in favor of each Lender requesting a Note.
- (b) <u>Closing Certificate</u>. The Administrative Agent shall have received a certificate of the Borrower, dated the Closing Date, substantially in the form of <u>Exhibit F</u>, with appropriate insertions and attachments, satisfactory in form and substance to the Administrative Agent, executed by the President or any Vice President and the Secretary or any Assistant Secretary of the Borrower.
- (c) <u>Representations and Warranties</u>. Each of the representations and warranties made by the Borrower in or pursuant to the Loan Documents shall be true and correct in all material respects (other than any representation or warranty qualified as to "materiality", "Material Adverse Effect" or similar language, which shall be true and correct in all respects) on and as of the Closing Date as if made on and as of the Closing Date.
- (d) <u>Legal Opinions</u>. The Administrative Agent shall have received, with a counterpart for each Lender, the executed legal opinion of counsel to the Borrower (which may be delivered in part by in-house counsel to the Borrower), covering the matters set forth in <u>Exhibit G</u>. Each such legal opinion shall cover such other matters incident to the transactions contemplated by this Agreement as the Administrative Agent may reasonably require.

- (e) <u>Fees</u>. All accrued fees and reasonable out-of-pocket expenses (including the reasonable fees and expenses of counsel to the Arrangers) of the Arrangers through the Closing Date invoiced with reasonable detail at least three Business Days prior to the Closing Date in connection with the Loan Documents shall have been paid; <u>provided</u> that the Arrangers shall have provided an estimate and available reasonable detail five business days prior to the Closing Date.
- (f) <u>Financial Statements</u>. The Administrative Agent shall have received an unaudited consolidated balance sheet of the Borrower and its consolidated Subsidiaries and the related unaudited consolidated statements of operations and of cash flows for each fiscal quarter ended after December 31, 2020 (so long as such fiscal quarters have ended at least 40 days prior to the Closing Date). The Borrower's filing of any required unaudited financial statements with respect to the Borrower on Form 10-Q will satisfy the requirements under this paragraph.
- (g) No Default. No Default or Event of Default shall have occurred and be continuing on such date or after giving effect to the Loans requested to be made, or Letters of Credit requested to be issued.
- (h) No conflicts. No law or regulation shall be applicable in the reasonable judgment of the Lenders that restrains, prevents or imposes material adverse conditions upon the Transactions or this Agreement. All requisite governmental authorities and third parties shall have approved or consented to the Transactions and the other transactions contemplated hereby to the extent required (without the imposition of any materially burdensome condition or qualification in the reasonable judgment of the Lenders) and all such approvals shall be in full force and effect. The Transactions shall be in compliance with all applicable laws and regulations.
- (i) <u>Termination of Existing Credit Facilities</u>. Evidence that all principal amounts, interest, fees and other amounts owed under the Borrower's Existing Credit Facilities shall have been paid in full on or before the Closing Date and all commitments and agreements with respect thereto shall have been terminated.
- (j) <u>Know Your Customer Information</u>. The Administrative Agent shall have received at least three Business Days prior to the Closing Date all documentation and other information about the Borrower as has been reasonably requested by the Administrative Agent at least 10 Business Days prior to the Closing Date that is required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including without limitation the USA PATRIOT Act.

7.2. Conditions to Each Loan and Letter of Credit

- . The agreement of each Lender to make any Loan requested to be made by it on any date following the Closing Date, or to issue any Letter of Credit, is subject to the satisfaction of the following conditions precedent:
- (a) <u>Representations and Warranties</u>. Each of the representations and warranties (except for those made in <u>subsection 6.13</u> and <u>6.14</u>, which shall be made only on the Closing Date and on each Termination Date which a Borrower has requested be extended in accordance with <u>subsection 2.20</u>) made by the Borrower in or pursuant to the Loan Documents shall be true and correct in all material respects on and as of such date as if made on and as of such date.

- (b) No Default. No Default or Event of Default shall have occurred and be continuing on such date or after giving effect to the Loans requested to be made, or Letters of Credit requested to be issued.
- (c) Each borrowing by the Borrower hereunder, and each request by the Borrower for the issuance of a Letter of Credit shall constitute a representation and warranty by the Borrower as of the date thereof that the conditions contained in this subsection have been satisfied.

SECTION 8

AFFIRMATIVE COVENANTS

The Borrower hereby agrees that, so long as the Commitments (or any of them) remain in effect, any Letter of Credit is outstanding or any amount is owing to any Lender, and Issuing Lender or the Administrative Agent hereunder or under any other Loan Documents, the Borrower shall and (except in the case of delivery of financial information, reports and notices) shall cause each of its Subsidiaries to:

8.1. Financial Statements

- . Furnish to each Lender:
- (a) as soon as available, but in any event not later than 20 days after required to be filed with the Securities and Exchange Commission at the end of each fiscal year of the Borrower, a copy of the consolidated balance sheet of the Borrower and its consolidated Subsidiaries as at the end of such year and the related consolidated statements of operations and stockholders' equity and of cash flows for such year, setting forth in each case in comparative form the figures for the previous year, reported on without a "going concern" or like qualification or exception, or qualification arising out of the scope of the audit, by Ernst & Young LLP or other independent certified public accountants of nationally recognized standing;
- (b) as soon as available, but in any event not later than 15 days after required to be filed with the Securities and Exchange Commission at the end of each of the first three quarterly periods of each fiscal year of the Borrower commencing with the fiscal quarter ending on or about March 31, 2021, the unaudited consolidated balance sheet of the Borrower and its consolidated Subsidiaries as at the end of such quarter and the related unaudited consolidated statements of operations for such quarter and the portion of the fiscal year through the end of such quarter and of cash flows of the Borrower and its consolidated Subsidiaries for the portion of the fiscal year through the end of such quarter, setting forth in each case in comparative form the figures for the previous year, certified by a Responsible Officer as being fairly stated in all material respects (subject to normal year-end audit adjustments); and
- (c) all such financial statements shall be complete and correct in all material respects and shall be prepared in reasonable detail and in accordance with GAAP applied consistently throughout the periods reflected therein and with prior periods (except as approved by such accountants or officer, as the case may be, and disclosed therein); <u>provided</u>, that it is hereby acknowledged that the quarterly financial statements delivered pursuant to paragraph (b) above may not include all of the information and footnotes required by GAAP for complete annual financial statements.

Any financial statement required to be furnished pursuant to this <u>subsection 8.1</u> may be delivered electronically and if so delivered, shall be deemed to have been furnished on the earlier of the date (i) on which the Borrower posts such documents, or provides a link thereto on the Borrower's website at the

website address listed in <u>subsection 13.2(a)</u>, (ii) on which such documents are posted on the Securities and Exchange Commission's website (www.sec.gov), or (iii) on which such documents are posted on the Borrower's behalf on any website to which each Lender and Administrative Agent have access (whether a commercial, third-party website such as Intralinks or DebtDomain or whether sponsored by the Administrative Agent); <u>provided</u> that the Borrower shall give notice (which may be in the form of facsimile or electronic mail) of any such posting to the Administrative Agent (who shall then give notice of any such posting to the Lenders). Notwithstanding the foregoing, the Borrower shall deliver paper copies of any financial statement referred to in this <u>subsection 8.1</u> to the Administrative Agent if the Administrative Agent or any Lender requests the Borrower to furnish such paper copies until written notice to cease delivering such paper copies is given by the Administrative Agent.

8.2. Certificates; Other Information

- . Furnish to the Administrative Agent with sufficient copies for the Lenders:
- (a) concurrently with the delivery of the financial statements referred to in <u>subsections 8.1(a)</u> and <u>8.1(b)</u>, a certificate of a Responsible Officer stating that such Officer has obtained no knowledge of any Default or Event of Default that has occurred and is continuing except as specified in such certificate, and including calculations demonstrating compliance with <u>subsection 9.1</u> (which delivery may, unless the Administrative Agent, or a Lender requests executed originals, be by electronic communication including fax or email and shall be deemed to be an original authentic counterpart thereof for all purposes);
- (b) within ten days after the same are sent, copies of all financial statements and reports which the Borrower sends to its stockholders, and within five days after the same are filed, copies of all financial statements and reports which the Borrower may make to, or file with, the Securities and Exchange Commission or any successor or analogous Governmental Authority, and promptly after the same are issued, copies of all press releases issued by the Borrower; and
- (c) promptly, such additional financial and other information as any Lender may from time to time reasonably request, including information for purposes of compliance with applicable "know your customer" and anti-money laundering rules and regulations, including the Patriot Act and the Beneficial Ownership Regulation, as reasonably requested by the Lenders.

The Borrower hereby acknowledges that (a) the Administrative Agent and/or the Arrangers may make available to the Lenders and the Issuing Lender materials and/or information provided by or on behalf of the Borrower hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks or another similar electronic system (the "Platform") and (b) certain of the Lenders (each, a "Public Lender") may have personnel who do not wish to receive material non-public information with respect to the Borrower or its Affiliates, or the respective securities of any of the foregoing, and who may be engaged in investment and other market-related activities with respect to such Persons' securities. The Borrower hereby agrees that (a) all Borrower Materials that are to be made available to Public Lenders shall be clearly and conspicuously marked "PUBLIC" which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the first page thereof; (b) by marking Borrower Materials "PUBLIC," the Borrower shall be deemed to have authorized the Administrative Agent, the Arrangers, the Issuing Lender and the Lenders to treat such Borrower Materials as not containing any material non-public information with respect to the Borrower or its securities for purposes of United States Federal and state securities laws (provided, however, that to the extent such Borrower Materials constitute Information (as defined in subsection 13.14), they shall be treated as set forth in subsection 13.14); (c) all Borrower Materials marked "PUBLIC" are permitted to be made available through a portion of the

- 77 -

Platform designated "Public Side Information;" and (d) the Administrative Agent and the Arrangers shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" as being suitable only for posting on a portion of the Platform not designated "Public Side Information."

8.3. Payment of Obligations

. Pay, discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all its obligations of whatever nature, (other than where the amount or validity thereof is currently being contested in good faith by appropriate proceedings and reserves in conformity with GAAP with respect thereto have been provided on the books of the Borrower or its Subsidiaries, as the case may be) except to the extent that the failure to do so could not reasonably be expected to result in a Material Adverse Effect.

8.4. Conduct of Business and Maintenance of Existence

. (a) Continue to engage in business of the same general type as conducted by it on the Closing Date; (b) preserve, renew and keep in full force and effect its corporate existence (except as could not in the aggregate be reasonably expected to have a Material Adverse Effect); (c) take all reasonable action to maintain all rights, privileges and franchises necessary or desirable in the normal conduct of its business except as otherwise permitted pursuant to subsection 9.4; and (d) comply with all Requirements of Law except to the extent that failure to comply therewith could not, in the aggregate, be reasonably expected to have a Material Adverse Effect.

8.5. Maintenance of Property; Insurance

. Keep all property necessary in its business in good working order and condition except to the extent that failure to do so could not, in the aggregate, be reasonably expected to have a Material Adverse Effect; maintain with financially sound and reputable insurance companies insurance on all its property in at least such amounts and against at least such risks as are adequate for conducting its business; and furnish to each Lender, upon written request, full information as to the insurance carried.

8.6. Inspection of Property; Books and Records; Discussions

. Keep proper books of records and account in which full, true and correct entries in conformity with GAAP and all Requirements of Law shall be made of all dealings and transactions in relation to its business and activities; and permit representatives of any Lender (upon reasonable advance notice coordinated through the Administrative Agent) to visit and inspect any of its properties and examine and make abstracts from any of its books and records at any reasonable time and as often as may reasonably be desired and to discuss the business, operations, properties and financial and other condition of the Borrower and its Subsidiaries with officers and employees of the Borrower and its Subsidiaries and with its independent certified public accountants.

8.7. Notices

. Promptly give notice (unless available in the public filings or releases of the Borrower or its Subsidiaries) to the Administrative Agent and each Lender of:

(a) the occurrence of any Default or Event of Default;

- (b) any (i) default or event of default under any Contractual Obligation of the Borrower or any of its Subsidiaries or (ii) litigation, investigation or proceeding which may exist at any time involving the Borrower or any of its Subsidiaries, which in either case, could reasonably be expected to have a Material Adverse Effect; and
- (c) the following events, as soon as reasonably possible and in any event within 30 days after the Borrower knows of the event: (i) the occurrence or reasonably expected occurrence of any Reportable Event with respect to any Plan, a failure of the Borrower, its Subsidiaries or a Commonly Controlled Entity to make any required contribution to a Plan, the creation of any Lien in favor of the PBGC or a Plan, any determination that a Single Employer Plan is in "at risk" status (within the meaning of Section 430 of the Code or Section 303 of ERISA), or any withdrawal by the Borrower or a Commonly Controlled Entity from, or the termination or Insolvency of, any Multiemployer Plan or determination that any Multiemployer Plan is in "endangered" or "critical" status (within the meaning of Section 432 of the Code or Section 305 of ERISA); or (ii) the institution of proceedings or the taking of any other action by the PBGC, the Borrower, any Commonly Controlled Entity or any Multiemployer Plan with respect to the withdrawal by the Borrower or a Commonly Controlled Entity from, or the termination or Insolvency of, any Plan (other than the termination of any Single Employer Plan pursuant to Section 4041(b) of ERISA), or with respect to any determination that any Single Employer Plan is in "at risk" status or any such Multiemployer Plan is in "endangered" or "critical" status; where, in connection with any of the foregoing in clause (i) or (ii), only to the extent the amount of liability the Borrower or any Commonly Controlled Entity could reasonably be expected to have arising from an event has a Material Adverse Effect.

Each notice pursuant to this <u>subsection 8.7</u> shall be accompanied by a statement of a Responsible Officer setting forth details of the occurrence referred to therein and stating what action the Borrower proposes to take with respect thereto.

SECTION 9

NEGATIVE COVENANTS

The Borrower hereby agrees that, so long as the Commitments (or any of them) remain in effect, any Letter of Credit remains outstanding, or any amount is owing to any Lender, any Issuing Lender or the Administrative Agent hereunder or under any other Loan Documents, the Borrower shall not, and (except with respect to subsection 9.1) shall not permit any of its Subsidiaries to, directly or indirectly:

9.1. Financial Covenant

. <u>Consolidated Leverage Ratio</u>. Permit the Consolidated Leverage Ratio as at the last day of any period of four consecutive fiscal quarters of the Borrower to exceed the Maximum Leverage Ratio.

9.2. Limitation on Liens

- . Create, incur, assume or suffer to exist any Lien upon any of its property, assets or revenues, whether now owned or hereafter acquired, except for:
- (a) Liens for taxes not yet due or which are being contested in good faith by appropriate proceedings, <u>provided</u> that adequate reserves with respect thereto are maintained on the books of the Borrower or its Subsidiaries, as the case may be, in conformity with GAAP;

- (b) carriers', warehousemen's, mechanics', materialmen's, repairmen's or other like Liens arising in the ordinary course of business which are not overdue for a period of more than 60 days or which are being contested in good faith by appropriate proceedings;
- (c) pledges or deposits in connection with workers' compensation, unemployment insurance and other social security legislation and deposits securing liability to insurance carriers under insurance or self-insurance arrangements;
- (d) deposits to secure the performance of bids, trade contracts (other than for borrowed money), leases, statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business;
- (e) easements, rights-of-way, restrictions and other similar encumbrances incurred in the ordinary course of business which, in the aggregate, are not substantial in amount and which do not in any case materially detract from the value of the property subject thereto or materially interfere with the ordinary conduct of the business of the Borrower or such Subsidiary;
- (f) Liens in existence on the date hereof listed on <u>Schedule 9.2</u>, <u>provided</u> that no such Lien is spread to cover any additional property after the Closing Date and that the amount of Indebtedness secured thereby is not increased;
- (g) Liens securing Indebtedness of the Borrower and its Subsidiaries incurred to finance the acquisition of fixed or capital assets, provided that (i) such Liens shall be created substantially simultaneously with the acquisition of such fixed or capital assets, (ii) such Liens do not at any time encumber any property other than the property financed by such Indebtedness and (iii) the amount of Indebtedness secured thereby is not increased;
- (h) Liens on the property or assets of a corporation which becomes a Subsidiary after the date hereof, <u>provided</u> that (i) such Liens existed at the time such corporation became a Subsidiary and were not created in anticipation thereof, (ii) any such Lien is not spread to cover any property or assets of such corporation after the time such corporation becomes a Subsidiary, and (iii) the amount of Indebtedness secured thereby is not increased;
 - (i) Liens created pursuant to any Receivables Transaction permitted pursuant to subsection 9.3; and
- (j) Liens (not otherwise permitted hereunder) which secure obligations not exceeding (as to the Borrower and all Subsidiaries) the greater of (i) \$500,000,000 and (ii) 7.5% of the Consolidated Tangible Assets, in aggregate amount at any time.

9.3. Limitation on Indebtedness pursuant to Receivables Transactions

. Create, issue, incur, assume, become liable in respect of or suffer to exist any Indebtedness pursuant to any Receivables Transaction, except for Indebtedness pursuant to all Receivables Transactions in an aggregate principal amount not exceeding 20% of Consolidated Tangible Assets.

9.4. Limitation on Fundamental Changes

. Enter into any merger, consolidation or amalgamation, or liquidate, wind up or dissolve itself (or suffer any liquidation or dissolution), consummate a Division as the Dividing Person, or convey, sell,

lease, assign, transfer or otherwise Dispose of, all or substantially all of its property, business or assets, except:

- (a) any Subsidiary of the Borrower may be merged or consolidated with or into the Borrower (<u>provided</u> that the Borrower shall be the continuing or surviving corporation) or with or into any one or more wholly owned Subsidiaries of the Borrower (<u>provided</u> that the wholly owned Subsidiary or Subsidiaries shall be the continuing or surviving corporation);
- (b) any Subsidiary of the Borrower that is inactive or no longer needed in the Borrower's consolidated group structure may be liquidated, dissolved, or otherwise eliminated under applicable law, so long as any remaining significant assets of such Subsidiary are transferred to the Borrower or to another of Borrower's wholly owned Subsidiaries (as a liquidation distribution or otherwise);
- (c) the Borrower or any wholly owned Subsidiary of the Borrower may sell, lease, transfer or otherwise dispose of any or all of its assets (upon voluntary liquidation or otherwise) to the Borrower or any other wholly owned Subsidiary, and, so long as no Default or Event of Default shall have occurred and be continuing or would occur as a result thereof, the Borrower or any Subsidiary of the Borrower may sell, lease, transfer or otherwise dispose of any or all of its assets (upon voluntary liquidation or otherwise) to any non-wholly owned Subsidiary of the Borrower for fair market value;
- (d) any non-wholly owned Subsidiary of the Borrower may sell, lease, transfer or otherwise dispose of any or all of its assets (upon voluntary liquidation or otherwise) to the Borrower or any wholly owned Subsidiary of the Borrower for fair market value or may sell, lease, transfer or otherwise dispose of any or all of its assets (upon voluntary liquidation or otherwise) to any other non-wholly owned Subsidiary of the Borrower; and
- (e) the Borrower or any Subsidiary of the Borrower may be merged or consolidated with or into another Person; <u>provided</u> that the Borrower or such Subsidiary shall be the continuing or surviving corporation and no Default or Event of Default shall have occurred and be continuing or would occur as a result thereof (and, in the case of any such transaction involving a Subsidiary, such Subsidiary shall continue to be a Subsidiary or the Borrower shall have received fair market value therefor as determined by the Board of Directors of the Borrower); and <u>provided further</u> that the Borrower may not be merged or consolidated with or into any Subsidiary.

9.5. Limitation on Indebtedness of Subsidiaries

. Permit any Subsidiaries to, directly or indirectly, create, incur, assume or guaranty, or otherwise become or remain directly or indirectly liable with respect to, any Indebtedness in excess of 12.5% of Consolidated Tangible Assets in the aggregate at any time for all such Subsidiaries, except for (i) Indebtedness permitted by subsection 9.3 hereof, (ii) any Indebtedness of any Subsidiary of the Borrower owing to the Borrower or to any other Subsidiary of the Borrower, (iii) Indebtedness in existence on the date hereof listed on Schedule 9.5 and (iv) Indebtedness of American Medical Systems Europe B.V. for so long as American Medical Systems Europe B.V. is a foreign Subsidiary that owns no material assets (including no Capital Stock) and has no material operations.

SECTION 10

EVENTS OF DEFAULT

If any of the following events shall occur and be continuing:

- (a) The Borrower shall fail to pay any principal of any Loan or any Reimbursement Obligation when due in accordance with the terms thereof or hereof; or the Borrower shall fail to pay any interest on any Loan, or any fee or other amount payable hereunder, within five days after any such interest or other amount becomes due in accordance with the terms thereof or hereof; or
- (b) Any representation or warranty made or deemed made by the Borrower herein or in any other Loan Document or which is contained in any certificate, document or financial or other statement furnished by it at any time under or in connection with this Agreement shall prove to have been incorrect in any material respect on or as of the date made or deemed made; or
- (c) The Borrower shall default in the observance or performance of any covenant contained in <u>subsections 8.4(b)</u>, <u>8.7(a)</u> or in <u>Section 9</u>; or
 - (d) The Borrower shall default in the observance or performance of any agreement contained in Section 12; or
- (e) The Borrower shall default in the observance or performance of any other agreement contained in this Agreement (other than as provided above in this Section), and such default described in this clause (e) shall continue unremedied for a period of 30 days; or
- (f) The Borrower or any of its Subsidiaries shall: (i) default in any payment of principal of or interest of any Indebtedness (other than the Loans) or in the payment of any Guarantee Obligation, beyond the period of grace, if any, provided in the instrument or agreement under which such Indebtedness or Guarantee Obligation was created; or (ii) default in the observance or performance of any other agreement or condition relating to any such Indebtedness or Guarantee Obligation or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event shall occur or condition exist, the effect of which default or other event or condition is to cause, or to permit the holder or holders of such Indebtedness or beneficiary or beneficiaries of such Guarantee Obligation (or a trustee or agent on behalf of such holder or holders or beneficiary or beneficiaries) to cause, with the giving of notice if required, such Indebtedness to become due prior to its stated maturity or such Guarantee Obligation to become payable; <u>provided</u>, <u>however</u>, that no Default or Event of Default shall exist under this paragraph unless the aggregate amount of Indebtedness and/or Guarantee Obligations in respect of which any default or other event or condition referred to in this paragraph shall have occurred shall be equal to at least \$250,000,000; or
- (g) (i) The Borrower or any of its Subsidiaries shall commence any case, proceeding or other action (A) under any existing or future law of any jurisdiction, domestic or foreign, relating to bankruptcy, insolvency, examinership, court protection, reorganization or relief of debtors, seeking to have an order for relief entered with respect to it, or seeking to adjudicate it a bankrupt or insolvent, or seeking reorganization, arrangement, adjustment, winding-up, liquidation, dissolution, composition, examinership, court protection or other relief with respect to it or its debts, or (B) seeking appointment of an examiner, receiver, trustee, custodian, conservator or other similar official for it or for all or any substantial part of its assets, or the Borrower or any of its Subsidiaries shall make a general assignment for the benefit of its creditors; or (ii) there shall be commenced against the Borrower or any of its Subsidiaries any case, proceeding or other action of a nature referred to in clause (i) above which (A) results in the entry of an order for relief or any such adjudication or appointment or (B) remains undismissed, undischarged or unbonded for a period of 60 days; or (iii) there shall be commenced against the Borrower or any of its Subsidiaries any case, proceeding or other action seeking issuance of a warrant of attachment, execution, distraint or similar process against all or any substantial part of its assets which

results in the entry of an order for any such relief which shall not have been vacated, discharged, or stayed or bonded pending appeal within 60 days from the entry thereof; or (iv) the Borrower or any of its Subsidiaries shall take any action in furtherance of, or indicating its consent to, approval of, or acquiescence in, any of the acts set forth in clause (i), (ii), or (iii) above; or (v) the Borrower or any of its Subsidiaries shall generally not or shall admit in writing its inability to, pay its debts as they become due; or

- (h) (i) The Borrower, its Subsidiaries or any Commonly Controlled Entity engages in any "prohibited transaction" (as defined in Section 406 of ERISA or Section 4975 of the Code) involving any Plan, (ii) any Plan fails to meet the minimum funding standards described in Section 302 of ERISA or any Lien in favor of the PBGC or a Plan arises on the assets of the Borrower or any Commonly Controlled Entity, (iii) a Reportable Event occurs with respect to, or proceedings shall commence to have a trustee appointed, or a trustee shall be appointed, to administer or to terminate, any Single Employer Plan, which Reportable Event or commencement of proceedings or appointment of a trustee results in the termination of such Plan for purposes of Title IV of ERISA, (iv) any Single Employer Plan terminates for purposes of Title IV of ERISA, (v) a determination that any Single Employer Plan is, or is expected to be, in "at risk" status (within the meaning of Section 430 of the Code or Section 303 of ERISA), (vi) the Borrower or any Commonly Controlled Entity incurs any liability in connection with a withdrawal from, or the Insolvency of, a Multiemployer Plan, or a determination that any Multiemployer Plan is in "endangered" or "critical" status (within the meaning of Section 432 of the Code or Section 305 of ERISA), or (vii) any other event or condition occurs or exist with respect to a Plan; and in each case in clauses (i) through (vii) above, such event or condition, together with all other such events or conditions, if any, could reasonably be expected to have a Material Adverse Effect; or
- (i) Any one judgment or decree shall be entered against the Borrower or any of its Subsidiaries involving in the aggregate a liability (not paid or in excess of the amount recoverable by insurance) of \$250,000,000 (net of any related tax benefit) or more, and such judgment or decree shall not have been vacated, discharged, stayed or appealed (as long as enforcement is effectively stayed during such appeal or such appeal is bonded, if required) within 60 days from the entry thereof; or
- (j) (i) Any Person or "group" (within the meaning of Section 13(d) or 14(d) of the Securities Exchange Act of 1934, as amended) (A) shall have acquired beneficial ownership of 40% or more of any outstanding class of Capital Stock having ordinary voting power in the election of directors of the Borrower (other than Peter M. Nicholas and John E. Abele or any of their affiliated trust holdings) or (B) shall obtain the power (whether or not exercised) to elect a majority of the Borrower's directors; or (ii) the Board of Directors of the Borrower shall not consist of a majority of Continuing Directors; "Continuing Directors" shall mean the directors of the Borrower on the Closing Date and each other director, if such other director's nomination for election to the Board of Directors of the Borrower is recommended by a majority of the then Continuing Directors;

then, and in any such event, (A) if such event is an Event of Default specified in clause (i) or (ii) of paragraph (g) above with respect to the Borrower, automatically the Commitments shall immediately terminate and the Loans hereunder (with accrued interest thereon) and all other amounts owing under this Agreement and the other Loan Documents (including, without limitation, all amounts of L/C Obligations, whether or not the beneficiaries of the then outstanding Letters of Credit shall have presented the documents required thereunder) shall immediately become due and payable, and (B) if such event is any other Event of Default, either or both of the following actions may be taken: (i) with the consent of the Majority Lenders, the Administrative Agent may, or upon the request of the Majority Lenders, the Administrative Agent shall, by notice to the Borrower declare the Commitments to be terminated

forthwith, whereupon the Commitments shall immediately terminate; and (ii) with the consent of the Majority Lenders, the Administrative Agent may, or upon the request of the Majority Lenders, the Administrative Agent shall, by notice to the Borrower, declare the Loans hereunder (with accrued interest thereon) and all other amounts owing under this Agreement and the other Loan Documents (including, without limitation, all amounts of L/C Obligations, whether or not the beneficiaries of the then outstanding Letters of Credit shall have presented the documents required thereunder) to be due and payable forthwith, whereupon the same shall immediately become due and payable. In the case of each Letter of Credit with respect to which presentment for honor shall not have occurred at the time of an acceleration pursuant to this paragraph, the Borrower shall at such time Cash Collateralize the L/C Obligations in an amount (in the currency in which such Letter of Credit is denominated) equal to the then undrawn and unexpired amount of such Letter of Credit. Amounts held as Cash Collateral shall be applied by the Administrative Agent in accordance with subsection 2.18.

SECTION 11

THE AGENTS

11.1. Appointment

. Each Lender hereby irrevocably designates and appoints the Administrative Agent as the agent of such Lender under this Agreement and the other Loan Documents, and each Lender irrevocably authorizes the Administrative Agent, in such capacity, to take such action on its behalf under the provisions of this Agreement and the other Loan Documents and to exercise such powers and perform such duties as are expressly delegated to the Administrative Agent by the terms of this Agreement and the other Loan Documents, together with such other powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary elsewhere in this Agreement, the Administrative Agent shall not have any duties or responsibilities, except those expressly set forth herein, or any fiduciary relationship with any Lender, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against the Administrative Agent.

11.2. Delegation of Duties

. The Administrative Agent may execute any of its duties under this Agreement and the other Loan Documents by or through agents or attorneys-in-fact and shall be entitled to advice of counsel concerning all matters pertaining to such duties. The Administrative Agent shall not be responsible for the negligence or misconduct of any agents or attorneys in-fact selected by it with reasonable care.

11.3. Exculpatory Provisions

. Neither any Agent nor any of its officers, directors, employees, agents, attorneys-in-fact or Affiliates shall be (i) liable for any action lawfully taken or omitted to be taken by it or such Person under or in connection with this Agreement or any other Loan Document (except for its or such Person's own gross negligence or willful misconduct) or (ii) responsible in any manner to any of the Lenders for any recitals, statements, representations or warranties made by the Borrower or any officer thereof contained in this Agreement or any other Loan Document or in any certificate, report, statement or other document referred to or provided for in, or received by such Agent under or in connection with, this Agreement or any other Loan Document or for the value, validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any other Loan Document or for any failure of the Borrower or any

Foreign Subsidiary Borrower to perform its obligations hereunder or thereunder. No Agent shall be under any obligation to any Lender to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Loan Document, or to inspect the properties, books or records of the Borrower or any Foreign Subsidiary Borrower.

11.4. Reliance by Administrative Agent

. The Administrative Agent shall be entitled to rely, and shall be fully protected in relying, upon any Note, writing, resolution, notice, consent, certificate, affidavit, letter, telecopy, telex or teletype message, statement, order or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons and upon advice and statements of legal counsel (including, without limitation, counsel to the Borrower or any Foreign Subsidiary Borrower), independent accountants and other experts selected by the Administrative Agent. The Administrative Agent may deem and treat the payee of any Note as the owner thereof for all purposes unless a written notice of assignment, negotiation or transfer thereof shall have been filed with the Administrative Agent. The Administrative Agent shall be fully justified in failing or refusing to take any action under this Agreement or any other Loan Document unless it shall first receive such advice or concurrence of the Majority Lenders as it deems appropriate or it shall first be indemnified to its satisfaction by the Lenders against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. The Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement and the other Loan Documents in accordance with a request of the Majority Lenders, and such request and any action taken or failure to act pursuant thereto shall be binding upon all the Lenders and all future holders of the Loans.

11.5. Notice of Default

. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default hereunder unless the Administrative Agent has received notice from a Lender or the Borrower referring to this Agreement, describing such Default or Event of Default and stating that such notice is a "notice of default". In the event that the Administrative Agent receives such a notice, the Administrative Agent shall give notice thereof to the Lenders. The Administrative Agent shall take such action with respect to such Default or Event of Default as shall be reasonably directed by the Majority Lenders; <u>provided</u> that unless and until the Administrative Agent shall have received such directions, the Administrative Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable in the best interests of the Lenders.

11.6. Non-Reliance on Administrative Agent and Other Lenders

. Each Lender expressly acknowledges that neither the Administrative Agent nor any of its officers, directors, employees, agents, attorneys-in-fact or Affiliates has made any representations or warranties to it and that no act by the Administrative Agent hereafter taken, including any review of the affairs of the Borrower or any Foreign Subsidiary Borrower, shall be deemed to constitute any representation or warranty by the Administrative Agent to any Lender. Each Lender represents to the Administrative Agent that it has, independently and without reliance upon the Administrative Agent or any other Lender, and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, operations, property, financial and other condition and creditworthiness of the Borrower and the Foreign Subsidiary Borrowers, if any, and made its own decision to make its Loans hereunder and enter into this Agreement. Each Lender also represents that it

- 85 -

will, independently and without reliance upon the Administrative Agent or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make such investigation as it deems necessary to inform itself as to the business, operations, property, financial and other condition and creditworthiness of the Borrower. Except for notices, reports and other documents expressly required to be furnished to the Lenders by the Administrative Agent hereunder, the Administrative Agent shall not have any duty or responsibility to provide any Lender with any credit or other information concerning the business, operations, property, condition (financial or otherwise), prospects or creditworthiness of the Borrower or any Foreign Subsidiary Borrower which may come into the possession of the Administrative Agent or any of its officers, directors, employees, agents, attorneys-in-fact or Affiliates.

11.7. Indemnification

. The Lenders agree to indemnify the Administrative Agent (or sub-agent), in its capacity, any Issuing Lender, in its capacity, and any Related Party acting for the Administrative Agent (or any sub-agent) or any Issuing Lender in connection with such capacity (to the extent not reimbursed by the Borrower and without limiting the obligation of the Borrower and the Foreign Subsidiary Borrowers, if any, to do so), ratably according to their respective Aggregate Exposure Percentages in effect on the date on which indemnification is sought (or, if indemnification is sought after the date upon which the Commitments shall have terminated and the Loans shall have been paid in full, ratably in accordance with such Aggregate Exposure Percentages immediately prior to such date), from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind whatsoever which may at any time (including, without limitation, at any time following the payment of the Loans) be imposed on, incurred by or asserted against the Administrative Agent (or any subagent), any Issuing Lender or such Related Party in any way relating to or arising out of, the Commitments, this Agreement, any of the other Loan Documents or any documents contemplated by or referred to herein or therein or the transactions contemplated hereby or thereby or any action taken or omitted by the Administrative Agent (or any sub-agent), any Issuing Lender or such Related Party under or in connection with any of the foregoing; provided that no Lender shall be liable for the payment of any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements which are found by a final and nonappealable decision of a court of competent jurisdiction to have resulted from the gross negligence or willful misconduct of the Administrative Agent (or sub-agent), any Issuing Lender or any Related Party acting for the Administrative Agent (or any sub-agent) or any Issuing Lender in connection with such capacity. The agreements in this subsection shall survive the payment of the Loans and all other amounts payable hereunder.

11.8. Administrative Agent in Its Individual Capacity

. The Administrative Agent and its Affiliates may make loans to, accept deposits from and generally engage in any kind of business with the Borrower or the Foreign Subsidiary Borrower, if any, as though the Administrative Agent were not the Administrative Agent hereunder and under the other Loan Documents. With respect to the Loans made by it and with respect to any Letter of Credit issued or participated in by it, the Administrative Agent shall have the same rights and powers under this Agreement and the other Loan Documents as any Lender and may exercise the same as though it were not the Administrative Agent, and the terms "Lender" and "Lenders" shall include the Administrative Agent in its individual capacity.

11.9. Successor Administrative Agent

- 86 -

. The Administrative Agent may resign as Administrative Agent upon 10 days' notice to the Lenders. If the Administrative Agent shall resign as Administrative Agent under this Agreement and the other Loan Documents, then the Majority Lenders shall appoint from among the Lenders a successor agent for the Lenders, which successor agent (provided that it shall have been approved by the Borrower), shall succeed to the rights, powers and duties of the Administrative Agent hereunder. Upon the earlier of (i) the 10 day period following the Administrative Agent's notice of resignation to the Lenders and (ii) the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Administrative Agent, and the retiring Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section). The term "Administrative Agent" shall mean such successor agent, and the former Administrative Agent's rights, powers and duties as Administrative Agent shall be terminated, without any other or further act or deed on the part of such former Administrative Agent or any of the parties to this Agreement or any holders of the Loans. After any retiring Administrative Agent's resignation as Administrative Agent, the provisions of this Section 11 shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Administrative Agent under this Agreement and the other Loan Documents.

Any resignation by Wells Fargo as Administrative Agent pursuant to this section shall also constitute its resignation as Issuing Lender. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, (a) such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring Issuing Lender, (b) the retiring Issuing Lender shall be discharged from all of their respective duties and obligations hereunder or under the other Loan Documents, and (c) the successor Issuing Lender shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or make other arrangements satisfactory to the retiring Issuing Lender to effectively assume the obligations of the retiring Issuing Lender with respect to such Letters of Credit.

11.10. The Arrangers, the Bookrunner and the Documentation Agents

. None of the Arrangers, the Bookrunner or the Documentation Agents shall have any right, power, obligation, liability, responsibility or duty under this Agreement other than those applicable to all Lenders as such. Without limiting the foregoing, none of the Arrangers, the Bookrunner or the Documentation Agents shall have or be deemed to have any fiduciary relationship with any Lender. Each Lender acknowledges that it has not relied, and will not rely, on the Arrangers, the Bookrunner or the Documentation Agents in deciding to enter into this Agreement or in taking or not taking any action hereunder.

11.11. Certain ERISA Matters

.

(a) Each Lender (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto to the date such Person ceases being a Lender party hereto, for the benefit of the Administrative Agent, the Arrangers, and the other Agents and their respective Affiliates, and not, for the avoidance of doubt, to or for the benefit of the Borrower, that at least one of the following is and will be true:

- 87 -

- (i) such Lender is not using "plan assets" (within the meaning of the Plan Asset Regulations) of one or more Benefit Plans in connection with the Loans or the Commitments,
- (ii) the transaction exemption set forth in one or more PTEs, such as PTE 84-14 (a class exemption for certain transactions determined by independent qualified professional asset managers), PTE 95-60 (a class exemption for certain transactions involving insurance company general accounts), PTE 90-1 (a class exemption for certain transactions involving insurance company pooled separate accounts), PTE 91-38 (a class exemption for certain transactions involving bank collective investment funds) or PTE 96-23 (a class exemption for certain transactions determined by in-house asset managers), is applicable with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Commitments and this Agreement,
- (iii) (A) such Lender is an investment fund managed by a "Qualified Professional Asset Manager" (within the meaning of Part VI of PTE 84-14), (B) such Qualified Professional Asset Manager made the investment decision on behalf of such Lender to enter into, participate in, administer and perform the Loans, the Commitments and this Agreement, (C) the entrance into, participation in, administration of and performance of the Loans, the Commitments and this Agreement satisfies the requirements of sub-sections (b) through (g) of Part I of PTE 84-14 and (D) to the best knowledge of such Lender, the requirements of subsection (a) of Part I of PTE 84-14 are satisfied with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Commitments and this Agreement, or
- (iv) such other representation, warranty and covenant as may be agreed in writing between the Administrative Agent, in its sole discretion, and such Lender.
- (b) In addition, unless either (1) sub-clause (i) in the immediately preceding clause (a) is true with respect to a Lender or (2) a Lender has not provided another representation, warranty and covenant in accordance with sub-clause (iv) in the immediately preceding clause (a), such Lender further (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto, for the benefit of, the Administrative Agent, the Arrangers, and the other Agents and their respective Affiliates, and not, for the avoidance of doubt, to or for the benefit of the Borrower, that none of the Administrative Agent, or the Arrangers, or any other Agents or any of their respective Affiliates is a fiduciary with respect to the assets of such Lender involved in the Loans, the Commitments and this Agreement (including in connection with the reservation or exercise of any rights by the Administrative Agent under this Agreement, any Loan Document or any documents related to hereto or thereto).

11.12. Accidental Payments

.

(a) Each Lender, each Issuing Lender, each other Secured Party and any other party hereto hereby severally agrees that if (i) the Administrative Agent notifies (which such notice shall be conclusive absent manifest error) such Lender or Issuing Lender or any other Secured Party (or the Lender Affiliate of a Secured Party) or any other Person that the Administrative Agent has determined in its sole discretion that has received funds on behalf of a Lender, Issuing Lender, Secured Party or other Person (each such recipient, a "Payment Recipient") from the Administrative Agent or any of its Affiliates were erroneously transmitted to, or otherwise erroneously or mistakenly received by, such Payment Recipient (whether or

not known to such Payment Recipient) or (ii) any Payment Recipient receives any payment from the Administrative Agent (or any of its Affiliates) (x) that is in a different amount than, or on a different date from, that specified in a notice of payment, prepayment or repayment sent by the Administrative Agent (or any of its Affiliates) with respect to such payment, prepayment or repayment, (y) that was not preceded or accompanied by a notice of payment, prepayment or repayment sent by the Administrative Agent (or any of its Affiliates) with respect to such payment, prepayment or repayment (in whole or in part) then, in each case, an error in payment shall be presumed to have been made (any such amounts specified in clauses (i) or (ii) of this Section 11.12(a), whether received as a payment, prepayment or repayment of principal, interest, fees, distribution or otherwise; individually and collectively, an "Erroneous Payment") then such Payment Recipient is deemed to have knowledge of such error at the time of its receipt of such Erroneous Payment; provided that nothing in this Section shall require the Administrative Agent to provide any of the notices specified in clauses (i) or (ii) above. Each Payment Recipient shall not assert any right or claim to the Erroneous Payment, and hereby waives any claim, counterclaim, defense or right of set-off or recoupment with respect to any demand, claim or counterclaim by the Administrative Agent for the return of any Erroneous Payments, including without limitation waiver of any defense based on "discharge for value" or any similar doctrine.

- (b) Without limiting the immediately preceding <u>clause (a)</u>, each Payment Recipient agrees that, in the case of clause (a)(ii) above, it shall promptly (and, in all events, within one Business Day of its knowledge (or deemed knowledge) of such error) notify the Administrative Agent in writing of such occurrence.
- (c) In the case of either clause (a)(i) or (a)(ii) above, such Erroneous Payment shall at all times remain the property of the Administrative Agent and shall be segregated by the Payment Recipient and held in trust for the benefit of the Administrative Agent, and upon demand from the Administrative Agent such Payment Recipient shall (or, with respect to any Payment Recipient who received such funds on its behalf shall cause such Payment Recipient to), promptly, but in all events no later than one Business Day thereafter, return to the Administrative Agent the amount of any such Erroneous Payment (or portion thereof) as to which such a demand was made in same day funds and in the currency so received, together with interest thereon in respect of each day from and including the date such Erroneous Payment (or portion thereof) was received by such Payment Recipient to the date such amount is repaid to the Administrative Agent at the greater of the Federal Funds Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation from time to time in effect.
- (d) In the event that an Erroneous Payment (or portion thereof) is not recovered by the Administrative Agent for any reason, after demand therefor by the Administrative Agent in accordance with immediately preceding clause (c), from any Lender or Issuing Lender that is a Payment Recipient (such unrecovered amount as to such Lender or Issuing Lender, an "Erroneous Payment Return Deficiency"), then at the sole discretion of the Administrative Agent and upon the Administrative Agent's written notice to such Payment Recipient (i) such Payment Recipient shall be deemed to have assigned its Loans (but not its Commitments) to the Administrative Agent or, at the option of the Administrative Agent, the Administrative Agent's lending affiliate, in a principal amount equal to the Erroneous Payment Return Deficiency (or such lesser amount as the Administrative Agent may specify) (such assignment of the Loans (but not Commitments), the "Erroneous Payment Deficiency Assignment Deficiency Assignment of the Administrative Agent or its lending affiliate as the assignee of such Erroneous Payment Deficiency Assignment, and the Administrative Agent may reflect in the Register its ownership interest in

- 89 -

the Loans subject to the Erroneous Payment Deficiency Assignment. As to any Erroneous Payment Deficiency Assignment, the provisions of this clause (d) shall govern in the event of any conflict with the terms and conditions of Section 13.6. For the avoidance of doubt, no Erroneous Payment Deficiency Assignment will reduce the Commitments of any Lender or Issuing Bank and such Commitments shall remain available in accordance with the terms of this Agreement.

- (e) Each party hereto hereby agrees that (x) in the event an Erroneous Payment (or portion thereof) is not recovered from any Payment Recipient that has received such Erroneous Payment (or portion thereof) for any reason, the Administrative Agent shall be subrogated to all the rights of such Payment Recipient with respect to such amount, (y) the receipt of an Erroneous Payment by a Payment Recipient shall not for the purpose of this Agreement be treated as a payment, prepayment, repayment, discharge or other satisfaction any Obligations owed by the Borrower or any other Credit Party and (z) to the extent that an Erroneous Payment was in any way or at any time credited as payment or satisfaction of any of the Obligations, the Obligations or any part thereof that were so credited, and all rights of the Payment Recipient, as the case may be, shall be reinstated and continue in full force and effect as if such payment or satisfaction had never been received.
- (f) Each Payment Recipient hereby authorizes the Administrative Agent to set off, net and apply any and all amounts at any time owing to such Payment Recipient under any Loan Document, or otherwise payable or distributable by the Administrative Agent to such Payment Recipient from any source, against any amount due to the Administrative Agent under pursuant to this Section 11.12 or under the indemnification provisions of this Agreement.
- (g) Each party's obligations under this <u>Section 11.12</u> shall survive the resignation or replacement of the Administrative Agent or any transfer of right or obligations by, or the replacement of, a Lender, the termination of the Commitments or the repayment, satisfaction or discharge of all Obligations (or any portion thereof) under any Loan Document.

SECTION 12

GUARANTEE

12.1. Guarantee

.

- (a) The Borrower hereby unconditionally and irrevocably guarantees to the Administrative Agent, for the ratable benefit of the Administrative Agent and the Lenders and their respective successors, endorsees, transferees and assigns, the prompt and complete payment and performance by the Foreign Subsidiary Borrowers (if any) when due (whether at the stated maturity, by acceleration or otherwise) of the Obligations.
- (b) No payment or payments made by the Borrower or any other Person or received or collected by the Administrative Agent or any Lender from the Borrower or any other Person by virtue of any action or proceeding or any set-off or appropriation or application, at any time or from time to time, in reduction of or in payment of the Obligations shall be deemed to modify, reduce, release or otherwise affect the liability of the Borrower hereunder which shall, notwithstanding any such payment or

payments, remain liable hereunder for the Obligations until the Obligations are paid in full and the Commitments are terminated.

(c) The Borrower agrees that whenever, at any time, or from time to time, it shall make any payment to the Administrative Agent or any Lender on account of its liability hereunder, it will notify the Administrative Agent and such Lender in writing that such payment is made under this Section for such purpose.

12.2. No Subrogation

Notwithstanding any payment or payments made by the Borrower hereunder, or any set-off or application of funds of the Borrower by the Administrative Agent or any Lender, the Borrower shall not be entitled to be subrogated to any of the rights of the Administrative Agent or any Lender against the Foreign Subsidiary Borrowers or against any collateral security or guarantee or right of offset held by the Administrative Agent or any Lender for the payment of the Obligations, nor shall the Borrower seek or be entitled to seek any contribution or reimbursement from the Foreign Subsidiary Borrowers in respect of payments made by the Borrower hereunder, until all amounts owing to the Administrative Agent and the Local Currency Lenders by the Foreign Subsidiary Borrowers on account of the Obligations are paid in full and the Commitments are terminated. If any amount shall be paid to the Borrower on account of such subrogation rights at any time when all of the Obligations shall not have been paid in full, such amount shall be held by the Borrower in trust for the Administrative Agent and the Lenders, segregated from other funds of the Borrower, and shall, forthwith upon receipt by the Borrower, be turned over to the Administrative Agent in the exact form received by the Borrower (duly indorsed by the Borrower to the Administrative Agent, if required), to be applied against the Obligations, whether matured or unmatured, in such order as Administrative Agent may determine. The provisions of this paragraph shall continue to be effective after the termination of this Agreement, the payment in full of the Obligations and the termination of the Commitments.

12.3. Amendments, etc. with respect to the Obligations; Waiver of Rights

. The Borrower shall remain obligated hereunder notwithstanding that, without any reservation of rights against the Borrower, and without notice to or further assent by the Borrower, any demand for payment of any of the Obligations made by the Administrative Agent or any Lender may be rescinded by the Administrative Agent or such Lender, and any of the Obligations continued, and the Obligations, or the liability of any other party upon or for any part thereof, or any collateral security or guarantee therefor or right of offset with respect thereto, may, from time to time, in whole or in part, be renewed, extended, amended, modified, accelerated, compromised, waived, surrendered or released by the Administrative Agent or any Lender, and any Loan Documents and any other documents executed and delivered in connection therewith may be amended, modified, supplemented or terminated, in whole or in part, in accordance with the provisions thereof as the Administrative Agent (or the requisite Lenders, as the case may be) may deem advisable from time to time, and any collateral security, guarantee or right of offset at any time held by the Administrative Agent or any Lender for the payment of the Obligations may be sold, exchanged, waived, surrendered or released. None of the Administrative Agent or any Lender shall have any obligation to protect, secure, perfect or insure any Lien at any time held by it as security for the Obligations or for this Agreement or any property subject thereto. When making any demand hereunder against the Borrower, the Administrative Agent or any Lender may, but shall be under no obligation to, make a similar demand on the Foreign Subsidiary Borrowers or any other guarantor, and any failure by the Administrative Agent or any Lender to make any such demand or to collect any payments from a Foreign Subsidiary Borrower or any such other guarantor or any release of a Foreign Subsidiary Borrower

or such other guarantor shall not relieve the Borrower of its obligations or liabilities hereunder, and shall not impair or affect the rights and remedies, express or implied, or as a matter of law, of the Administrative Agent or any Lender against the Borrower. For the purposes hereof "demand" shall include the commencement and continuance of any legal proceedings.

12.4. Guarantee Absolute and Unconditional

. The Borrower waives any and all notice of the creation, renewal, extension or accrual of any of the Obligations and notice of or proof of reliance by the Administrative Agent or any Lender upon this Agreement or acceptance of this Agreement; the Obligations, and any of them, shall conclusively be deemed to have been created, contracted or incurred, or renewed, extended, amended or waived, in reliance upon this Agreement; and all dealings between the Foreign Subsidiary Borrowers and the Borrower, on the one hand, and the Administrative Agent and the Lenders, on the other, shall likewise be conclusively presumed to have been had or consummated in reliance upon this Agreement. The Borrower waives diligence, presentment, protest, demand for payment and notice of default or nonpayment to or upon the Foreign Subsidiary Borrowers and the Borrower with respect to the Obligations. This Section 12 shall be construed as a continuing, absolute and unconditional guarantee of payment without regard to (a) the validity, regularity or enforceability of this Agreement, any other Loan Document, any of the Obligations or any other collateral security therefor or guarantee or right of offset with respect thereto at any time or from time to time held by the Administrative Agent or any Lender, (b) any defense, set-off or counterclaim (other than a defense of payment or performance) which may at any time be available to or be asserted by the Foreign Subsidiary Borrowers against the Administrative Agent or any Lender, (c) any law, regulation, decree or order of any jurisdiction, or any other event, affecting the Obligations or any Lender's rights with respect thereto, including, without limitation: (i) the application of any such law, regulation, decree or order, including any prior approval, that would prevent the exchange of a non-Dollar currency for Dollars or the remittance of funds outside of such jurisdiction or the unavailability of Dollars in any legal exchange market in such jurisdiction in accordance with normal commercial practice; or (ii) a declaration of banking moratorium or any suspension of payments by banks in such jurisdiction or the imposition by such jurisdiction or any governmental authority thereof of any moratorium on the required rescheduling or restructuring of, or required approval of payments on, any indebtedness in such jurisdiction; or (iii) any expropriation, confiscation, nationalization or requisition by such country or any governmental authority that directly or indirectly deprives the companies in such jurisdiction of any payment obligation under the Obligations; or (iv) any war (whether or not declared), insurrection, revolution, hostile act, civil strife or similar events occurring in such jurisdiction that has the same effect as the events described in clause (i), (ii) or (iii) above (in each of the cases contemplated in clauses (i) through (iv) above, to the extent occurring or existing on or at any time after the date of this Agreement), or (d) any other circumstance whatsoever (with or without notice to or knowledge of the Foreign Subsidiary Borrowers or the Borrower) which constitutes, or might be construed to constitute, an equitable or legal discharge of the Foreign Subsidiary Borrowers for the Obligations, or of the Borrower under this Section 12, in bankruptcy or in any other instance. When pursuing its rights and remedies hereunder against the Borrower, the Administrative Agent and any Lender may, but shall be under no obligation to, pursue such rights and remedies as it may have against the Foreign Subsidiary Borrowers or any other Person or against any collateral security or guarantee for the Obligations or any right of offset with respect thereto, and any failure by the Administrative Agent or any Lender to pursue such other rights or remedies or to collect any payments from the Borrower or any such other Person or to realize upon any such collateral security or guarantee or to exercise any such right of offset, or any release of the Foreign Subsidiary Borrowers or any such other Person or of any such collateral security, guarantee or right of offset, shall not relieve the Borrower of any liability hereunder, and shall not impair or affect the rights and remedies, whether express, implied or available as a matter of law, of the Administrative Agent

- 92 -

or any Lender against the Borrower. This Section 12 shall remain in full force and effect and be binding in accordance with and to the extent of its terms upon the Borrower and its successors and assigns, and shall inure to the benefit of the Administrative Agent and the Lenders, and their respective successors, endorsees, transferees and assigns, until all the Obligations and the obligations of the Borrower under this Agreement shall have been satisfied by payment in full and the Commitments shall be terminated, notwithstanding that from time to time during the term of this Agreement the Foreign Subsidiary Borrowers may be free from any Obligations.

12.5. Reinstatement

. This <u>Section 12</u> shall continue to be effective, or be reinstated, as the case may be, if at any time payment, or any part thereof, of any of the Obligations is rescinded or must otherwise be restored or returned by the Administrative Agent or any Lender upon the insolvency, bankruptcy, dissolution, liquidation or reorganization of the Borrower or upon or as a result of the appointment of a receiver, intervenor or conservator of, or trustee or similar officer for, the Borrower or any substantial part of its property, or otherwise, all as though such payments had not been made.

12.6. Payments

. The Borrower hereby agrees that all payments required to be made by it hereunder will be made to the Administrative Agent without set-off or counterclaim in accordance with the terms of the Obligations, including, without limitation, in the currency in which payment is due, provided that if a payment is due in a currency other than Dollars and/or at a place other than the United States, and such payment is not made as and when agreed, the Borrower will, upon the Administrative Agent's request, either (i) make payment in such non- Dollar currency and at the place where such payment is payable or (ii) pay the Administrative Agent in Dollars at the Administrative Agent's office referred in Subsection 13.2. In the event of a payment pursuant to clause (ii) above, the Borrower will pay the Administrative Agent the Dollar Equivalent of the amount of such payment on the date the Borrower makes such payment.

12.7. "Lenders"

. For all purposes of this Section 12, the term "Lenders" shall be deemed to include Local Currency Lenders.

SECTION 13

MISCELLANEOUS

13.1. Amendments and Waivers

•

(a) Except as provided in paragraphs (b) and (d) of this <u>subsection 13.1</u>, neither this Agreement nor any other Loan Document, nor any terms hereof or thereof may be amended, supplemented or modified except in accordance with the provisions of this subsection. The Majority Lenders may, or, with the written consent of the Majority Lenders, the Administrative Agent may, from time to time, (a) enter into with the Borrower written amendments, supplements or modifications hereto and to the other Loan Documents for the purpose of adding any provisions to this Agreement or the other Loan Documents or changing in any manner the rights of the Lenders or of the Borrower hereunder or

thereunder or (b) waive, on such terms and conditions as the Majority Lenders or the Administrative Agent, as the case may be, may specify in such instrument, any of the requirements of this Agreement or the other Loan Documents or any Default or Event of Default and its consequences; provided, however, that no such waiver and no such amendment, supplement or modification shall (i) reduce the amount or extend the scheduled date of maturity of any Loan or reduce the stated rate or amount of any interest or fee payable hereunder (except that any amendment or modification of defined terms used in the financial covenants in this Agreement shall not constitute a reduction in the rate of interest or fees for purposes of this clause (i)) or extend the scheduled date of any payment thereof or increase the amount or extend the expiration date of any Lender's Multicurrency Commitment, Revolving Credit Commitment or Issuing Lender Commitment, in each case without the consent of each Lender or Issuing Lender, as applicable, directly affected thereby, (ii) amend, modify or waive any provision of this subsection or reduce the percentages specified in the definitions, of Majority Lenders or Majority Multicurrency Lenders, or consent to the assignment or transfer by the Borrower of any of its rights and obligations under this Agreement and the other Loan Documents or release, subordinate or otherwise materially limit the Borrower's liability with respect to the guarantee set forth in Section 12, in each case without the written consent of all the Lenders, (iii) amend, modify or waive any provision of Section 11 without the written consent of the then Administrative Agent, (iv) waive any condition in subsection 7.2 without the written consent of the Majority Lenders, (v) modify the pro rata distribution of payments, proceeds or fees payable to the Lenders (except in connection with an amend and extend transaction offered ratably to all Lenders under a tranche of Commitments) without the consent of each Lender directly affected thereby or (vi) amend, modify or waive any provision of Section 5 without the written consent of each Lender directly adversely affected thereby. Any such waiver and any such amendment, supplement or modification shall apply equally to each of the Lenders and shall be binding upon the Borrower, the Lenders, the Administrative Agent and all future holders of the Loans. In the case of any waiver, the Borrower, the Lenders and the Administrative Agent shall be restored to their former positions and rights hereunder and under the other Loan Documents, and any Default or Event of Default waived shall be deemed to be cured and not continuing; no such waiver shall extend to any subsequent or other Default or Event of Default or impair any right consequent thereon.

- (b) In addition to amendments effected pursuant to the foregoing paragraph (a), (i) Schedule II may be amended to change administrative information contained therein with the approval of the Majority Multicurrency Lenders, upon execution and delivery by the Borrower and the Administrative Agent of a written amendment providing for such amendment, (ii) additional freely-convertible euro currencies may be added as Available Foreign Currencies, upon execution and delivery by the Borrower, the Administrative Agent and the Majority Multicurrency Lenders of an amendment providing for such addition and (iii) this Agreement may be amended by the Borrower and the Administrative Agent to implement increases in the Revolving Credit Commitments pursuant to subsection 2.17.
- (c) The Administrative Agent shall give prompt written notice to each Lender of any amendment effected pursuant to <u>subsection</u> 13.1(b).
- (d) Notwithstanding the provisions of this <u>subsection 13.1</u>, any Local Currency Facility may be amended, supplemented or otherwise modified in accordance with its terms so long as after giving effect thereto either (x) such Local Currency Facility ceases to be a "<u>Local Currency Facility</u>" and the Borrower so notifies the Administrative Agent or (y) the Local Currency Facility continues to meet the requirements of a Local Currency Facility set forth herein.
- (e) Notwithstanding the provisions of this <u>subsection 13.1</u>, no Defaulting Lender shall have any right to approve or disapprove any amendment, waiver or consent hereunder (and any amendment,

- 94 -

waiver or consent which by its terms requires the consent of all Lenders or each affected Lender may be effected with the consent of the applicable Lenders other than Defaulting Lenders), except that (x) the Commitment of any Defaulting Lender may not be increased or extended and the principal amount owed to such Lender may not be reduced other than pursuant to payments made by a Borrower with respect thereto without the consent of such Lender and (y) any waiver, amendment or modification requiring the consent of all Lenders or each affected Lender that by its terms affects any Defaulting Lender more adversely than other affected Lenders shall require the consent of such Defaulting Lender.

- (f) Notwithstanding the foregoing, the Administrative Agent, with the consent of the Borrower, may amend, modify or supplement any Loan Document without the consent of any Lender or the Majority Lenders in order to correct, amend or cure any ambiguity, inconsistency or defect or correct any typographical error or manifest error in any Loan Document.
- (g) If any Lender is a Non-Consenting Lender, then the Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Section 13.6 and payment of amounts due to such Lender under this Agreement), all of its interests, rights (other than its existing rights to payments pursuant to Section 3.9 or Section 3.10) and obligations under this Agreement and the related Loan Documents to an assignee that shall assume such obligations (which assignee may be another Lender or an Affiliate of a Lender, if a Lender or such Affiliate accepts such assignment); provided that the applicable assignee shall have consented to the applicable amendment, waiver or consent.

13.2. Notices

•

(a) General. All notices, requests and demands to or upon the respective parties hereto to be effective shall be in writing (including by facsimile transmission) and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made (a) in the case of delivery by hand, when delivered, (b) in the case of delivery by mail, three days after being deposited in the mails, postage prepaid, or (c) in the case of delivery by facsimile transmission, when sent and receipt has been confirmed, addressed as follows in the case of the Borrower and the Administrative Agent, and as set forth in Schedule I in the case of the other parties hereto, or to such other address as may be hereafter notified by the respective parties hereto:

If to the Borrower:

Boston Scientific Corporation
300 Boston Scientific Way
Marlborough, Massachusetts 01752
Attention: Daniel J. Brennan
Executive Vice President and Chief Financial Officer and
Robert J. Castagna
Vice President and Treasurer
Fax: 508-683-4410
www.bostonscientific.com

with a copy to:

Boston Scientific Corporation 300 Boston Scientific Way Marlborough, Massachusetts 01752 Attention: Vance Brown

Vice President and Chief Corporate Counsel

Fax: 508-683-5389

Email: legalnotices@bsci.com

If to the Administrative Agent (for payments and notices of borrowings, etc. in Dollars):

Wells Fargo Bank, National Association MAC D01109-019 1525 W. W.T. Harris Blvd. Charlotte, North Carolina 28262 Attention: Syndication Agency Services

Facsimile No.: 844-879-5899

E-mail: agencyservices.requests@wellsfargo.com

If to the Administrative Agent (for payments and notices of borrowings, etc. in non-Dollars):

Wells Fargo Bank, National Association MAC D01109-019 1525 W. W.T. Harris Blvd. Charlotte, North Carolina 28262 Attention: Syndication Agency Services

Facsimile No.: 844-879-5899

E-mail: agencyservices.requests@wellsfargo.com

If to the Administrative Agent (for other notices):

Wells Fargo Bank, National Association 301 S. College 550 South Tryon Street, 142th Floor Charlotte, NC 28202

Attn:

Telephone: (704) 715-4361

E-mail:

For Letters of Credit, to the address provided by the applicable Issuing Lender to the Borrower for such purpose.

<u>provided</u> that any notice, request or demand to or upon the Administrative Agent or the Lenders pursuant to <u>subsection 2.2</u>, <u>2.6</u>, <u>2.9</u>, <u>2.14</u>, <u>2.15</u>, or <u>3.2</u> shall not be effective until received.

(b) Notices and other communications to the Lenders and the Issuing Lender hereunder may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by the Administrative Agent, <u>provided</u> that the foregoing shall

not apply to notices to any Lender or the Issuing Lender pursuant to Section 2, 3, 4 and 5 if such Lender or the Issuing Lender, as applicable, has notified the Administrative Agent that it is incapable of receiving notices under such Section by electronic communication. The Administrative Agent or the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it, provided that approval of such procedures may be limited to particular notices or communications.

Unless the Administrative Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), <u>provided</u> that if such notice or other communication is not sent during the normal business hours of the recipient, such notice or communication shall be deemed to have been sent at the opening of business on the next business day for the recipient, and (ii) notices or communications posted to an Internet or intranet website shall be deemed received upon the deemed receipt by the intended recipient at its e-mail address as described in the foregoing clause (i) of notification that such notice or communication is available and identifying the website address therefor.

(c) The Platform. THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE." THE AGENT PARTIES (AS DEFINED BELOW) DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE BORROWER MATERIALS OR THE ADEQUACY OF THE PLATFORM, AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS IN OR OMISSIONS FROM THE BORROWER MATERIALS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY ANY AGENT PARTY IN CONNECTION WITH THE BORROWER MATERIALS OR THE PLATFORM. In no event shall the Administrative Agent or any of its Affiliates (collectively, the "Agent Parties") have any liability to the Borrower, any Lender, the Issuing Lender or any other Person for losses, claims, damages, liabilities or expenses of any kind (whether in tort, contract or otherwise) arising out of the Borrower's or the Administrative Agent's transmission of Borrower Materials through the Internet, except to the extent that such losses, claims, damages, liabilities or expenses are determined by a court of competent jurisdiction by a final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Agent Party; provided, however, that in no event shall any Agent Party have any liability to the Borrower, any Lender, the Issuing Lender or any other Person for indirect, special, incidental, consequential or punitive damages (as opposed to direct or actual damages).

13.3. No Waiver; Cumulative Remedies

. No failure to exercise and no delay in exercising, on the part of the Administrative Agent or any Lender, any right, remedy, power or privilege hereunder or under the other Loan Documents shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

13.4. Survival of Representations and Warranties

- 97 -

. All representations and warranties made hereunder, in the other Loan Documents and in any document, certificate or statement delivered pursuant hereto or in connection herewith shall survive the execution and delivery of this Agreement and the making of the Loans hereunder.

13.5. Payment of Expenses and Taxes

. The Borrower agrees (a) to pay or reimburse each of the Arrangers and the Administrative Agent for all their reasonable and documented out-of-pocket costs and expenses incurred in connection with the development, preparation and execution of, and any amendment, supplement or modification to, this Agreement and the other Loan Documents and any other documents prepared in connection herewith or therewith, and the consummation and administration of the transactions contemplated hereby and thereby, including, without limitation, the reasonable and documented fees and disbursements of outside counsel (including one local counsel in each applicable jurisdiction) to the Administrative Agent and the Arrangers, (b) to pay or reimburse each Lender, each Issuing Lender, the Arrangers and the Administrative Agent for all its costs and expenses incurred in connection with the enforcement or preservation of any rights under this Agreement, the other Loan Documents and any such other documents, including, without limitation, the documented fees and disbursements of counsel (including the allocated fees and expenses of in house counsel) to each Lender and Issuing Lender and of counsel to the Administrative Agent and the Arrangers, provided, that in connection with any workout or restructuring, the Borrower shall pay the fees and disbursements of (i) one U.S. counsel for the Administrative Agent, the Arrangers, the Issuing Lenders and the Lenders pursuant to this clause (b) and (ii) one counsel to the Administrative Agent, the Arrangers, the Issuing Lenders and the Lenders in the jurisdiction of each Foreign Subsidiary Borrower pursuant to this clause (b), and, in the case of an actual or perceived conflict of interest where the indemnified party affected by such conflict informs the Borrower of such conflict and thereafter, retains its own counsel, of another firm of counsel for such affected indemnified party, (c) to pay, indemnify, and hold each Lender, each Issuing Lender and the Administrative Agent and each of their affiliates and their respective officer, directors, employees, agents and advisors (each, an "indemnified party") harmless from, any and all recording and filing fees and any and all liabilities with respect to, or resulting from any delay in paying, stamp, excise and other similar taxes, if any, which may be payable or determined to be payable in connection with the execution and delivery of, or consummation or administration of any of the transactions contemplated by, or any amendment, supplement or modification of, or any waiver or consent under or in respect of, this Agreement, the other Loan Documents and any such other documents, and (d) to pay, indemnify, and hold each indemnified party harmless from and against any and all other liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever incurred by any indemnified party or asserted against any indemnified party by any third party or by the Borrower or any of its Subsidiaries arising out of, in connection with or as a result of the Transactions (or any transactions related thereto), or the execution, delivery, enforcement, performance and administration of this Agreement, the other Loan Documents and such other documents, including, without limitation, any of the foregoing relating to the violation of, noncompliance with or liability under, any Environmental Law applicable to the operations of the Borrower, any of its Subsidiaries or any of the Properties (all the foregoing in this clause (d), collectively, the "indemnified liabilities"), provided that the Borrower shall have no obligation hereunder to any indemnified party with respect to indemnified liabilities arising from the gross negligence or willful misconduct of such indemnified party determined in a court of competent jurisdiction in a final nonappealable judgment. The agreements in this subsection shall survive repayment of the Loans and all other amounts payable hereunder and the termination of this Agreement.

13.6. Successors and Assigns; Participations and Assignments

- 98 -

- (a) This Agreement shall be binding upon and inure to the benefit of the Borrower, the Lenders, the Administrative Agent and their respective successors and assigns, except that the Borrower may not assign or transfer any of its rights or obligations under this Agreement without the prior written consent of each Lender.
- (b) Any Lender, other than a Conduit Lender, may, in the ordinary course of its commercial banking business and in accordance with applicable law, at any time sell to one or more banks or other entities (other than a Defaulting Lender, the Borrower or its Affiliates or Subsidiaries or any natural Person) ("Participants") participating interests in any Loan owing to such Lender, any Commitment of such Lender or any other interest of such Lender hereunder and under the other Loan Documents. In the event of any such sale by a Lender of a participating interest to a Participant, such Lender's obligations under this Agreement to the other parties to this Agreement shall remain unchanged, such Lender shall remain solely responsible for the performance thereof, such Lender shall remain the holder of any such Loan for all purposes under this Agreement and the other Loan Documents, and the Borrower and the Administrative Agent shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement and the other Loan Documents. No Lender shall be entitled to create in favor of any Participant, in the participation agreement pursuant to which such Participant's participating interest shall be created or otherwise, any right to vote on, consent to or approve any matter relating to this Agreement or any other Loan Document except for those specified in clauses (i) and (ii) of the proviso to subsection 13.1(a). The Borrower agrees that if amounts outstanding under this Agreement are due or unpaid, or shall have been declared or shall have become due and payable upon the occurrence of an Event of Default, each Participant shall, to the maximum extent permitted by applicable law, be deemed to have the right of set-off in respect of its participating interest in amounts owing under this Agreement to the same extent as if the amount of its participating interest were owing directly to it as a Lender under this Agreement, provided that, in purchasing such participating interest, such Participant shall be deemed to have agreed to share with the Lenders the proceeds thereof as provided in subsection 13.7(a) as fully as if it were a Lender hereunder. The Borrower also agrees that each Participant shall be entitled to the benefits of subsections 3.9, 3.10 and 3.11 with respect to its participation in the Commitments and the Loans outstanding from time to time as if it was a Lender; provided that, in the case of subsection 3.10, such Participant shall have complied with the requirements of said subsection, and provided, further, that no Participant shall be entitled to receive any greater amount pursuant to any such subsection than the transferor Lender would have been entitled to receive in respect of the amount of the participation transferred by such transferor Lender to such Participant had no such transfer occurred. Each Lender that sells a participation, acting solely for this purpose as a nonfiduciary agent of the Borrower, shall maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Loans or other obligations under this Agreement (the "Participant Register"); provided, that no Lender shall have any obligation to disclose all or any portion of the Participant Register to any Person (including the identity of any Participant or any information relating to a Participant's interest in any Commitments, Loans or its other Obligations under any Loan Document) except to the extent that such disclosure is necessary to establish that such Commitment, Loan or other Obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender, the Borrower and the Administrative Agent shall treat each person whose name is recorded in the Participant Register pursuant to the terms hereof as the owner of such participation for all purposes of this Agreement, notwithstanding notice to the contrary. For the avoidance of doubt, the Administrative Agent (in its capacity as Administrative Agent) shall have no responsibility for maintaining a Participant Register.

- 99 -

- (c) Any Lender, other than a Conduit Lender, may, in the ordinary course of its commercial banking business and in accordance with applicable law, at any time and from time to time, assign (i) to any Lender or any Lender Affiliate with the consent (in each case, not to be unreasonably withheld, delayed or conditioned) of the Administrative Agent and the Issuing Lender (in the case of assignments of Revolving Commitments) or (ii) with the consent of the Borrower (unless an Event of Default shall have occurred and be continuing), the Administrative Agent and Issuing Lender (which consent in each case shall not be unreasonably withheld, delayed or conditioned), to an additional bank, financial institution, or other entity (an "Assignee") all or any part of its rights and obligations under this Agreement and the other Loan Documents pursuant to an Assignment and Assumption, substantially in the form of Exhibit H, executed by such Assignee, such assigning Lender (and, in the case of an Assignee that is not a Lender or a Lender Affiliate, by the Administrative Agent and the Issuing Lender) and delivered to the Administrative Agent for its acceptance and recording in the Register, provided that, except in the case of an assignment to a Lender or a Lender Affiliate or an assignment of the entire remaining amount of the assigning Lender's Commitments or Loans, the amount of the Commitments or Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent) shall not be less than \$5,000,000 (or such lesser amount as may be agreed to by the Borrower and the Administrative Agent), and provided further that no such assignment shall be made to (A) any Defaulting Lender or any of its Subsidiaries, (B) the Borrower or any of its Affiliates or Subsidiaries or (C) any natural person. Upon such execution, delivery, acceptance and recording, from and after the effective date determined pursuant to such Assignment and Assumption, (x) the Assignee thereunder shall be a party hereto and, to the extent provided in such Assignment and Assumption, have the rights and obligations of a Lender hereunder with Commitments as set forth therein, and (y) the assigning Lender thereunder shall, to the extent provided in such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all or the remaining portion of an assigning Lender's rights and obligations under this Agreement, such assigning Lender shall cease to be a party hereto but shall continue to be entitled to the indemnity and expense reimbursement provisions in the Loan Documents arising out of the period prior to the assignment). Each assignment by a Lender of any portion of its Revolving Credit Commitment shall be accompanied by assignment by such Lender to the same Assignee of the same percentage of such Lender's Multicurrency Commitment. Notwithstanding the foregoing, any Conduit Lender may assign at any time to its designating Lender hereunder without the consent of the Borrower or the Administrative Agent any or all of the Loans it may have funded hereunder and pursuant to its designation agreement and without regard to the limitations set forth in the first sentence of this subsection 13.6(c).
- (d) The Administrative Agent, acting solely for this purpose as a non-fiduciary agent of the Borrower, shall maintain at the address of the Administrative Agent referred to in <u>subsection 13.2</u> a copy of each Assignment and Assumption delivered to it and a register (the "<u>Register</u>") for the recordation of the names and addresses of the Lenders and the Commitments of, and principal amount (and stated interest) of the Loans owing to, each Lender from time to time. The entries in the Register shall be conclusive, in the absence of manifest error, and the Borrower, the Administrative Agent and the Lenders shall treat each Person whose name is recorded in the Register as the owner of a Loan or other obligation hereunder as the owner thereof for all purposes of this Agreement and the other Loan Documents, notwithstanding any notice to the contrary. Any assignment of any Loan or other obligation hereunder not evidenced by a Note shall be effective only upon appropriate entries with respect thereto being made in the Register. In addition, the Administrative Agent shall maintain on the Register information regarding the designation, and revocation of designation, of any Lender as a Defaulting Lender. The Register shall be available for inspection by the Borrower or any Lender at any reasonable time and from time to time upon reasonable prior notice.

- 100 -

- (e) Upon its receipt of an Assignment and Assumption executed by an assigning Lender and an Assignee (and, in the case of an Assignee that is not then a Lender or a Lender Affiliate, by the Borrower (if required) and the Administrative Agent) together with payment to the Administrative Agent of a registration and processing fee of \$4,000, the Administrative Agent shall (i) promptly accept such Assignment and Assumption and (ii) on the effective date determined pursuant thereto record the information contained therein in the Register and give notice of such acceptance and recordation to the Lenders and the Borrower; provided that the Administrative Agent may, in its sole discretion, elect to waive such registration and processing fee in the case of any assignment.
- (f) The Borrower authorizes each Lender to disclose to any Participant or Assignee (each, a "<u>Transferee</u>") and any prospective Transferee, subject to the provisions of <u>subsection 13.14</u>, any and all financial information in such Lender's possession concerning the Borrower and its Affiliates which has been delivered to such Lender by or on behalf of the Borrower pursuant to this Agreement or which has been delivered to such Lender by or on behalf of the Borrower in connection with such Lender's credit evaluation of the Borrower and its Affiliates prior to becoming a party to this Agreement.
- (g) Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank or any central bank having jurisdiction over such Lender in accordance with applicable law; <u>provided</u> that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.
- (h) In connection with any assignment of rights and obligations of any Defaulting Lender hereunder, no such assignment shall (unless otherwise agreed by the Borrower and the Administrative Agent) be effective unless and until, in addition to the other conditions thereto set forth herein, the parties to the assignment shall make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrower and the Administrative Agent, the applicable pro rata share of Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (x) pay and satisfy in full all payment liabilities then owed by such Defaulting Lender to the Administrative Agent or any Lender hereunder (and interest accrued thereon) and (y) acquire (and fund as appropriate) its full pro rata share of all applicable Loans and participations in Letters of Credit in accordance with its Revolving Credit Commitment Percentage. Notwithstanding the foregoing, in the event that any assignment of rights and obligations of any Defaulting Lender hereunder shall become effective under applicable law without compliance with the provisions of this paragraph, then the assignee of such interest shall be deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.

13.7. Adjustments; Set-off

(a) If any Lender (a "<u>benefited Lender</u>") shall at any time receive any payment of all or part of its Loans or the Reimbursement Obligations owing to it (or any participation therein arising pursuant to <u>subsection 13.15</u>) then due and owing, or interest thereon, or receive any collateral in respect thereof (whether voluntarily or involuntarily, by set-off, pursuant to events or proceedings of the nature referred to in <u>subsection 10(g)</u>, or otherwise), in a greater proportion than any such payment to or collateral

received by any other Lender (other than to the extent expressly provided herein or by court order), if any, in respect of such other Lender's Loans or the Reimbursement Obligations owing to it (or any participation therein arising pursuant to <u>subsection 13.15</u>) then due and owing, or interest thereon, such benefited Lender shall purchase for cash from the other Lenders a participating interest in such portion of each such other Lender's Loans or the Reimbursement Obligations owing to it, or shall provide such other Lenders with the benefits of any such collateral, or the proceeds thereof, as shall be necessary to cause such benefited Lender to share the excess payment or benefits of such collateral or proceeds ratably with each of the Lenders; <u>provided</u>, <u>however</u>, that if all or any portion of such excess payment or benefits is thereafter recovered from such benefited Lender, such purchase shall be rescinded, and the purchase price and benefits returned, to the extent of such recovery, but without interest; and <u>provided further</u> that in the event that any Defaulting Lender shall exercise any such right of setoff, (x) all amounts so set off shall be paid over immediately to the Administrative Agent for further application in accordance with the provisions of <u>subsection 2.19</u> and, pending such payment, shall be segregated by such Defaulting Lender from its other funds and deemed held in trust for the benefit of the Administrative Agent and the Lenders, and (y) the Defaulting Lender shall provide promptly to the Administrative Agent a statement describing in reasonable detail the Obligations owing to such Defaulting Lender as to which it exercised such right of setoff.

(b) In addition to any rights and remedies of the Lenders provided by law, each Lender shall have the right, without prior notice to the Borrower, any such notice being expressly waived by the Borrower to the extent permitted by applicable law, upon any amount becoming due and payable by the Borrower hereunder (whether at the stated maturity, by acceleration or otherwise) to set-off and appropriate and apply against such amount any and all deposits (general or special, time or demand, provisional or final), in any currency, and any other credits, indebtedness or claims, in any currency, in each case whether direct or indirect, absolute or contingent, matured or unmatured, at any time held or owing by such Lender or any Affiliate, branch or agency thereof to or for the credit or the account of the Borrower. Each Lender agrees promptly to notify the Borrower and the Administrative Agent after any such set-off and application made by such Lender, provided that the failure to give such notice shall not affect the validity of such set-off and application.

13.8. Counterparts

. This Agreement may be executed by one or more of the parties to this Agreement on any number of separate counterparts (including by facsimile transmission or in electronic (i.e., "pdf" or "tif") format), and all of said counterparts taken together shall be deemed to constitute one and the same instrument. A set of the copies of this Agreement signed by all the parties shall be lodged with the Borrower and the Administrative Agent.

13.9. Severability

. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction. Without limiting the foregoing provisions of this Section 13.9, if and to the extent that the enforceability of any provisions in this Agreement relating to Defaulting Lenders shall be limited by Debtor Relief Laws or a Bail-In Action, as determined in good faith by the Administrative Agent or the Issuing Lender, as applicable, then such provisions shall be deemed to be in effect only to the extent not so limited.

- 102 -

13.10. Integration

. This Agreement and the other Loan Documents represent the agreement of the Borrower, the Administrative Agent and the Lenders with respect to the subject matter hereof, and there are no promises, undertakings, representations or warranties by the Administrative Agent or any Lender relative to subject matter hereof not expressly set forth or referred to herein or in the other Loan Documents.

13.11. GOVERNING LAW

. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER AND ANY CLAIM, CONTROVERSY, DISPUTE, PROCEEDING OR CAUSE OF ACTION (WHETHER IN CONTRACT, TORT OR OTHERWISE AND WHETHER AT LAW OR IN EQUITY) BASED UPON, ARISING OUT OF OR RELATING TO THIS AMENDMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK.

13.12. Submission To Jurisdiction and Waivers

- . The Borrower hereby irrevocably and unconditionally:
- (a) submits for itself and its property in any legal action or proceeding (whether in contract, tort or otherwise and whether at law or in equity) relating to this Agreement and the other Loan Documents to which it is a party, or for recognition and enforcement of any judgment in respect thereof, to the exclusive general jurisdiction of the Courts of the State of New York sitting in New York County, Borough of Manhattan, the courts of the United States of America for the Southern District of New York, and appellate courts from any thereof;
- (b) consents that any such action or proceeding may be brought in such courts and waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same;
- (c) agrees that service of process in any such action or proceeding may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to the Borrower at its address set forth in <u>subsection 13.2</u> or at such other address of which the Administrative Agent shall have been notified pursuant thereto;
- (d) agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law or shall limit the right to sue in any other jurisdiction; and
- (e) waives, to the maximum extent not prohibited by law, any right it may have to claim or recover in any legal action or proceeding referred to in this subsection any special, exemplary, punitive or consequential damages.

13.13. Acknowledgements

. The Borrower hereby acknowledges that:

- (a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Loan Documents;
- (b) neither the Administrative Agent nor any Lender has any fiduciary relationship with or duty to the Borrower arising out of or in connection with this Agreement or any of the other Loan Documents, and the relationship between the Administrative Agent and Lenders, on the one hand, and the Borrower, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and
- (c) no joint venture is created hereby or by the other Loan Documents or otherwise exists by virtue of the transactions contemplated hereby among the Lenders or among the Borrower and the Lenders.

13.14. Confidentiality

. Each Lender agrees to keep confidential any Information (a) provided to it by or on behalf of the Borrower or any of its Subsidiaries pursuant to or in connection with this Agreement or (b) obtained by such Lender based on a review of the books and records of the Borrower or any of its Subsidiaries; provided that nothing herein shall prevent any Lender from disclosing any such Information (i) to the Administrative Agent, the Issuing Lender or any other Lender, (ii) to any Transferee or prospective Transferee which receives such Information having been made aware of the confidential nature thereof and having agreed to abide by the provisions of this subsection 13.14, (iii) to its employees, directors, agents, attorneys, accountants and other professional advisors, and to its and its Affiliates' respective employees, officers, directors, agents, attorneys, accountants and other professional advisors who are directed to be bound by the provisions of this subsection 13.14 and who have a need for such Information in connection with this Agreement or other transactions or proposed transactions with the Borrower, (iv) upon the request or demand of any self-regulatory authority or any Governmental Authority having jurisdiction or oversight over such Lender or in connection with any assignment or pledge permitted under subsection 13.6(g), (v) in response to any order of any court, any self-regulatory authority or other Governmental Authority or as may otherwise be required pursuant to any Requirement of Law, (vi) subject to an agreement to comply with the provisions of this subsection, to any actual or prospective counter-party (or its advisors) to any Hedge Agreement, (vii) which has been publicly disclosed other than in breach of this Agreement, (viii) in connection with the exercise of any remedy hereunder, or (ix) with the written consent of either a Responsible Officer or the treasurer of the Borrower or (x) to the extent required by a potential or actual insurer or reinsurer in connection with providing insurance, reinsurance or credit risk mitigation coverage under which payments are to be made or may be made by reference to this Agreement.

For purposes of this Section, "<u>Information</u>" means all information received from either Borrower or any Subsidiary relating to either Borrower or any Subsidiary or any of their respective businesses, other than any such information that is available to the Administrative Agent, any Lender or the Issuing Lender on a nonconfidential basis prior to disclosure by the Borrower or any Subsidiary and other than information pertaining to this Agreement routinely provided by arrangers to data service providers, including league table providers, that serve the lending industry, <u>provided</u> that, in the case of information received from the Borrower or any Subsidiary after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

Each of the Administrative Agent, the Lenders and the Issuing Lender acknowledges that (a) the Information may include material non-public information concerning the Borrower or a Subsidiary, as the case may be, (b) it has developed compliance procedures regarding the use of material non-public information and (c) it will handle such material non-public information in accordance with applicable Law, including Federal and state securities laws.

13.15. Loan Conversion/Participations

.

- (a) (i) On any Conversion Date, to the extent not otherwise prohibited by a Requirement of Law or otherwise, all Loans (other than CAF Advances) outstanding in any currency other than Dollars ("Loans to be Converted") shall be converted into Revolving Credit Loans denominated in Dollars (calculated on the basis of the relevant Exchange Rates as of the Business Day immediately preceding the Conversion Date) ("Converted Loans") and (ii) on the Conversion Date (A) each Lender severally, unconditionally and irrevocably agrees that it shall purchase in Dollars a participating interest in such Converted Loans in an amount equal to its Conversion Sharing Percentage of the outstanding principal amount of the Converted Loans and (B) to the extent necessary to cause the Committed Outstandings Percentage of each Lender to equal its Revolving Credit Commitment Percentage (calculated immediately prior to the termination or expiration of the Revolving Credit Commitments), each Lender severally, unconditionally and irrevocably agrees that it shall purchase or sell a participating interest in Revolving Credit Loans then outstanding. Each Lender will immediately transfer to the Administrative Agent, in immediately available funds, the amounts of its participation(s), and the proceeds of such participation(s) shall be distributed by the Administrative Agent to each Lender from which a participating interest is being purchased in the amount(s) provided for in the preceding sentence. All Converted Loans shall bear interest at the rate which would otherwise be applicable to ABR Loans.
- (b) If, for any reason, the Loans to be Converted may not be converted into Dollars in the manner contemplated by paragraph (a) of this <u>subsection 13.15</u>, (i) effective on such Conversion Date, each Lender severally, unconditionally and irrevocably agrees that it shall purchase a participating interest in such Loans to be Converted, as the case may be, in an amount equal to its Conversion Sharing Percentage of such Loans to be Converted, and (ii) each Lender shall purchase or sell participating interests as provided in paragraph (a)(ii)(B) of this <u>subsection 13.15</u>. Each such Lender will immediately transfer to the appropriate Administrative Agent, in immediately available funds, the amount(s) of its participation(s), and the proceeds of such participation(s) shall be distributed by the Administrative Agent to each relevant Lender in the amount(s) provided for in the preceding sentence.
- (c) To the extent any Non-Excluded Taxes are required to be withheld from any amounts payable by a Lender to another Lender in connection with its participating interest in any Converted Loan, the Borrower shall be required to pay increased amounts to the Lender receiving such payments to the same extent they would be required under <u>subsection 3.10</u> if the Borrower were making payments directly to such Lender.
- (d) Any time when the actions contemplated by paragraph (a) or (b) of this <u>subsection 13.15</u> have been taken, upon the notice of any Lender to the Borrower the following shall occur (which notice shall not be required to be declared at any time an Event of Default under <u>subsection 10(g)</u> has occurred): (i) the Borrower (through the guarantee contained in <u>Section 12</u>) shall automatically be deemed to have assumed the Local Currency Loans which are Converted Loans in which such Lender holds a participation, and (ii) such Local Currency Loans shall be assigned by the relevant Lender holding such

Local Currency Loans or obligations to the Lender who gave the notice requesting such assumption by the Borrower.

(e) If a Defaulting Lender exists on the Conversion Date, the Administrative Agent may, in consultation with one or more other Revolving Lenders which are not Defaulting Lenders, revise this <u>subsection 13.15</u> and related definitions in order to allocate payments for the benefit of such Defaulting Lender in a manner consistent with <u>subsection 2.19</u>.

13.16. Judgment

•

- (a) If for the purpose of obtaining judgment in any court it is necessary to convert a sum due hereunder in one currency into another currency, the parties hereto agree, to the fullest extent that they may effectively do so, that the rate of exchange used shall be that at which in accordance with normal banking procedures the Administrative Agent could purchase the first currency with such other currency in the city in which it normally conducts its foreign exchange operation for the first currency on the Business Day preceding the day on which final judgment is given.
- (b) The obligation of the Borrower in respect of any sum due from it to any Lender hereunder shall, notwithstanding any judgment in a currency (the "Judgment Currency") other than that in which such sum is denominated in accordance with the applicable provisions of this Agreement (the "Agreement Currency"), be discharged only to the extent that on the Business Day following receipt by such Lender of any sum adjudged to be so due in the Judgment Currency such Lender may in accordance with normal banking procedures purchase the Agreement Currency with the Judgment Currency; if the amount of Agreement Currency so purchased is less than the sum originally due to such Lender in the Agreement Currency, the Borrower agrees notwithstanding any such judgment to indemnify such Lender against such loss, and if the amount of the Agreement Currency so purchased exceeds the sum originally due to any Lender, such Lender agrees to remit to the Borrower such excess.

13.17. WAIVERS OF JURY TRIAL

. THE BORROWER, THE ADMINISTRATIVE AGENT AND THE LENDERS HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVE, TO THE FULLEST EXTENT PERMITTED BY LAW, TRIAL BY JURY IN ANY LEGAL ACTION OR PROCEEDING (WHETHER IN CONTRACT, TORT OR OTHERWISE AND WHETHER AT LAW OR IN EQUITY) RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT AND FOR ANY COUNTERCLAIM THEREIN.

13.18. USA Patriot Act Notice

Each Lender that is subject to the requirements of the Patriot Act and the requirements of the Beneficial Ownership Regulation and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies the Borrower that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "Patriot Act") and the requirements of the Beneficial Ownership Regulation, it is required to obtain, verify and record information that identifies the Borrower and each Foreign Subsidiary Borrower, which information includes the name and address of the Borrower and each Foreign Subsidiary Borrower and other information that will allow such Lender or the Administrative Agent, as applicable, to identify the Borrower and each Foreign Subsidiary Borrower in accordance with the Patriot Act and the Beneficial Ownership Regulation.

13.19. No Advisory or Fiduciary Responsibility

. In connection with all aspects of each transaction contemplated hereby (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document), the Borrower acknowledges and agrees, and acknowledges its Affiliates' understanding, that: (a) (i) the arranging and other services regarding this Agreement provided by the Administrative Agent and the Arrangers are arm's-length commercial transactions between the Borrower and its Affiliates, on the one hand, and the Administrative Agent and the Arrangers, on the other hand, (ii) the Borrower has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate, and (iii) the Borrower is capable of evaluating, and understands and accepts, the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents; (b) (i) the Administrative Agent and the Arrangers each is and has been acting solely as a principal and, except as expressly agreed in writing by the relevant parties, has not been, is not, and will not be acting as an advisor, agent or fiduciary for the Borrower or any of its respective Affiliates, or any other Person and (ii) neither the Administrative Agent nor the Arrangers has any obligation to the Borrower or any of its respective Affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein and in the other Loan Documents; and (c) the Administrative Agent and the Arrangers and their respective Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of the Borrower and its Affiliates, and neither the Administrative Agent nor the Arrangers has any obligation to disclose any of such interests to the Borrower or any of its Affiliates.

13.20. Acknowledgement and Consent to Bail-In of Affected Financial Institutions

- . Notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any Affected Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the Write-Down and Conversion Powers of the applicable Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:
- (a) the application of any Write-Down and Conversion Powers by the applicable Resolution Authority to any such liabilities arising hereunder which may be payable to it by any party hereto that is an Affected Financial Institution; and
- (b) the effects of any Bail-In Action on any such liability, including, if applicable:
 - (i) a reduction in full or in part or cancellation of any such liability;
- (ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such Affected Financial Institution, its parent undertaking, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or
- (iii) the variation of the terms of such liability in connection with the exercise of the Write-Down and Conversion Powers of the applicable Resolution Authority.

13.21. Acknowledgement Regarding Any Supported QFCs

. To the extent that the Loan Documents <u>provide</u> support, through a guarantee or otherwise, for Hedge Agreements or any other agreement or instrument that is a QFC (such support, "<u>QFC Credit</u>

Support" and, each such QFC, a "Supported QFC"), the parties acknowledge and agree as follows with respect to the resolution power of the Federal Deposit Insurance Corporation under the Federal Deposit Insurance Act and Title II of the Dodd-Frank Wall Street Reform and Consumer Protection Act (together with the regulations promulgated thereunder, the "U.S. Special Resolution Regimes") in respect of such Supported QFC and QFC Credit Support (with the provisions below applicable notwithstanding that the Loan Documents and any Supported QFC may in fact be stated to be governed by the laws of the State of New York and/or of the United States or any other state of the United States):

- (a) In the event a Covered Entity that is party to a Supported QFC (each, a "Covered Party") becomes subject to a proceeding under a U.S. Special Resolution Regime, the transfer of such Supported QFC and the benefit of such QFC Credit Support (and any interest and obligation in or under such Supported QFC and such QFC Credit Support, and any rights in property securing such Supported QFC or such QFC Credit Support) from such Covered Party will be effective to the same extent as the transfer would be effective under the U.S. Special Resolution Regime if the Supported QFC and such QFC Credit Support (and any such interest, obligation and rights in property) were governed by the laws of the United States or a state of the United States. In the event a Covered Party or a BHC Act Affiliate of a Covered Party becomes subject to a proceeding under a U.S. Special Resolution Regime, Default Rights under the Loan Documents that might otherwise apply to such Supported QFC or any QFC Credit Support that may be exercised against such Covered Party are permitted to be exercised to no greater extent than such Default Rights could be exercised under the U.S. Special Resolution Regime if the Supported QFC and the Loan Documents were governed by the laws of the United States or a state of the United States. Without limitation of the foregoing, it is understood and agreed that rights and remedies of the parties with respect to a Defaulting Lender shall in no event affect the rights of any Covered Party with respect to a Supported QFC or any QFC Credit Support.
 - (b) As used in this <u>Section 13.21</u>, the following terms have the following meanings:

"BHC Act Affiliate" of a party means an "affiliate" (as such term is defined under, and interpreted in accordance with, 12 U.S.C. 1841(k)) of such party.

"Covered Entity" means any of the following:

- (i) a "covered entity" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 252.82(b);
- (ii) a "covered bank" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 47.3(b); or
- (iii) a "covered FSI" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 382.2(b).

"<u>Default Right</u>" has the meaning assigned to that term in, and shall be interpreted in accordance with, 12 C.F.R. §§ 252.81, 47.2 or 382.1, as applicable.

"QFC" has the meaning assigned to the term "qualified financial contract" in, and shall be interpreted in accordance with, 12 U.S.C. 5390(c)(8)(D).

[Signature Pages Follow]

IN WITNESS WHEREOF, the parties hereto have caused th	is Agreement to be duly	y executed and delivered	d by their proper and	duly
authorized officers as of the day and year first above written.				

BOSTON SCIENTIFIC CORPORATION

By:		
Name:		
Title:		
	Signature Page to	
	Boston Scientific Credit Agreement	

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By:
Name:
Title:
WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender
By:
Name:
Title:

Signature Page to
Boston Scientific Credit Agreement

Insert Lender's Name, as a Lender [and as an Issuing Lender]		
By: Name: Title:		
Signature Page to Boston Scientific Credit Agreement		

SCHEDULE I

NAMES AND COMMITMENTS OF LENDERS

Name	Revolving Credit Commitment	Multicurrency Commitment	Issuing Lender Commitment
Wells Fargo Bank, National Association	\$ 275,000,000 <u>317,000,000</u>	\$80,000,000	\$83,333,333
Bank of America, N.A.	\$ 275,000,000 <u>317,000,000</u>	\$ 80,000,000 <u>92,240,000</u>	\$83,333,333
Barclays Bank PLC	\$275,000,000	\$80,000,000	\$83,333,333
Citibank, N.A.	\$ 275,000,000 <u>317,000,000</u>	\$ 80,000,000 <u>92,240,000</u>	\$83,333,333
JPMorgan Chase Bank, N.A.	\$ 275,000,000 <u>317,000,000</u>	\$80,000,000	\$83,333,333
Société Générale	\$275,000,000	\$80,000,000	\$83,333,333
BNP Paribas	\$115,000,000	\$33,440,000	
DNB Capital LLC Deutsche Bank AG New York			
Branch	\$115,000,000	\$33,440,000	
Goldman Sachs Bank USA	\$115,000,000	\$33,440,000	
MUFGRoyal Bank, Ltd. of Canada	\$115,000,000	\$33,440,000	
Royal Standard Chartered Bank-of Canada	\$115,000,000	\$33,440,000	
The Bank of Nova Scotia	\$115,000,000	\$33,440,000	
The Toronto-Dominion Bank, New York Branch	\$115,000,000	\$33,440,000	
U.S Bank National Association	\$115,000,000	\$33,440,000	
Allied IrishMUFG Banks, p.l.e, Ltd.	\$ 45,000,000 <u>43,000,000</u>	\$ 13,120,000 <u>12,480,000</u>	
Deutsche Bank AG New York Branch	\$45,000,000	\$13,120,000	
Intesa Sanpaolo S.p.A., New York Branch	\$45,000,000	\$13,120,000	
Standard Chartered Bank	\$45,000,000	\$13,120,000	
Total	\$2,750,000,000	\$800,000,000	\$500,000,000

CERTIFICATIONS

- I, Michael F. Mahoney, certify that:
 - I have reviewed this Quarterly Report on Form 10-Q of Boston Scientific Corporation;
 - Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
 - Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
 - The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our a) supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about c) the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
 - The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are a) reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's b) internal control over financial reporting.

Date: August 1, 2024 /s/ Michael F. Mahoney

> Michael F. Mahoney Chief Executive Officer

CERTIFICATIONS

I, Daniel J. Brennan, certify that:

- 1 I have reviewed this Quarterly Report on Form 10-Q of Boston Scientific Corporation;
- 2 Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3 Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5 The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 1, 2024 /s/ Daniel J. Brennan

Daniel J. Brennan

Executive Vice President and Chief Financial Officer

CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q of Boston Scientific Corporation (the "Company") for the period ended June 30, 2024 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned Chief Executive Officer of the Company hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that based on his knowledge:

- (1) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of Boston Scientific Corporation.

This certification shall not be deemed "filed" for any purpose, nor shall it be deemed to be incorporated by reference into any filing under the Securities Act of 1933 or the Securities Exchange Act of 1934 regardless of any general incorporation language in such filing.

By: /s/ Michael F. Mahoney

Michael F. Mahoney Chief Executive Officer

August 1, 2024

CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q of Boston Scientific Corporation (the "Company") for the period ended June 30, 2024 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned Chief Financial Officer of the Company hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that based on his knowledge:

- (1) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of Boston Scientific Corporation.

This certification shall not be deemed "filed" for any purpose, nor shall it be deemed to be incorporated by reference into any filing under the Securities Act of 1933 or the Securities Exchange Act of 1934 regardless of any general incorporation language in such filing.

By: /s/ Daniel J. Brennan

Daniel J. Brennan

Executive Vice President and Chief Financial Officer

August 1, 2024